UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

Form 10-Q

(Mark One)

X

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF

THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended March 31, 2011

or

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF

THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from to

Commission File Number: 001-14236

FelCor Lodging Trust Incorporated

(Exact name of registrant as specified in its charter)

Maryland

(State or other jurisdiction of incorporation or organization)

545 E. John Carpenter Freeway, Suite 1300, Irving, Texas

(Address of principal executive offices)

(972) 444-4900

(Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15 (d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. \boxtimes Yes \square No

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (\$232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). \Box Yes \Box No

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See definition of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer Non-accelerated filer (Do not check if a smaller reporting company) Accelerated filer \boxtimes Smaller reporting company \square

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). □ Yes ⊠ No

At April 21, 2011, the registrant had issued and outstanding 124,479,397 shares of common stock.

75-2541756 (I.R.S. Employer Identification No.)

75062

(Zip Code)

INDEX

Page

PART I - FINANCIAL INFORMATION

Item 1.	Financial Statements	3
	Consolidated Balance Sheets March 31, 2011 and December 31, 2010 (unaudited)	3
	Consolidated Statements of Operations – For the Three Months Ended March 31, 2011 and 2010 (unaudited)	4
	Consolidated Statements of Comprehensive Loss – For the Three Months Ended March 31, 2011 and 2010 (unaudited)	5
	Consolidated Statements of Changes in Equity – For the Three Months Ended March 31, 2011 and 2010 (unaudited)	6
	Consolidated Statements of Cash Flows – For the Three Months Ended March 31, 2011 and 2010 (unaudited)	7
	Notes to Consolidated Financial Statements	8
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	19
	General	19
	Results of Operations	21
	Non-GAAP Financial Measures	22
	Pro Rata Share of Rooms Owned	27
	Hotel Portfolio Composition	28
	Hotel Operating Statistics	29
	Hotel Portfolio	31
	Liquidity and Capital Resources	33
	Inflation	36
	Seasonality	36
	Disclosure Regarding Forward-Looking Statements	36
Item 3.	Quantitative and Qualitative Disclosures about Market Risk	37
Item 4.	Controls and Procedures	37

PART II - OTHER INFORMATION

Item 6. Exhibits	38
SIGNATURE	39

PART I -- FINANCIAL INFORMATION

Item 1. **Financial Statements.**

FELCOR LODGING TRUST INCORPORATED

CONSOLIDATED BALANCE SHEETS

(unaudited, in thousands)

	March 31, 2011	December 31, 2010
Assets		
Investment in hotels, net of accumulated depreciation of \$998,506 and \$982,564 at March 31, 2011 and December 31, 2010, respectively	\$ 1,960,848	\$ 1,985,779
Investment in unconsolidated entities	73,972	75,920
Hotel held for sale	18,533	
Cash and cash equivalents	91,040	200,972
Restricted cash	19,254	16,702
Accounts receivable, net of allowance for doubtful accounts of \$683 and \$696 at March 31, 2011 and December 31, 2010, respectively	36,878	27,851
Deferred expenses, net of accumulated amortization of \$14,863 and \$17,892 at March 31, 2011 and December 31, 2010, respectively	22,245	19,940
Other assets	25,438	32,271
Total assets	\$ 2,248,208	\$ 2,359,435
Liabilities and Equity		
Debt, net of discount of \$50,432 and \$53,193 at March 31, 2011 and December 31, 2010, respectively	\$ 1,466,798	\$ 1,548,309
Distributions payable	76,293	76,293
Accrued expenses and other liabilities	154,478	144,451
Total liabilities	1,697,569	1,769,053
Commitments and contingencies		
Redeemable noncontrolling interests in FelCor LP at redemption value, 285 units issued and outstanding at March 31, 2011 and December 31, 2010	1,745	2,004
Equity:		
Preferred stock, \$0.01 par value, 20,000 shares authorized:		
Series A Cumulative Convertible Preferred Stock, 12,880 shares, liquidation value of \$322,011, issued and outstanding at March 31, 2011 and December 31, 2010	309,362	309,362
Series C Cumulative Redeemable Preferred Stock, 68 shares, liquidation value of \$169,950, issued and outstanding at March 31, 2011 and December 31, 2010	169,412	169,412
Common stock, \$0.01 par value, 200,000 shares authorized and 96,872 shares issued at March 31, 2011, and 101,038 shares issued, including shares in treasury, at December 31, 2010	969	1,010
Additional paid-in capital	2,191,278	2,190,308
Accumulated other comprehensive income	27,745	26,457
Accumulated deficit	(2,169,344)	-
Less: Common stock in treasury, at cost, of 4,156 shares at December 31, 2010		(73,341)
Total FelCor stockholders' equity	529,422	568,583
Noncontrolling interests in other partnerships	19,472	19,795
Total equity	548,894	588,378
Total liabilities and equity	\$ 2,248,208	\$ 2,359,435

CONSOLIDATED STATEMENTS OF OPERATIONS For the Three Months Ended March 31, 2011 and 2010 (unaudited, in thousands, except for per share data)

	Three Months Ended March 31,		
	2011	2010	
Revenues:			
Hotel operating revenue	\$ 235,105	\$ 216,758	
Other revenue	225	365	
Total revenues	235,330	217,123	
Expenses:			
Hotel departmental expenses	85,422	77,672	
Other property related costs	69,457	62,702	
Management and franchise fees	10,942	10,145	
Taxes, insurance and lease expense	20,723	22,379	
Corporate expenses	9,537	9,847	
Depreciation and amortization	35,317	36,284	
Other expenses	631	561	
Total operating expenses	232,029	219,590	
Operating income (loss)	3,301	(2,467)	
Interest expense, net	(33,765)	(35,403)	
Extinguishment of debt	(245)		
Loss before equity in loss from unconsolidated entities	(30,709)	(37,870)	
Equity in loss from unconsolidated entities	(1,583)	(1,474)	
Gain on involuntary conversion	150		
Loss from continuing operations	(32,142)	(39,344)	
Discontinued operations	416	(23,598)	
Net loss	(31,726)	(62,942)	
Net loss (income) attributable to noncontrolling interests in other partnerships	(58)	229	
Net loss attributable to redeemable noncontrolling interests in FelCor LP	120	325	
Net loss attributable to FelCor	(31,664)	(62,388)	
Preferred dividends	(9,678)	(9,678)	
Net loss attributable to FelCor common stockholders	\$ (41,342)	\$ (72,066)	
Basic and diluted per common share data:			
Loss from continuing operations	\$ (0.44)	\$ (0.77)	
Net loss	<u>_</u>	\$ (1.14)	
Basic and diluted weighted average common shares outstanding	95,350	63,475	

CONSOLIDATED STATEMENTS OF COMPREHENSIVE LOSS For the Three Months Ended March 31, 2011 and 2010 (unaudited, in thousands)

	Three Months Ended March 31,		
		2011	2010
Net loss	\$	(31,726)	\$ (62,942)
Foreign currency translation adjustment		1,292	2,079
Comprehensive loss		(30,434)	(60,863)
Comprehensive loss (income) attributable to noncontrolling interests in other partnerships		(58)	229
Comprehensive loss attributable to redeemable noncontrolling interests in FelCor LP		116	316
Comprehensive loss attributable to FelCor	\$	(30,376)	\$ (60,318)

FELCOR LODGING TRUST INCORPORATED CONSOLIDATED STATEMENTS OF CHANGES IN EQUITY

For the Three Months Ended March 31, 2011 and 2010

(unaudited, in thousands)

	Preferred Stock		Preferred Stock			Common Stock			_		Accumulated		ı				Noncontrolling					
	Number of Shares	1	Amount	Number of Shares		Amount		Additional Paid-in Capital		Other Comprehensive Income	A	Accumulated Deficit]	Freasury Stock	I	Interests in Other Partnerships		omprehensive come (Loss)	Total Equity			
Balance at December 31, 2009	12,948	\$	478,774	69,413	\$	694	\$	2,021,837	\$	23,528	\$	(1,792,822)	\$	(71,895)	\$	22,583			\$	682,699		
Issuance of stock awards	_		_	_		—		(229)		_		_		297		_				68		
Amortization of stock awards			—	_		_		1,413				—		—		—				1,413		
Forfeiture of stock awards				_		_		149				—		(631)		—				(482)		
Allocation to redeemable noncontrolling interests	—		_	—		_		(935)		—		_		_		_				(935)		
Contribution from noncontrolling interests	_		—	_		—		_				_		—		—				_		
Distribution to noncontrolling interests	—		—			—		_		—		—		—		(602)				(602)		
Other			-	-		_		_				(10)		-		-				(10)		
Preferred dividends accrued:												(6.270)								(6.070)		
\$0.4875 per Series A preferred share			_					_				(6,279)				—				(6,279)		
\$0.50 per Series C depositary preferred share	—			—		—		—		—		(3,399)		—		—				(3,399)		
Comprehensive loss:																						
Foreign exchange translation	—		—	—				—		2,070		—		—		—	\$	2,070				
Net income (loss)			—	—		_		—				(62,388)		—		(229)		(62,617)				
Comprehensive loss																	\$	(60,547)		(60,547)		
Balance at March 31, 2010	12,948	\$	478,774	69,413	\$	694	\$	2,022,235	\$	25,598	\$	(1,864,898)	\$	(72,229)	\$	21,752	_		\$	611,926		
Balance at December 31, 2010	12,948	\$	478,774	101,038	\$	1,010	\$	2,190,308	\$	26,457	\$	(2,054,625)	\$	(73,341)	\$	19,795			\$	588,378		
Retirement of treasury stock	—			(4,156)		(41)		—		—		(73,300)		73,341		—				—		
Amortization of stock awards	—		—	—		—		827				—		—		—				827		
Forfeiture of stock awards	_		_	(10)		_		_		—		(75)		_		_				(75)		
Allocation to redeemable noncontrolling interests	_		_	_		_		143		_		_		_		_				143		
Contribution from noncontrolling interests	_		_	_		_		_		_		_		_		64				64		
Distribution to noncontrolling interests	_			—		_		_		—		_		—		(445)				(445)		
Other			_	_				_				(2)				_				(2)		
Preferred dividends accrued:																						
\$0.4875 per Series A preferred share	_		_	_		_		_		_		(6,279)		_		_				(6,279)		
\$0.50 per Series C depositary preferred share	—		_	—		_		_		—		(3,399)		_		_				(3,399)		
Comprehensive loss:																						
Foreign exchange translation	_							_		1,288							\$	1,288				
Net income (loss)			_					—				(31,664)		—		58		(31,606)				
Comprehensive loss																	\$	(30,318)		(30,318)		
Balance at March 31, 2011	12,948	\$	478,774	96,872	\$	969	\$	2,191,278	\$	27,745	\$	(2,169,344)	\$		\$	19,472			\$	548,894		

CONSOLIDATED STATEMENTS OF CASH FLOWS For the Three Months Ended March 31, 2011 and 2010 (unaudited, in thousands)

	Three Months Ended March 31,			
		2011		2010
Cash flows from operating activities:				
Net loss	\$	(31,726)	\$	(62,942)
Adjustments to reconcile net loss to net cash provided by operating activities:				
Depreciation and amortization		35,671		37,598
Amortization of deferred financing fees and debt discount		4,715		4,131
Amortization of unearned officers' and directors' compensation		1,803		1,616
Equity in loss from unconsolidated entities		1,583		1,474
Distributions of income from unconsolidated entities		165		142
Extinguishment of debt		252		—
Impairment loss				21,060
Changes in assets and liabilities:				
Accounts receivable		(9,316)		(7,269)
Restricted cash – operations		(458)		984
Other assets		(3,090)		(1,879)
Accrued expenses and other liabilities		<u>6,206</u> 5,805		<u>33,597</u> 28,512
Net cash flow provided by operating activities Cash flows from investing activities:		5,805		20,312
Improvements and additions to hotels		(15,038)		(8,200)
Additions to condominium project		(13,038)		(110)
		. ,		
Change in restricted cash – investing		(2,094)		(1,219)
Insurance proceeds		11		
Distributions from unconsolidated entities		200		559
Contributions to unconsolidated entities		(16.096)		(300)
Net cash flow used in investing activities		(16,986)		(9,270)
Cash flows from financing activities:				0.1
Proceeds from borrowings		185,040		81
Repayment of borrowings		(269,318)		(4,783)
Payment of deferred financing fees		(4,491)		(1,695)
Distributions paid to noncontrolling interests		(445)		(602)
Contributions from noncontrolling interests		64		
Distributions paid to preferred stockholders		(9,678)		
Net cash flow used in financing activities		(98,828)		(6,999)
Effect of exchange rate changes on cash		77		234
Net change in cash and cash equivalents		(109,932)		12,477
Cash and cash equivalents at beginning of periods		200,972		263,531
Cash and cash equivalents at end of periods	\$		\$	276,008
Cash and cash equivalents at end of periods	\$	91,040	ψ	270,008
Supplemental cash flow information – interest paid	\$	12,095	\$	14,166

FELCOR LODGING TRUST INCORPORATED NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

1. Organization

FelCor Lodging Trust Incorporated (NYSE:FCH), or FelCor, is a Maryland corporation, operating as a real estate investment trust, or REIT. We are the sole general partner of, and the owner of a greater than 99% partnership interest in, FelCor Lodging Limited Partnership, or FelCor LP, through which we held ownership interests in i) 81 hotels in continuing operations with approximately 23,000 rooms and ii) one hotel designated as held for sale at March 31, 2011. At March 31, 2011, we had an aggregate of 97,156,344 shares and units outstanding, consisting of 96,871,619 shares of FelCor common stock and 284,725 FelCor LP units not owned by FelCor.

Of the 81 hotels included in continuing operations we owned a 100% interest in 63 hotels, a 90% interest in entities owning three hotels, an 82% interest in an entity owning one hotel, a 60% interest in an entity owning one hotel and a 50% interest in entities owning 13 hotels. We consolidate our real estate interests in the 68 hotels in which we held greater than 50% ownership interests, and we record the real estate interests of the 13 hotels in which we held 50% ownership interests using the equity method.

At March 31, 2011, 80 of the 81 hotels were leased to operating lessees, and one 50%-owned hotel was operated without a lease. We held majority interests and had direct or indirect controlling interests in all of the operating lessees. Because we owned controlling interests in these lessees (including lessees of 12 of the 13 hotels in which we owned 50% of the real estate interests), we consolidated our lessee interests in these hotels (we refer to these 80 hotels as our Consolidated Hotels) and reflect 100% of those hotels' revenues and expenses on our statement of operations. Of our Consolidated Hotels, we owned 50% of the real estate interests in each of 12 hotels (we accounted for the ownership in our real estate interests of these hotels by the equity method) and majority real estate interests in these hotels (we consolidate our real estate interest in these hotels).

The following table illustrates the distribution of our 80 Consolidated Hotels among our brands at March 31, 2011:

Brand	Hotels	Rooms
Embassy Suites Hotels [®]	44	11,450
Holiday Inn [®]	15	5,154
Sheraton [®] and Westin [®]	8	2,774
Doubletree [®] and Hilton [®]	9	2,030
Marriott [®] and Renaissance [®]	3	1,321
Fairmont [®]	1	383
Total	80	23,112

At March 31, 2011, our Consolidated Hotels were located in the United States (78 hotels in 22 states) and Canada (two hotels in Ontario), with concentrations in California (15 hotels), Florida (12 hotels) and Texas (11 hotels). Approximately 54% of our hotel room revenues were generated from hotels in these three states during the first three months of 2011.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

1. **Organization** — (continued)

At March 31, 2011, of our 80 Consolidated Hotels: (i) subsidiaries of Hilton Hotels Corporation, or Hilton, managed 52 hotels, (ii) subsidiaries of InterContinental Hotels Group, or IHG, managed 15 hotels, (iii) subsidiaries of Starwood Hotels & Resorts Worldwide Inc., or Starwood, managed eight hotels, (iv) subsidiaries of Marriott International Inc., or Marriott, managed three hotels, (v) a subsidiary of Fairmont Hotels and Resorts, or Fairmont, managed one hotel, and (vi) an independent management company managed one hotel.

Our hotels managed by Marriott are accounted for on a fiscal year comprised of 52 or 53 weeks ending on the Friday closest to December 31. Our three-month period ending March 31, 2011 and 2010 includes the results of operations for our Marriott-managed hotels for the 12 week period ending March 25, 2011 and March 26, 2010, respectively.

The information in our consolidated financial statements for the three months ended March 31, 2011 and 2010 is unaudited. Preparing financial statements in conformity with accounting principles generally accepted in the United States of America, or GAAP, requires management to make estimates and assumptions that affect the amounts reported in the financial statements and accompanying notes. Actual results could differ from those estimates. The accompanying financial statements for the three months ended March 31, 2011 and 2010, include adjustments based on management's estimates (consisting of normal and recurring accruals), which we consider necessary for a fair presentation of the results for the periods. The financial information should be read in conjunction with the consolidated financial statements for the year ended December 31, 2010, included in our Annual Report on Form 10-K. Operating results for the three months ended March 31, 2011 are not necessarily indicative of actual operating results for the entire year.

2. Change in Accounting Estimate

Effective January 1, 2011, we reclassified certain inventory (such as china, glass, silver, and linen) aggregating \$10.3 million to investment in hotels from other assets. We believe this classification to be preferable to its prior classification because we receive a long-term benefit (approximately three years) from these inventories, similar to other long-term physical assets, which are classified as investment in hotels. This change was considered a change in accounting estimate inseparable from a change in accounting policy, and will result in changes to our depreciation expense prospectively. As a result, existing inventories will be amortized over a three-year period. Prospectively, significant additions in conjunction with major renovations, expansions or changes in brand standards for these inventories will be capitalized at acquisition, and depreciated over a three year estimated useful life. Minor replacement or replenishment of inventory will be expensed when purchased.

3. Reacquired Stock

Effective January 1, 2011, we have removed previously reacquired capital stock shown as treasury stock from our balance sheet and recorded the related amounts as a reduction of common stock (at \$0.01 par value per share) and an increase in accumulated deficit. Prior to 2011, our accounting records included treasury stock. We changed our accounting presentation to be consistent with Maryland law (Maryland is our domicile), which does not contemplate treasury stock. This change in accounting policy was recorded for book purposes as a retirement of treasury stock. Any capital stock reacquired in the future for any purpose will be recorded as a reduction of common stock (at \$0.01 par value per share) and an increase in accumulated deficit.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

4. Investment in Unconsolidated Entities

We owned 50% interests in joint ventures that owned 13 hotels at March 31, 2011 and December 31, 2010. We also own a 50% interest in entities that own real estate in Myrtle Beach, South Carolina and provide condominium management services. We account for our investments in these unconsolidated entities under the equity method. We do not have any majority-owned subsidiaries that are not consolidated in our financial statements. We make adjustments to our equity in income from unconsolidated entities related to the difference between our basis in investment in unconsolidated entities compared to the historical basis of the assets recorded by the joint ventures.

The following table summarizes combined balance sheet information for our unconsolidated entities (in thousands):

	Ma	March 31, 2011		cember 31, 2010
Investment in hotels, net of accumulated depreciation	\$	187,242	\$	192,584
Total assets	\$	205,259	\$	209,742
Debt	\$	153,622	\$	154,590
Total liabilities	\$	157,653	\$	159,171
Equity	\$	47,606	\$	50,571

Our unconsolidated entities' debt at March 31, 2011 and December 31, 2010 consisted entirely of non-recourse mortgage debt.

The following table sets forth summarized combined statement of operations information for our unconsolidated entities (in thousands):

	Three Mon Marc	
	2011	2010
Total revenues	\$ 11,688	\$ 12,739
Net loss	\$ (2,235)	\$ (3,136)
Net loss attributable to FelCor	\$ (1,118)	\$ (1,568)
Gain on joint venture liquidation		559
Depreciation of cost in excess of book value	(465)	(465)
Equity in loss from unconsolidated entities	\$ (1,583)	\$ (1,474)

The following table summarizes the components of our investment in unconsolidated entities (in thousands):

	Γ	March 31, 2011	De	cember 31, 2010
Hotel-related investments	\$	14,875	\$	15,736
Cost in excess of book value of hotel investments		50,169		50,634
Land and condominium investments		8,928		9,550
	\$	73,972	\$	75,920

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

4. Investment in Unconsolidated Entities — (continued)

The following table summarizes the components of our equity in loss from unconsolidated entities (in thousands):

		onths Ended ch 31,
	2011	2010
Hotel investments	\$ (962) \$ (905)
Other investments	(621) (569)
Equity in loss from unconsolidated entities	\$ (1,583) \$ (1,474)

5. Debt

Consolidated debt consisted of the following (dollars in thousands):

	Encumbered Hotels	Interest Rate (%)	Maturity Date	March 31, 2011	December 31, 2010
Secured line of credit ^(a)	11 hotels	L + 4.50	August 2014 ^(b)	\$ 145,000	\$
Mortgage debt					
Mortgage debt	12 hotels	L + 0.93 ^(c)	November 2011	250,000	250,000
Mortgage debt	9 hotels	$L + 5.10^{(d)}$	April 2015	212,000	212,000
Mortgage debt	7 hotels	9.02	April 2014	112,109	113,220
Mortgage debt	5 hotels ^(e)	6.66	June - August 2014	68,744	69,206
Mortgage debt	1 hotel	8.77	May 2013	27,770	27,770
Mortgage debt	1 hotel	5.81	July 2016	11,210	11,321
Mortgage debt	1 hotel	6.15	June 2011	7,473	7,800
Other		4.25	May 2011	563	524
Senior notes					
Senior secured notes ^(f)	14 hotels	10.00	October 2014	585,573	582,821
Senior notes		9.00	June 2011	46,356	46,347
Retired debt					227,300
Total	61 hotels			\$1,466,798	\$ 1,548,309

(a) The outstanding balance on the line of credit was paid subsequent to March 31, 2011. We currently have full availability under our \$225 million line of credit.

- (b) This loan can be extended for one year (to 2015), subject to satisfying certain conditions.
- (c) We purchased an interest rate cap that caps LIBOR at 7.8% and expires November 2011 for a \$250 million notional amount.
- (d) LIBOR for this loan is subject to a 3% floor. We purchased an interest rate cap that caps LIBOR at 5.0% and expires May 2012 for a \$212 million notional amount.
- (e) The hotels securing this debt are subject to separate loan agreements and are not cross-collateralized.
- (f) These notes have \$636 million in aggregate principal outstanding and were sold at a discount that provides an effective yield of 12.875% before transaction costs.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

5. **Debt** — (continued)

In March 2011, we closed a \$225 million secured, revolving line of credit with a group of seven banks. At closing, we repaid two secured loans, totaling \$198.3 million and \$28.8 million, with a combination of \$52.1 million of cash on hand and funds drawn under our new line of credit. The repaid loans would have matured in 2013 and 2012 (including extensions), respectively, and were secured by mortgages on 11 hotels. Those same hotels now secure repayment of amounts outstanding under the line of credit. The credit facility bears interest equal to LIBOR, plus 4.5%, with no LIBOR floor.

We reported \$33.8 million and \$35.4 million of interest expense for the three months ended March 31, 2011 and 2010, respectively, which is net of: (i) interest income of \$41,000 and \$105,000 and (ii) capitalized interest of \$198,000 and \$145,000, respectively.

6. Hotel Operating Revenue, Departmental Expenses, and Other Property Operating Costs

Hotel operating revenue from continuing operations was comprised of the following (in thousands):

	Thr	Three Months Ended March 31,				
		2011	2010			
Room revenue	\$	184,366	\$	170,287		
Food and beverage revenue		38,039		33,555		
Other operating departments		12,700		12,916		
Total hotel operating revenue	\$	235,105	\$	216,758		

Nearly all of our revenue is comprised of hotel operating revenue, which includes room revenue, food and beverage revenue, and revenue from other hotel operating departments (such as telephones, parking and business centers). These revenues are recorded net of any sales or occupancy taxes collected from our guests. All rebates or discounts are recorded, when allowed, as a reduction in revenue, and there are no material contingent obligations with respect to rebates or discounts offered by us. All revenues are recorded on an accrual basis, as earned. Appropriate allowances are made for doubtful accounts, which are recorded as a bad debt expense. The remainder of our revenue was derived from other sources.

Hotel departmental expenses from continuing operations were comprised of the following (in thousands):

	Three Months Ended March 31,						
	2011				20	10	
	Amount		% of Total Hotel Operating Revenue		Amount	% of Total Hotel Operating Revenue	
Room	\$	49,528	21.1%	\$	45,480	21.0%	
Food and beverage		29,859	12.7		26,254	12.1	
Other operating departments		6,035	2.5		5,938	2.7	
Total hotel departmental expenses	\$	85,422	36.3%	\$	77,672	35.8%	

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

6. Hotel Operating Revenue, Departmental Expenses, and Other Property Operating Costs — (continued)

Other property operating costs from continuing operations were comprised of the following (in thousands):

	Three Months Ended March 31,						
		20	11		20	10	
	% of Total Hotel Operating Amount Revenue			Amount	% of Total Hotel Operating Revenue		
Hotel general and administrative expense	\$	23,063	9.8%	\$	20,225	9.3%	
Marketing		21,356	9.1		18,698	8.6	
Repair and maintenance		13,095	5.6		12,323	5.7	
Utilities		11,943	5.0		11,456	5.3	
Total other property operating costs	\$	69,457	29.5%	\$	62,702	28.9%	

7. Taxes, Insurance and Lease Expense

Taxes, insurance and lease expense from continuing operations were comprised of the following (in thousands):

	Th	Three Months Ended March 31,				
		2011		2010		
Hotel lease expense ^(a)	\$	8,304	\$	7,758		
Land lease expense ^(b)		2,235		2,155		
Real estate and other taxes		7,419		8,968		
Property insurance, general liability insurance and other		2,765		3,498		
Total taxes, insurance and lease expense	\$	20,723	\$	22,379		

(a) Hotel lease expense is recorded by the consolidated operating lessees of 12 hotels owned by unconsolidated entities, and is partially (generally 49%) offset through noncontrolling interests in other partnerships. Our 50% share of the corresponding lease income is recorded through equity in income from unconsolidated entities. Hotel lease expense includes percentage rent of \$2.9 million and \$2.4 million for the three months ended March 31, 2011 and 2010, respectively.

(b) Land lease expense includes percentage rent of \$722,000 and \$631,000 for the three months ended March 31, 2011 and 2010, respectively.

FELCOR LODGING TRUST INCORPORATED NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

8. Impairment

Our hotels comprise operations and cash flows that can clearly be distinguished, operationally and for financial reporting purposes, from the remainder of our operations. Accordingly, we consider our hotels to be components for purposes of determining impairment charges and reporting discontinued operations.

During the quarter ended March 31, 2010, we recorded a \$21.1 million impairment charge with regard to two hotels, which were subsequently transferred to their lenders in full satisfaction of the related debt. This impairment charge is reflected in discontinued operations.

For our 2010 impairment charges, we estimated the hotels' fair value by using estimated future cash flows, terminal values based on the projected cash flows and capitalization rates in the range of what is reported in industry publications for operationally similar assets and other available market information. The cash flows used for determining the fair values were discounted using market-based discounts generally used for operationally similar assets. The inputs used to determine the fair values of these hotels are classified as Level 3 under the authoritative guidance for fair value measurements.

We may record additional impairment charges if operating results of individual hotels are materially different from our forecasts, the economy and lodging industry weakens, or we shorten our contemplated holding period for additional hotels.

9. Discontinued Operations

We had one hotel held for sale at March 31, 2011. We consider a sale to be probable within the next twelve months (for purposes of determining whether a hotel is held for sale) when a buyer completes its due diligence review of the asset, we have an executed contract for sale, and we have received a substantial non-refundable deposit.

Discontinued operations include results of operations for one hotel designated as held for sale at March 31, 2011, one hotel sold in 2010, and two hotels transferred to lenders in satisfaction of debt in 2010. The following table summarizes the condensed financial information for those hotels (in thousands):

	Three Months Ended March 31,					
		2011	2010			
Hotel operating revenue	\$	2,241	\$	9,289		
Operating expenses		(1,737)		(32,050) ^(a)		
Operating income (loss) from discontinued operations	\$	504	\$	(22,761)		
Interest expense, net		(81)		(837)		
Extinguishment of debt		(7)				
Income (loss) from discontinued operations	\$	416	\$	(23,598)		

(a) Includes an impairment charge of \$21.1 million.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

10. Loss Per Share

The following table sets forth the computation of basic and diluted loss per share (in thousands, except per share data):

	Three Months Ended March 31,			
		2011		2010
Numerator:				
Net loss attributable to FelCor	\$	(31,664)	\$	(62,388)
Discontinued operations attributable to FelCor		(415)		23,492
Loss from continuing operations attributable to FelCor		(32,079)		(38,896)
Less: Preferred dividends		(9,678)		(9,678)
Numerator for continuing operations attributable to FelCor common stockholders		(41,757)		(48,574)
Discontinued operations attributable to FelCor		415		(23,492)
Numerator for basic and diluted loss attributable to FelCor common stockholders	\$	(41,342)	\$	(72,066)
Denominator:				
Denominator for basic and diluted loss per share		95,350		63,475
Basic and diluted loss per share data:				
Loss from continuing operations	\$	(0.44)	\$	(0.77)
Discontinued operations	\$		\$	(0.37)
Net loss	\$	(0.43)	\$	(1.14)
			_	

Securities that could potentially dilute earnings per share in the future that were not included in the computation of diluted loss per share, because they would have been antidilutive for the periods presented, are as follows (in thousands):

	Three Mont March	
	2011	2010
Series A convertible preferred shares	9,985	9,985

Series A preferred dividends that would be excluded from net loss attributable to FelCor common stockholders, if these Series A preferred shares were dilutive, were \$6.3 million for the three months ended March 31, 2011 and 2010.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

11. Preferred Dividends

In January 2011, we reinstated our current quarterly preferred dividend and paid current quarterly dividends in January 2011. In March 2011, we declared current quarterly preferred dividends for the first quarter of \$9.7 million, which are payable on May 2, 2011. We are unable to pay any common dividends unless and until all accrued and current preferred dividends are paid. Our Board of Directors will determine whether and when to declare future dividends (including the accrued but unpaid preferred dividends) based upon various factors, including operating results, economic conditions, other operation trends, our financial condition and capital requirements, as well as minimum REIT distribution requirements. We had \$76.3 million of aggregate accrued dividends payable on our Series A and Series C preferred stock at March 31, 2011 and December 31, 2010.

12. Noncontrolling Interests

We record the noncontrolling interests of other consolidated partnerships as a separate component of equity in the condensed consolidated balance sheets. Additionally, the condensed consolidated statements of operations separately present earnings and other comprehensive income attributable to controlling and non-controlling interests. We adjust the noncontrolling interests of FelCor LP each period so that the carrying value equals the greater of its carrying value based on the accumulation of historical cost or its redemption value. The historical cost of the noncontrolling interests of FelCor LP is based on the proportional relationship between the carrying value of equity associated with our common stockholders relative to that of the unitholders of FelCor LP. Net income (loss) is allocated to the noncontrolling interests of FelCor LP based on their weighted average ownership percentage during the period. At March 31, 2011, approximately \$1.7 million of cash or FelCor common stock, at our option, would be paid to the noncontrolling interests of FelCor LP if the partnership were terminated. This balance is calculated based on the 284,725 partnership units held by third parties, valued at the March 31, 2011 closing price for our common stock (\$6.13/share), which we have assumed would be equal to the value provided to outside partners upon liquidation of FelCor LP on that date.

	Three	Three Months Ended March 31,			
	2	011	2010		
Balance at beginning of period	\$	2,004 \$	1,062		
Redemption value allocation		(143)	935		
Comprehensive income (loss):					
Foreign exchange translation		4	9		
Net loss		(120)	(325)		
Balance at end of period	\$	1,745 \$	1,681		

The changes in redeemable noncontrolling interests for the three months ended March 31, 2011 and 2010 are shown below (in thousands):

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

13. Fair Value of Financial Instruments

Disclosures about fair value of our financial instruments are based on pertinent information available to management as of March 31, 2011. Considerable judgment is necessary to interpret market data and develop estimated fair value. Accordingly, the estimates presented herein are not necessarily indicative of the amounts that we could realize on disposition of the financial instruments. The use of different market assumptions and/or estimation methodologies may have a material effect on estimated fair value amounts.

Our estimates of the fair value of (i) accounts receivable, accounts payable and accrued expenses approximate carrying value due to the relatively short maturity of these instruments; (ii) our publicly-traded debt is based on observable market data; and (iii) our debt that is not traded publicly is based on estimated effective borrowing rates for debt with similar terms, loan to estimated fair value and remaining maturities (the estimated fair value of all our debt was \$1.6 billion at March 31, 2011).

14. Subsequent Events

We began marketing 14 non-strategic hotels for sale during the fourth quarter of 2010 and, as of the date of this filing, we had six of these hotels under contract for sale (including the hotel reported in discontinued operations for the quarter ended March 31, 2011) for estimated gross proceeds of \$114 million.

In April 2011, we agreed to acquire two midtown Manhattan hotels, Royalton and Morgans (282 guest rooms, in total), for \$140.0 million. We expect to close this transaction in the second quarter of 2011.

In April 2011, we received approximately \$159 million of aggregate net proceeds (after underwriting discounts and commissions) from a public offering of 27.6 million shares of our common stock. Proceeds from this offering were used, on a temporary basis, to repay funds drawn on our line of credit.

In April 2011, our wholly-owned subsidiary sold \$525.0 million in aggregate principal amount of its 6.75% senior secured notes due 2019. The transaction is expected to settle on May 10, 2011, at which time, the gross proceeds will be placed in escrow. Funds held in escrow will be released to us when we acquire Royalton and Morgans and FelCor LP assumes the initial issuer's obligations under those notes. We expect net proceeds after initial purchasers' discount and expenses to be approximately \$512 million. If the proceeds are not released from escrow for any reason, the notes will be mandatorily redeemed (the redemption price would be the face amount of the notes, plus accrued interest through redemption).

Between our recent equity and debt offerings, we raised approximately \$671 million in aggregate net proceeds. We expect to use those net proceeds as follows:

• \$140.0 million of the net proceeds from our debt offering will be used to fund our purchase of Royalton and Morgans.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

14. Subsequent Events — (continued)

- We contemplate using a substantial portion of the net proceeds from our recent equity offering to redeem up to approximately \$145 million of our 10% notes during the second quarter (the contractual redemption price is 110% of the face amount (*i.e.*, up to \$159 million), plus accrued but unpaid interest to the redemption date). Since a substantial portion of the net proceeds of our equity offering were used temporarily to repay outstanding borrowings under our line of credit, we will re-draw those funds under our line of credit to pay the aggregate redemption price.
- We will use a portion of the net proceeds from our debt offering to repay funds drawn under our line of credit (at March 31, 2011, we had \$145.0 million outstanding). As a consequence, our full \$225 million line of credit will be available for general corporate purposes, including repayment of other debt and to fund future acquisitions.
- We will use \$46.4 million of those net proceeds to repay the remainder of our 9% senior secured notes due 2011 when they mature in the second quarter.
- The remaining net proceeds from the two offerings will be used for general corporate purposes, including repayment of higher-cost debt (which has the dual effect of extending maturity by as much as five years and reducing our weighted average cost of debt) and future acquisitions that will help build long-term stockholder value.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

General

Beginning in 2010, a combination of improved demand and historically low new lodging supply growth fueled a demand-driven recovery. As the recovery moved into the first quarter of 2011, occupancy continued to grow with increased vigor and moderate supply growth continued. The continued supply and demand imbalance drives improved average daily rate, or ADR, for us and the industry. ADR growth indicates a second critical stage of the lodging recovery.

Our hotels are taking advantage of the growth in corporate and premium segments to remix their customer base and replace lower-rated business with premium customers. Additionally, our hotels are increasing rates, where appropriate. Our first quarter of 2011 revenue per available room, or RevPAR, improved 6.3%, compared to the same period in the prior year. The improved RevPAR was driven by improvements in ADR (4.0%) and occupancy (2.3%). As ADR becomes a larger contributor to RevPAR growth, Hotel EBITDA margin generally improves because we are charging more for the same room. In the first quarter of 2011, Hotel EBITDA margin improved by 198 basis points, compared to the same period last year. We expect ADR will contribute significantly to RevPAR growth and improved Hotel EBITDA margin through 2011.

As part of our long-term strategic plan:

- We began marketing 14 non-strategic hotels for sale during the fourth quarter of 2010 and, as of the date of this filing, we had six of these hotels under contract for sale (including the hotel reported in discontinued operations for the quarter ended March 31, 2011) for estimated gross proceeds of \$114 million.
- In April 2011, we agreed to acquire two midtown Manhattan hotels, Royalton and Morgans (282 guest rooms, in total), for \$140.0 million. We expect to close this transaction in the second quarter of 2011.
- In April 2011, we received approximately \$159 million of aggregate net proceeds (after underwriting discounts and commissions) from a public offering of 27.6 million shares of our common stock. Proceeds from this offering were used, on a temporary basis, to repay funds drawn on our line of credit.
- In April 2011, our wholly-owned subsidiary sold \$525.0 million in aggregate principal amount of its 6.75% senior secured notes due 2019. The transaction is expected to settle on May 10, 2011, at which time, the the gross proceeds will be placed in escrow. Funds held in escrow will be released to us when we acquire Royalton and Morgans and FelCor LP assumes the initial issuer's obligations under those notes. We expect net proceeds after initial purchasers' discount and expenses to be approximately \$512 million. If the proceeds are not released from escrow for any reason, the notes will be mandatorily redeemed (the redemption price would be the face amount of the notes, plus accrued interest through redemption).
- Between our recent equity and debt offerings, we raised approximately \$671 million in aggregate net proceeds. We expect to use those net proceeds as follows:
 - \$140.0 million of the net proceeds from our debt offering will be used to fund our purchase of Royalton and Morgans.

- We contemplate using a substantial portion of the net proceeds from our recent equity offering to redeem up to approximately \$145 million of our 10% notes during the second quarter (the contractual redemption price is 110% of the face amount (*i.e.*, up to \$159 million), plus accrued but unpaid interest to the redemption date). Since a substantial portion of the net proceeds of our equity offering were used temporarily to repay outstanding borrowings under our line of credit, we will re-draw those funds under our line of credit to pay the aggregate redemption price.
- We will use a portion of the net proceeds from our debt offering to repay funds drawn under our line of credit (at March 31, 2011, we had \$145.0 million outstanding). As a consequence, our full \$225 million line of credit will be available for general corporate purposes, including repayment of other debt and to fund future acquisitions.
- We will use \$46.4 million of those net proceeds to repay the remainder of our 9% senior secured notes due 2011 when they mature in the second quarter.
- The remaining net proceeds from the two offerings will be used for general corporate purposes, including repayment of higher-cost debt (which has the dual effect of extending maturity by as much as five years and reducing our weighted average cost of debt) and future acquisitions that will help build long-term stockholder value.

In January 2011, we reinstated our current quarterly preferred dividend and paid current quarterly dividends in January 2011. In March 2011, we declared current quarterly dividends. We are unable to pay any common dividends unless and until all accrued and current preferred dividends are paid. Our Board of Directors will determine whether and when to declare future dividends (including the accrued but unpaid preferred dividends) based upon various factors, including operating results, economic conditions, other operation trends, our financial condition and capital requirements, as well as minimum REIT distribution requirements.

Results of Operations

Comparison of the Three Months ended March 31, 2011 and 2010

For the three months ended March 31, 2011, we recorded a \$41.3 million net loss attributable to common stockholders, compared to a \$72.1 million net loss attributable to common stockholders for the same period in 2010. (In the first quarter of 2010, we recorded a \$21.1 million impairment.)

In the first quarter of 2011:

- *Total revenue* was \$235.3 million, an increase of \$18.2 million, or 8.4%, compared to the same period in 2010. The increased revenue principally reflects a 6.3% increase in same-store RevPAR, which includes a 2.3% increase in occupancy and a 4.0% increase in ADR. The Fairmont Copley Plaza, which we acquired in August 2010, contributed \$6.7 million.
- Hotel departmental expenses increased \$7.8 million compared to the same period in 2010 reflecting higher occupancies and \$4.6 million of hotel departmental expenses from the Fairmont Copley Plaza. As a percentage of total revenue, hotel departmental expenses increased from 35.8% to 36.3%, compared to the same period in 2010. The changes in departmental expenses as a percent of revenue are primarily due to the mix and nature of the business of the Fairmont Copley Plaza, which receives a significant portion of its revenue from food and beverage operations. Food and beverage revenue generally has much higher expenses as a percent of revenue.
- *Other property related costs* increased \$6.8 million reflecting higher occupancies, as well as \$2.6 million of other property related costs at the Fairmont Copley Plaza. As a percentage of total revenue, other property related costs remained essentially flat compared to the same period in 2010.
- *Management and franchise fees* increased \$797,000 compared to the same period in 2010 reflecting higher revenues (which serve as the basis for determining such fees), including \$201,000 of management and franchise fees from the Fairmont Copley Plaza. As a percent of total revenue, management and franchise fees remained essentially unchanged from the same period in 2010.
- *Taxes, insurance and lease expense* decreased \$1.7 million compared to the same period in 2010. As a percent of total revenue, taxes, insurance and lease expense decreased to 8.8% compared to 10.3% for the same period in 2010. Decreases in insurance, estimated Canadian taxes and favorable resolution of property tax appeals were partially offset by \$721,000 of incremental expenses at the Fairmont Copley Plaza.
- Depreciation and amortization expense decreased \$967,000 compared to the same period in 2010. As a percent of total revenue, depreciation and amortization expense decreased to 15.0% compared to 16.7% for the same period in 2010. Our asset values, the basis from which we calculate depreciation, declined between the end of the first quarter of 2010 and the first quarter of 2011 as a result of impairment charges recorded in late 2010. As a consequence, our first quarter depreciation expense declined from 2010 to 2011. However, this decline was partially offset by \$691,000 of depreciation expense related to the Fairmont Copley Plaza, which we did not own in the first quarter of 2010.
- *Net interest expense* decreased \$1.6 million, compared to the same period in 2010. This decrease primarily reflects a decrease in our average debt balances during those periods.
- *Discontinued operations* in 2010 reflect the \$21.1 million impairment charge (noted above) related to two hotels transferred to lenders in 2010 in full satisfaction of the related debt.

Non-GAAP Financial Measures

We refer in this report to certain "non-GAAP financial measures." These measures, including FFO, Adjusted FFO, EBITDA, Adjusted EBITDA, Hotel EBITDA, and Hotel EBITDA margin, are measures of our financial performance that are not calculated and presented in accordance with GAAP. The following tables reconcile these non-GAAP measures to the most comparable GAAP financial measure. Immediately following the reconciliations, we include a discussion of why we believe these measures are useful supplemental measures of our performance and the limitations of such measures.

The following table details our computation of FFO and Adjusted FFO (in thousands, except for per share data):

	Three Months Ended March 31,						
		2011					
	Dollars	Shares	Per Share Amount	Dollars	Shares	Per Share Amount	
Net loss	\$ (31,726)			\$ (62,942)			
Noncontrolling interests	62			554			
Preferred dividends	(9,678)			(9,678)			
Numerator for basic and diluted loss attributable to common stockholders	(41,342)	95,350	(0.43)	(72,066)	63,475	(1.14)	
Depreciation and amortization	35,317		0.37	36,284		0.57	
Depreciation, discontinued operations and unconsolidated entities	3,581		0.04	4,977		0.08	
Gain on sale of unconsolidated entities	_	_	_	(559)	_	(0.01)	
Noncontrolling interests in FelCor LP	(120)	285	(0.01)	(325)	295		
Gain on involuntary conversion	(150)			—			
FFO	(2,714)	95,635	(0.03)	(31,689)	63,770	(0.50)	
Impairment loss, discontinued operations and unconsolidated entities	_		_	21,060		0.33	
Acquisition costs	119					_	
Extinguishment of debt, including discontinued operations	252		0.01		_		
Adjusted FFO	\$ (2,343)	95,635	\$ (0.02)	\$ (10,629)	63,770	\$ (0.17)	

Reconciliation of Net Loss to FFO and Adjusted FFO

(in thousands, except per share data)

The following table details our computation of EBITDA, Adjusted EBITDA, and Hotel EBITDA (in thousands):

Reconciliation of Net Loss to EBITDA, Adjusted EBITDA, and Hotel EBITDA (in thousands)

	Three Months Ended March 31,			
		2011	2010	
Net loss	\$	(31,726)	\$ (62,942)	
Depreciation and amortization		35,317	36,284	
Depreciation, discontinued operations and unconsolidated entities		3,581	4,977	
Interest expense		33,806	35,508	
Interest expense, discontinued operations and unconsolidated entities		1,209	2,337	
Amortization of stock compensation		1,803	1,616	
Noncontrolling interests in other partnerships		(58)	229	
EBITDA		43,932	18,009	
Impairment loss, discontinued operations and unconsolidated entities			21,060	
Extinguishment of debt, including discontinued operations		252		
Acquisition costs		119		
Gain on involuntary conversion		(150)		
Gain on sale of unconsolidated subsidiary			(559)	
Adjusted EBITDA		44,153	38,510	
Other revenue		(225)	(365)	
Adjusted EBITDA from acquired hotels			(2,078)	
Equity in income from unconsolidated subsidiaries (excluding interest and depreciation)		(3,341)	(2,984)	
Noncontrolling interests in other partnerships (excluding interest and depreciation)		626	392	
Consolidated hotel lease expense		8,304	7,758	
Unconsolidated taxes, insurance and lease expense		(1,684)	(1,692)	
Interest income		(41)	(105)	
Other expenses (excluding acquisition costs)		512	561	
Corporate expenses (excluding amortization expense of stock compensation)		7,734	8,231	
Adjusted EBITDA from disposed hotels		(857)	(380)	
Hotel EBITDA	\$	55,181	\$ 47,848	

The following tables detail our computation of Hotel EBITDA, Hotel EBITDA margin, hotel operating expenses and the reconciliation of hotel operating expenses to total operating expenses with respect to 80 of our Consolidated Hotels at the dates presented.

Hotel EBITDA and Hotel EBITDA Margin

(dollars in thousands)

	Th	Three Months Ended March 31,					
		2011	2010				
Total revenues	\$	235,330	\$	217,123			
Other revenue		(225)		(365)			
Hotel operating revenue		235,105		216,758			
Add: revenue from acquired hotels				5,855			
Same-store hotel operating revenue		235,105		222,613			
Same-store hotel operating expenses		(179,924)		(174,765)			
Hotel EBITDA	\$	55,181	\$	47,848			
Hotel EBITDA margin ^(a)		23.5%		21.5%			

(a) Hotel EBITDA as a percentage of same-store hotel operating revenue.

Reconciliation of Total Operating Expenses to Same-Store Hotel Operating Expenses (in thousands)

	Thr	Three Months Ended March 31,			
		2011		2010	
Total operating expenses	\$	232,029	\$	219,590	
Unconsolidated taxes, insurance and lease expense		1,684		1,692	
Consolidated hotel lease expense		(8,304)		(7,758)	
Corporate expenses		(9,537)		(9,847)	
Depreciation and amortization		(35,317)		(36,284)	
Other expenses		(631)		(561)	
Acquired hotel expenses				7,933	
Hotel operating expenses	\$	179,924	\$	174,765	

Reconciliation of Ratio of Operating Income (Loss) to Total Revenues to Hotel EBITDA Margin

		Three Months Ended March 31,		
	2011	2010		
Ratio of operating income (loss) to total revenues	1.4%	(1.1)%		
Other revenue	(0.1)	(0.2)		
Acquired hotel revenue		2.7		
Unconsolidated taxes, insurance and lease expense	(0.7)	(0.8)		
Consolidated lease expense	3.5	3.5		
Other expenses	0.3	0.3		
Corporate expenses	4.1	4.4		
Depreciation and amortization	15.0	16.3		
Acquired hotel expenses		(3.6)		
Hotel EBITDA margin	23.5%	21.5 %		

Substantially all of our non-current assets consist of real estate. Historical cost accounting for real estate assets implicitly assumes that the value of real estate assets diminishes predictably over time. Since real estate values instead have historically risen or fallen with market conditions, most industry investors consider supplemental measures of performance, which are not measures of operating performance under GAAP, to be helpful in evaluating a real estate company's operations. These supplemental measures are not measures of operating performance under GAAP. However, we consider these non-GAAP measures to be supplemental measures of a hotel REIT's performance and should be considered along with, but not as an alternative to, net income (loss) attributable to FelCor as a measure of our operating performance.

FFO and EBITDA

The White Paper on Funds From Operations approved by the Board of Governors of the National Association of Real Estate Investment Trusts ("NAREIT"), defines FFO as net income or loss attributable to parent (computed in accordance with GAAP), excluding gains or losses from sales of property, plus depreciation and amortization, and after adjustments for unconsolidated partnerships and joint ventures. Adjustments for unconsolidated partnerships and joint ventures are calculated to reflect FFO on the same basis. We compute FFO in accordance with standards established by NAREIT. This may not be comparable to FFO reported by other REITs that do not define the term in accordance with the current NAREIT definition or that interpret the current NAREIT definition differently than we do.

EBITDA is a commonly used measure of performance in many industries. We define EBITDA as net income or loss attributable to parent (computed in accordance with GAAP) plus interest expenses, income taxes, depreciation and amortization, and after adjustments for unconsolidated partnerships and joint ventures. Adjustments for unconsolidated partnerships and joint ventures are calculated to reflect EBITDA on the same basis.

Adjustments to FFO and EBITDA

We adjust FFO and EBITDA when evaluating our performance because management believes that the exclusion of certain additional items, including but not limited to those described below, provides useful supplemental information to investors regarding our ongoing operating performance and that the presentation of Adjusted FFO, and Adjusted EBITDA when combined with GAAP net income attributable to FelCor, EBITDA and FFO, is beneficial to an investor's better understanding of our operating performance.

- *Gains and losses related to extinguishment of debt and interest rate swaps* We exclude gains and losses related to extinguishment of debt and interest rate swaps from FFO and EBITDA because we believe that it is not indicative of ongoing operating performance of our hotel assets. This also represents an acceleration of interest expense or a reduction of interest expense, and interest expense is excluded from EBITDA.
- *Impairment losses* We exclude the effect of impairment losses and gains or losses on disposition of assets in computing Adjusted FFO and Adjusted EBITDA because we believe that including these is not consistent with reflecting the ongoing performance of our remaining assets. Additionally, we believe that impairment charges and gains or losses on disposition of assets represent accelerated depreciation, or excess depreciation, and depreciation is excluded from FFO by the NAREIT definition and from EBITDA.
- *Cumulative effect of a change in accounting principle* Infrequently, the Financial Accounting Standards Board promulgates new accounting standards that require the consolidated statements of operations to reflect the cumulative effect of a change in accounting principle. We exclude these one-time adjustments in computing Adjusted FFO and Adjusted EBITDA because they do not reflect our actual performance for that period.

In addition, to derive Adjusted EBITDA we exclude gains or losses on the sale of depreciable assets because we believe that including them in EBITDA is not consistent with reflecting the ongoing performance of our remaining assets. Additionally, the gain or loss on sale of depreciable assets represents either accelerated depreciation or excess depreciation in previous periods, and depreciation is excluded from EBITDA.

Hotel EBITDA and Hotel EBITDA Margin

Hotel EBITDA and Hotel EBITDA margin are commonly used measures of performance in the hotel industry and give investors a more complete understanding of the operating results over which our individual hotels and operating managers have direct control. We believe that Hotel EBITDA and Hotel EBITDA margin are useful to investors by providing greater transparency with respect to two significant measures used by us in our financial and operational decision-making. Additionally, using these measures facilitates comparisons with other hotel REITs and hotel owners. We present Hotel EBITDA and Hotel EBITDA margin by eliminating from continuing operations all revenues and expenses not directly associated with hotel operations including but not limited to corporate-level expenses; impairment losses; gains or losses on disposition of assets; and gains and losses related to extinguishment of debt. We eliminate corporate-level costs and expenses because we believe property-level results provide investors with supplemental information into the ongoing operational performance of our hotels and the effectiveness of management on a property-level basis. We exclude the effect of impairment losses, gains or losses on disposition of assets, and gains or losses related to extinguishment of debt because we believe that including these is not consistent with reflecting the ongoing performance of our remaining assets. We also eliminate consolidated percentage rent paid to unconsolidated entities, which is effectively eliminated by noncontrolling interests and equity in income from unconsolidated subsidiaries, and include the cost of unconsolidated taxes, insurance and lease expense, to reflect the entire operating costs applicable to our hotels. Hotel EBITDA and Hotel EBITDA margins are presented on a same-store basis.

Limitations of Non-GAAP Measures

Our management and Board of Directors use Hotel EBITDA and Hotel EBITDA margin to evaluate the performance of our hotels and to facilitate comparisons between us and other hotel owners, in evaluating hotel-level performance and the operating efficiency of our hotel managers.

The use of these non-GAAP financial measures has certain limitations. Hotel EBITDA and Hotel EBITDA margin, as presented by us, may not be comparable to these measures as calculated by other companies. These measures do not reflect certain expenses that we incurred and will incur, such as depreciation and amortization, interest and capital expenditures. Management compensates for these limitations by separately considering the impact of these excluded items to the extent they are material to operating decisions or assessments of our operating performance. Our reconciliations to the most comparable GAAP financial measures, and our consolidated statements of operations and cash flows, include interest expense, capital expenditures, and other excluded items, all of which should be considered when evaluating our performance, as well as the usefulness of our non-GAAP financial measures.

These non-GAAP financial measures are used in addition to and in conjunction with results presented in accordance with GAAP. They should not be considered as alternatives to operating profit, cash flow from operations, or any other operating performance measure prescribed by GAAP. Hotel EBITDA and Hotel EBITDA margin reflect additional ways of viewing our operations that we believe, when viewed with our GAAP results and the reconciliations to the corresponding GAAP financial measures, provide a more complete understanding of factors and trends affecting our business than could be obtained absent this disclosure. We strongly encourage investors to review our financial information in its entirety and not to rely on a single financial measure.

Pro Rata Share of Rooms Owned

The following table sets forth, at March 31, 2011, our pro rata share of hotel rooms, included in continuing operations, after giving consideration to the portion of rooms attributed to our partners in our consolidated and unconsolidated joint ventures:

	Hotels	Room Count at March 31, 2011
Consolidated Hotels	80	23,112
Unconsolidated hotel operations	1	171
Total hotels	81	23,283
50% joint ventures	13	(1,573)
60% joint venture	1	(214)
82% joint venture	1	(40)
90% joint ventures	3	(68)
Pro rata rooms attributed to joint venture partners		(1,895)
Pro rata share of rooms owned		21,388

Hotel Portfolio Composition

The following table illustrates the distribution of 80 Consolidated Hotels in continuing operations by brand, market and location at March 31, 2011.

Brand	Hotels	Rooms	% of Total	% of 2010 Hotel EBITDA ^(a)
Embassy Suites Hotels	44	11,450	50	58
Holiday Inn	15	5,154	22	18
Sheraton and Westin	8	2,774	12	9
Doubletree	7	1,471	6	7
Renaissance and Marriott	3	1,321	6	3
Hilton	2	559	2	3
Fairmont	1	383	2	2
Market				
South Florida	5	1,439	6	7
Los Angeles area	4	899	4	6
San Francisco area	6	2,138	9	6
Atlanta	5	1,462	6	6
Dallas	4	1,333	б	5
Boston	3	915	4	5
Minneapolis	3	736	3	4
Philadelphia	2	729	3	4
Orlando	4	1,038	5	4
Central California Coast	2	408	2	4
Myrtle Beach	2	640	3	4
New Orleans	2	744	3	4
San Antonio	3	874	4	3
San Diego	1	600	3	3
Other	34	9,157	39	35
Location				
Urban	21	6,741	29	32
Suburban	31	7,656	33	29
Airport	18	5,788	25	23
Resort	10	2,927	13	16

(a) Hotel EBITDA is a non-GAAP financial measure. A detailed reconciliation and further discussion of Hotel EBITDA is contained in the "Non-GAAP Financial Measures" section of Management's Discussion and Analysis of Financial Condition and Results of Operations in Item 2 of this Quarterly Report on Form 10-Q.

Hotel Operating Statistics

The following tables set forth occupancy, ADR and RevPAR for the three months ended March 31, 2011 and 2010, and the percentage changes therein for the periods presented, for our 80 Consolidated Hotels included in continuing operations.

Operating Statistics by Brand

		Occupancy (%)			
	Three Mon Marc	nths Ended ch 31,			
	2011	2010	%Variance		
Embassy Suites Hotels	72.5	70.8	2.3		
Holiday Inn	68.0	67.6	0.6		
Sheraton and Westin	66.7	64.0	4.3		
Doubletree	71.4	70.0	1.9		
Renaissance and Marriott	71.0	65.3	8.6		
Hilton	42.5	46.2	(8.0)		
Fairmont	53.0	51.7	2.6		
Total hotels	69.6	68.0	2.3		

		ADR (\$)			
		nths Ended ch 31,			
	2011	2010	%Variance		
Embassy Suites Hotels	130.49	129.42	0.8		
Holiday Inn	110.89	104.30	6.3		
Sheraton and Westin	109.51	103.71	5.6		
Doubletree	131.94	118.75	11.1		
Renaissance and Marriott	196.66	183.84	7.0		
Hilton	98.10	95.75	2.4		
Fairmont	199.71	174.05	14.7		
Total hotels	127.88	123.02	4.0		

		RevPAR (\$)			
		nths Ended ch 31,			
	2011	2011 2010			
Embassy Suites Hotels	94.57	91.66	3.2		
Holiday Inn	75.41	70.52	6.9		
Sheraton and Westin	73.07	66.37	10.1		
Doubletree	94.15	83.12	13.3		
Renaissance and Marriott	139.54	120.08	16.2		
Hilton	41.65	44.21	(5.8)		
Fairmont	105.82	89.91	17.7		
Total hotels	88.97	83.67	6.3		

Operating Statistics for Our Top Markets

		%Variance
		(2.3)
		4.7
68.3	65.3	4.6
73.9	75.2	(1.8)
72.9	65.4	11.4
72.7	67.0	8.4
57.8	60.4	(4.3)
82.9	80.9	2.4
68.6	69.7	(1.6)
40.8	44.1	(7.5)
70.0	68.7	1.8
68.6	66.5	3.2
73.9	74.7	(1.0)
73.8	71.5	3.2
	ADR (\$)	
Three Months I	. ,	
2011	2010	%Variance
		(3.4)
		4.5
		9.3
		0.6
		9.4
		(2.6)
		11.4
		3.6
		(3.1)
		2.5
		8.2
		6.7
		(3.2)
122.03		6.0
Three Months I		
	,	0/37
		%Variance
		(5.6)
		9.4
		14.3
		(1.2)
		21.9
		5.6
		6.6
		6.1
		(4.7)
		(5.2)
100.32	91.04	10.2
100.72	91.52	10.1
70.39	73.46	(4.2)
90.08	82.33	9.4
	2011 83.2 73.8 68.3 73.9 72.9 72.7 57.8 82.9 68.6 40.8 70.0 68.6 40.8 70.0 68.6 73.9 73.8 Three Months I 2011 158.05 138.26 134.09 106.06 123.63 122.52 124.14 118.54 133.87 98.75 143.29 146.90 95.21 122.03 Three Months I 2011 131.51 101.99 91.53 78.40 90.09 89.01 71.77 98.27 91.81 40.31 100.32 100.72	83.2 85.1 73.8 70.5 68.3 65.3 73.9 75.2 72.9 65.4 72.7 67.0 57.8 60.4 82.9 80.9 68.6 69.7 40.8 44.1 70.0 68.7 68.6 66.5 73.9 74.7 73.8 71.5 ADR (\$) Three Months Ended March 31, 2011 2010 158.05 163.64 138.26 132.32 134.09 122.73 106.06 105.48 123.63 112.99 122.52 125.73 124.14 111.42 118.54 114.47 133.87 138.16 98.75 96.37 143.29 132.43 146.90 137.72 95.21 98.33 122.03 115.09 Coll 2010

Hotel Portfolio

The following table sets forth certain descriptive information regarding the hotels in which we held ownership interest at March 31, 2011.

Consolidated Hotel	Brand	<u>State</u>	<u>Rooms</u>	% Owned (a)
Birmingham	Embassy Suites Hotel	AL	242	
Phoenix – Biltmore	Embassy Suites Hotel	AZ	232	
Phoenix – Crescent	Sheraton	AZ	342	
Anaheim – North	Embassy Suites Hotel	CA	222	
Dana Point – Doheny Beach	Doubletree Guest Suites	CA	196	
Indian Wells – Esmeralda Resort & Spa	Renaissance Resort	CA	560	
Los Angeles – International Airport/South	Embassy Suites Hotel	CA	349	
Milpitas – Silicon Valley	Embassy Suites Hotel	CA	266	
Napa Valley	Embassy Suites Hotel	CA	205	
Oxnard – Mandalay Beach – Hotel & Resort	Embassy Suites Hotel	CA	248	
San Diego – On the Bay	Holiday Inn	CA	600	
San Francisco – Airport/Waterfront	Embassy Suites Hotel	CA	340	
San Francisco – Airport/South San Francisco	Embassy Suites Hotel	CA	312	
San Francisco – Fisherman's Wharf	Holiday Inn	CA	585	
San Francisco – Union Square	Marriott	CA	400	
San Rafael – Marin County	Embassy Suites Hotel	CA	235	50%
Santa Barbara – Goleta	Holiday Inn	CA	160	
Santa Monica Beach – at the Pier	Holiday Inn	CA	132	
Wilmington	Doubletree	DE	244	90%
Boca Raton	Embassy Suites Hotel	FL	263	
Deerfield Beach – Resort & Spa	Embassy Suites Hotel	FL	244	
Ft. Lauderdale – 17th Street	Embassy Suites Hotel	FL	361	
Ft. Lauderdale – Cypress Creek	Sheraton Suites	FL	253	
Jacksonville – Baymeadows	Embassy Suites Hotel	FL	277	
Miami – International Airport	Embassy Suites Hotel	FL	318	
Orlando – International Airport	Holiday Inn	FL	288	
Orlando – International Drive South/Convention	Embassy Suites Hotel	FL	244	
Orlando– North	Embassy Suites Hotel	FL	277	
Orlando – Walt Disney World Resort	Doubletree Guest Suites	FL	229	
St. Petersburg – Vinoy Resort & Golf Club	Renaissance Resort	FL	361	
Tampa – Tampa Bay	Doubletree Guest Suites	FL	203	
Atlanta – Airport	Embassy Suites Hotel	GA	232	
Atlanta – Buckhead	Embassy Suites Hotel	GA	316	
Atlanta – Galleria	Sheraton Suites	GA	278	
Atlanta – Gateway – Atlanta Airport	Sheraton	GA	395	
Atlanta – Perimeter Center	Embassy Suites Hotel	GA	241	50%
Chicago – Lombard/Oak Brook	Embassy Suites Hotel	IL	262	50%
Chicago – Gateway – O'Hare	Sheraton Suites	IL	296	0070
Indianapolis – North	Embassy Suites Hotel	IN	221	82%
Kansas City – Overland Park	Embassy Suites Hotel	KS	199	50%
Lexington – Lexington Green	Hilton Suites	KY	174	2070
Baton Rouge	Embassy Suites Hotel	LA	223	
Daton Rouge	Embassy Suites Hotel	LA	223	

Hotel Portfolio (continued)

Consolidated Hotel	<u>Brand</u>	<u>State</u>	<u>Rooms</u>	<u>% Owned</u> (a)
New Orleans – Convention Center	Embassy Suites Hotel	LA	370	
New Orleans – French Quarter	Holiday Inn	LA	374	
Boston – at Beacon Hill	Holiday Inn	MA	303	
Boston – Copley Plaza	Fairmont	MA	383	
Boston – Marlborough	Embassy Suites Hotel	MA	229	
Baltimore – at BWI Airport	Embassy Suites Hotel	MD	251	90%
Bloomington	Embassy Suites Hotel	MN	218	
Minneapolis – Airport	Embassy Suites Hotel	MN	310	
St. Paul – Downtown	Embassy Suites Hotel	MN	208	
Kansas City – Plaza	Embassy Suites Hotel	MO	266	50%
Charlotte	Embassy Suites Hotel	NC	274	50%
Charlotte – SouthPark	Doubletree Guest Suites	NC	208	
Raleigh/Durham	Doubletree Guest Suites	NC	203	
Raleigh – Crabtree	Embassy Suites Hotel	NC	225	50%
Parsippany	Embassy Suites Hotel	NJ	274	50%
Secaucus – Meadowlands	Embassy Suites Hotel	NJ	261	50%
Philadelphia – Historic District	Holiday Inn	PA	364	
Philadelphia – Society Hill	Sheraton	PA	365	
Pittsburgh – at University Center (Oakland)	Holiday Inn	PA	251	
Charleston – Mills House	Holiday Inn	SC	214	
Myrtle Beach – Oceanfront Resort	Embassy Suites Hotel	SC	255	
Myrtle Beach Resort	Hilton	SC	385	
Nashville – Airport – Opryland Area	Embassy Suites Hotel	TN	296	
Nashville – Opryland – Airport (Briley Parkway)	Holiday Inn	TN	383	
Austin	Doubletree Guest Suites	TX	188	90%
Austin – Central	Embassy Suites Hotel	ΤX	260	50%
Corpus Christi	Embassy Suites Hotel	TX	150	
Dallas – DFW International Airport South	Embassy Suites Hotel	ΤX	305	
Dallas – Love Field	Embassy Suites Hotel	TX	248	
Dallas – Market Center	Embassy Suites Hotel	ΤX	244	
Dallas – Park Central	Westin	TX	536	60%
Houston – Medical Center	Holiday Inn	ΤX	287	
San Antonio – International Airport	Embassy Suites Hotel	TX	261	50%
San Antonio – International Airport	Holiday Inn	ΤX	397	
San Antonio – NW I-10	Embassy Suites Hotel	TX	216	50%
Burlington Hotel & Conference Center	Sheraton	VT	309	
Canada				
Foronto – Airport	Holiday Inn	Ontario	446	
Toronto – Yorkdale	Holiday Inn	Ontario	370	
Unconsolidated Hotel				
New Orleans – French Quarter – Chateau LeMoyne	Holiday Inn	LA	171	50%
Hotel Held for Sale				
Phoenix – Tempe	Embassy Suites Hotel	AZ	224	

(a) We own 100% of the real estate interests unless otherwise noted.

Liquidity and Capital Resources

Operating Activities

During the first quarter of 2011, cash provided by operating activities (primarily hotel operations) was \$5.8 million, \$22.7 million less than the same period in 2010. This decrease is due primarily to the 2011 payment of liquidated damages to IHG, increased hotel bonus payments and decreases in other accrued liabilities, which offset increases from hotel operations. At March 31, 2011, we had \$91.0 million of cash and cash equivalents, including approximately \$41.0 million held under management agreements to meet working capital needs.

The lodging recovery that began in 2010 has continued into 2011. ADR and occupancy have improved significantly in the first quarter of 2011 resulting in a 6.3% increase in our RevPAR. We expect our 2011 RevPAR will increase from 6% to 8% compared to 2010, which assumes continued occupancy and ADR growth. We expect to generate \$72 million to \$81 million of cash from operating activities in 2011.

We are subject to increases in hotel operating expenses, including wage and benefit costs, repair and maintenance expenses, utilities and insurance expenses that can fluctuate disproportionately to revenues. Some of these operating expenses are difficult to predict and control, which lends volatility to our operating results. From 2008 to 2010, we implemented extensive cost containment initiatives at our hotels, including reducing headcount and improving productivity and energy efficiency. If RevPAR decreases and/or Hotel EBITDA margins shrink, our operations, earnings and/or cash flow could be materially adversely affected.

Investing Activities

During the first three months of 2011, cash used in investing activities increased \$7.7 million, compared to the same period in 2010, due primarily to increased capital expenditures made in 2011. In the first quarter of 2011, we completed approximately \$15.0 million of capital improvements at our hotels, and we expect to spend approximately \$45 million in non-discretionary capital and \$35 million to \$40 million in discretionary capital in 2011 (including more than \$20 million at the Fairmont Copley Plaza), which investments will be funded from operating cash flow, cash on hand or draws under our line of credit.

We agreed to acquire two midtown Manhattan hotels, the Royalton and Morgans hotels (282 guest rooms in total), for \$140.0 million. The hotels require limited initial capital (both hotels are in excellent condition and are recently renovated). We expect to close this transaction in the second quarter of 2011, with funds from the recent sale of senior notes.

As part of our strategic plan, we intend to sell non-strategic hotels that do not meet our investment criteria, thereby freeing our capital for redeployment (*e.g.*, reduce overall leverage, acquire other hotels and invest in remaining FelCor properties that generate a higher return on invested capital). We began marketing 14 hotels for sale during the fourth quarter of 2010, and we expect to sell a majority of these hotels during 2011. We are currently under contract to sell six of the 14 hotels for total gross proceeds of approximately \$114 million. We have also identified an additional 21 non-strategic hotels. We will continue to monitor the transaction environment and will bring these additional hotels to market at the appropriate time.

Financing Activities

During the first quarter of 2011, cash used in financing activities increased by \$91.8 million compared to the same period in 2010, due primarily to repayment of \$227 million of secured debt using cash on hand and funds drawn under our new line of credit, and payment of preferred dividends in 2011. We expect to pay approximately \$8 million in normally occurring principal payments and \$39 million in preferred dividends in 2011, which payments will be funded from operating cash flow, cash on hand or draws under our line of credit. We expect to repay, refinance or extend the debt maturing in 2011.

In January 2011, we reinstated our current quarterly preferred dividend and paid current quarterly dividends in January 2011. In March 2011, we declared current quarterly dividends. We are unable to pay any common dividends unless and until all accrued and current preferred dividends are paid. Our Board of Directors will determine whether and when to declare future dividends (including the accrued but unpaid preferred dividends) based upon various factors, including operating results, economic conditions, other operation trends, our financial condition and capital requirements, as well as minimum REIT distribution requirements.

Senior Secured Note Offering. In April 2011, our wholly-owned subsidiary sold \$525.0 million in aggregate principal amount of its 6.75% senior secured notes due 2019. The transaction is expected to settle on May 10, 2011, at which time, the gross proceeds will be placed in escrow. Funds held in escrow will be released to us when we acquire Royalton and Morgans and FelCor LP assumes the initial issuer's obligations under those notes. We expect net proceeds after initial purchasers' discount and expenses to be approximately \$512 million. If the proceeds are not released from escrow for any reason, the notes will be mandatorily redeemed (the redemption price would be the face amount of the notes, plus accrued interest through redemption).

Common Stock Offering. In April 2011, we received approximately \$159 million of aggregate net proceeds (after underwriting discounts and commissions) from a public offering of 27.6 million shares of our common stock. Proceeds from this offering were used, on a temporary basis, to repay funds drawn on our line of credit.

Between our recent equity and debt offerings, we raised approximately \$671 million in aggregate net proceeds. We expect to use those net proceeds as follows:

- \$140.0 million of the net proceeds from our debt offering will be used to fund our purchase of Royalton and Morgans.
- We contemplate using a substantial portion of the net proceeds from our recent equity offering to redeem up to approximately \$145 million of our 10% notes during the second quarter (the contractual redemption price is 110% of the face amount (*i.e.*, up to \$159 million), plus accrued but unpaid interest to the redemption date). Since a substantial portion of the net proceeds of our equity offering were used temporarily to repay outstanding borrowings under our line of credit, we will re-draw those funds under our line of credit to pay the aggregate redemption price.
- We will use a portion of the net proceeds from our debt offering to repay funds drawn under our line of credit (at March 31, 2011, we had \$145.0 million outstanding). As a consequence, our full \$225 million line of credit will be available for general corporate purposes, including repayment of other debt and to fund future acquisitions.

- We will use \$46.4 million of those net proceeds to repay the remainder of our 9% senior secured notes due 2011 when they mature in the second quarter.
- The remaining net proceeds from the two offerings will be used for general corporate purposes, including repayment of higher-cost debt (which has the dual effect of extending maturity by as much as five years and reducing our weighted average cost of debt) and future acquisitions that will help build long-term stockholder value.

Line of credit. In March 2011, we closed a \$225 million secured, revolving line of credit. At closing, we repaid two secured loans, totaling \$198.3 million and \$28.8 million, with a combination of \$52.1 million of cash on hand and funds drawn under the new line of credit. The repaid loans would have matured in 2013 and 2012 (including extensions), respectively, and were secured by mortgages on 11 hotels. Those same hotels now secure repayment of amounts outstanding under the line of credit. The credit facility bears interest equal to LIBOR, plus 4.5%, with no LIBOR floor.

Secured Debt. At March 31, 2011, we had a total of \$1.5 billion of consolidated secured debt (including \$145 million drawn on our line of credit) with 61 encumbered consolidated hotels with a \$1.5 billion aggregate net book value (including 14 hotels that are collateral for our senior secured notes).

Except in the case of our senior secured notes, our mortgage debt is generally recourse solely to the specific hotels securing the debt except in case of fraud, misapplication of funds and certain other limited recourse carve-out provisions, which could extend recourse to us. Much of our secured debt allows us to substitute collateral under certain conditions and is prepayable, subject to various prepayment, yield maintenance or defeasance obligations.

Most of our secured debt (other than our senior secured notes) includes lock-box arrangements under certain circumstances. We are permitted to spend an amount required to cover our budgeted hotel operating expenses, taxes, debt service, insurance and capital expenditure reserves even if revenues are flowing through a lock-box in cases where a specified debt service coverage ratio is not met. With the exception of one hotel, all of our consolidated loans subject to lock-box provisions currently exceed the applicable minimum debt service coverage ratios.

Senior Notes. Our senior notes require that we satisfy total leverage, secured leverage and interest coverage tests in order to: (i) incur additional indebtedness, except to refinance maturing debt with replacement debt, as defined under our indentures; (ii) pay dividends in excess of the minimum distributions required to meet the REIT qualification test; (iii) repurchase capital stock; or (iv) merge. At March 31, 2011, we exceeded the relevant minimum thresholds. These notes are guaranteed by us, and payment of our 10% notes are secured by a pledge of the limited partner interests in FelCor LP owned by FelCor. In addition, payment of our senior notes are secured by combinations of first lien mortgages and related security interests and/or negative pledges on up to 14 hotels (for our 10% notes) and up to six hotels (for our 6.75% notes), and pledges of equity interests in certain subsidiaries of FelCor LP.

Interest Rate Caps. To fulfill requirements under certain loans, we entered into interest rate cap agreements with aggregate notional amounts of \$639.2 million at March 31, 2011 and December 31, 2010. These interest rate caps were not designated as hedges and had insignificant fair values at both March 31, 2011 and December 31, 2010, resulting in no significant net earnings impact.

Inflation

Operators of hotels, in general, possess the ability to adjust room rates daily to reflect the effects of inflation. Competition may, however, require us to reduce room rates in the near term and may limit our ability to raise room rates in the future. We are also subject to the risk that inflation will cause increases in hotel operating expenses disproportionately to revenues. If competition requires us to reduce room rates or limits our ability to raise room rates in the future, we may not be able to adjust our room rates to reflect the effects of inflation in full, in which case our operating results and liquidity could be adversely affected.

Seasonality

The lodging business is seasonal in nature. Generally, hotel revenues are greater in the second and third calendar quarters than in the first and fourth calendar quarters, although this may not be true for hotels in major tourist destinations. Revenues for hotels in tourist areas generally are substantially greater during tourist season than other times of the year. Seasonal variations in revenue at our hotels can be expected to cause quarterly fluctuations in our revenues. Quarterly earnings also may be adversely affected by events beyond our control, such as extreme weather conditions, economic factors and other considerations affecting travel. To the extent that cash flow from operations is insufficient during any quarter, due to temporary or seasonal fluctuations in revenues, we may utilize cash on hand or borrowings to satisfy our obligations.

Disclosure Regarding Forward-Looking Statements

This report and the documents incorporated by reference in this report include forward-looking statements that involve a number of risks and uncertainties. Forward-looking statements can be identified by the use of forward-looking terminology, such as "believes," "expects," "anticipates," "may," "will," "should," "seeks," or other variations of these terms (including their use in the negative), or by discussions of strategies, plans or intentions. A number of factors could cause actual results to differ materially from those anticipated by these forward-looking statements. Certain of these risks and uncertainties are described in greater detail under "Risk Factors" in our Annual Report on Form 10-K or in our other filings with the Securities and Exchange Commission, or the SEC.

These forward-looking statements are necessarily dependent upon assumptions and estimates that may prove to be incorrect. Accordingly, while we believe that the plans, intentions and expectations reflected in these forward-looking statements are reasonable, we cannot assure you that deviations from these plans, intentions or expectations will not be material. The forward-looking statements included in this report, and all subsequent written and oral forward-looking statements attributable to us or persons acting on our behalf, are expressly qualified in their entirety by the risk factors and cautionary statements discussed in our filings to the SEC. We undertake no obligation to publicly update any forward-looking statements to reflect future circumstances or changes in our expectations.

Item 3. Quantitative and Qualitative Disclosures about Market Risk.

At March 31, 2011, approximately 59% of our consolidated debt had fixed interest rates.

The following table provides information about our financial instruments that are sensitive to changes in interest rates. For debt obligations, the table presents scheduled maturities and weighted average interest rates, by maturity dates. The fair value of our fixed rate debt indicates the estimated principal amount of debt having the same debt service requirements that could have been borrowed at the date presented, at then current market interest rates.

Expected Maturity Date at March 31, 2011 (dollars in thousands)

	Expected Maturity Date								
	2011	2012	2013	2014	2015	Th	ereafter	Total	Fair Value
Liabilities									
Fixed-rate:									
Debt	\$ 58,237	\$ 4,566	\$32,712	\$804,775	\$ 564	\$	8,813	\$ 909,667	\$1,023,223
Average interest rate	8.56%	7.68%	8.61%	9.61%	5.81%		5.81%	9.45%	
Floating-rate:									
Debt	251,706	1,832	1,985	147,153	204,887			607,563	\$ 604,251
Average interest rate ^(a)	1.34%	8.10%	8.10%	7.93%	9.22%		_	5.64%	
Total debt	\$309,943	\$ 6,398	\$34,697	\$951,928	\$ 205,451	\$	8,813	\$ 1,517,230	
Average interest rate	2.70%	7.80%	8.58%	9.35%	9.21%		5.81%	7.93%	
Net discount								(50,432)	
Total debt								\$ 1,466,798	

(a) The average floating interest rate represents the implied forward rates in the yield curve at March 31, 2011.

Item 4. Controls and Procedures.

(a) Evaluation of disclosure controls and procedures.

Under the supervision and with the participation of our management, including our chief executive officer and chief financial officer, we conducted an evaluation of the effectiveness of the design and operation of our disclosure controls and procedures (as defined in Rule 13a-15(e) promulgated under the Securities Exchange Act of 1934) as of the end of the period covered by this report (the "Evaluation Date"). Based on this evaluation, our chief executive officer and chief financial officer concluded, as of the Evaluation Date, that our disclosure controls and procedures were effective, such that the information relating to us required to be disclosed in our reports is recorded, processed, summarized and reported within the time periods specified in SEC rules and forms, and is accumulated and communicated to our management, including our chief executive officer, as appropriate, to allow timely decisions regarding required disclosures.

(b) Changes in internal control over financial reporting.

There have not been any changes in our internal control over financial reporting (as defined in Rule 13a-15 (f) promulgated under the Securities Exchange Act of 1934) during the quarter covered by this report that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II -- OTHER INFORMATION

Item 6. Exhibits.

The following exhibits are furnished in accordance with the provisions of Item 601 of Regulation S-K:

Exhibit Number	Description of Exhibit
10.1	Revolving Credit Agreement dated as of March 4, 2011, among FelCor/JPM Hospitality (SPE), L.L.C., DJONT/JPM Hospitality Leasing (SPE), L.L.C., FelCor/JPM Boca Raton Hotel, L.L.C., and DJONT/JPM Boca Raton Leasing, L.L.C., as borrowers, and JPMorgan Chase Bank, N.A., as administrative agent, and the lenders that are parties thereto.
10.2	Form of Revolving Note under the Revolving Credit Agreement, each dated as of March 4, 2011, made by FelCor/JPM Hospitality (SPE), L.L.C., DJONT/JPM Hospitality Leasing (SPE), L.L.C., FelCor/JPM Boca Raton Hotel, L.L.C., and DJONT/JPM Boca Raton Leasing, L.L.C., for the benefit of the lenders.
10.3	Guaranty Agreement to the Revolving Credit Agreement dated as of March 4, 2011, by FelCor Lodging Trust Incorporated and FelCor Lodging Limited Partnership in favor of JPMorgan Chase Bank, N.A., as administrative agent, on behalf of the lenders.
10.4	Form of Fee and Leasehold Mortgage, Assignment of Leases and Rents, Security Agreement and Fixture Filing under the Revolving Credit Agreement dated as of March 4, 2011, granted by FelCor/JPM Hospitality (SPE), L.L.C., DJONT/JPM Hospitality Leasing (SPE), L.L.C., FelCor/JPM Boca Raton Hotel, L.L.C., and/or DJONT/JPM Boca Raton Leasing, L.L.C. for the benefit of JPMorgan Chase Bank, N.A., as administrative agent for the lenders.
31.1	Certification of the Chief Executive Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
31.2	Certification of the Chief Financial Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
32	Certification of the Chief Executive Officer and Chief Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

FELCOR LODGING TRUST INCORPORATED

Date: May 4, 2011

By: /s/ Lester C. Johnson

Name:Lester C. JohnsonTitle:Senior Vice President, Chief Accounting Officer

REVOLVING CREDIT AGREEMENT

Dated as of March 4, 2011

among

FELCOR/JPM HOSPITALITY (SPE), L.L.C.,

DJONT/JPM HOSPITALITY LEASING (SPE), L.L.C.,

FELCOR/JPM BOCA RATON HOTEL, L.L.C., and

DJONT/JPM BOCA RATON LEASING, L.L.C.

each as a Borrower,

JPMORGAN CHASE BANK, N.A., as Administrative Agent

and The Other Lenders Party Hereto

and BANK OF AMERICA, N.A., as Syndication Agent

and J.P. MORGAN SECURITIES, LLC and Merrill Lynch, Pierce, Fenner & Smith Incorporated as Joint Lead Arrangers and Co-Bookrunners

TABLE OF CONTENTS

Cure	. 4
rther Assurances lot Affected	
-i-	

ARTICLE I	DEFINITIONS AND ACCOUNTING TERMS	1
1.01.	Defined Terms	1
1.02.	Other Interpretive Provisions	37
1.03.	Classification of Loans and Borrowings	38
1.04.	Accounting Terms	38
1.05.	Rounding	38
1.06.	Times of Day	39
1.07.	Joint and Several Obligations	39
ARTICLE II	THE COMMITMENTS AND LOANS	40
2.01.	Commitments	40
2.02.	Loans and Borrowings	40
2.03.	Requests for Revolving Borrowings	40
2.04.	Swingline Loans	41
2.05.	Funding of Borrowings	42
2.06.	Interest Elections	43
2.07.	Termination and Reduction of Commitments	44
2.08.	Repayment of Loans; Evidence of Debt	45
2.09.	Prepayment of Loans	46
2.10.	Interest	47
2.11.	Fees	47
2.12.	Computation of Interest and Fees	48
2.13.	Evidence of Debt	48
2.14.	Payments Generally; Administrative Agent's Clawback	48
2.15.	Sharing of Payments by Lenders	49
2.16.	Extension of Maturity Date	50
2.17.	Collateral Property Releases	52
2.18.	Release on Payment in Full	54
2.19.	Incremental Commitments	54
ARTICLE III	CASH MANAGEMENT	56
3.01.	Establishment of Accounts	56
3.02.	Account Name	57
3.03.	Eligible Accounts	58
3.04.	Permitted Investments	58
3.05.	Transfers from Account	58
3.06.	Transfer To and Disbursements from the Lockbox Account	58
3.07.	Lockbox Trigger Event Cure	60
3.08.	Control	60
3.09.	Security Interest	60
3.10.	Rights on Default	61
3.11.	Financing Statement; Further Assurances	61
3.12.	Borrowers Obligations Not Affected	62

TABLE OF CONTENTS (continued)

<u>Page</u>

3.13.	Deposit Accounts	62
ARTICLE IV	TAXES, YIELD PROTECTION AND ILLEGALITY	62
4.01.	Taxes	62
4.02.	Illegality	64
4.03.	Inability to Determine Rates	65
4.04.	Increased Costs; Reserves on Eurodollar Loans	65
4.05.	Compensation for Losses	66
4.06.	Mitigation Obligations; Replacement of Lenders	67
4.07.	Survival	67
4.08.	Defaulting Lender	67
ARTICLE V	CONDITIONS PRECEDENT TO LOANS	69
5.01.	Conditions to Closing Date	69
5.02.	Each Credit Event	74
ARTICLE VI	REPRESENTATIONS AND WARRANTIES	75
6.01.	Existence, Qualification and Power	75
6.02.	Proceedings	75
6.03.	No Conflicts	75
6.04.	Litigation	76
6.05.	Agreements	76
6.06.	Solvency	76
6.07.	Full and Accurate Disclosure	77
6.08.	No Plan Assets	77
6.09.	Compliance with Legal Requirements	77
6.10.	Financial Information	77
6.11.	Condemnation	78
6.12.	Federal Reserve Regulations; Foreign Asset Control Regulations	78
6.13.	Utilities and Public Access	78
6.14.	Foreign Person	78
6.15.	Fee and Leasehold Ownership	78
6.16.	Separate Tax Lots; Assessments	79
6.17.	Enforceability	79
6.18.	No Prior Assignment	79
6.19.	Insurance	79
6.20.	Use of Property	79
6.21.	Certificate of Occupancy; Licenses	79
6.22.	Flood Zone	79
6.23.	Physical Condition	80
6.24.	Boundaries	80
6.25.	Leases	80
6.26.	Survey	81

TABLE OF CONTENTS (continued)

		Page
6.27.	Filing and Recording Taxes	81
6.28.	Franchise Agreements; Property Improvement Plans	81
6.29.	Management Agreements	81
6.30.	Illegal Activity	81
6.31.	No Change in Facts or Circumstances; Disclosure	81
6.32.	Investment Company Act	82
6.33.	Principal Place of Business; State of Organization; Tax Identification Number	82
6.34.	Single Purpose Entity	82
6.35.	Business Purposes	87
6.36.	Taxes	87
6.37.	Environmental Representations and Warranties	87
6.38.	Ground Lease Representations	88
6.39.	Operating Lease Representations	90
6.40.	Liens	91
6.41.	Service Contracts	91
6.42.	Personal Property Leasing and Financing	91
6.43.	Reciprocal Agreements	91
6.44.	Survival of Representations	92
ARTICLE VII	AFFIRMATIVE COVENANTS	92
7.01.	Existence; Compliance with Legal Requirements	92
7.02.	Taxes and Other Charges	93
7.03.	Litigation	94
7.04.	Access to Collateral Properties	94
7.05.	Notice of Default	94
7.06.	Cooperation in Legal Proceedings	94
7.07.	Award and Insurance Benefits	94
7.08.	Further Assurances	94
7.09.	Mortgage and Intangible Taxes	95
7.10.	Financial Reporting	95
7.11.	Business and Operations	98
7.12.	Costs of Enforcement	98
7.13.	Estoppel Statements	99
7.14.	Use of Proceeds	100
7.15.	Performance by Borrowers	100
7.16.	Leasing Matters	101
7.17.	Management Agreements	103
7.18.	Environmental Covenants	105
7.19.	Alterations	106
7.20.	Franchise Agreement	107
7.21.	Operating Lease	109
7.22.	OFAC	109
7.23.	Ground Leases	109

TABLE OF CONTENTS (continued)

7.24.	O&M Program	113
ARTICLE VIII	NEGATIVE COVENANTS	113
8.01.	Indebtedness	113
8.02.	Investments	113
8.03.	Liens	114
8.04.	Personal Property Leasing and Financing	115
8.05.	Operation and Service Agreements	115
8.06.	Restricted Payments	116
8.07.	Financial Covenants	116
8.08.	Dispositions	117
8.09.	Dissolution	117
8.10.	No Subsidiaries	118
8.11.	Burdensome Agreements	118
8.12.	Change In Business	118
8.13.	Debt Cancellation	118
8.14.	Zoning	118
8.15.	No Joint Assessment	118
8.16.	Name, Identity, Structure, or Principal Place of Business	119
8.17.	ERISA	119
8.18.	Affiliate Transactions	120
8.19.	Transfers	120
8.20.	REA	121
ARTICLE IX	INSURANCE; CASUALTY; CONDEMNATION	121
9.01.	Insurance	121
9.02.	Casualty	126
9.03.	Condemnation	127
9.04.	Restoration	127
ARTICLE X	EVENTS OF DEFAULT AND REMEDIES	132
10.01.	Events of Default	132
10.02.	Remedies upon Event of Default	137
10.03.	Application of Funds	137
ARTICLE XI	ADMINISTRATIVE AGENT	138
11.01.	Appointment and Authority	138
11.02.	Rights as a Lender	138
11.03.	Exculpatory Provisions	139
11.04.	Reliance by Administrative Agent	139
11.05.	Delegation of Duties	140
11.06.	Resignation of Administrative Agent	140
11.07.	Non-Reliance on Administrative Agent and Other Lenders	141

TABLE OF CONTENTS (continued)

Page

11.08.	No Other Duties, Etc	141
11.09.	Administrative Agent May File Proofs of Claim	141
11.10.	Collateral and Guaranty Matters	142
ARTICLE XII	MISCELLANEOUS	142
12.01.	Amendments, Etc	142
12.02.	Notices; Effectiveness; Electronic Communications	144
12.03.	No Waiver; Cumulative Remedies	146
12.04.	Expenses; Indemnity; Damage Waiver	146
12.05.	Payments Set Aside	148
12.06.	Successors and Assigns	149
12.07.	Treatment of Certain Information; Confidentiality	152
12.08.	Right of Setoff	153
12.09.	Exculpation	154
12.10.	Interest Rate Limitation	154
12.11.	Counterparts; Integration; Effectiveness	154
12.12.	Survival of Representations and Warranties	155
12.13.	Severability	155
12.14.	Replacement of Lenders	155
12.15.	Governing Law; Jurisdiction; Etc	156
12.16.	Waiver of Jury Trial	157
12.17.	No Advisory or Fiduciary Responsibility	157
12.18.	USA PATRIOT Act Notice	158
12.19.	Discretion	158
12.20.	Offsets, Counterclaims and Defenses	158
12.21.	No Joint Venture or Partnership, No Third Party Beneficiaries	158
12.22.	Publicity	159
12.23.	Waiver of Marshalling of Assets	159
12.24.	Conflict; Construction of Documents; Reliance	159
12.25.	Brokers and Financial Advisors	160
12.26.	Time of the Essence	160
12.27.	ENTIRE AGREEMENT	160

SCHEDULES

2.01		Commitments and Applicable Percentages
3.01(a)	_	Property Account Banks; List of Accounts
6.01		Existence of Borrower
6.11	_	Condemnation Matters
6.15		Collateral Properties
6.21		Licenses
6.25	—	Operating Leases
6.33	_	Borrowers Principal Place of Business, State of Organization; Tax Identification Number
6.37		Environmental Matters
6.38		Ground Lease Exceptions
6.39		Rent Arrearage
6.41		Service Contracts
6.42		Personal Property Leases and Financings
7.25		Certain Post-Closing Obligations
8.03	_	Existing Liens
12.02	_	Administrative Agent's Office, Certain Addresses for Notices
Х	—	Allocated Loan Amount

EXHIBITS

Exhibit A	—	Form of Borrowing Request
Exhibit B-1		Form of Revolving Note
Exhibit B-2		Form of Swingline Note
Exhibit C		Form of Assignment and Assumption
Exhibit D		Form of Facility Extension Request
Exhibit E-1		Form of Annual Responsible Officer's Certificate
Exhibit E-2		Form of Quarterly Responsible Officer's Certificate
Exhibit E-3		Form of [Extension][Collateral Release] Responsible Officer's Certificate
Exhibit G		Form of Accounts Notice
Exhibit H		Survey Requirements
Exhibit I		Operating Statement

REVOLVING CREDIT AGREEMENT

This REVOLVING CREDIT AGREEMENT ("<u>Agreement</u>") is entered into as of March 4, 2011, among FELCOR/JPM HOSPITALITY (SPE), L.L.C., a Delaware limited liability company ("<u>Hospitality Owner</u>"), DJONT/JPM HOSPITALITY LEASING (SPE), L.L.C., a Delaware limited liability company ("<u>Hospitality Operating Lessee</u>"), FELCOR/JPM BOCA RATON HOTEL, L.L.C., a Delaware limited liability company ("<u>Boca Owner</u>", and with Hospitality Owner, the "<u>Owners</u>") and DJONT/JPM BOCA RATON LEASING, L.L.C., a Delaware limited liability company ("<u>Boca Operating Lessee</u>" and with Hospitality Operating Lessee, the "<u>Lessees</u>"; Hospitality Owner, Hospitality Operating Lessee, Boca Owner and Boca Operating Lessee to be referred to collectively as "<u>Borrowers</u>" and each a "<u>Borrower</u>"), each lender from time to time party hereto (collectively, the "<u>Lenders</u>" and individually, a "<u>Lender</u>"), and JPMORGAN CHASE BANK, N.A., as Administrative Agent.

PRELIMINARY STATEMENTS:

Each Borrower is a special purpose entity owned directly or indirectly by FelCor Lodging Limited Partnership ("<u>FelCor Op</u>") to make and administer various investments in the Collateral Properties (as defined below).

Borrowers have requested that the Lenders provide a revolving credit facility to provide for Borrowers' working capital and other lawful corporate purposes (including payment of dividends constituting Restricted Payments permitted hereunder), and the Lenders have indicated their willingness to lend to Borrowers, on a joint and several basis, on the terms and subject to the conditions set forth herein.

In consideration of the mutual covenants and agreements herein contained, the parties hereto covenant and agree as follows:

ARTICLE I DEFINITIONS AND ACCOUNTING TERMS

1.01 <u>Defined Terms</u>. As used in this Agreement, the following terms shall have the meanings set forth below:

"<u>ABR</u>", when used in reference to any Loan or Borrowing, refers to whether such Loan, or the Loans comprising such Borrowing, are bearing interest at a rate determined by reference to the Alternate Base Rate.

"<u>Acceptable Appraisal</u>" means a written appraisal (i) prepared by a qualified professional independent MAI appraiser acceptable to Administrative Agent, (ii) meeting Administrative Agent's customary independent appraisal requirements and otherwise acceptable to Administrative Agent as to form, assumptions, substance and appraisal date, (iii) prepared in compliance with FIRREA, and (iv) prepared prior to the Closing Date or in connection with the Extension, or from time to time pursuant to Section 7.08(c) or at Borrowers' option.

"<u>Account</u> <u>Collateral</u>" means: (a) the Accounts, and all cash, checks, drafts, certificates and instruments, if any, from time to time deposited or held in the Accounts from time to time; (b) any and all amounts invested in Permitted Investments; (c) all interest, dividends, cash, instruments and other property from time to time received, receivable or otherwise payable in respect of, or in exchange for, any or all of the foregoing; and (d) to the extent not covered by clauses (a) - (c) above, all "proceeds" (as defined under the UCC) of any or all of the foregoing.

"<u>Accounts</u>" means the Property Account, the Lockbox Account and any Reserve Accounts.

"<u>Accounts Receivable</u>" has the meaning specified in Article 1 of the Mortgage with respect to each Collateral Property.

"<u>Act</u>" has the meaning specified in <u>Section 6.34</u>.

"Adjusted LIBO Rate" means, with respect to any Eurodollar Borrowing for any Interest Period, an interest rate per annum (rounded upwards, if necessary, to the next 1/100 of 1%) equal to (a) the LIBO Rate for such Interest Period multiplied by (b) the Statutory Reserve Rate.

"<u>Administrative Agent</u>" means JPMC in its capacity as administrative agent under any of the Loan Documents, or any successor administrative agent.

"<u>Administrative Agent's Office</u>" means Administrative Agent's address and, as appropriate, account as set forth on Schedule 12.02, or such other address or account as Administrative Agent may from time to time notify to Borrowers and the Lenders.

"<u>Administrative</u> <u>Questionnaire</u>" means an Administrative Questionnaire in a form supplied by Administrative Agent.

"<u>Affiliate</u>" means, with respect to any Person, another Person that directly, or indirectly through one or more intermediaries, Controls or is Controlled by or is under common Control with the Person specified.

"Affiliate Ground Leases" means the Boca Raton Affiliate Ground Lease and the Philadelphia Affiliate Ground Lease.

"Affiliate Ground Lessors" means the Boca Raton Affiliate Ground Lessor and the Philadelphia Affiliate Ground Lessor.

"<u>Affiliated Manager</u>" means any property manager which is an Affiliate of any Borrower, its Principal or Guarantor, or in which any Borrower, its Principal or Guarantor has, directly or indirectly, any legal, beneficial or economic Equity Interest.

"<u>Agent Parties</u>" has the meaning specified in <u>Section 12.02(c)</u>.

"<u>Aggregate Commitments</u>" means the aggregate Commitments of all Lenders.

"<u>Agreement</u>" has the meaning specified in the introductory paragraph.

"<u>Allocated Loan Amount</u>" means for a Collateral Property as of any date of determination: (i) from the Closing Date through the date of receipt by Administrative Agent of new Acceptable Appraisals for each of the Collateral Properties, the amount for each Collateral Property set forth in Schedule X, and (ii) from and after the date of receipt by Administrative Agent of new Acceptable Appraisals for each of the Collateral Properties:

(a) the ratio of (i) the Appraised Value of such Collateral Property to (ii) the aggregate Appraised Value of all Collateral Properties, expressed as a percentage; <u>multiplied</u> <u>by</u>

(b) an amount equal to the Aggregate Commitments.

"Alternate Base Rate" means, for any day, a rate per annum equal to the greatest of (a) the Prime Rate in effect on such day, (b) the Federal Funds Effective Rate in effect on such day plus ½ of 1% and (c) the Adjusted LIBO Rate for a one month Interest Period on such day (or if such day is not a Business Day, the immediately preceding Business Day) plus 1%, provided that, for the avoidance of doubt, the Adjusted LIBO Rate for any day shall be based on the rate appearing on Page LIBOR 01 of the Reuters screen (or on any successor or substitute page of such page) at approximately 11:00 a.m. London time on such day. Any change in the Alternate Base Rate due to a change in the Prime Rate, the Federal Funds Effective Rate or the Adjusted LIBO Rate shall be effective from and including the effective date of such change in the Prime Rate, the Federal Funds Effective Rate or the Adjusted LIBO Rate, respectively.

"<u>ALTA</u>" means the American Land Title Association, or any successor thereto.

"<u>Annual Budget</u>" means an operating budget for each Collateral Property prepared by the appropriate Manager and approved by the applicable Borrower, including all planned Capital Expenditures and FF&E budgeted information for each Collateral Property, prepared by the applicable Borrower for the applicable fiscal year or other period.

"<u>Applicable Margin</u>" means, for any day, with respect to any ABR Loans, 3.50% per annum or with respect to any Eurodollar Loan, 4.50% per annum, as the case may be.

"Applicable Percentage" means, with respect to any Lender, the percentage of the Aggregate Commitments represented by such Lender's Commitment (carried out to the ninth decimal place); provided that in the case of Section 4.08(b) when a Defaulting Lender shall exist, "Applicable Percentage" shall mean the percentage of the Aggregate Commitments (disregarding any Defaulting Lender's Commitment) represented by such Lender's Commitment. If the Commitments have terminated or expired, the Applicable Percentages shall be determined based upon the Commitments most recently in effect, giving effect to any assignments and to any Lender's status as a Defaulting Lender at the time of determination.

"<u>Appraisal Triggered Release</u>" has the meaning specified in <u>Section 2.17(a)</u>.

"<u>Appraised Value</u>" means the "as is" market value for a Collateral Property set forth in the most recent Acceptable Appraisal. "<u>Approved Annual Budget</u>" has the meaning specified in <u>Section 7.10(e)</u>.

"<u>Approved Fund</u>" means any Fund that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers or manages a Lender.

"<u>Arrangers</u>" means J.P. Morgan Securities, LLC. and Merrill Lynch, Pierce, Fenner & Smith Incorporated, in their capacity as Joint Lead Arrangers and Co-Bookrunners.

"<u>Assignee Group</u>" means two or more Eligible Assignees that are Affiliates of one another or two or more Approved Funds managed by the same investment advisor.

"<u>Assignment and Assumption</u>" means an assignment and assumption entered into by a Lender and an Eligible Assignee (with the consent of any party whose consent is required by <u>Section 12.06(b)</u>), and accepted by Administrative Agent, in substantially the form of Exhibit C or any other form approved by Administrative Agent.

"<u>Assignment of Leases and Rents</u>" means with respect to each Collateral Property that certain first priority Assignment of Leases and Rents, dated as of the date hereof, from the relevant Borrower, as assignor to Administrative Agent, as assignee, assigning to Administrative Agent on behalf of Lenders all of such Borrower's interest in and to the Leases and Rents of such Collateral Property as security for the Obligations.

"<u>Attributable Indebtedness</u>" means, on any date, (a) in respect of any Capitalized Lease of any Person, the capitalized amount thereof that would appear on a balance sheet of such Person prepared as of such date in accordance with GAAP, (b) in respect of any Synthetic Lease Obligation, the capitalized amount of the remaining lease or similar payments under the relevant lease or other applicable agreement or instrument that would appear on a balance sheet of such Person prepared as of such date in accordance with GAAP if such lease or other agreement or instrument were accounted for as a Capitalized Lease and (c) all Synthetic Debt of such Person.

"<u>Availability Period</u>" means the period from and including the Closing Date to but excluding the earlier of the Maturity Date and the date of termination of the Commitments.

"<u>Award</u>" means any compensation paid by any Governmental Authority in connection with a Condemnation in respect of all or any part of any Collateral Property.

"<u>Bankruptcy Code</u>" means the Bankruptcy Reform Act of 1978, as heretofore and hereafter amended, and codified as 11 U.S.C. sections 101 et seq.

"<u>Bankruptcy Event</u>" means, with respect to any Person, such Person becomes the subject of a bankruptcy or insolvency proceeding, or has had a receiver, conservator, trustee, administrator, custodian, assignee for the benefit of creditors or similar Person charged with the reorganization or liquidation of its business appointed for it, or, in the good faith determination of Administrative Agent, has taken any action in furtherance of, or indicating its consent to, approval of, or acquiescence in, any such proceeding or appointment, provided that a Bankruptcy Event shall not result solely by virtue of any ownership interest, or the acquisition of any ownership interest, in such Person by a Governmental Authority or instrumentality thereof, provided, further, that such

ownership interest does not result in or provide such Person with immunity from the jurisdiction of courts within the United States or from the enforcement of judgments or writs of attachment on its assets or permit such Person (or such Governmental Authority or instrumentality) to reject, repudiate, disavow or disaffirm any contracts or agreements made by such Person.

"<u>Base Management Fee</u>" means the base management fee to be paid under the applicable Management Agreement, which shall not include any incentive or similar performance based fees pursuant to such agreement.

"<u>Board</u>" means the Board of Governors of the Federal Reserve System of the United States of America.

"Boca Borrowers" means Boca Owner and Boca Operating Lessee.

"Boca Operating Lessee" has the meaning specified in the introductory paragraph.

"Boca Owner" has the meaning specified in the introductory paragraph.

"<u>Boca Raton Affiliate Ground Lease</u>" means the Ground Lease dated October 21, 2003, as amended and assigned, by and between FelCor Op, as landlord, and Boca Owner, as tenant, with respect to the Boca Raton Property.

"<u>Boca Raton Affiliate Ground Lessee</u>" means Boca Owner in its capacity as tenant under the <u>Boca Raton Affiliate Ground Lease</u>.

"<u>Boca Raton Affiliate Ground Lessor</u>" means FelCor Op in its capacity as landlord under the Boca Raton Affiliate Ground Lease.

"<u>Boca Raton Franchise Agreement</u>" means the Embassy Suites License Agreement dated February 28, 1996, as amended and assigned, by and between Boca Operating Lessee, as licensee, and HLT Existing Franchise Holding L.L.C., as licensor, with respect to the Boca Raton Property.

"<u>Boca Raton Management Agreement</u>" means the Management Agreement dated as of February 28, 2006, as amended and assigned, by and between Boca Operating Lessee, as "Owner," and Embassy Suites Management LLC, as "Manager," with respect to the Boca Raton Property.

"<u>Boca Raton Operating Lease</u>" means the Lease Agreement dated as of February 28, 1996, as amended and assigned, by and between Boca Owner and Boca Operating Lessee for the Boca Raton Property.

"<u>Boca Raton Property</u>" means the property located at 661 NW 53rd Street, Boca Raton, Florida 33487 and commonly known as Embassy Suites, Boca Raton.

"<u>Borrower</u>" and "<u>Borrowers</u>" have the meanings specified in the introductory paragraph hereto.

"<u>Borrower</u> <u>Materials</u>" means information and materials made available by Administrative Agent to Lenders which were provided by or on behalf of Borrowers.

"Borrower <u>Representative</u>" means Hospitality Owner, in its capacity as agent and as representative for all Borrowers.

"<u>Borrowing</u>" means (a) Revolving Loans of the same Type, made, converted or continued on the same date and, in the case of Eurodollar Loans, as to which a single Interest Period is in effect and (b) any Swingline Loan.

"<u>Borrowing Request</u>" means a request by Borrowers for a Revolving Borrowing in accordance with Section 2.03, in the form of Exhibit A.

"<u>Business Day</u>" shall mean any day that is not a Saturday, Sunday or any other day on which commercial banks in New York City are authorized or required by law to remain closed; provided that, when used in connection with a Eurodollar Loan, the term "Business Day" shall also exclude any day on which banks are not open for dealings in dollar deposits in the London interbank market.

"Business Party" has the meaning specified in Section 6.34(x).

"<u>Capital Expenditures</u>" means, with respect to any Person for any period, any expenditure in respect of the purchase or other acquisition of any fixed or capital asset (excluding normal replacements and maintenance which are properly charged to current operations).

"<u>Capital Expenditures Reserve Account</u>" has the meaning specified in Section 3.01 (c)(iii).

"<u>Capitalized Leases</u>" means all leases that have been or should be, in accordance with GAAP, recorded as capitalized leases.

"<u>cash</u>" means coin or currency of the United States of America or immediately available federal funds, including such funds delivered by wire transfer.

"Casualty" has the meaning specified in Section 9.02.

"<u>Casualty Consultant</u>" has the meaning specified in Section 9.04(e).

"Casualty Retainage" has the meaning specified in Section 9.04(f).

"<u>Certification of Non-Foreign Status</u>" means an affidavit, signed under penalty of perjury by an authorized officer of each Borrower, stating (a) that such Borrower is not a "foreign corporation," "foreign partnership," "foreign trust," or "foreign estate," as those terms are defined in the Code and the regulations promulgated thereunder, (b) such Borrower's U.S. employer identification number, and (c) the address of such Borrower's principal place of business. Such affidavit shall be consistent with the requirements of the regulations promulgated under Section 1445 of the Code, and shall otherwise be in form and substance acceptable to Administrative Agent.

"<u>Change of Control</u>" means (a) the acquisition of ownership, directly or indirectly, beneficially or of record, by any Person or group (within the meaning of the Securities Exchange Act of 1934 and the rules of the Securities and Exchange Commission thereunder as in effect on the date hereof), of Equity Interests representing more than 49.99% of the then-exercisable aggregate ordinary voting power represented by the issued and outstanding Equity Interests of FelCor Trust; (b) occupation of a majority of the seats (other than vacant seats) on the board of directors of the FelCor Trust by Persons who were neither (i) nominated by the board of directors of FelCor Trust nor (ii) appointed by directors so nominated; or (c) the acquisition of direct or indirect Control of FelCor Trust by any Person or group.

"<u>Change in Law</u>" means the occurrence, after the date of this Agreement, of any of the following: (a) the adoption or taking effect of any law, rule, regulation or treaty, (b) any change in any law, rule, regulation or treaty or in the administration, interpretation or application thereof by any Governmental Authority or (c) the making or issuance of any request, guideline or directive (whether or not having the force of law) by any Governmental Authority; <u>provided</u> however that notwithstanding anything herein to the contrary, the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder or issued in connection therewith shall be deemed to be a "Change in Law" regardless of the date enacted, adopted or issued.

"<u>Charlotte Management Agreement</u>" means the Management Agreement dated as of July 12, 2002, as amended and assigned, by and between Hospitality Operating Lessee, as "Owner," and DT Management, Inc., as "Manager," with respect to the Charlotte Property.

"<u>Charlotte Operating Lease</u>" means the Operating Lease dated as of July 12, 2002, as amended and assigned, by and between Hospitality Owner and Hospitality Operating Lessee for the Charlotte Property.

"<u>Charlotte Property</u>" means the property located at 6300 Morrison Blvd., Charlotte, North Carolina 28211 and commonly known as Doubletree Charlotte, North Carolina.

"<u>Class</u>," when used in reference to any Loan or Borrowing, refers to whether such Loan, or the Loans comprising such Borrowings, are Revolving Loans or Swingline Loans.

"<u>Closing Date</u>" means the first date all the conditions precedent in Section 5.01 are satisfied or waived in accordance with Section 12.01.

"Code" means the Internal Revenue Code of 1986.

"<u>Collateral</u>" means (a) all of the "<u>Collateral</u>," "<u>Personal Property</u>" and "<u>Property</u>" referred to in the Mortgages, (b) all Account Collateral, and (c) all of the other property that is or is intended under the terms of the Loan Documents, including the Pledge Agreements, to be subject to Liens in favor of Administrative Agent for the benefit of the Secured Parties.

"<u>Collateral Document</u>" means, collectively each Mortgage, the Operating Lease Subordination Agreements, each Pledge Agreement, each Assignment of Leases and Rents each Lockbox Account Agreement and all financing statements, instruments, documents or filings necessary to establish or maintain Liens in favor of the Secured Parties.

"<u>Collateral Property</u>" means, each hotel project listed on Schedule 6.15 together with all "Property" defined in the Mortgages with respect to such project.

"<u>Commitment</u>" means, with respect to each Lender, the commitment of such Lender to make Revolving Loans and to acquire participations in Swingline Loans hereunder, expressed as an amount representing the maximum aggregate amount of such Lender's Revolving Credit Exposure hereunder, as such commitment may be (a) reduced from time to time pursuant to Section 2.07, (b) increased from time to time pursuant to Section 2.19, or (c) reduced or increased from time to time pursuant to assignments by or to such Lender pursuant to Section 12.06. The initial amount of each Lender's Commitment is set forth on Schedule 2.01, or in the Assignment and Assumption pursuant to which such Lender shall have assumed its Commitment, as applicable. The initial aggregate amount of the Lenders' Commitments is \$225,000,000.

"<u>Condemnation</u>" means a temporary or permanent taking by any Governmental Authority as the result or in lieu or in anticipation of the exercise of the right of condemnation or eminent domain, of all or any part of any Collateral Property, or any interest therein or right accruing thereto, including any right of access thereto or any change of grade affecting such Collateral Property or any part thereof.

"<u>Condemnation Proceeds</u>" has the meaning specified in Section 9.04(b).

"Constituent Member" has the meaning specified in Section 12.09.

"<u>Consumer Price Index</u>" means the Consumer Price Index for all Urban Consumers, U.S. City Average, published by the United States Bureau of Labor Statistics.

"<u>Contractual</u> <u>Obligation</u>" means, as to any Person, any provision of any security issued by such Person or of any agreement, instrument or other undertaking to which such Person is a party or by which it or any of its property is bound.

"<u>Control</u>" means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ability to exercise voting power, by contract or otherwise. "<u>Controlling</u>" and "<u>Controlled</u>" have meanings correlative thereto.

"Credit Party" means Administrative Agent, Swingline Lender or any other Lender.

"<u>Dana Point Management Agreement</u>" means the Management Agreement dated as of July 1, 2006, as amended and assigned, by and between Boca Operating Lessee, as "Owner," and Doubletree Management LLC, as "Manager," with respect to the Dana Point Property. "<u>Dana Point Operating Lease</u>" means the Lease Agreement dated as of February 20, 1997, as amended and assigned, by and between Boca Owner and Boca Operating Lessee for the Dana Point Property.

"<u>Dana Point Property</u>" means the property located at 34402 Pacific Coast Highway, Dana Point, California 92629 and commonly known as Doubletree Guest Suites Doheny Beach-Dana Point.

"Debt Service" means, for any period, the sum of (a) all interest, premium payments, debt discount, fees, charges and related expenses in connection with borrowed money (including capitalized interest) or in connection with the deferred purchase price of assets, in each case to the extent treated as interest in accordance with GAAP, in each case, of or by Borrowers, for such period plus (b) any and all scheduled repayments of principal during such period in respect of Indebtedness that becomes due and payable or that are to become (or deemed to become) due and payable during such period pursuant to any agreement or instrument to which a Borrower is a party relating to (i) the borrowing of money or the obtaining of credit, including the issuance of notes or bonds, and (ii) the deferred purchase price of assets (other than trade payables incurred in the ordinary course of business). Demand obligations shall be deemed to be due and payable during any fiscal period during which such obligations are outstanding.

"<u>Debt Service Coverage Ratio</u>" means, as of the last day of any fiscal quarter, the ratio in which:

(a) the numerator is EBITDA for the four (4) fiscal quarters ending on such last day minus Capital Expenditure reserves equal to four percent (4%) of Gross Income from Operations for such period; and

(b) the denominator is the aggregate amount of principal and interest payments on the Loans calculated for such period using the Outstanding Amount at the time of such calculation, assuming (x) a thirty (30) year amortization schedule, and (y) an interest rate per annum equal to (i) in every context except the Extension, the greater of (A) seven percent (7.0%) and (B) the actual interest rate on the Loans at such time (if Loans of more than one Type and/or of multiple Adjusted LIBO Rates are then in effect, the effective blended rate on the aggregate of such Loans) at the time of such calculation (the "<u>Current Interest Rate</u>"), and (ii) solely in the context of the Extension, the greatest of (A) seven percent (7.0%), (B) the rate for seven (7) year Treasury securities plus four percent (4%), and (C) the Current Interest Rate.

For avoidance of doubt, the "pro forma" calculation of Debt Service Coverage Ratio as of December 31, 2010, shall compare principal and interest payments on the Loans calculated as of such date (as if the Aggregate Commitments as of the Closing Date were then already advanced) to the actual EBITDA of the Collateral Properties for the four (4) fiscal quarters ending on December 31, 2010.

"<u>Debtor Relief Laws</u>" means the Bankruptcy Code, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief Laws of the United States or other applicable jurisdictions from time to time in effect and affecting the rights of creditors generally.

"<u>Default</u>" means any event or condition that constitutes an Event of Default or that, with the giving of any notice, the passage of time, or both, would be an Event of Default.

"<u>Default Rate</u>" means an interest rate equal to (a) for ABR Loans, the Alternate Base Rate plus the Applicable Margin applicable to ABR Loans plus two percent (2.0%); and (b) for Eurodollar Loans, the Adjusted LIBO Rate plus the Applicable Margin applicable to Eurodollar Loans plus two percent (2.0%). Without duplication of the foregoing, with respect to the interest rate for overdue interest, fees and other amounts, "Default Rate" means the Alternate Base Rate plus the Applicable Margin applicable to ABR Loans plus two percent (2.0%).

"Defaulting Lender" means any Lender (as reasonably determined by the Administrative Agent) that (a) has failed, within three Business Days of the date required to be funded or paid, to (i) fund any portion of its Revolving Loans, (ii) fund any portion of its participations in Swingline Loans or (iii) pay over to any Credit Party any other amount required to be paid by it hereunder (other than a de minimus amount), unless, in the case of clause (i) above, such Lender notifies Administrative Agent in writing that such failure is the result of such Lender's good faith determination that a condition precedent to funding (specifically identified and including the particular default, if any) has not been satisfied, (b) has notified Borrowers or any Credit Party in writing, or has made a public statement to the effect, that it does not intend or expect to comply with any of its funding obligations under this Agreement (unless such writing or public statement indicates that such position is based on such Lender's good faith determination that a condition precedent (specifically identified and including the particular default, if any) to funding a loan under this Agreement cannot be satisfied) or generally under other agreements in which it commits to extend credit, (c) has failed, within three Business Days after request by a Credit Party, acting in good faith, to provide a certification in writing from an authorized officer of such Lender that it will comply with its obligations (and is financially able to meet such obligations) to fund prospective Revolving Loans and participations in then outstanding Swingline Loans under this Agreement, provided that such Lender shall cease to be a Defaulting Lender pursuant to this clause (c) upon such Credit Party's receipt of such certification in form and substance satisfactory to it and Administrative Agent, or (d) has become the subject of a Bankruptcy Event.

"<u>Disposition</u>" or "<u>Dispose</u>" means the sale, transfer, license, lease or other disposition (including any sale and leaseback transaction) of any property by any Person (or the granting of any option or other right to do any of the foregoing), including any sale, assignment, transfer or other disposal, with or without recourse, of any notes or accounts receivable or any rights and claims associated therewith.

"Dollar" and "\$" mean lawful money of the United States.

"<u>EBITDA</u>" means, at any date of determination, for any period an amount equal to (a) the aggregate net income (or loss) of Borrowers for such period, plus (b) the following to the extent deducted in calculating such net income (loss): (i) Debt Service; (ii) the provision for Federal, state, local and foreign income taxes payable; (iii) depreciation and amortization expense; and (iv) other non-recurring expenses reducing such net income (loss) which do not represent a cash item in such period or any future period (in each case of or by Borrowers for such period), and (v) amortization of intangibles for such period, all extraordinary nonrecurring items of expense, such as employee severance expenses and hurricane losses (net of insurance proceeds), non-cash impairment charges, non cash write-offs of deferred financing costs (provided, however, that in the case of clause (iv) and clause (v) above, Borrowers will use the same amounts for "non-recurring expenses" as Borrowers' Affiliates have included in public disclosure as a reconciling item to "Adjusted EBITDA", allocated to the Collateral Properties to the extent applicable to the Collateral Properties), and costs, premiums and penalties arising by contract in connection with the prepayment of indebtedness, <u>minus</u> (c) the following to the extent included in calculating such net income (loss): (i) Federal, state, local and foreign income tax credits, and (ii) all non-recurring non-cash items increasing net income (loss) (in each case of or by Borrowers for such period), if and to the extent that such amounts of "non-recurring non-cash items" have been included by Borrowers' Affiliates in public disclosures as a reconciling item to "Adjusted EBITDA", provided that the calculation of EBITDA for the Boca Borrowers for the fiscal quarters ending on or prior to September 30, 2010 shall be based on actual EBITDA of the Collateral Properties owned by the Boca Borrowers during such fiscal quarters, and the calculation of EBITDA from the fiscal quarters ending after September 30, 2010 shall be based on EBITDA of the Boca Borrowers.

"<u>Eligible Account</u>" shall mean a separate and identifiable account from all other funds held by the holding institution that is either (a) an account or accounts maintained with a federal or State-chartered depository institution or trust company which complies with the definition of Eligible Institution or (b) a segregated trust account or accounts maintained with a federal or State chartered depository institution or trust company acting in its fiduciary capacity which, in the case of a State chartered depository institution or trust company, is subject to regulations substantially similar to 12 C.F.R.§9.10(b), having in either case a combined capital and surplus of at least \$50,000,000 and subject to supervision or examination by federal and State authority. An Eligible Account will not be evidenced by a certificate of deposit, passbook or other instrument.

"<u>Eligible Assignee</u>" means any Person that meets the requirements to be an assignee under Section 12.06(b)(i), (<u>iii</u>), (<u>v</u>) and (<u>vi</u>) (subject to such consents, if any, as may be required under Section 12.06(b)(iii)).

"<u>Eligible Institution</u>" shall mean a depository institution or trust company, insured by the Federal Deposit Insurance Corporation, (a) the short term unsecured debt obligations or commercial paper of which are rated at least A-1 by S&P, P-1 by Moody's and F-1 by Fitch in the case of accounts in which funds are held for thirty (30) days or less, or (b) the long term unsecured debt obligations of which are rated at least A- by Fitch and S&P and A3 by Moody's in the case of accounts in which funds are held for more than thirty (30) days.

"Emergency Repairs" has the meaning specified in Section 9.04(b).

"<u>Environmental Indemnity</u>" means that certain indemnity executed by Guarantor and each Borrower in favor of the Secured Parties in form and substance acceptable to the Administrative Agent. "<u>Environmental Laws</u>" means any and all Federal, state, local, and foreign statutes, laws, regulations, ordinances, rules, judgments, orders, decrees, permits, concessions, grants, franchises, licenses, agreements or governmental restrictions relating to pollution and the protection of the environment or the release of any materials into the environment, including those related to hazardous substances or wastes, air emissions and discharges to waste or public systems.

"Environmental Liability" means any liability, contingent or otherwise (including any liability for damages, costs of environmental remediation, fines, penalties or indemnities), of a Borrower directly or indirectly resulting from or based upon (a) violation of any Environmental Law, (b) the generation, use, handling, transportation, storage, treatment or disposal of any Hazardous Materials, (c) exposure to any Hazardous Materials, (d) the release or threatened release of any Hazardous Materials into the environment or (e) any contract, agreement or other consensual arrangement pursuant to which liability is assumed or imposed with respect to any of the foregoing.

"<u>Environmental Lien</u>" has the meaning specified in Section 7.18(a).

"Environmental Report" has the meaning specified in Section 5.01(a)(xiii).

"Equity Interests" means, with respect to any Person, all of the shares of capital stock of (or other ownership or profit interests in) such Person, all of the warrants, options or other rights for the purchase or acquisition from such Person of shares of capital stock of (or other ownership or profit interests in) such Person, all of the securities convertible into or exchangeable for shares of capital stock of (or other ownership or profit interests in) such Person or warrants, rights or options for the purchase or acquisition from such Person of such shares (or such other interests), and all of the other ownership or profit interests in such Person (including partnership, member or trust interests therein), whether voting or nonvoting, and whether or not such shares, warrants, options, rights or other interests are outstanding on any date of determination.

"Equity <u>Owner</u>" has the meaning specified in Section 6.34(x).

"ERISA" means the Employee Retirement Income Security Act of 1974.

"<u>ERISA Affiliate</u>" means any trade or business (whether or not incorporated) under common control with a Borrower within the meaning of Section 414(b) or (c) of the Code (and Sections 414(m) and (o) of the Code for purposes of provisions relating to Section 412 of the Code).

"<u>Eurodollar</u>", when used in reference to any Loan or Borrowing, refers to whether such Loan, or the Loans comprising such Borrowing, are bearing interest at a rate determined by reference to the Adjusted LIBO Rate.

"Event of Default" has the meaning specified in Section 10.01.

"Excluded Taxes" means, with respect to Administrative Agent, any Lender or any other recipient of any payment to be made by or on account of any obligation of Borrowers hereunder, (a) taxes imposed on or measured by its overall net income (however denominated), and franchise taxes imposed on it (in lieu of net income taxes), by the jurisdiction (or any political subdivision thereof) under the laws of which such recipient is organized or in which its principal office is located or, in the case of any Lender, in which its applicable Lending Office is located, (b) any branch profits taxes imposed by the United States or any similar tax imposed by any other jurisdiction in which such Borrower is located and (c) in the case of a Foreign Lender (other than an assignee pursuant to a request by Borrowers under Section 12.14), any withholding tax that is imposed on amounts payable to such Foreign Lender at the time such Foreign Lender's failure or inability (other than as a result of a Change in Law) to comply with Section 4.01(d), except to the extent that such Foreign Lender (or its assignor, if any) was entitled, at the time of designation of a new Lending Office (or assignment), to receive additional amounts from such Borrower with respect to such withholding tax pursuant to Section 4.01(a).

"<u>Extension</u>" means the extension of the Stated Maturity Date pursuant to Section 2.16.

"Extension Effective Date" has the meaning specified in Section 2.16.

"<u>Extension Period</u>" means the period from the Extension Effective Date through the Stated Maturity Date as extended pursuant to Section 2.16.

"<u>Facility</u>" means at any time, the aggregate amount of Commitments at such time or the aggregate principal amount of the Loans of all Lenders outstanding at such time, as applicable.

"<u>Facility Extension Request</u>" means a request for extension of the Maturity Date in the form of Exhibit D.

"Federal Funds Rate" means, for any day, the weighted average (rounded upwards, if necessary, to the next 1/100 of 1%) of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers, as published on the next succeeding Business Day by the Federal Reserve Bank of New York, or, if such rate is not so published for any day that is a Business Day, the average (rounded upwards, if necessary, to the next 1/100 of 1%) of the quotations for such day for such transactions received by Administrative Agent from three Federal funds brokers of recognized standing selected by it.

"<u>Fee Letter</u>" means the letter agreement; dated January 25, 2011, among Borrowers, Administrative Agent and the Arrangers.

"FelCor Op" has the meaning assigned in the preliminary statements.

"<u>FelCor TRS</u>" means FelCor TRS Holdings, L.L.C., a Delaware limited liability .

company.

"FelCor Trust" means FelCor Lodging Trust Incorporated.

"FF&E" means all furniture, furnishings, fixtures and equipment required for the operation of the Collateral Properties, including, without limitation, lobby furniture, carpeting, draperies, paintings, bedspreads, television sets, office furniture and equipment such as safes, cash registers, and accounting, duplicating and communication equipment, telephone systems, back and front of the house computerized systems, guest room furniture, specialized hotel equipment such as equipment required for the operation of kitchens, laundries, the front desk, dry cleaning facilities, bar and cocktail lounges, restaurants, recreational facilities as they may exist from time to time, and decorative lighting, material handling equipment and cleaning and engineering equipment and all other fixtures, equipment, apparatus and personal property needed for such purposes; but excluding, (a) Collateral Property building equipment and systems (including, but not limited to, the heating, ventilating and air conditioning system, elevators, electrical distribution system, life safety systems and plumbing), (b) other fixtures attached to and forming part of the Improvements (including, but not limited to, lighting fixtures and bars) installed during construction of the Collateral Properties (but replacements thereof shall be included) and (c) Operating Equipment and Supplies.

"<u>FF&E Expenditures</u>" means amounts expended for the purchase, replacement and/ or the installation of FF&E at a Collateral Property or any Capital Expenditures at a Collateral Property.

"<u>FIRREA</u>" means the Financial Institutions Recovery, Reform and Enforcement Act of 1989, as amended from time to time.

"<u>Fiscal Year</u>" means each twelve (12) month period commencing on January 1 and ending on December 31, during the term of the Loan.

"Fitch" means Fitch, Inc.

"Flood Insurance Acts" has the meaning specified in Section 9.01(a)(vii).

"Flood Insurance Policies" has the meaning specified in Section 9.01(a)(vii).

"<u>Foreign Lender</u>" means any Lender that is organized under the laws of a jurisdiction other than that in which Borrowers are resident for tax purposes. For purposes of this definition, the United States, each state thereof and the District of Columbia shall be deemed to constitute a single jurisdiction.

"<u>Franchise Agreements</u>" means the Boca Raton Franchise Agreement, the Mandalay Beach Franchise Agreement and the Nashville Franchise Agreement, and "<u>Franchise Agreement</u>" means any one of the Franchise Agreements.

"<u>Franchised Properties</u>" means (i) the Boca Raton Property, (ii) the Mandalay Beach Property and (iii) the Nashville Property.

"<u>Franchisor</u>" means, for each Franchise Agreement, the franchisor identified in the definition of each Franchise Agreement, or such successor Franchisor which is identified in any Replacement Franchise Agreement.

"<u>Fund</u>" means any Person (other than a natural person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its activities.

"<u>GAAP</u>" means generally accepted accounting principles in the United States set forth in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or such other principles as may be approved by a significant segment of the accounting profession in the United States, that are applicable to the circumstances as of the date of determination, consistently applied.

"<u>Governmental Authority</u>" means the government of the United States or any other nation, or of any political subdivision thereof, whether state or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government (including any supra-national bodies such as the European Union or the European Central Bank).

"<u>Gross Income from Operations</u>" means, for any period, all income, room revenues, food and beverage revenue, telephone revenue, computed in accordance with GAAP derived from the ownership and operation of the Collateral Properties from whatever source, including, but not limited to, the Rents, utility charges, service fees or charges, license fees, parking fees, rent concessions or credits, and other required pass-throughs, but excluding sales, use and occupancy or other taxes on receipts required to be accounted for by Borrowers to any Governmental Authority, interest on credit accounts, refunds and uncollectible accounts, sales of furniture, fixtures and equipment, Insurance Proceeds (other than business interruption or other loss of income insurance), Awards, unforfeited security deposits, utility and other similar deposits, escalations, forfeited security deposits. Gross income shall not be diminished as a result of the Mortgages or the creation of any intervening estate or interest in a Collateral Property or any part thereof.

"<u>Ground Lease Properties</u>" means the Collateral Properties so identified on Schedule 6.15, and "<u>Ground Lease Property</u>" means any of them.

"<u>Ground Leases</u>" means the Boca Raton Affiliate Ground Lease, the Myrtle Beach Golf Course Lease, the Pittsburgh Ground Lease, the Philadelphia Affiliate Ground Lease and the Houston Ground Lease, and "<u>Ground Lease</u>" means any one of the Ground Leases.

"<u>Ground Lessee</u>" means (i) the Boca Affiliate Ground Lessee, and (ii) Hospitality Owner, in its capacity as ground lessee, under each of the other Ground Leases (other than the Boca Raton Affiliate Ground Lease).

"<u>Ground Lessors</u>" means the Boca Raton Affiliate Ground Lessor, the Myrtle Beach Golf Course Lessor, the Pittsburgh Ground Lessor, the Philadelphia Affiliate Ground Lessor and the Houston Ground Lessor, and "<u>Ground Lessor</u>" means any one of the Ground Lessors.

"Guarantee" means, as to any Person, (a) any obligation, contingent or otherwise, of such Person guaranteeing or having the economic effect of guaranteeing any Indebtedness or other obligation payable or performable by another Person (the "primary obligor") in any manner, whether directly or indirectly, and including any obligation of such Person, direct or indirect, (i) to purchase or pay (or advance or supply funds for the purchase or payment of) such Indebtedness or other obligation, (ii) to purchase or lease property, securities or services for the purpose of assuring the obligee in respect of such Indebtedness or other obligation of the payment or performance of such Indebtedness or other obligation, (iii) to maintain working capital, equity capital or any other financial statement condition or liquidity or level of income or cash flow of the primary obligor so as to enable the primary obligor to pay such Indebtedness or other obligation, or (iv) entered into for the purpose of assuring in any other manner the obligee in respect of such Indebtedness or other obligation of the payment or performance thereof or to protect such obligee against loss in respect thereof (in whole or in part), or (b) any Lien on any assets of such Person securing any Indebtedness or other obligation of any other Person, whether or not such Indebtedness or other obligation is assumed by such Person (or any right, contingent or otherwise, of any holder of such Indebtedness to obtain any such Lien). The amount of any Guarantee shall be deemed to be an amount equal to the stated or determinable amount of the related primary obligation, or portion thereof, in respect of which such Guarantee is made or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof as determined by the guaranteeing Person in good faith. The term "Guarantee" as a verb has a corresponding meaning.

"<u>Guarantor</u>" means FelCor Trust, FelCor Op and each other Person that shall be required to execute and deliver a Guaranty pursuant to this Agreement.

"<u>Guaranty</u>" means, collectively, the recourse guaranty made by Guarantor in favor of Administrative Agent and Lenders, and any other Guarantee executed by any other person from time to time in favor of Administrative Agent and Lenders, in each case, in form and substance acceptable to the Administrative Agent.

"<u>Hazardous Materials</u>" shall mean petroleum and petroleum products and compounds containing them, including gasoline, diesel fuel and oil; explosives; flammable materials; radioactive materials; polychlorinated biphenyls and compounds containing them; toxic mold; lead and lead-based paint; asbestos or asbestos-containing materials in any form that is or could become friable; underground or above-ground storage tanks, whether empty or containing any substance; any substance the presence of which on the Collateral Property is prohibited by any federal, state or local authority; any substance that requires special handling; and any other material or substance now or in the future defined as a "*hazardous substance*," "*hazardous material*," "*hazardous waste*," "*toxic substance*," "*toxic pollutant*," "*contaminant*," "*pollutant*" or other words of similar import within the meaning of any Environmental Law.

"Hospitality Borrowers" means Hospitality Owner and Hospitality Operating Lessee.

"<u>Hospitality</u> <u>Operating</u> <u>Lessee</u>" has the meaning specified in the introductory paragraph.

"Hospitality Owner" has the meaning specified in the introductory paragraph.

"<u>Houston Ground Lease</u>" means the Ground Lease dated as of July 5, 1962, as amended and assigned, between Houston Ground Lessor and Hospitality Owner.

"<u>Houston Ground Lessor</u>" means, collectively, J. Gordon Zuber, the M. M. Feld, Jr. Trust, the Feld Family Venture, Joseph B. Zuber, Bernice Feld, Rhoda Dreyfus, Noonie Z. Schmidt, Bert L. Zuber, Harry A. Zuber, Maurice Robinowitz, CLT, LLC and Louis B. Marks, and Bonnie Zuber Beerman, and their respective successors and assigns, and any successor thereto.

"<u>Houston Management Agreement</u>" means the Management Agreement dated as of June 30, 2001, as amended and assigned, by and between Hospitality Operating Lessee, as "Owner," and IHG, as "Manager," with respect to the Houston Property.

"<u>Houston Operating Lease</u>" means the Operating Lease dated as of July 27, 1998, as amended and assigned, by and between Hospitality Owner and Hospitality Operating Lessee for the Houston Property.

"<u>Houston</u> P<u>roperty</u>" means the property located at 6800 Main Street, Houston, TX 77030 and commonly known as Holiday Inn Hotel and Suites Houston, Texas.

"<u>IHG</u>" shall mean Intercontinental Hotels Group Resources, Inc., successor to Bristol Management L.P.

"<u>Improvements</u>" shall have the meaning specified in the related Mortgage with respect to each Collateral Property.

"Incentive Management Fee" means any incentive or similar performance based fees payable to any Manager pursuant to a Management Agreement.

"Incremental Limit" has the meaning specified in Section 2.19(a).

"Increased Amount Date" has the meaning specified in Section 2.19(a).

"<u>Indebtedness</u>" means, as to any Person at a particular time, without duplication, all of the following, whether or not included as indebtedness or liabilities in accordance with GAAP:

(a) all obligations of such Person for borrowed money and all obligations of such Person evidenced by bonds, debentures, notes, loan agreements or other similar instruments;

(b) the maximum amount of all direct or contingent obligations of such Person arising under letters of credit (including standby and commercial), bankers' acceptances, bank guaranties, surety bonds and similar instruments;

(c) net obligations of such Person under any Swap Contract;

(d) all obligations of such Person to pay the deferred purchase price of property or services (other than trade accounts payable in the ordinary course of business and not past due for more than sixty (60) days after the date on which such trade account was created);

(e) indebtedness (excluding prepaid interest thereon) secured by a Lien on property owned or being purchased by such Person (including indebtedness arising under conditional sales or other title retention agreements), whether or not such indebtedness shall have been assumed by such Person or is limited in recourse;

(f) all Attributable Indebtedness in respect of Capitalized Leases and Synthetic Lease Obligations of such Person and all Synthetic Debt of such Person;

(g) all obligations of such Person to purchase, redeem, retire, defease or otherwise make any payment in respect of any Equity Interest in such Person or any other Person or any warrant, right or option to acquire such Equity Interest, valued, in the case of a redeemable preferred interest, at the greater of its voluntary or involuntary liquidation preference plus accrued and unpaid dividends; and

(h) all Guarantees of such Person in respect of any of the foregoing.

For all purposes hereof, the Indebtedness of any Person shall include the Indebtedness of any partnership or joint venture (other than a joint venture that is itself a corporation or limited liability company) in which such Person is a general partner or a joint venturer, unless such Indebtedness is expressly made non-recourse to such Person. The amount of any net obligation under any Swap Contract on any date shall be deemed to be the Swap Termination Value thereof as of such date.

"Indemnified Taxes" means Taxes other than Excluded Taxes.

"<u>Indemnitees</u>" has the meaning specified in Section 12.04(b).

"Independent Director" has the meaning specified in Section 6.34(x).

"Information" has the meaning specified in Section 12.07.

"Insurance Premium Account" has the meaning specified in Section 3.01(c)(ii).

"Insurance Premiums" has the meaning specified in Section 9.01(b).

"Insurance Proceeds" has the meaning specified in Section 9.04(b).

"Interest Election Request" means a request by Borrowers to convert or continue a Revolving Borrowing in accordance with Section 2.06.

"Interest Period" means with respect to any Eurodollar Borrowing, the period commencing on the date of such Borrowing and ending on the numerically corresponding day in the calendar month that is one, two, three or six months thereafter, as Borrowers may elect (or such

other period as may be elected by Borrowers and accepted by each Lender in its discretion); provided, that (i) if any Interest Period would end on a day other than a Business Day, such Interest Period shall be extended to the next succeeding Business Day unless such next succeeding Business Day would fall in the next calendar month, in which case such Interest Period shall end on the next preceding Business Day, (ii) any Interest Period pertaining to a Eurodollar Borrowing that commences on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the last calendar month of such Interest Period and (iii) no Interest Period shall extend beyond the Maturity Date. For purposes hereof, the date of a Borrowing initially shall be the date on which such Borrowing is made and, in the case of a Revolving Borrowing, thereafter shall be the effective date of the most recent conversion or continuation of such Borrowing.

"Investment" means, as to any Person, any direct or indirect acquisition or investment by such Person, whether by means of (a) the purchase or other acquisition of Equity Interests of another Person, (b) a loan, advance or capital contribution to, Guarantee or assumption of debt of, or purchase or other acquisition of any other debt or interest in, another Person, or (c) the purchase or other acquisition (in one transaction or a series of transactions) of assets of another Person that constitute a business unit or all or a substantial part of the business of, such Person.

"JPMC" means JPMorgan Chase Bank, N.A. and its successors.

"<u>Laws</u>" means, collectively, all international, foreign, Federal, state and local statutes, treaties, rules, guidelines, regulations, ordinances, codes and administrative or judicial precedents or authorities, including the interpretation or administration thereof by any Governmental Authority charged with the enforcement, interpretation or administration thereof, and all applicable administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authority, in each case whether or not having the force of law.

"<u>Leases</u>" has the meaning specified in Article I of the Mortgage with respect to each Collateral Property, but excluding the Operating Lease and Ground Lease with respect to such Collateral Property.

"Legal Requirements" means, with respect to each Collateral Property, Laws affecting any Borrower or any Manager (and for which any such party is liable under a Ground Lease) with respect to any Collateral Property or any part thereof, or the ownership, zoning, construction, use, alteration, occupancy or operation thereof, or any part thereof, whether now or hereafter enacted and in force, and all material permits, licenses and authorizations and regulations relating thereto, and all material covenants, REAs, agreements, restrictions and encumbrances contained in any instruments, either of record or known to any Borrower, at any time in force affecting such Collateral Property or any part thereof, including, without limitation, any which may (a) require repairs, modifications or alterations in or to such Collateral Property or any part thereof, or (b) in any way limit the use and enjoyment thereof.

"Lender" has the meaning specified in the introductory paragraph.

"<u>Lending Office</u>" means, as to any Lender, the office or offices of such Lender described as such in such Lender's Administrative Questionnaire, or such other office or offices as a Lender may from time to time notify Borrowers and Administrative Agent.

"LIBO Rate" means, with respect to any Eurodollar Borrowing for any Interest Period, the rate appearing on Page LIBOR 01 of the Reuters screen (or on any successor or substitute page of such page, providing rate quotations comparable to those currently provided on such page of such Service, as determined by Administrative Agent from time to time for purposes of providing quotations of interest rates applicable to dollar deposits in the London interbank market) at approximately 11:00 a.m., London time, two Business Days prior to the commencement of such Interest Period, as the rate for dollar deposits with a maturity comparable to such Interest Period. In the event that such rate is not available at such time for any reason, then the "LIBO Rate" with respect to such Eurodollar Borrowing for such Interest Period shall be the rate at which dollar deposits of \$5,000,000 and for a maturity comparable to such Interest Period are offered by the principal London office of Administrative Agent in immediately available funds in the London interbank market at approximately 11:00 a.m., London time, two Business Days prior to the commencement of such interbank market at approximately 11:00 a.m., London time, two Business Days prior to the commencement of such interbank market at approximately 11:00 a.m., London time, two Business Days prior to the commencement of such interbank market at approximately 11:00 a.m., London time, two Business Days prior to the commencement of such Interest Period.

"Licenses" has the meaning specified in Section 6.21.

"<u>Lien</u>" means any mortgage, pledge, hypothecation, assignment, deposit arrangement, lien (statutory or other), encumbrance, charge, deed of trust or other security interest, or preferential arrangement in the nature of a security interest of any kind or nature whatsoever (including any conditional sale or other title retention agreement, any easement, right of way or other encumbrance on title to real property, and any financing lease having substantially the same economic effect as any of the foregoing).

"<u>LLC Agreement</u>" has the meaning specified in Section 6.34.

"Loan" means an extension of credit by a Lender to Borrowers under Article II, including any incremental Loans made pursuant to Section 2.19.

"<u>Loan Documents</u>" means, collectively, this Agreement, the Notes, each Guaranty, the Collateral Documents, the Environmental Indemnity and all other documents executed and/or delivered in connection with the Facility.

"<u>Loan Parties</u>" means, collectively, each Borrower and the Guarantor(s), and "<u>Loan</u> <u>Party</u>" means any one of the Loan Parties.

"<u>Loan to Value Ratio</u>" means, as of any date of determination the ratio of (a) the amount of the Aggregate Commitments (as may be permanently reduced from time to time pursuant to Section 2.07) to (b) the aggregate Appraised Value of all Collateral Properties.

"Lockbox Account" has the meaning specified in Section 3.01(b).

"Lockbox Account Agreement" has the meaning specified in Section 3.01(b).

"Lockbox Bank" means JPMC.

"<u>Lockbox Period</u>" means either a Lockbox Period (SPE) or a Lockbox Period (Up), as applicable; provided that if a Lockbox Period (SPE) and a Lockbox Period (Up) shall coexist, the Lockbox Period shall continue until a Lockbox Trigger Event Cure has occurred for both Lockbox Trigger Events.

"<u>Lockbox Period</u> (SPE)" means the period commencing upon the occurrence of a Lockbox Trigger Event (SPE) and ending upon the occurrence of a Lockbox Trigger Event Cure (SPE).

"<u>Lockbox Period</u> (Up)" means the period commencing upon the occurrence of a Lockbox Trigger Event (Up) and ending upon the occurrence of a Lockbox Trigger Event Cure (Up).

"<u>Lockbox Trigger Event</u>" means either a Lockbox Trigger Event (SPE) or a Lockbox Trigger Event (Up), as applicable.

"Lockbox Trigger Event (SPE)" means the occurrence of an Event of Default.

"Lockbox Trigger Event (Up)" means, solely to the extent the same does not constitute an Event of Default hereunder, the occurrence of any of the following events with respect to FelCor Op and/or FelCor Trust: (A) filing by such Person of a petition under Section 301 of the Bankruptcy Code or otherwise voluntary commencement by such Person of any proceeding or filing by such Person of any petition seeking liquidation, reorganization or other relief under any similar Federal, state or foreign bankruptcy, insolvency, receivership or similar law now or hereafter in effect, (B) any involuntary proceeding seeking liquidation, reorganization or other relief in respect to such Person or its debts, or of a substantial part of its assets, under any Federal, state or foreign bankruptcy, insolvency, receivership or similar law now or hereafter in effect, (C) application by such Person for or consent to the appointment of a receiver, trustee, custodian, sequestrator, conservator or similar official for such Person or for a substantial part of its assets, (D) filing by such Person of an answer admitting the material allegations of a petition filed against it in any such proceeding, (E) making of a general assignment for the benefit of creditors by such Person or (F) taking of any action for the purpose of effecting any of the foregoing by such Person.

"<u>Lockbox Trigger Event Cure</u>" means either a Lockbox Trigger Event Cure (SPE) or a Lockbox Trigger Event Cure (Up), as applicable.

"<u>Lockbox Trigger Event Cure (SPE)</u>" means with respect to a Lockbox Trigger Event (SPE), the cure of such Event of Default and no other Event of Default shall have occurred and be continuing.

"<u>Lockbox Trigger Event Cure (Up)</u>" means with respect to a Lockbox Trigger Event (Up), the dismissal of such filing without prejudice.

"<u>Major Lease</u>" means: (a) any Operating Lease, (b) any lease which together with all other leases to the same tenant and to all Affiliates of such tenant, (i) provides for ten percent (10%) or more of the total gross income for such Collateral Property, (ii) covers five percent (5%) or more of the total space at such Collateral Property, in the aggregate and (iii) provides for a lease term of more than five (5) years including options to renew; and (c) any instrument guaranteeing or providing credit support for any Major Lease.

"<u>Management Agreements</u>" means, the Boca Raton Management Agreement, the Dana Point Management Agreement, the Charlotte Management Agreement, the Houston Management Agreement, the Mandalay Beach Management Agreement, the Myrtle Beach Management Agreement, the Myrtle Beach Golf Course Management Agreement, the Nashville Management Agreement, the Philadelphia Management Agreement, the Pittsburgh Management Agreement, the Santa Barbara Management Agreement and the Santa Monica Management Agreement or, if the context requires, any Replacement Management Agreement Agreement" means any of the Management Agreements.

"<u>Management Fee</u>" means, collectively, the Base Management Fees and any Incentive Management Fees payable to any Manager pursuant to its applicable Management Agreement.

"<u>Manager</u>" means, for each Collateral Property, the property manager identified in the definition of each Management Agreement, or, if the context requires, any other Qualified Manager who is managing any Collateral Property in accordance with the terms and provisions of this Agreement.

"<u>Manager Account</u>" means such account as any Manager may from time to time designate by written notice to Administrative Agent.

"<u>Manager's Consent and Subordination of Management Agreement</u>" means, for each Collateral Property, a consent of Manager and subordination in form and substance satisfactory to Administrative Agent.

"<u>Mandalay Beach Franchise Agreement</u>" means the Embassy Suites License Agreement dated December 13, 1996, as amended and assigned, by and between Hospitality Operating Lessee, as licensee, and HLTExisting Franchise Holding L.L.C., as licensor, with respect to the Mandalay Beach Property.

"<u>Mandalay Beach Management Agreement</u>" means the Management Agreement dated as of May 8, 2006, as amended and assigned, by and between Hospitality Operating Lessee, as "Owner," and Embassy Suites Management LLC, as "Manager," with respect to the Mandalay Beach Property.

"<u>Mandalay Beach Operating Lease</u>" means the Operating Lease dated as of May 8, 1996, as amended and assigned, by and between Hospitality Owner and Hospitality Operating Lessee for the Mandalay Beach Property.

"<u>Mandalay Beach Property</u>" means the property located at 2101 Mandalay Beach Road, Oxnard, CA 93035 and commonly known as Embassy Suites Mandalay Beach, California. "<u>Material Adverse Effect</u>" means (a) a material adverse change in, or a material adverse effect upon, the business, assets, properties or condition (financial or otherwise) of Guarantor and Borrowers, taken as a whole (which shall not include any such change or effect that affects the hotel or hospitality industries generally); (b) a material impairment of the rights and remedies of Administrative Agent or any Lender under any Loan Document, or of the ability of any Loan Party to perform its material obligations under any Loan Document to which it is a party; or (c) a material adverse effect upon the legality, validity, binding effect or enforceability against any Loan Party of any Loan Document to which it is a party.

"<u>Material Indebtedness</u>" means Recourse Indebtedness (other than the Loans), or obligations in respect of one or more Swap Contracts, of any one or more of the Guarantor and its Subsidiaries in an aggregate principal amount exceeding \$50,000,000. For purposes of determining Material Indebtedness, the "principal amount" of the obligations of the Guarantor or any Subsidiary in respect of any Swap Contract at any time shall be the maximum aggregate amount (giving effect to any netting agreements) that the Guarantor or such Subsidiary would be required to pay if such Swap Contract were terminated at such time.

"<u>Material Property Event</u>" means, with respect to any Collateral Property, the occurrence of any event or circumstance that could reasonably be expected to result in a (a) material adverse effect with respect to the financial condition or the operations of such Collateral Property, (b) material adverse effect on the Appraised Value of such Collateral Property (which shall not include any such change in Appraised Value due to a general decline in the relevant market for such real property) or (c) material adverse effect on the title of such Collateral Property.

"<u>Material Title Defects</u>" means, with respect to any Collateral Property, defects, Liens (other than Liens for local real estate taxes and similar local governmental charges and Permitted Liens), and other encumbrances in the nature of easements, servitudes, restrictions, and rights-of-way that would customarily be deemed unacceptable title exceptions for a prudent lender (*i.e.*, a prudent lender would reasonably determine that such exceptions, individually or in the aggregate, materially impair the title, ownership, use, value or operations of the Collateral Property in question) or which could reasonably be expected to result in a Material Property Event).

"<u>Maturity Date</u>" means the earliest of (a) the Stated Maturity Date, and (b) the date upon which Administrative Agent declares the Obligations due and payable after the occurrence of an Event of Default.

"Maximum Rate" has the meaning specified in Section 12.10.

"<u>Member</u>" has the meaning specified in Section 6.34.

"Monthly Capital Expenditures Reserve Deposit" shall mean the greater of (i) one twelfth (1/12) of the amount for FF&E Expenditures set forth in the Approved Annual Budget for such Fiscal Year; and (ii) the quotient obtained by dividing (A) the aggregate Gross Income from Operations for the Properties still subject to the Lien of a Security Instrument for the preceding calendar year (as reflected in Borrower's annual operating statements as approved and accepted by Administrative Agent) multiplied by four percent (4%) by (B) twelve (12). The Monthly Capital Expenditures Reserve Deposit shall be adjusted annually.

"<u>Monthly Insurance Premium Deposit</u>" means one-twelfth of the Insurance Premiums that Administrative Agent determines based on the Approved Annual Budget from time to time will be payable for the renewal of the coverage afforded by the Policies upon the expiration thereof in order to accumulate with Administrative Agent sufficient funds to pay all such Insurance Premiums as and when due, together with such additional amount as may be specified by Administrative Agent in any given month to meet the Insurance Premiums due and payable that month.

"<u>Monthly Tax Deposit</u>" means one-twelfth of the Taxes that Administrative Agent determines based on the Approved Annual Budget from time to time will be payable in order to accumulate with Administrative Agent sufficient funds to pay all such Taxes as and when due, together with such additional amount as may be specified by Administrative Agent in any given month to meet the Taxes due and payable that month.

"Moody's" means Moody's Investors Service, Inc.

"<u>Mortgage</u>" means, with respect to each Collateral Property, that certain first priority (Fee and/or Leasehold) Mortgage (or Deed of Trust), Assignment of Leases and Rents, Security Agreement and Fixture Filing executed and delivered by a Borrower (and where applicable, Philadelphia Affiliate Ground Lessor and Boca Raton Affiliate Ground Lessor) with respect to any Collateral Property in favor of Administrative Agent for the benefit of the Secured Parties as security for the Obligations and encumbering such Collateral Property.

"<u>MSA</u>" means metropolitan statistical area.

"<u>Myrtle Beach Golf Course</u>" means the Arcadian Shores Golf Club, located in Myrtle Beach, South Carolina.

"<u>Myrtle Beach Golf Course Lease</u>" means the Ground Lease dated as of March 1, 1972, as amended and assigned, by and between Myrtle Beach Golf Course Lessor and Hospitality Owner for the Myrtle Beach Golf Course.

"<u>Myrtle Beach Golf Course Lessor</u>" means Burroughs & Chapin Company, Inc., and any successor thereto.

"<u>Myrtle Beach Golf Course Management Agreement</u>" means the Golf Facility Management Agreement dated as of May 12, 2009, by and between Hospitality Operating Lessee, as owner, and, Myrtle Beach Golf Course Manager, as manager, with respect to the Myrtle Beach Golf Course.

"<u>Myrtle Beach Management Agreement</u>" means the Management Agreement dated as of July 19, 2002, as amended and assigned, by and between Hospitality Operating Lessee, as "Owner," and Hilton Hotels Corporation, as "Manager," with respect to the Myrtle Beach Property, but excluding the Myrtle Beach Golf Course for periods during the effectiveness of the Myrtle Beach Golf Course Management Agreement. "<u>Myrtle Beach Golf Course Manager</u>" means Burroughs & Chapin Golf Management Company, LLC, and any successor thereto.

"<u>Myrtle Beach Operating Lease</u>" means the Operating Lease dated as of July 19, 2002, as amended and assigned, by and between Hospitality Owner and Hospitality Operating Lessee for the Myrtle Beach Property.

"<u>Myrtle Beach Property</u>" means the property located at 10000 Beach Club Dr., Myrtle Beach, South Carolina 29572 and commonly known as Hilton Myrtle Beach, South Carolina, including the Myrtle Beach Golf Course.

"<u>Nashville Franchise Agreement</u>" means the Embassy Suites License Agreement dated July 28, 1994, as amended and assigned, by and between Hospitality Operating Lessee, as licensee, and HLT Existing Franchise Holding L.L.C., as licensor, with respect to the Nashville Property.

"<u>Nashville Management Agreement</u>" means the Management Agreement dated as of July 28, 2004, as amended and assigned, by and between Hospitality Operating Lessee, as "Owner," and Embassy Suites Management, LLC, as "Manager," with respect to the Nashville Property.

"<u>Nashville Operating Lease</u>" means the Operating Lease dated as of July 28, 1994, as amended and assigned, by and between Hospitality Owner and Hospitality Operating Lessee for the Nashville Property.

"<u>Nashville</u> <u>Property</u>" means the property located at 10 Century Blvd., Nashville, Tennessee 37214 and commonly known as Embassy Suites Nashville, Tennessee.

"<u>Net Operating Income</u>" means the amount obtained by subtracting Operating Expenses from Gross Income from Operations.

"<u>Net Proceeds</u>" has the meaning specified in Section 9.04(b).

"<u>Net Proceeds Deficiency</u>" has the meaning specified in Section 9.04(h).

"<u>Net Release Proceeds</u>" means (a) in connection with any sale of any of the Collateral Property the proceeds thereof received by the applicable Borrower in the form of cash, net of attorneys' fees, accountants' fees, investment banking fees, proceeds of rental and business interruption insurance, and other customary fees and expenses actually incurred in connection therewith, not to exceed six percent (6%) of the total proceeds and net of taxes paid or reasonably estimated to be payable as a result thereof (after taking into account any available tax credits or deductions and any tax sharing arrangements); provided that such 6% cap shall exclude amounts payable as termination fees under any applicable Management Agreements and (b) in connection with any refinancing of any of the Collateral Properties, the cash proceeds received by the applicable Borrower from such refinancing, net of attorneys' fees, investment banking fees, accountants' fees, underwriting discounts, and commissions and other customary fees and expenses actually incurred in connection therewith, not to exceed three percent (3%) of the total proceeds. "<u>Nonrecourse Indebtedness</u>" means, with respect to a Person, Indebtedness for borrowed money in respect of which recourse for payment (except for customary exceptions for fraud, misapplication of funds, environmental indemnities, violation of "special purpose entity" covenants, bankruptcy, insolvency, receivership or other similar events and other similar exceptions to recourse liability until a claim is made with respect thereto, and then such Indebtedness shall not constitute "Nonrecourse Indebtedness" only to the extent of the amount of such claim) is contractually limited to specific assets of such Person encumbered by a Lien securing such Indebtedness.

"<u>Note</u>" means a promissory note made by Borrowers in favor of a Lender evidencing Loans made by such Lender substantially in the form of Exhibit B-1.

"<u>O&M Program</u>" means, with respect to each of the Dana Point Property, the Houston Property, the Myrtle Beach Property and the Mandalay Beach Property, the asbestos operations and maintenance program required for such Collateral Property under Section 7.24, as developed by Borrowers and approved by Administrative Agent, as the same may be amended, replaced, supplemented or otherwise modified from time to time with the consent of Administrative Agent, which shall not be unreasonably withheld.

"<u>Obligations</u>" means all advances to, and debts, liabilities, obligations, covenants and duties of, any Loan Party arising under any Loan Document or otherwise with respect to any Loan, whether direct or indirect (including those acquired by assumption), absolute or contingent, due or to become due, now existing or hereafter arising and including interest and fees that accrue after the commencement by or against any Loan Party or any Subsidiary thereof of any proceeding under any Debtor Relief Laws naming such Person as the debtor in such proceeding, regardless of whether such interest and fees are allowed claims in such proceeding.

"<u>Officer's Certificate</u>" means the certificate of a Responsible Officer of the applicable Borrower, which shall be in form and substance reasonably satisfactory to Administrative Agent, and in the case of Sections 2.16(b), 2.17(g), 5.01(a)(viii), 5.02(e) and 7.10(b) and (c), shall be in the form of Exhibits E-1, <u>E-</u>2 or <u>E</u>-3, as applicable.

"<u>Operating Equipment and Supplies</u>" means all chinaware, glassware, linens, silverware, tools, kitchen utensils, uniforms, engineering and housekeeping tools and utensils, food and beverage items, fuel, soap, mechanical stores, cleaning supplies and materials, matches, stationary, paper supplies, laundry supplies, food service preparation utensils, housekeeping supplies, accounting supplies and other immediately consumable items used in the operation of a Collateral Property.

"<u>Operating Expenses</u>" means the total of all expenditures, computed in accordance with GAAP, of whatever kind relating to the operation, maintenance and management of the Collateral Properties that are incurred on a regular monthly or other periodic basis, including without limitation, utilities, ordinary repairs and maintenance, insurance premiums, license fees, property taxes and assessments, advertising expenses, Management Fees, franchise fees, payroll and related taxes, computer processing charges, operational equipment or other lease payments permitted hereunder, and other similar costs, but excluding depreciation, Debt Service and Capital Expenditures. "<u>Operating Lease Subordination Agreements</u>" means subordination agreements in form and substance satisfactory to Administrative Agent relating to each Operating Lease.

"<u>Operating Leases</u>" means the Boca Raton Operating Lease, the Dana Point Operating Lease, the Charlotte Operating Lease, the Houston Operating Lease, the Mandalay Beach Operating Lease, the Myrtle Beach Operating Lease, the Nashville Operating Lease, the Philadelphia Operating Lease, the Pittsburgh Operating Lease, the Santa Barbara Operating Lease and the Santa Monica Operating Lease, and "<u>Operating Lease</u>" means any one of the Operating Leases.

"<u>Operating Lessees</u>" has the meaning specified in the introductory paragraph.

"Organizational Documents" means, (a) with respect to any corporation, the certificate or articles of incorporation and the bylaws (or equivalent or comparable constitutive documents with respect to any non-U.S. jurisdiction); (b) with respect to any limited liability company, the certificate or articles of formation or organization and operating agreement; and (c) with respect to any partnership, joint venture, trust or other form of business entity, the partnership, joint venture or other applicable agreement of formation or organization and any agreement, instrument, filing or notice with respect thereto filed in connection with its formation or organization and, if applicable, any certificate or articles of formation or organization of such entity.

"<u>Other Charges</u>" means all personal property taxes, ground rents, maintenance charges, impositions other than Taxes and any other charges, including, without limitation, vault charges and license fees for the use of vaults, chutes and similar areas adjoining any Collateral Property, now or hereafter levied or assessed or imposed against such Collateral Property or any part thereof.

"<u>Other Taxes</u>" means all present or future stamp or documentary taxes or any other excise or property taxes, charges or similar levies arising from any payment made hereunder or under any other Loan Document or from the execution, delivery or enforcement of, or otherwise with respect to, this Agreement or any other Loan Document.

"<u>Outstanding Amount</u>" means the aggregate outstanding principal amount of Loans after giving effect to any Borrowings and prepayments or repayments as the case may be, occurring on such date.

"<u>Overhead Sharing Agreement</u>" means that certain Overhead Sharing Agreement dated as of January 1, 2011, by and among Borrowers, FelCor Op and the certain other parties thereto.

"Owners" has the meaning specified in the introductory paragraph.

"<u>Parent</u>" means, with respect to any Lender, any Person as to which such Lender is, directly or indirectly, a subsidiary.

"<u>Participant</u>" has the meaning specified in Section 12.06(d).

"Patriot Act" has the meaning specified in Section 12.18.

"<u>Payment Date</u>" means the first (1st) day of each calendar month, or if such day is not a Business Day, the immediately succeeding Business Day, and the Maturity Date.

"Permitted Investments" means:

(a) direct obligations of, or obligations the principal of and interest on which are unconditionally guaranteed by, the United States of America (or by any agency thereof to the extent such obligations are backed by the full faith and credit of the United States of America), in each case maturing within one year from the date of acquisition thereof;

(b) investments in commercial paper maturing within 270 days from the date of acquisition thereof and having, at such date of acquisition, the highest credit rating obtainable from S&P or from Moody's;

(c) investments in certificates of deposit, banker's acceptances and time deposits maturing within 180 days from the date of acquisition thereof issued or guaranteed by or placed with, and money market deposit accounts issued or offered by, any domestic office of any commercial bank organized under the laws of the United States of America or any State thereof which has a combined capital and surplus and undivided profits of not less than \$500,000,000;

(d) fully collateralized repurchase agreements with a term of not more than 30 days for securities described in clause (a) above and entered into with a financial institution satisfying the criteria described in clause (c) above; and

(e) money market funds that (i) comply with the criteria set forth in Securities and Exchange Commission Rule 2a-7 under the Investment Company Act of 1940, (ii) are rated AAA by S&P and Aaa by Moody's and (iii) have portfolio assets of at least \$5,000,000,000.

"<u>Permitted Lien</u>" has the meaning specified in Section 8.03.

"Permitted Personal Property Lien" has the meaning specified in Section 8.04.

"<u>Person</u>" means any natural person, corporation, limited liability company, trust, joint venture, association, company, partnership, Governmental Authority or other entity.

"<u>Personal Property</u>" has the meaning specified in Article I of the Mortgage with respect to each Collateral Property.

"<u>Philadelphia Affiliate Ground Lease</u>" means the Ground Lease, dated as of June 5, 2009, between the Philadelphia Affiliate Ground Lessor and Hospitality Owner.

"<u>Philadelphia Affiliate Ground Lessor</u>" means FelCor Pennsylvania Company, L.L.C., a Delaware limited liability company, and any successor thereto as permitted pursuant to this Agreement.

"<u>Philadelphia Management Agreement</u>" means the Management Agreement dated as of June 30, 2001, as amended and assigned, by and between Hospitality Operating Lessee, as "Owner," and IHG, as "Manager," with respect to the Philadelphia Property.

"<u>Philadelphia</u> <u>Operating Lease</u>" means the Operating Lease dated as of July 29, 1998, as amended and assigned, by and between Hospitality Owner and Hospitality Operating Lessee for the Philadelphia Property.

"<u>Philadelphia Property</u>" means the property located at 400 Arch Street, Philadelphia, PA 19106 and commonly known as Holiday Inn Philadelphia, Pennsylvania.

"<u>Physical Condition Report</u>" means, with respect to each Collateral Property, a structural engineering report prepared by a company satisfactory to Administrative Agent regarding the physical condition of such Collateral Property, satisfactory in form and substance to Administrative Agent in its reasonable discretion, which report shall, among other things, (a) confirm that such Collateral Property and its use complies, in all material respects, with all applicable Legal Requirements (including, without limitation, zoning, subdivision and building laws) and (b) include a copy of a final certificate of occupancy with respect to all Improvements on such Collateral Property.

"<u>Pittsburgh Ground Lease</u>" means the Ground Lease dated as of January 29, 1988 as amended and assigned, between Pittsburgh Ground Lessor and Hospitality Owner.

"<u>Pittsburgh Ground Lessor</u>" means the Trustees of the Masonic Temple Trust and The Masonic Fund Society for the County of Allegheny, and any successor thereto.

"<u>Pittsburgh Management Agreement</u>" means the Management Agreement dated as of June 30, 2001, as amended by First Amendment to Management Agreement dated as of July 1, 2001 as further amended and assigned, by and between Hospitality Operating Lessee, as "Owner," and IHG, as "Manager," with respect to the Pittsburgh Property.

"<u>Pittsburgh Operating Lease</u>" means the Lease Agreement dated as of October 27, 1998, as amended and assigned, by and between Hospitality Owner and Hospitality Operating Lessee for the Pittsburgh Property.

"<u>Pittsburgh Property</u>" means the property located at 100 Lytton Avenue, Pittsburgh, Pennsylvania 15213 and commonly known as Holiday Inn Pittsburgh at University Center. "<u>Plan</u>" means any "employee benefit plan" (as such term is defined in Section 3(3) of ERISA) established by a Borrower or, with respect to any such plan that is subject to Section 412 of the Code or Title IV of ERISA, any ERISA Affiliate.

"<u>Plan Assets</u>" has the meaning set forth in 29 C.F.R. section 25 10.3-101, as modified by Section 3(42) of ERISA.

"Platform" means IntraLinks or another similar electronic system.

"<u>Pledge Agreement</u>" means with respect to each Borrower's Principal, that certain Pledge Agreement executed and delivered by such Principal with respect to its equity interest in the applicable Borrower, in favor of Administrative Agent for the benefit of the Secured Parties.

"Policy" has the meaning specified in Section 9.01(b).

"<u>Prime Rate</u>" means the rate of interest per annum publicly announced from time to time by JPMorgan Chase Bank, N.A. as its prime rate in effect at its office located at 270 Park Avenue, New York, New York; each change in the Prime Rate shall be effective from and including the date such change is publicly announced as being effective.

"<u>Principal</u>" means, with respect to any limited liability company, its sole member; and with respect to any limited partnership, its general partner.

"Property Accounts" has the meaning specified in Section 3.01(a).

"<u>Qualified Insurer</u>" has the meaning specified in Section 9.01(b).

"<u>Qualified Manager</u>" means any Manager or another reputable and experienced professional management organization (a) which manages, together with its Affiliates, one hundred fifty (150) properties of a type, quality and size similar to the Collateral Properties, totaling in the aggregate no less than 30,000 guest rooms and (b) prior to whose employment as manager of a Collateral Property shall have been approved by Administrative Agent, which approval shall not be unreasonably withheld, delayed or conditioned. For purposes of this Agreement, Aimbridge Hospitality, Davidson Hotel Company, Highgate Holdings, Inc., Interstate Management Company L.L.C. and Wyndham Worldwide Corporation are deemed to be Qualified Managers.

"<u>Rating Agencies</u>" means each of S&P, Moody's, and Fitch, and any other nationallyrecognized statistical rating agency which has been approved by Administrative Agent.

"<u>REAs</u>" means each construction, operation, and reciprocal easement agreements or similar agreements (including any separate agreements or other agreements between a Loan Party and one or more other parties to any REA with respect to such REA) affecting any Collateral Property or portion thereof, and "<u>REA</u>" means any one of the REAs.

"<u>Recourse Indebtedness</u>" means any Indebtedness which is not Nonrecourse Indebtedness. If any Indebtedness is partially Nonrecourse Indebtedness and partially Recourse Indebtedness, only the portion that is Recourse Indebtedness shall be included as Recourse Indebtedness for purposes hereof. "<u>Register</u>" has the meaning specified in Section 12.06(c).

"<u>Related Parties</u>" means, with respect to any Person, such Person's Affiliates and the partners, members, directors, officers, employees, agents and advisors of such Person and of such Person's Affiliates.

"<u>Release</u>" of any Hazardous Materials means any release, deposit, discharge, emission, leaking, spilling, seeping, migrating, injecting, pumping, pouring, emptying, escaping, dumping, disposing or other movement of Hazardous Materials.

"<u>Release Price</u>" means for a Collateral Property as of any date of determination, an amount equal to one hundred ten percent (110%) of the Allocated Loan Amount for such Collateral Property.

"<u>Renewal Lease</u>" has the meaning specified in Section 7.16(a).

"<u>Rents</u>" has the meaning specified in Article I of the Mortgage with respect to each Collateral Property.

"Replacement Franchise Agreement" means, with respect to any Franchised Property, (a) either (i) a franchise agreement with the applicable Franchisor substantially in the same form and substance as the initial Franchise Agreement for such Franchised Property and on the same or more favorable terms to the applicable Operating Lessee, or (ii) a franchise agreement with the applicable Franchisor, which franchise agreement shall be acceptable to Administrative Agent in form and substance, and (b) a franchisor estoppel and recognition agreement or other "comfort letter" substantially in the form delivered to Administrative Agent on the date hereof (or such other form acceptable to Administrative Agent), executed and delivered to Administrative Agent by the applicable Operating Lessee and the applicable Franchisor at Borrowers' expense; provided, however, with respect to any expiring or replacement Franchise Agreement, Borrowers shall notify Administrative Agent, but shall not be required to obtain Administrative Agent's consent in the event that the Franchise Agreement in effect on the date hereof is extended on the same or more favorable terms to the applicable Operating Lessee as prior to the expiration thereof, provided further that if such Franchise Agreement is modified, extended supplemented or replaced a franchisor estoppel and recognition or other "comfort letter" shall be provided as described in (b) above.

"<u>Replacement Management Agreement</u>" means, collectively, (a) either (i) a management agreement with a Qualified Manager substantially in the same form and substance as the initial Management Agreement for such Collateral Property and on the same or more favorable terms to the applicable Operating Lessee, or (ii) a management agreement with a Qualified Manager, which management agreement shall be acceptable to Administrative Agent in form and substance, (b) if requested by Administrative Agent, a conditional assignment of management agreement in such form acceptable to Administrative Agent executed and delivered to Administrative Agent by the applicable Operating Lessee and such Qualified Manager at Borrowers' expense; provided, however, with respect to any expiring or replacement Management Agreement, Borrowers shall notify Administrative Agent but shall not be required to obtain Administrative Agent's consent in the event that the Management Agreement in effect on the date hereof is extended on the same or more favorable terms to the applicable Operating Lessee as prior to the expiration thereof.

"<u>Required Lenders</u>" means, as of any date of determination, Lenders having Revolving Credit Exposure and unused Commitments representing more than fifty and one hundredth percent (50.01%) of the sum of the total Revolving Credit Exposures and unused Commitments at such time; provided that the portion of the Revolving Credit Exposures and unused Commitments held or deemed held by, any Defaulting Lender shall be excluded for purposes of making a determination of Required Lenders.

"<u>Reserve Accounts</u>" has the meaning specified in Section 3.01(c).

"<u>Responsible Officer</u>" means the president or any vice president of a Loan Party. Any document delivered hereunder that is signed by a Responsible Officer of a Loan Party shall be conclusively presumed to have been authorized by all necessary organizational action on the part of such Loan Party and such Responsible Officer shall be conclusively presumed to have acted on behalf of such Loan Party.

"<u>Restoration</u>" means the repair and restoration of a Collateral Property after a Casualty or Condemnation as nearly as possible to the condition the Collateral Property was in immediately prior to such Casualty or Condemnation, with such alterations as may be approved by Administrative Agent, and in accordance with applicable Laws and the requirements of any applicable Management Agreement, Franchise Agreement and Operating Lease.

"<u>Restricted Party</u>" means any Borrower, any Principal of a Borrower or any Affiliated Manager or any shareholder, partner or member or any direct or indirect legal or beneficial owner of, any Borrower, any Principal of a Borrower or any Affiliated Manager; provided, however, that for purposes of Section 8.03 only in no event shall (a) FelCor Trust or FelCor Op be deemed a Restricted Party, nor (b) any direct or indirect beneficial owner of the entities listed in (a) be deemed a Restricted Party solely because of its direct or indirect beneficial ownership of such entities.

"<u>Restricted Payment</u>" means any dividend or other distribution (whether in cash, securities or other property) with respect to any capital stock or other Equity Interest of any Person or any of its Subsidiaries, or any payment (whether in cash, securities or other property), including any sinking fund or similar deposit, on account of the purchase, redemption, retirement, defeasance, acquisition, cancellation or termination of any such capital stock or other Equity Interest, or on account of any return of capital to any Person's stockholders, partners or members (or the equivalent of any thereof), or any option, warrant or other right to acquire any such dividend or other distribution or payment, or any intercompany advance to an Affiliate of a Person as reflected on the books and records of such Person.

"<u>Revolving Credit Exposure</u>" means, with respect to any Lender at any time, the sum of the outstanding principal amount of such Lender's Revolving Loans and Swingline Exposure at such time.

"Revolving Loan" means a Loan made pursuant to Section 2.01.

"S&P" means Standard & Poor's Ratings Services, a division of McGraw-Hill, Inc.

"Sale Triggered Release" has the meaning specified in Section 2.17.

"<u>Santa Barbara Management Agreement</u>" means the Management Agreement dated as of June 30, 2001, as amended by First Amendment to Management Agreement dated as of July 1, 2001 as further amended and assigned, by and between Hospitality Operating Lessee, as "Owner," and IHG, as "Manager," with respect to the Santa Barbara Property.

"<u>Santa Barbara Operating Lease</u>" means the Lease Agreement dated as of July 27, 1998, as amended and assigned, by and between Hospitality Owner and Hospitality Operating Lessee for the Santa Barbara Property.

"<u>Santa Barbara</u> Pr<u>operty</u>" means the property located at 5650 Calle Real, Goleta, California 93117 and commonly known as Holiday Inn Santa Barbara/Goleta.

"Santa Monica Management Agreement" means the Management Agreement dated as of March 11, 2004, as amended and assigned, by and between Hospitality Operating Lessee, as "Owner," and IHG, as "Manager," with respect to the Santa Monica Property.

"Santa Monica Operating Lease" means the Operating Lease dated as of March 1, 2004, as amended and assigned, by and between Hospitality Owner and Hospitality Operating Lessee for the Santa Monica Property.

"<u>Santa Monica Property</u>" means the property located at 120 Colorado Avenue, Santa Monica, California 90401 and commonly known as Holiday Inn Santa Monica, California.

"<u>Secured Parties</u>" means, collectively, Administrative Agent, Lenders, each co-agent or sub-agent appointed by Administrative Agent from time to time pursuant to Section 11.05, and the other Persons the Obligations owing to which are or are purported to be secured by the Collateral under the terms of the Collateral Documents.

"Security Deposit" has the meaning specified in Section 7.16(e).

"<u>Senior Secured Notes Indenture</u>" means the Indenture dated as of October 1, 2009, among the FelCor Op (as assignee of FelCor Escrow Holdings, L.L.C.), FelCor Trust and its subsidiaries, as guarantors, and U.S. Bank National Association, as trustee and collateral agent, as modified by a First Supplemental Indenture dated as of October 12, 2009, a Second Supplemental Indenture dated as of October 13, 2009, a Third Supplemental Indenture dated as of March 23, 2010 and a Fourth Supplemental Indenture dated as of March 3, 2011, pursuant to which the Senior Secured Notes are issued.

"<u>Senior Secured Notes</u>" means the 10% Senior Secured Notes Due 2014 of FelCor Op, issued pursuant to the Senior Secured Notes Indenture. "Special Member" has the meaning specified in Section 6.34.

"<u>Stated Maturity Date</u>" means, August 1, 2014, or such later date as may be extended pursuant to Section 2.16.

"<u>Statutory Reserve Rate</u>" means a fraction (expressed as a decimal), the numerator of which is the number one and the denominator of which is the number one minus the aggregate of the maximum reserve percentages (including any marginal, special, emergency or supplemental reserves) expressed as a decimal established by the Board to which Administrative Agent is subject for eurocurrency funding (currently referred to as "Eurocurrency Liabilities" in Regulation D of the Board). Such reserve percentages shall include those imposed pursuant to such Regulation D. Eurodollar Loans shall be deemed to constitute eurocurrency funding and to be subject to such reserve requirements without benefit of or credit for proration, exemptions or offsets that may be available from time to time to any Lender under such Regulation D or any comparable regulation. The Statutory Reserve Rate shall be adjusted automatically on and as of the effective date of any change in any reserve percentage.

"<u>Subsidiary</u>" of a Person means a corporation, partnership, joint venture, limited liability company or other business entity of which a majority of the shares of securities or other interests having ordinary voting power for the election of directors or other governing body (other than securities or interests having such power only by reason of the happening of a contingency) are at the time beneficially owned, or the management of which is otherwise controlled, directly, or indirectly through one or more intermediaries, or both, by such Person. Unless otherwise specified, all references herein to a "<u>Subsidiary</u>" or to "<u>Subsidiaries</u>" shall refer to a Subsidiary or Subsidiaries of Borrowers.

"Super-Majority Lenders" means, as of any date of determination, Lenders having Revolving Credit Exposure and unused Commitments representing more than sixty-six and twothirds percent (66-2/3%) of the sum of the total Revolving Credit Exposures and unused Commitments at such time; provided that the portion of the Revolving Credit Exposures and unused Commitments held or deemed held by, any Defaulting Lender shall be excluded for purposes of making a determination of Super-Majority Lenders.

"<u>Survey</u>" has the meaning specified in Section 5.01(a)(x).

"Swap Contract" means (a) any and all rate swap transactions, basis swaps, credit derivative transactions, forward rate transactions, commodity swaps, commodity options, forward commodity contracts, equity or equity index swaps or options, bond or bond price or bond index swaps or options or forward bond or forward bond price or forward bond index transactions, interest rate options, forward foreign exchange transactions, cap transactions, floor transactions, collar transactions, currency swap transactions, cross-currency rate swap transactions, currency options, spot contracts, or any other similar transactions or any combination of any of the foregoing (including any options to enter into any of the foregoing), whether or not any such transaction is governed by or subject to any master agreement, and (b) any and all transactions of any kind, and the related confirmations, which are subject to the terms and conditions of, or governed by, any form of master agreement published by the International Swaps and Derivatives Association, Inc., any International Foreign Exchange Master Agreement, or any other master agreement, including any such obligations or liabilities under any master agreement.

"Swap Termination Value" means, in respect of any one or more Swap Contracts, after taking into account the effect of any legally enforceable netting agreement relating to such Swap Contracts, (a) from the date such Swap Contracts have been closed out and termination value (s) determined in accordance therewith, such termination value(s), and (b) for any date prior to the date referenced in clause (a), the amount(s) determined as the mark-to-market value(s) for such Swap Contracts, as determined based upon one or more mid-market or other readily available quotations provided by any recognized dealer in such Swap Contracts (which may include a Lender or any Affiliate of a Lender).

"<u>Swingline Exposure</u>" means, at any time, the aggregate principal amount of all Swingline Loans outstanding at such time. The Swingline Exposure of any Lender at any time shall be its Applicable Percentage of the total Swingline Exposure at such time.

"Swingline Lender" means JPMorgan Chase Bank, in its capacity as lender of Swingline Loans hereunder.

"Swingline Loan" means a Loan made pursuant to Section 2.04.

"<u>Swingline Note</u>" means a promissory note made by Borrowers in favor of Swingline Lender evidencing Swingline Loans made by such Swingline Lender substantially in the form of Exhibit B-2.

"<u>Synthetic Debt</u>" means, with respect to any Person as of any date of determination thereof, all obligations of such Person in respect of transactions entered into by such Person that are intended to function primarily as a borrowing of funds (including any minority interest transactions that function primarily as a borrowing) but are not otherwise included in the definition of "<u>Indebtedness</u>" or as a liability on the consolidated balance sheet of such Person and its Subsidiaries in accordance with GAAP.

"<u>Synthetic Lease Obligation</u>" means the monetary obligation of a Person under (a) a so-called synthetic, off-balance sheet or tax retention lease, or (b) an agreement for the use or possession of property (including sale and leaseback transactions), in each case, creating obligations that do not appear on the balance sheet of such Person but which, upon the application of any Debtor Relief Laws to such Person, would be characterized as the indebtedness of such Person (without regard to accounting treatment).

"Tax Account" has the meaning specified in Section 3.01(c)(i).

"<u>Taxes</u>" means all present or future taxes, levies, imposts, duties, deductions, withholdings, assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.

"<u>Terrorism Exclusion</u>" has the meaning specified in Section 9.01(a)(x).

"<u>Terrorism Insurance</u>" has the meaning specified in Section 9.01(a)(x).

"<u>Terrorism Insurance Cap</u>" has the meaning specified in Section 9.01(a)(x).

(a)(x).

"Title Company" means such title insurance companies acceptable to Administrative

"Terrorism Insurance Required Amount" has the meaning specified in Section 9.01

Agent.

"<u>Title Insurance Commitments</u>" means the commitments to issue the Title Policies, issued by the Title Company for each Collateral Property, along with copies of all instruments creating or evidencing exceptions or encumbrances to title.

"<u>Title Policies</u>" means an ALTA or equivalent form of Mortgagee Title Policy from the Title Company and insuring the priority and sufficiency of the Mortgages as first Liens upon the applicable Collateral Properties, (a) in an aggregate amount acceptable to Administrative Agent, (b) showing all easements or other matters affecting the Collateral Properties, all subject only to such exceptions or qualifications as are reasonably acceptable to Administrative Agent, (c) insuring the priority of Secured Party's Liens granted by the Mortgages against all possible contractors', suppliers, and mechanics' lien claims that heretofore or hereafter arise, as well as survey matters which could result in a Material Title Defect, and (d) to the extent available, containing any customary endorsements or assurances that Administrative Agent acting on behalf of the Secured Parties may request for protection of its interests including, but not limited to (i) zoning endorsements, (ii) variable rate endorsements, (iii) usury endorsements, (iv) comprehensive endorsements, (v) access endorsements, insuring that there will be at least one location at each Collateral Property with unlimited vehicular ingress and egress to an adjacent street, and (vi) other customary endorsements requested by Administrative Agent and its counsel.

"Transfer" means, with respect to any property, rights, or interests, any sale, deed, conveyance, lease, mortgage, grant, bargain, encumbrance, pledge, assignment, grant of options with respect to, installment sales contracts for, or other transfer or disposition, in whole or in part, with respect to any legal or beneficial interest therein, directly or indirectly, voluntarily or involuntarily, by operation of law or otherwise, and whether or not for consideration or of record. With respect to Restricted Parties, the term "Transfer" shall include the following: (a) if a Restricted Party is a corporation, any merger, consolidation or sale or pledge of such corporation's stock or the creation or issuance of new stock; (b) if a Restricted Party is a limited or general partnership or joint venture, any merger or consolidation or the change, removal, resignation or addition of a general partner or the sale or pledge of the partnership interest of any general partner or any profits or proceeds relating to such partnership interest, or the sale or pledge of limited partnership interests or any profits or proceeds relating to such limited partnership interests or the creation or issuance of new limited partnership interests; (c) if a Restricted Party is a limited liability company, any merger or consolidation or the change, removal, resignation or addition of a managing member (or if no managing member, any member) or the sale or pledge of the membership interest of a managing member (or if no managing member, any member) or any profits or proceeds relating to such membership interest, or the sale or pledge of non-managing membership interests or the creation or issuance of new non-managing membership interests; and (d) if a Restricted Party is a trust or nominee trust, any merger, consolidation or the sale or pledge of the legal or beneficial interest in a Restricted Party or the creation or issuance of new legal or beneficial interests.

"<u>Type</u>", when used in reference to any Loan or Borrowing, refers to whether the rate of interest on such Loan, or on the Loans comprising such Borrowing, is determined by reference to the Adjusted LIBO Rate or the Alternate Base Rate.

"<u>UCC</u>" means the Uniform Commercial Code as in effect in the State of New York; provided that, if perfection or the effect of perfection or non-perfection or the priority of any security interest in any Collateral is governed by the Uniform Commercial Code as in effect in a jurisdiction other than the State of New York, "<u>UCC</u>" means the Uniform Commercial Code as in effect from time to time in such other jurisdiction for purposes of the provisions hereof relating to such perfection, effect of perfection or non-perfection or priority.

"<u>Unaudited Financial Statements</u>" means, if available, the pro-forma unaudited balance sheets of each Borrower and its Subsidiaries for the Fiscal Year ended December 31, 2010, and the related statements of income or operations for such period of each Borrower.

"<u>United States</u>" and "<u>U.S.</u>" mean the United States of America.

"2009 <u>Term Loan</u>" means the Indebtedness of the Hospitality Borrowers outstanding under that certain Term Loan Agreement, dated as of June 12, 2009, by and among the Hospitality Borrowers, Administrative Agent and the lenders party thereto.

"2010 Term Loan" means the Indebtedness of the Boca Borrowers outstanding under that certain Term Loan Agreement, dated as of August 19, 2010, by and among the Boca Borrowers, Administrative Agent and the lenders party thereto.

1.0.2. <u>Other Interpretive Provisions</u>. With reference to this Agreement and each other Loan Document, unless otherwise specified herein or in such other Loan Document:

The definitions of terms herein shall apply equally to the singular and plural (a) forms of the terms defined. Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms. The words "include," "includes" and "including" shall be deemed to be followed by the phrase "without limitation." The word "will" shall be construed to have the same meaning and effect as the word "shall." Unless the context requires otherwise, (i) any definition of or reference to any agreement, instrument or other document (including any Organizational Document) shall be construed as referring to such agreement, instrument or other document as from time to time amended, supplemented or otherwise modified (subject to any restrictions on such amendments, supplements or modifications set forth herein or in any other Loan Document), (ii) any reference herein to any Person shall be construed to include such Person's successors and assigns, (iii) the words "herein," "hereof" and "hereunder," and words of similar import when used in any Loan Document, shall be construed to refer to such Loan Document in its entirety and not to any particular provision thereof, (iv) all references in a Loan Document to Articles, Sections, Preliminary Statements, Exhibits and Schedules shall be construed to refer to Articles and Sections of, and Preliminary Statements, Exhibits and Schedules to, the Loan Document in which such references appear, (v) any reference to any law shall include all statutory and regulatory provisions consolidating, amending, replacing or interpreting such law and any reference to any law or regulation shall, unless otherwise specified, refer to such law or regulation as amended, modified or supplemented from time

to time, and (vi) the words "<u>asset</u>" and "<u>property</u>" shall be construed to have the same meaning and effect and to refer to any and all tangible and intangible assets and properties, including cash, securities, accounts and contract rights.

(i) In the computation of periods of time from a specified date to a later specified date, the word "<u>from</u>" means "<u>from and including</u>;" the words "to" and "until" each mean "<u>to but excluding</u>;" and the word "<u>through</u>" means "<u>to and including</u>."

(ii) Section headings herein and in the other Loan Documents are included for convenience of reference only and shall not affect the interpretation of this Agreement or any other Loan Document.

1.0.3. <u>Classification of Loans and Borrowings</u>. For purposes of this Agreement, Loans may be classified and referred to by Class (e.g., a "Revolving Loan") or by Type (e.g., a "Eurodollar Loan") or by Class and Type (e.g., a "Eurodollar Revolving Loan"). Borrowings also may be classified and referred to by Class (e.g., a "Revolving Borrowing") or by Type (e.g., a "Eurodollar Borrowing") or by Class and Type (e.g., a "Eurodollar Revolving Borrowing").

1.0.4. <u>Accounting Terms</u>. (a) <u>Generally</u>. All accounting terms not specifically or completely defined herein shall be construed in conformity with, and all financial data (including financial ratios and other financial calculations) required to be submitted pursuant to this Agreement shall be prepared in conformity with, GAAP applied on a consistent basis, as in effect from time to time, applied in a manner consistent with that used in preparing the Unaudited Financial Statements, except as otherwise specifically prescribed herein.

(b) <u>Changes in GAAP</u>. If at any time any change in GAAP occurring after the Closing Date would affect Borrowers in their computation of any financial ratio or requirement set forth in any Loan Document, and any Borrower or the Required Lenders shall so request, Administrative Agent, the Lenders and Borrowers shall negotiate in good faith to amend such ratio or requirement to preserve the original intent thereof in light of such change in GAAP (subject to the approval of the Required Lenders); provided that, until so amended, (i) such ratio or requirement shall continue to be computed in accordance with GAAP prior to such change therein and (ii) Borrowers shall provide to Administrative Agent and the Lenders financial statements and other documents required under this Agreement or as reasonably requested hereunder setting forth a reconciliation between calculations of such ratio or requirement made before and after giving effect to such change in GAAP.

1.05. <u>Rounding</u>. Any financial ratios required to be maintained by Borrowers pursuant to this Agreement shall be calculated by dividing the appropriate component by the other component, carrying the result to one place more than the number of places by which such ratio is expressed herein and rounding the result up or down to the nearest number (with a rounding-up if there is no nearest number).

1.06. <u>Times of Day</u>. Unless otherwise specified, all references herein to times of day shall be references to Eastern time (daylight or standard, as applicable).

1.07. Joint and Several Obligations.

(a) All representations contained herein shall be deemed individually made by each Borrower, and each of the covenants, agreements, and obligations set forth herein shall be deemed to be the joint and separate covenants, agreements, and obligations of each Borrower. Any notice, request, consent, report, or other information or agreement delivered by any Borrower shall be deemed to be ratified by, consented to, and also delivered by the other Borrowers. Each Borrower recognizes and agrees that each covenant and agreement of a "Borrower" and "Borrowers" in this Agreement and in any other Loan Document shall create a joint and several obligation of such entities, which may be enforced against such entities jointly, or against each entity separately.

(b) Each Borrower hereby irrevocably and unconditionally agrees: (i) that it is jointly and severally liable to Administrative Agent and Lenders for the full and prompt payment of the Obligations and the performance by each Borrower of its obligations hereunder in accordance with the terms hereof; (ii) to fully and promptly perform all of its obligations hereunder with respect to the Obligations; and (iii) as a primary obligation to indemnify Administrative Agent and Lenders on demand for and against any loss (including losses due to Administrative Agent's or any Lender's negligence but excluding losses determined by a court of competent jurisdiction by a final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of Administrative Agent or any Lender) actually incurred by such Administrative Agent or any Lender as a result of any of the obligations of any one or more of Borrowers being or becoming void, voidable, unenforceable, or ineffective for any reason whatsoever, whether or not known to Administrative Agent, any Lender, or any other Person.

(c) Each Borrower waives as to each of the other Borrowers: (i) the amendment, extension, renewal, compromise, discharge, acceleration or otherwise changing the time for payment of, or any other terms with respect to, the Obligations or any part thereof or any substitution of Collateral (to the extent the foregoing has occurred in accordance with any applicable requirements of Section 12.01); (ii) any defense arising by reason of any disability or other defense of any of the other Borrowers or the cessation from any cause whatsoever (including any act or omission of the other Borrowers or Principal of any Borrower) of the liability of the other Borrowers; (iii) any right to require Administrative Agent or any Lender to proceed against or exhaust any security for the Obligations, or pursue any other remedy in Administrative Agent or any Lender's power whatsoever; and (iv) to the fullest extent permitted by law, any and all other defenses or benefits that may be derived from or afforded by applicable law limiting the liability of or exonerating guarantors or sureties.

ARTICLE II THE COMMITMENTS AND LOANS

2.01. <u>Commitments</u>. Subject to the terms and conditions set forth herein, each Lender agrees to make Revolving Loans to Borrowers from time to time during the Availability Period in an aggregate principal amount that will not result in (a) such Lender's Revolving Credit Exposure exceeding such Lender's Commitment or (b) the sum of the total Revolving Credit Exposures exceeding the Aggregate Commitments. Within the foregoing limits and subject to the terms and conditions set forth herein, Borrowers may borrow, prepay and reborrow Revolving Loans.

2.02. Loans and Borrowings. i) Each Revolving Loan shall be made as part of a Borrowing consisting of Revolving Loans made by the Lenders ratably in accordance with their respective Commitments. The failure of any Lender to make any Loan required to be made by it shall not relieve any other Lender of its obligations hereunder; provided that the Commitments of the Lenders are several and no Lender shall be responsible for any other Lender's failure to make Loans as required.

(b) Subject to Section 2.12, each Revolving Borrowing shall be comprised entirely of ABR Loans or Eurodollar Loans as Borrowers may request in accordance herewith. Each Swingline Loan shall be an ABR Loan. Each Lender at its option may make any Eurodollar Loan by causing any domestic or foreign branch or Affiliate of such Lender to make such Loan; provided that any exercise of such option shall not affect the obligation of Borrowers to repay such Loan in accord-ance with the terms of this Agreement.

(c) At the commencement of each Interest Period for any Eurodollar Revolving Borrowing, such Borrowing shall be in an aggregate amount that is an integral multiple of \$1,000,000 and not less than \$5,000,000. At the time that each ABR Revolving Borrowing is made, such Borrowing shall be in an aggregate amount that is an integral multiple of \$100,000 and not less than \$500,000. Borrowings of more than one Type and Class may be outstanding at the same time; provided that there shall not at any time be more than a total of five (5) Eurodollar Revolving Borrowings outstanding.

(d) Notwithstanding any other provision of this Agreement, Borrowers shall not be entitled to request, or to elect to convert or continue, any Borrowing if the Interest Period requested with respect thereto would end after the Maturity Date.

2.03. <u>Requests for Revolving Borrowings</u>. To request a Revolving Borrowing, Borrowers shall notify Administrative Agent of such request by telephone (a) in the case of a Eurodollar Borrowing, not later than 11:00 a.m., New York City time, three Business Days before the date of the proposed Borrowing or (b) in the case of an ABR Borrowing, not later than 11:00 a.m., New York City time, one Business Day before the date of the proposed Borrowing. Each such telephonic Borrowing Request shall be irrevocable and shall be confirmed promptly by hand delivery, electronic delivery or telecopy to Administrative Agent of a written Borrowing Request in a form approved by Administrative Agent and signed by Borrowers. Each such telephonic and written Borrowing Request shall specify the following information in compliance with Section 2.02:

(i) the aggregate amount of the requested Borrowing;

(ii) the date of such Borrowing, which shall be a Business Day;

(iii) whether such Borrowing is to be an ABR Borrowing or a Eurodollar Borrowing;

(iv) in the case of a Eurodollar Borrowing, the initial Interest Period to be applicable thereto, which shall be a period contemplated by the definition of the term "Interest Period"; and

(v) the location and number of Borrowers' account to which funds are to be disbursed, which shall comply with the requirements of Section 2.05.

If no election as to the Type of Revolving Borrowing is specified, then the requested Revolving Borrowing shall be an ABR Borrowing. If no Interest Period is specified with respect to any requested Eurodollar Revolving Borrowing, then Borrowers shall be deemed to have selected an Interest Period of one month's duration. Promptly following receipt of a Borrowing Request in accordance with this Section, Administrative Agent shall advise each Lender of the details thereof and of the amount of such Lender's Loan to be made as part of the requested Borrowing.

2.04. <u>Swingline Loans</u>. (a) Subject to the terms and conditions set forth herein, Swingline Lender agrees to make Swingline Loans to Borrowers from time to time during the Availability Period, in an aggregate principal amount at any time outstanding that will not result in (i) the aggregate principal amount of outstanding Swingline Loans exceeding \$25,000,000 or (ii) the sum of the total Revolving Credit Exposures exceeding the Aggregate Commitments; provided that Swingline Lender shall not be required to make a Swingline Loan to refinance an outstanding Swingline Loan. Within the foregoing limits and subject to the terms and conditions set forth herein, Borrowers may borrow, prepay and reborrow Swingline Loans.

(b) To request a Swingline Loan, Borrowers shall notify Administrative Agent of such request by telephone (confirmed by electronic delivery or telecopy), not later than 12:00 noon, New York City time, on the day of a proposed Swingline Loan. Each such notice shall be irrevocable and shall specify the requested date (which shall be a Business Day) and amount of the requested Swingline Loan. Administrative Agent will promptly advise Swingline Lender of any such notice received from Borrowers. Swingline Lender shall make each Swingline Loan available to Borrowers by means of a credit to the general deposit account of Borrowers or their designee with Swingline Lender by 3:00 p.m., New York City time, on the requested date of such Swingline Loan.

(c) Swingline Lender may by written notice given to Administrative Agent not later than 10:00 a.m., New York City time, on any Business Day require the Lenders to acquire participations on such Business Day in all or a portion of the Swingline Loans outstanding. Such notice shall specify the aggregate amount of Swingline Loans in which Lenders will participate. Promptly upon receipt of such notice, Administrative Agent will give notice thereof to each Lender, specifying in such notice such Lender's Applicable Percentage of such Swingline Loan or Loans. Each Lender hereby absolutely and unconditionally agrees, upon receipt of notice as provided above, to pay to Administrative

Agent, for the account of Swingline Lender, such Lender's Applicable Percentage of such Swingline Loan or Loans. Each Lender acknowledges and agrees that its obligation to acquire participations in Swingline Loans pursuant to this paragraph is absolute and unconditional and shall not be affected by any circumstance whatsoever, including the occurrence and continuance of a Default or reduction or termination of the Commitments, and that each such payment shall be made without any offset, abatement, withholding or reduction whatsoever. Each Lender shall comply with its obligation under this paragraph by wire transfer of immediately available funds, in the same manner as provided in Section 2.05 with respect to Loans made by such Lender (and Section 2.05 shall apply, mutatis mutandis, to the payment obligations of the Lenders), and Administrative Agent shall promptly pay to Swingline Lender the amounts so received by it from the Lenders. Administrative Agent shall notify Borrowers of any participations in any Swingline Loan acquired pursuant to this paragraph, and thereafter payments in respect of such Swingline Loan shall be made to Administrative Agent and not to Swingline Lender. Any amounts received by Swingline Lender from Borrowers (or other party on behalf of Borrowers) in respect of a Swingline Loan after receipt by Swingline Lender of the proceeds of a sale of participations therein shall be promptly remitted to Administrative Agent; any such amounts received by Administrative Agent shall be promptly remitted by Administrative Agent to the Lenders that shall have made their payments pursuant to this paragraph and to Swingline Lender, as their interests may appear; provided that any such payment so remitted shall be repaid to Swingline Lender or to Administrative Agent, as applicable, if and to the extent such payment is required to be refunded to Borrowers for any reason. The purchase of participations in a Swingline Loan pursuant to this paragraph shall not relieve Borrowers of any default in the payment thereof.

2.05. <u>Funding of Borrowings</u>. iii) Each Lender shall make each Loan to be made by it hereunder on the proposed date thereof by wire transfer of immediately available funds by 12:00 noon, New York City time, to the account of Administrative Agent most recently designated by it for such purpose by notice to the Lenders; provided that Swingline Loans shall be made as provided in Section 2.04. Administrative Agent will make such Loans available to Borrowers by promptly crediting the amounts so received, in like funds, to an account maintained with Administrative Agent in New York City and designated by Borrowers in the applicable Borrowing Request.

(b) Unless Administrative Agent shall have received notice from a Lender prior to the proposed date of any Borrowing that such Lender will not make available to Administrative Agent such Lender's share of such Borrowing, Administrative Agent may assume that such Lender has made such share available on such date in accordance with paragraph (a) of this Section and may, in reliance upon such assumption, make available to Borrowers a corresponding amount. In such event, if a Lender has not in fact made its share of the applicable Borrowing available to Administrative Agent, then the applicable Lender and Borrowers severally agree to pay to Administrative Agent forthwith on demand such corresponding amount with interest thereon, for each day from and including the date such amount is made available to Borrowers to but excluding the date of payment to Administrative Agent, at (i) in the case of such Lender, the greater of the Federal Funds Effective Rate and a rate determined by Administrative Agent in accordance with banking industry rules on interbank compensation or (ii) in the case of Borrowers, the interest rate applicable to ABR Loans. If such Lender pays such amount to Administrative Agent, then such amount shall constitute such Lender's Loan included in such Borrowing.

2.06. <u>Interest Elections</u>. iv) Each Revolving Borrowing initially shall be of the Type specified in the applicable Borrowing Request and, in the case of a Eurodollar Revolving Borrowing, shall have an initial Interest Period as specified in such Borrowing Request. Thereafter, Borrowers may elect to convert such Borrowing to a different Type or to continue such Borrowing and, in the case of a Eurodollar Revolving Borrowing, may elect Interest Periods therefor, all as provided in this Section. Borrowers may elect different options with respect to different portions of the affected Borrowing, in which case each such portion shall be allocated ratably among the Lenders holding the Loans comprising such Borrowing. This Section shall not apply to Swingline Borrowings, which may not be converted or continued.

(b) To make an election pursuant to this Section, Borrowers shall notify Administrative Agent of such election by telephone by the time that a Borrowing Request would be required under Section 2.03 if Borrowers were requesting a Revolving Borrowing of the Type resulting from such election to be made on the effective date of such election. Each such telephonic Interest Election Request shall be irrevocable and shall be confirmed promptly by hand delivery, electronic delivery or telecopy to Administrative Agent of a written Interest Election Request in a form approved by Administrative Agent and signed by Borrowers.

(c) Each telephonic and written Interest Election Request shall specify the following information in compliance with Section 2.02:

(i) the Borrowing to which such Interest Election Request applies and, if different options are being elected with respect to different portions thereof, the portions thereof to be allocated to each resulting Borrowing (in which case the information to be specified pursuant to clauses (iii) and (iv) below shall be specified for each resulting Borrowing);

(ii) the effective date of the election made pursuant to such Interest Election Request, which shall be a Business Day;

(iii) whether the resulting Borrowing is to be an ABR Borrowing or a Eurodollar Borrowing; and

(iv) if the resulting Borrowing is a Eurodollar Borrowing, the Interest Period to be applicable thereto after giving effect to such election, which shall be a period contemplated by the definition of the term "Interest Period". If any such Interest Election Request requests a Eurodollar Borrowing but does not specify an Interest Period, then Borrowers shall be deemed to have selected an Interest Period of one month's duration.

(d) Promptly following receipt of an Interest Election Request, Administrative Agent shall advise each Lender of the details thereof and of such Lender's portion of each resulting Borrowing.

(e) If Borrowers fail to deliver a timely Interest Election Request with respect to a Eurodollar Revolving Borrowing prior to the end of the Interest Period applicable thereto, then, unless such Borrowing is repaid as provided herein, at the end of such Interest Period such Borrowing shall be converted to an ABR Borrowing. Notwithstanding any contrary provision hereof, if an Event of Default has occurred and is continuing and Administrative Agent, at the request of the Required Lenders, so notifies Borrowers, then, so long as an Event of Default is continuing (i) no outstanding Revolving Borrowing may be converted to or continued as a Eurodollar Borrowing and (ii) unless repaid, each Eurodollar Revolving Borrowing shall be converted to an ABR Borrowing at the end of the Interest Period applicable thereto.

2.07. <u>Termination and Reduction of Commitments</u>. v) Unless previously terminated, the Commitments shall terminate on the Maturity Date.

(b) Borrowers may at any time voluntarily terminate, or from time to time reduce, the Commitments; provided that (i) each reduction of the Aggregate Commitments shall be in an amount that is an integral multiple of \$1,000,000 and not less than \$5,000,000, except for a one time reduction in the Aggregate Commitments in connection with Borrower's compliance with the covenant set forth in Section 8.07(b), which one time reduction may be any lesser amount as necessary to comply with such covenant, and (ii) Borrowers shall not voluntarily terminate or reduce the Aggregate Commitments if, after giving effect to any concurrent prepayment of the Loans in accordance with Section 2.08, Revolving Credit Exposures would exceed the Aggregate Commitments.

(c) Aggregate Commitments shall be permanently reduced from time to time, as follows:

(i) <u>Net Proceeds</u>. Upon the occurrence of a Casualty or Condemnation that results in (or could reasonably be expected to result in) a Material Property Event with respect to the applicable Collateral Property, by an amount equal to the Net Proceeds retained by the Administrative Agent and applied to prepay the Loans in accordance with Sections 2.09(b)(ii) and 9.04(j); and

(ii) <u>Release of Collateral</u>. (A) Simultaneously with any release of Collateral in accordance with Section 2.17(a)(i), by an amount equal to the Release Price of such Collateral Property, or (B) simultaneously with any release of Collateral in accordance with Section 2.17(k), by an amount required to cause compliance, all as further set forth in such Section 2.17;

If, after giving effect to any permanent reductions in Aggregate Commitments under this Section 2.07(c), Revolving Credit Exposures would exceed the Aggregate Commitments, then Borrowers shall repay Loans in accordance with Section 2.09(b)(i).

(d) Borrowers shall notify Administrative Agent of any election or requirement to terminate or reduce the Commitments under paragraph (b) or (c) of this Section at least three Business Days prior to the effective date of such termination or reduction, specifying such election and the effective date thereof. Promptly following receipt of any notice, Administrative Agent shall advise the Lenders of the contents thereof. Each notice delivered by Borrowers pursuant to this Section shall be irrevocable; provided that a notice of termination of the Commitments delivered by Borrowers may state that such notice is conditioned upon the effectiveness of other credit facilities, in which case such notice may be revoked by Borrowers (by notice to Administrative Agent on or prior to the specified effective date) if such condition is not satisfied. Any termination or reduction of the Commitments shall be permanent. Each reduction of the Commitments shall be made ratably among the Lenders in accordance with their respective Commitments.

2.08. <u>Repayment of Loans</u>; <u>Evidence of Debt.</u> vi) Borrowers hereby unconditionally promise to pay (i) to Administrative Agent for the account of each Lender the then unpaid principal amount of each Revolving Loan on the Maturity Date and (ii) to Swingline Lender the then unpaid principal amount of each Swingline Loan on the earlier of the Maturity Date and the first date after such Swingline Loan is made that is the 15th or last day of a calendar month and is at least three Business Days after such Swingline Loan is made; provided that on each date that a Revolving Borrowing is made, Borrowers shall repay all Swingline Loans then outstanding.

(b) Each Lender shall maintain in accordance with its usual practice an account or accounts evidencing the indebtedness of Borrowers to such Lender resulting from each Loan made by such Lender, including the amounts of principal and interest payable and paid to such Lender from time to time hereunder.

(c) Administrative Agent shall maintain accounts in which it shall record (i) the amount of each Loan made hereunder, the Class and Type thereof and the Interest Period applicable thereto, (ii) the amount of any principal or interest due and payable or to become due and payable from Borrowers to each Lender hereunder and (iii) the amount of any sum received by Administrative Agent hereunder for the account of the Lenders and each Lender's share thereof.

(d) The entries made in the accounts maintained pursuant to paragraph (b) or (c) of this Section shall be <u>prima facie</u> evidence of the existence and amounts of the obligations recorded therein; provided that the failure of any Lender or Administrative Agent to maintain such accounts or any error therein shall not in any manner affect the obligation of Borrowers to repay the Loans in accordance with the terms of this Agreement.

2.09. Prepayment of Loans.

(a) Optional Prepayment: Borrowers shall have the right at any time and from time to prepay any Borrowing in whole or in part, subject to prior notice in accordance with paragraph (c) of this Section.

(b) Mandatory Prepayment:

(i) If for any reason the Revolving Credit Exposures exceed the Aggregate Commitments then in effect, Borrowers shall promptly (and in any event, within one Business Day) repay Loans in an aggregate amount equal to such excess and shall give prior notice of such prepayment in accordance with paragraph (c) of this Section.

(ii) If the Administrative Agent shall determine under Section 9.04(j) that excess Net Proceeds are to be applied to repay Loans, the Administrative Agent shall notify Borrowers of such amount, and Borrowers shall promptly (and in any event, within one Business Day) inform the Administrative Agent of the Borrowings to which such prepayment shall be applied, in accordance with paragraph (c) of this Section.

(c) Prior Notice: Borrowers shall notify the Administrative Agent (and, in the case of prepayment of a Swingline Loan, Swingline Lender) by telephone (confirmed by telecopy) of any prepayment hereunder (i) in the case of prepayment of a Eurodollar Revolving Borrowing, not later than 11:00 a.m., New York City time, three Business Days before the date of prepayment, (ii) in the case of prepayment of an ABR Revolving Borrowing, not later than 11:00 a.m., New York City time, one Business Day before the date of prepayment or (iii) in the case of prepayment of a Swingline Loan, not later than 12:00 noon, New York City time, on the date of prepayment. Each such notice shall be irrevocable and shall specify the prepayment date and the principal amount of each Borrowing or portion thereof to be prepaid; provided that, if a notice of prepayment is given in connection with a conditional notice of termination of the Commitments as contemplated by Section 2.07, then such notice of prepayment may be revoked if such notice of termination is revoked in accordance with Section 2.07. Promptly following receipt of any such notice relating to a Revolving Borrowing, Administrative Agent shall advise the Lenders of the contents thereof. Each partial prepayment of any Revolving Borrowing shall be in an amount that would be permitted in the case of an advance of a Revolving Borrowing of the same Type as provided in Section 2.02, except to the extent required by Section 2.07(c) or in connection with a partial reduction of Aggregate Commitment permitted under Section 2.17. Each prepayment of a Revolving Borrowing shall be applied ratably to the Loans included in the prepaid Borrowing. Prepayments shall be accompanied by accrued interest to the extent required by Section 2.11.

2.10. <u>Interest</u>. vii) The Loans comprising each Eurodollar Borrowing shall bear interest at the Adjusted LIBO Rate for the Interest Period in effect for such Borrowing, plus the Applicable Margin. The Loans comprising each ABR Borrowing (including each Swingline Loan) shall bear interest at the Alternate Base Rate plus the Applicable Margin.

(b) Default Rate Interest.

(i) If any amount of principal of any Loan is not paid when due (without regard to any applicable grace periods), whether at stated maturity, by acceleration or otherwise, such amount shall thereafter bear interest at a fluctuating interest rate per annum at all times equal to the Default Rate to the fullest extent permitted by applicable Laws.

(ii) If any amount (other than principal of any Loan) payable by Borrowers under any Loan Document is not paid when due (without regard to any applicable grace periods), whether at stated maturity, by acceleration or otherwise, then upon the request of Required Lenders (or automatically upon the entry or deemed entry of an order for relief with respect to any Borrower under any Debtor Relief Laws) such amount shall thereafter bear interest at a fluctuating interest rate per annum at all times equal to the applicable Default Rate to the fullest extent permitted by applicable Laws.

(iii) Upon the request of Required Lenders, while any Event of Default exists, Borrowers shall pay interest on the principal amount of all outstanding Obligations hereunder at a fluctuating interest rate per annum at all times equal to the Default Rate to the fullest extent permitted by applicable Laws.

(iv) Accrued and unpaid interest on past due amounts (including interest on past due interest) shall be due and payable upon demand.

(c) Interest on each Loan shall be due and payable in arrears on each Payment Date applicable thereto and at such other times as may be specified herein. Interest hereunder shall be due and payable in accordance with the terms hereof before and after judgment, and before and after the commencement of any proceeding under any Debtor Relief Law.

2.11. <u>Fees</u>. (a) Borrowers jointly and severally agree to pay to Administrative Agent for the account of each Lender an unused fee, which shall accrue at the rate of 0.50% per annum on the average daily unused amount of the Commitment of such Lender during the period from and including Closing Date to the last day of the Availability Period. Swingline Loans shall not be counted as usage of the Commitment for the purposes of calculating the unused fee. Accrued unused fees shall be payable in arrears on the last day of March, June, September and December of each year and on the date on which the Commitments terminate, commencing on the first such date to occur after the date hereof. All unused fees shall be computed on the basis of a year of 360 days and shall be payable for the actual number of days elapsed (including the first day but excluding the last day).

(b) Borrowers jointly and severally agree to pay to Administrative Agent, for its own account, fees payable in the amounts and at the times separately agreed upon between Borrowers and Administrative Agent.

(c) All fees payable hereunder shall be paid on the dates due in immediately available funds to Administrative Agent for its own account or for distribution to the Lenders, as applicable. Fees paid shall not be refundable under any circumstances.

2.12. <u>Computation of Interest and Fees</u>. All computations of interest for ABR Loans when the Alternate Base Rate is determined by the Prime Rate shall be made on the basis of a year of 365 or 366 days, as the case may be, and actual days elapsed. All other computations of fees and interest shall be made on the basis of a 360-day year and actual days elapsed (which results in more fees or interest, as applicable, being paid than if computed on the basis of a 365-day year). Interest shall accrue on each Loan for the day on which the Loan is made, and shall not accrue on a Loan, or any portion thereof, for the day on which the Loan or such portion is paid, <u>provided</u> that any Loan that is repaid on the same day on which it is made shall, subject to <u>Section 2.09</u> bear interest for one day. Each determination by Administrative Agent of an interest rate or fee hereunder shall be conclusive and binding for all purposes, absent manifest error.

2.13. Evidence of Debt. The Loans made by each Lender shall be evidenced by one or more accounts or records maintained by such Lender and by Administrative Agent in the ordinary course of business. The accounts or records maintained by Administrative Agent and each Lender shall be conclusive absent manifest error of the amount of the Loans made by Lenders to Borrowers and the interest and payments thereon. Any failure to so record or any error in doing so shall not, however, limit or otherwise affect the obligation of Borrowers hereunder to pay any amount owing with respect to the Obligations. In the event of any conflict between the accounts and records maintained by any Lender and the accounts and records of Administrative Agent in respect of such matters, the accounts and records of Administrative Agent shall control in the absence of manifest error. Upon the request of any Lender made through Administrative Agent, Borrowers shall execute and deliver to such Lender (through Administrative Agent) a Note, which shall evidence such Lender's Loans in addition to such accounts or records. Each Lender may attach schedules to its Note and endorse thereon the date, Type (if applicable), amount and maturity of its Loans and payments with respect thereto.

2.14. <u>Payments Generally; Administrative Agent's Clawback</u>. ix) <u>General</u>. All payments to be made by Borrowers shall be made without condition or deduction for any counterclaim, defense, recoupment or setoff. Except as otherwise expressly provided herein, all payments by Borrowers hereunder shall be made to Administrative Agent, for the account of the respective Lenders to which such payment is owed, at Administrative Agent's Office in Dollars and in immediately available funds not later than 3:00 p.m. on the date specified herein. Administrative Agent will promptly distribute to each Lender its Applicable Percentage in respect of the Facility (or other applicable share as provided herein) of such payment in like funds as received by wire transfer to such Lender's Lending Office. All payments received by Administrative Agent after 3:00 p.m. shall be deemed received on the next succeeding Business Day and any applicable interest or fee shall continue to accrue. If any payment to be made by Borrowers shall come due on a day other than a Business Day, payment shall be made on the next following Business Day, and such extension of time shall be reflected on computing interest or fees, as the case may be.

(b) <u>Presumption by Administrative Agent</u>. Unless Administrative Agent shall have received notice from Borrowers prior to the time at which any payment is due to Administrative Agent for the account of the Lenders hereunder that Borrowers will not make such payment, Administrative Agent may assume that Borrowers have made such payment on such date in accordance herewith and may, in reliance upon such assumption, distribute to the Lenders the amount due. In such event, if Borrowers have not in fact made such payment, then each of the Lenders severally agree to repay to Administrative Agent forthwith on demand the amount so distributed to such Lender, in immediately available funds with interest thereon, for each day from and including the date such amount is distributed to it to but excluding the date of payment to Administrative Agent, at the greater of the Federal Funds Rate and a rate reasonably determined by Administrative Agent in accordance with banking industry rules on interbank compensation.

(c) <u>Obligations of Lenders Several</u>. The obligations of the Lenders hereunder to make Loans and to make payments pursuant to <u>Section 12.04(c)</u> are several and not joint. The failure of any Lender to make any Loan, to make any payment under <u>Section 12.04(c)</u> on any date required hereunder shall not relieve any other Lender of its corresponding obligation to do so on such date, and no Lender shall be responsible for the failure of any other Lender to so make its Loan or to make its payment under <u>Section 12.04(c)</u>.

(d) <u>Funding Source</u>. Nothing herein shall be deemed to obligate any Lender to obtain the funds for any Loan in any particular place or manner or to constitute a representation by any Lender that it has obtained or will obtain the funds for any Loan in any particular place or manner.

(e) <u>Insufficient Funds</u>. If at any time insufficient funds are received by and available to Administrative Agent to pay fully all amounts of principal, interest and fees then due hereunder, such funds shall be applied (i) first, toward payment of interest and fees then due hereunder, ratably among the parties entitled thereto in accordance with the amounts of interest and fees then due to such parties, and (ii) <u>second</u>, toward payment of principal then due hereunder, ratably among the parties entitled thereto in accordance with the amounts of principal then due to such parties.

2.15. <u>Sharing of Payments by Lenders</u>. If any Lender shall, by exercising any right of setoff or counterclaim or otherwise, obtain payment in respect of (a) Obligations in respect of the Facility due and payable to such Lender hereunder and under the other Loan Documents at such time in excess of its ratable share (according to the proportion of (i) the amount of such Obligations in respect of the Facility due and payable to all Lenders hereunder and under the other Loan Documents at such time) of payments on account of the Obligations in respect of the Facility due and payable to all Lenders hereunder at such time other Loan Documents at such time) of payments on account of the Obligations in respect of the Facility due and payable to all Lenders hereunder at such time other Loan Documents at such time or (b) Obligations in respect of the Facility owing (but not due and payable) to such Lender and under the other Loan Documents at such time in excess of its ratable share (according to the proportion of (i) the amount of such Obligations owing (but not due and payable) to such Lender at such time to (ii) the aggregate amount of the Obligations in respect of the Facility owing (but not due and payable) to such Lender at such time to (ii) the aggregate amount of the Obligations in respect of

the Facility owing (but not due and payable) to all Lenders hereunder and under the other Loan Parties at such time) of payment on account of the Obligations in respect of the Facility owing (but not due and payable) to all Lenders hereunder and under the other Loan Documents at such time obtained by all of the Lenders at such time then the Lender receiving such greater proportion shall (a) notify Administrative Agent of such fact, and (b) purchase (for cash at face value) participations in the Loans of the other Lenders, or make such other adjustments as shall be equitable, so that the benefit of all such payments shall be shared by Lenders ratably in accordance with the aggregate amount of Obligations then due and payable to the Lenders or owing (but not due and payable) to the Lenders, as the case may be, provided that:

(i) if any such participations or subparticipations are purchased and all or any portion of the payment giving rise thereto is recovered, such participations or subparticipations shall be rescinded and the purchase price restored to the extent of such recovery, without interest; and

(ii) the provisions of this Section shall not be construed to apply to (A) any payment made by Borrowers pursuant to and in accordance with the express terms of this Agreement or (B) any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its Loans to any assignee or participant, other than to Borrowers or any Borrowers' Affiliates or Subsidiaries (as to which the provisions of this Section shall apply).

Each Borrower consents to the foregoing and agrees, to the extent it may effectively do so under applicable law, that any Lender acquiring a participation pursuant to the foregoing arrangements may exercise against any such Borrower's rights of setoff and counterclaim with respect to such participation as fully as if such Lender were a direct creditor of such Borrower in the amount of such participation.

If any Lender shall fail to make any payment required to be made by it pursuant to Section 2.04(c) [funding swingline participations], 2.05(b) [agent loan pre-funding], 2.14(d) [agent pre-funding of borrower repayments] or 12.04(c) [lender indemnity], then Administrative Agent may, in its discretion and notwithstanding any contrary provision hereof, (i) apply any amounts thereafter received by Administrative Agent for the account of such Lender for the benefit of Administrative Agent or Swingline Lender to satisfy such Lender's obligations to it under such Section until all such unsatisfied obligations are fully paid, and/or (ii) hold any such amounts in a segregated account as cash collateral for, and application to, any future funding obligations of such Lender under any such Section, in the case of each of clauses (i) and (ii) above, in any order as determined by Administrative Agent in its discretion.

2.16. <u>Extension of Maturity Date</u>. So long as no Default or Event of Default has occurred and is continuing on the Stated Maturity Date that would otherwise have occurred without giving effect to such request (such date, the "<u>Extension Effective Date</u>"), Borrowers may on one occasion extend the Stated Maturity Date to a date that is not later than one (1) year after the applicable original Stated Maturity Date, upon:

(a) delivery of a Facility Extension Request to Administrative Agent and the Lenders at least thirty (30) days but no more than ninety (90) days, prior to the Stated Maturity Date;

(b) delivery to Administrative Agent and the Lenders of an Officer's Certificate, dated as of the Extension Effective Date, certifying that (i) all representations and warranties of Borrowers set forth in this Agreement shall be true and correct in all material respects as of the date on which Borrowers delivered the Facility Extension Request to Administrative Agent and on the Extension Effective Date, except to the extent such representations and warranties were made as of a specified date, in which case such representation and warranty shall have been true and correct as of such specified date; (ii) no Default or Event of Default has occurred and is continuing on the Extension Effective Date or after giving effect to any extension of the Stated Maturity Date on such date; and (iii) the calculations attached, and made in reasonable detail, support the determinations made in clauses (c) through (f), below;

(c) the Debt Service Coverage Ratio, calculated on a pro-forma basis as of the Extension Effective Date, shall equal or exceed 1.5 to 1.0, as evidenced by an Officer's Certificate delivered by Borrowers to Administrative Agent and the Lenders with supporting calculations, dated as of the Extension Effective Date;

(d) the Loan to Value Ratio for the Collateral Properties shall be no more than 55% based on updated Acceptable Appraisals obtained no more than ninety (90) days prior to the proposed Extension Effective Date (and after giving effect to any permanent reductions of the Aggregate Commitment that may be necessary to cause the Loan to Value Ratio to be equal to or less than 55%);

(e) (x) unrestricted cash and Permitted Investments of FelCor Trust and its Subsidiaries, unused and available Commitments and amounts available to be drawn under other credit facilities of FelCor Trust and its Subsidiaries shall equal or exceed (y) the sum of (A) the aggregate principal amount of all Recourse Indebtedness of FelCor Trust and its Subsidiaries (including, to the extent applicable, the outstanding principal amount of the Senior Secured Notes) that matures within one year of the Extension Effective Date, and (B) the aggregate amount of preferred, convertible or other securities of FelCor Trust and its Subsidiaries that require mandatory cash purchases, cash redemption or other cash payments within one-year of the Extension Effective Date;

(f) the Loan Parties shall continue to be in pro forma compliance with the restrictions on indebtedness set forth in Section 4.03 of the Senior Secured Notes Indenture;

(g) payment to Administrative Agent for the benefit of Lenders of a facility extension fee equal to one half of one percent (0.50%) of the Aggregate Commitments as of the applicable Extension Effective Date (after giving effect to any permanent reduction of the Aggregate Commitments being made in connection with such Extension); (h) payment by Borrowers of all other reasonable fees and expenses to Administrative Agent and the Lenders to the extent then due as of the applicable Extension Effective Date; and

(i) execution (and if required by local law, recordation) of any extension agreement in form as required by Administrative Agent with respect to each Collateral Property and endorsements and other assurances as Administrative Agent may reasonably require to conform the Title Policies (including "datedown" endorsements) and amendment to any Mortgages of the remaining Collateral Properties to reflect any change in respective Appraised Values of the Collateral Properties (together with payment by Borrowers of any fees or expenses, including any additional mortgage tax, in connection therewith).

The Extension shall be evidenced by delivery of written confirmation of the same by Administrative Agent to Borrowers. In no event shall the Maturity Date occur later than August 1, 2015. During the extended term of the Loans, all terms and conditions of the Loan Documents (other than the original Stated Maturity Date) shall continue to apply, except that Borrowers shall have no further right to extend the term of the Loans after the Extension.

2.17. <u>Collateral Property Releases</u>. Borrowers may obtain the release of a Collateral Property from the Lien of the Mortgage thereon (and related Collateral Documents) and the release of Borrowers' obligations under the Loan Documents with respect to such Collateral Property (other than those expressly stated to survive), but only upon the satisfaction of each of the following conditions:

(a) Such release is in connection with (i) a sale or refinancing of such Collateral Property and Administrative Agent shall have received evidence that such Collateral Property shall be conveyed to a Person other than a Borrower (in the case of a sale) ("<u>Sale</u> <u>Triggered Release</u>"), or (ii) the receipt of new Acceptable Appraisals for the Collateral Properties, as further described in clause (d)(ii)(y) below ("<u>Appraisal Triggered Release</u>");

(b) (i) No Event of Default has occurred and is continuing, and no Default or Event of Default would exist after giving effect to the release of such Collateral Property, (ii) Borrowers, FelCor Trust and FelCor Op are in compliance, and after giving effect to the release of such Collateral Property, will be in compliance, with the covenants set forth in the Senior Secured Notes Indenture, and (iii) the release will not violate the terms of this Agreement or any other Loan Document or the Senior Secured Notes Indenture;

(c) Borrowers shall provide Administrative Agent and the Lenders with at least thirty (30) days but no more than forty-five (45) days prior written notice of its request to obtain a release of the Collateral Property, which notice shall be revocable by Borrowers at any time prior to the effectiveness of such release;

(d) After giving effect to such release, the Loan to Value Ratio for the remaining Collateral Properties shall be (i) no greater than the Loan to Value Ratio in effect immediately prior to such release, and (ii) in any event, equal to or less than sixty percent (60%); in each case as determined based on (A) the Aggregate Commitments after such release and prepayment of Loans and permanent reduction of Commitments, and (B) the Appraised Value of all remaining Collateral Properties, where:

(x) in connection with a Sale Triggered Release, such Appraised Value shall be based on (1) Acceptable Appraisals for each of the Collateral Properties delivered in connection with the Closing Date or the Extension Date, or (2) a new set of Acceptable Appraisals for each of the Collateral Properties otherwise obtained or delivered in accordance herewith; and

(y) in connection with an Appraisal Triggered Release, such Appraised Value (and the corresponding Allocated Loan Amount) shall be based on Acceptable Appraisals for each of the Collateral Properties prepared within six (6) months of the proposed release date of any such Collateral Property.

(e) After giving effect to such release,

(i) the remaining Collateral Properties shall be sufficient to result in a proforma Debt Service Coverage Ratio of not less than the Debt Service Coverage Ratio in effect immediately prior to such release (as determined based on (A) the Aggregate Commitments after such release and prepayment of Loans and permanent reduction of Commitments, and (B) the Net Operating Income of the remaining Collateral Properties); and

(ii) no Collateral Property shall comprise more than twenty percent (20%) of the aggregate Net Operating Income of all the Collateral Properties as of the release date, except for the Mandalay Beach Property, the Net Operating Income of which shall not comprise more than twenty five percent (25%) of the aggregate Net Operating Income of all the Collateral Properties as of such release date.

(f) Simultaneously with such release, Administrative Agent shall have received a wire transfer of immediately available federal funds in an amount equal to the Release Price for the applicable Collateral Property, to the extent required by Section 2.09(b)(i) (if Revolving Credit Exposure is then in excess of Aggregate Commitments, as a result of Aggregate Commitments being reduced in accordance with Section 2.07(c)(ii)), together with (i) all accrued and unpaid interest on the amount of principal being prepaid through and including the date of such prepayment, (ii) any amounts required to be paid pursuant to Section 4.05, if any, and (iii) all other sums due under this Agreement, the Notes or the other Loan Documents in connection with such release;

(g) Borrowers shall submit to Administrative Agent, not less than five (5) days prior to the date of such release, (i) a release of Lien (and related Loan Documents) for such Collateral Property for execution by Administrative Agent in a form appropriate for recordation in the state in which such Collateral Property is located and shall contain standard

provisions, if any, protecting the rights of Administrative Agent and Lenders, (ii) all other documentation Administrative Agent reasonably requires to be delivered by Borrowers in connection with such release, (iii) an Officer's Certificate certifying that (A) such documentation is in compliance with all applicable Legal Requirements, (B) the release will not impair or otherwise adversely affect the Liens, security interests and other rights of Administrative Agent and Lenders under the Loan Documents not being released (or as to the parties to the Loan Documents and Collateral Properties subject to the Loan Documents not being released), (C) the release is in compliance with clauses (d) and (e) above, together with computations in reasonable detail evidencing such compliance, and (D) (i) no Event of Default has occurred and is continuing, and no Default or Event of Default would exist after giving effect to the release of such Collateral Property; (ii) Borrowers, FelCor Trust and FelCor Op are in compliance, and after giving effect to the release of such Collateral Property, will be in compliance, with the covenants set forth in the Senior Secured Notes Indenture; and (iii) the release will not violate the terms of this Agreement or any other Loan Document or the Senior Secured Notes Indenture;

(h) Based on the Allocated Loan Amounts of the remaining Collateral Properties, Administrative Agent may obtain at Borrowers' expense such additional endorsements to its title policies to reflect such revised valuations, as Administrative Agent may reasonably require;

(i) Administrative Agent shall have received payment of all of its reasonable out of pocket costs and expenses, including due diligence review costs and reasonable counsel fees and disbursements incurred in connection with the release of such Collateral Property from the Lien of the applicable Mortgage and the review and approval of the documents and information required to be delivered in connection therewith;

(j) Administrative Agent shall have received a true, correct and complete copy of the purchase agreement or refinancing loan agreement, as applicable; and

(k) if Borrowers fail to meet the Loan to Value Ratio or the Debt Service Coverage Ratio tests set forth in clauses (d) and (e) above, Borrowers may partially reduce the Aggregate Commitments pursuant to Section 2.07(c) to comply with such tests.

2.18. <u>Release on Payment in Full</u>.

Administrative Agent shall, upon the written request and at the expense of Borrowers, upon payment in full of all Obligations, release the Lien of the Mortgages and any other Loan Document on each Collateral Property not theretofore released.

2.19. Incremental Commitments.

(a) Borrowers may, by written notice to the Administrative Agent on up to two (2) occasions during the period from the Closing Date to the Extension Effective Date, request incremental Commitments in an amount not to exceed the aggregate amount of \$15,000,000 from one or more additional Lenders (which may include any existing Lender) willing to provide such incremental Commitments in their own discretion; provided, that each incremental Lender shall be subject to the approval of the Administrative Agent (which approval shall not be unreasonably withheld) unless such incremental Lender is a Lender, an Affiliate of a Lender or an Approved Fund. Such notice shall set forth (i) the amount of the incremental Commitments being requested, (ii) the aggregate amount of all incremental Commitments, which when taken together with all other incremental Commitments, shall not exceed \$15,000,000 in the aggregate (the "Incremental Limit"), and (iii) the date on which such incremental Commitments are requested to become effective (the "Increased Amount Date"). The Administrative Agent and/or its Affiliates shall use commercially reasonable efforts, with the assistance of Borrowers, to arrange a syndicate of Lenders willing to hold the requested incremental Commitments.

(b) Borrowers and each incremental Lender shall execute and deliver to the Administrative Agent such documentation as the Administrative Agent shall reasonably specify to evidence the incremental Commitment of such incremental Lender. Each such documentation shall specify the terms of the applicable incremental Commitments; provided, that from and after the effectiveness of each amendment or other documentation, the associated incremental Commitments shall thereafter be Commitments with the same terms as the Commitments (including as to pricing and maturity). Each of the parties hereto hereby agrees that, upon the effectiveness of any such documentation, this Agreement shall be amended to the extent (but only to the extent) necessary to reflect the existence and terms of the incremental Commitments evidenced thereby, and new Notes shall be issued and Borrowers shall make such borrowings and repayments as shall be necessary to effect the reallocation of the Commitments, in each case without the consent of the Lenders other than those Lenders with incremental Commitments. The fees payable by Borrowers upon any such incremental Commitments shall be agreed upon by the Administrative Agent, the Lenders with incremental Commitments and Borrowers at the time of such increase. Nothing in this Section 2.19 shall constitute or be deemed to constitute an agreement by any Lender to increase its Commitments hereunder.

Notwithstanding the foregoing, no incremental Commitment shall become (c) effective under this Section 2.19 unless (i) on the date of such effectiveness, the conditions set forth in Section 5.02 shall be satisfied and the Administrative Agent shall have received a certificate to that effect dated such date and executed by a Responsible Officer of Borrowers, (ii) the Administrative Agent shall have received customary legal opinions, board resolutions and other customary closing certificates and documentation as required by the relevant amendment or other documentation and, to the extent required by the Administrative Agent, consistent with those delivered on the Closing Date under Section 5.01 and such additional customary documents and filings as the Administrative Agent may reasonably require, including amendments to Mortgages and date downs to, and incremental increases in the amounts of coverage under, the various Title Policies, and (iii) Borrowers shall be in pro forma compliance with the covenants set forth in Section 8.07 and the Loan to Value Ratio for all of the Collateral Properties, shall be no more than sixty percent (60%), in each case after giving effect to such incremental Commitments, the Loans to be made thereunder and the application of the proceeds therefrom as if made and applied on such date, and the Administrative Agent shall have received a certificate to that effect dated such date and executed by a Responsible Officer of Borrowers, showing such calculations in reasonable detail.

(d) Each of the parties hereto hereby agrees that the Administrative Agent may take any and all action as may be reasonably necessary to ensure that all Loans in respect of incremental Commitments, when originally made, are included in each Borrowing of outstanding Loans on a pro rata basis. Borrowers agree that Section 4.05 shall apply to any conversion of Eurodollar Loans to ABR Loans reasonably required by the Lenders to effect the foregoing.

ARTICLE III CASH MANAGEMENT

3.01. Establishment of Accounts.

Each Owner or Operating Lessee, as applicable, shall cause the Manager to (a) establish one or more accounts (the "Property Accounts"), and to establish or identify one or more Manager Accounts, into which the applicable Borrower shall, and shall direct each Manager to deposit all Gross Income from Operations (together with any other revenue received by Borrowers). Each Property Account will be held in the name of a Borrower for the benefit of Administrative Agent and each Manager Account will be held in the name of Manager. In no case shall any Borrower originate any instructions with respect to any Property Account or Manager Account. Borrowers agree that prior to the Closing Date, Administrative Agent will enter into an agreement with each Manager, pursuant to which each Manager will agree to comply with instructions originated by Administrative Agent (upon notice by Administrative Agent of the occurrence of an Event of Default) directing disposition of the funds owed to Borrowers under the Management Agreements without further consent by Borrowers, such agreements to be in form and substance satisfactory to Administrative Agent. Schedule 3.01(a) lists each Property Account by account number and institution.

(b) Upon an Event of Default, Administrative Agent, Borrowers (or Administrative Agent on behalf of Borrowers, under its power of attorney granted pursuant to Section 3.10 below) and Lockbox Bank shall complete and execute all paperwork to establish an account with Lockbox Bank (the "Lockbox Account"), which agreement: (i) shall be in form and substance mutually (and reasonably, in the case of Borrowers) satisfactory to Administrative Agent and Borrowers, (ii) shall provide that Lockbox Bank will comply with instructions originated by Administrative Agent for the benefit of Secured Parties directing disposition of the funds in the Lockbox Account without further consent by Borrowers (the "Lockbox Account Agreement"), and (iii) shall be effective immediately. During any Lockbox Period, after receipt of notice from Administrative Agent, each Manager shall transfer all amounts that would have otherwise been transferred to Borrowers under the Management Agreements, on a weekly basis into the Lockbox Account in accordance with Section 3.05.

(c) In addition, Administrative Agent shall cause the Lockbox Bank to establish the following Accounts (which may be book entry sub-accounts, collectively, the "<u>Reserve</u> <u>Accounts</u>") into which during any Lockbox Period amounts in the Lockbox Account shall be deposited or allocated in accordance with <u>Section 3.06(a)</u>:

(i) An account with Lockbox Bank into which the Monthly Tax Deposit, if any, shall be deposited (the "<u>Tax Account</u>");

(ii) An account with Lockbox Bank into which the Monthly Insurance Premium Deposit, if any, shall be deposited (the "<u>Insurance Premium Account</u>"); and

(iii) An account with Lockbox Bank into which the Monthly Capital Expenditures Reserve Deposit, if any, shall be deposited (the "<u>Capital Expenditures</u> <u>Reserve Account</u>").

Each Borrower and Manager shall (and each Operating Lessee shall direct (d) the applicable Manager to) deposit all Accounts Receivable for each Collateral Property, any other Gross Income from Operations and any other revenue relating to the Collateral Properties received or collected by such Borrower or Manager (on behalf of a Borrower) into the applicable Property Account or a Manager Account. For so long as any Obligations are outstanding, no Borrower nor any other Person shall maintain any other accounts (other than property level petty cash or similar type accounts at each Collateral Property with a total aggregate balance of less than \$125,000 for all Collateral Properties other than the Property Accounts and Manager Accounts (each of which petty cash, Property Accounts and Manager Accounts shall be controlled by the respective Manager), into which revenues from the ownership and operation of the Collateral Properties are deposited. No Borrower nor any other Person shall (A) open any other such account with respect to the deposit of income in connection with the Collateral Properties or (B) direct any existing or future third party payors to make payments in any manner other than as set forth in this Section 3.01 (d), without the prior written consent of Administrative Agent.

(e) Any Gross Income from Operations from the Collateral Properties held by Borrowers or Managers shall be deemed to be Collateral and shall be held in trust by such Person for the benefit, and as the property, of Administrative Agent for the benefit of the Secured Parties and shall not be commingled with any other funds or property of Borrowers or Managers, except as contemplated by the Management Agreements.

3.02. Account Name.

(a) The Accounts (other than the Manager Account, the Lockbox Account and each Reserve Account) shall each be in the name of a Borrower for the benefit of Administrative Agent. Each Reserve Account and the Lockbox Account shall be in the name of Administrative Agent for the benefit of the Secured Parties.

(b) In the event Administrative Agent is replaced, each Borrower acknowledges that each Reserve Account and the Lockbox Account shall be changed to be in the name of such new Administrative Agent.

3.03. Eligible Accounts.

Each Borrower shall, and each Borrower shall cause the applicable Manager and Eligible Institution holding such Account to, maintain each Property Account and Manager Account as an Eligible Account. Administrative Agent shall cause the applicable Eligible Institution holding the Lockbox Account to maintain such Account as an Eligible Account. If any Bank holding a Property Account or a Manager Account shall cease to be an Eligible Institution, provided a Lockbox Period is not then in effect, Borrowers and Manager shall select an Eligible Institution as a replacement therefore, subject to Administrative Agent's approval. If a Lockbox Period is in effect, Administrative Agent (in consultation with Managers, as applicable) may select the Eligible Institution that shall act as the replacement.

3.04. <u>Permitted Investments</u>.

Borrowers acknowledge that the only Permitted Investment which the Lockbox Bank may offer on the Lockbox Account and any Reserve Accounts is an interest bearing escrow account (bearing interest at a money market rate as determined by the Lockbox Bank). Administrative Agent and Lenders shall not be liable for any loss sustained on the investment of any funds deposited in the Accounts. All income earned from Permitted Investments shall be for the account of Borrowers. Borrowers hereby irrevocably authorize and direct the applicable Eligible Institution to hold any income earned from Permitted Investments as part of the Accounts. Borrowers shall be responsible for payment of any federal, state or local income or other tax applicable to income earned from Permitted Investments.

3.05. <u>Transfers from Account</u>. Borrowers shall cause each Manager to transfer funds from the Property Account and the Manager Accounts:

(i) if not in a Lockbox Period, to Borrowers pursuant to the applicable Management Agreement;

(ii) during any Lockbox Period, as directed in writing by Administrative Agent, after payment of Property related expenses in accordance with the Management Agreements (other than any Incentive Management Fees) to the Lockbox Account.

3.06. <u>Transfer To and Disbursements from the Lockbox Account</u>. Administrative Agent and Borrowers agree that during the continuance of a Lockbox Period, the following provisions shall apply (in accordance with the Lockbox Account Agreement):

(a) During any Lockbox Period (but prior to enforcement by Administrative Agent following an Event of Default, as set forth in <u>Section 3.10</u> below), on the date that is five days after each Payment Date (and if such day is not a Business Day then the preceding

day which is a Business Day), the Lockbox Bank shall withdraw, disburse and reserve funds in the Lockbox Account and the applicable Reserve Accounts, in the following order of priority:

(i) First:

(1) the Lockbox Bank shall: (A) transfer the Monthly Tax Deposit from the Lockbox Account to the Tax Account, and (B) disburse from the Tax Account, funds sufficient to pay Taxes and Other Charges then due and payable, in accordance with the Approved Annual Budget for such calendar month (or such other amount as may be approved by Administrative Agent) to Administrative Agent to pay such expenses;

(2) the Lockbox Bank shall: (A) transfer an amount equal to the Monthly Insurance Premium Deposit from the Lockbox Account to the Insurance Premium Account, and (B) disburse from the Insurance Premium Account, the Insurance Premiums then due and payable to FelCor Trust to pay Borrowers' pro rata portion of such expenses under the blanket insurance coverage policy by FelCor Trust, which disbursement shall be made upon evidence from FelCor Trust of payment of such Insurance Premiums, provided that if Borrowers shall have ceased to participate in such blanket insurance coverage policy, amounts in the Insurance Premium Account may be withdrawn by Administrative Agent to make payments of any Insurance Premiums then due and payable;

(3) the Lockbox Bank shall: (A) transfer the Monthly Capital Expenditures Reserve Deposit from the Lockbox Account to the Capital Expenditures Reserve Account, and (B) disburse funds from the Capital Expenditures Reserve Account to Borrowers (or to Manager to fund the "Capital Renewals Reserve" (as defined under the Management Agreements), if required under either of the Management Agreements) at the direction of Administrative Agent, which direction shall permit disbursement of funds sufficient to pay for any FF&E Expenditures (other than costs set forth on Schedule 9.05) submitted by Borrowers in accordance with the Approved Annual Budget (and such other amounts as approved by Administrative Agent in its discretion), which Borrowers submission shall include any documentation required by the Lockbox Account Agreement;

(ii) <u>Second</u>, the Lockbox Bank shall disburse funds sufficient to pay the Debt Service related to the Obligations from the Lockbox Account to Administrative Agent for application to Debt Service;

(iii) <u>Third</u>, the Lockbox Bank shall disburse to itself funds sufficient to pay for customary and reasonable fees and expenses incurred in connection with this Agreement and the accounts established hereunder; and

(iv) <u>Fourth</u>, disburse to Administrative Agent, funds sufficient for Administrative Agent to pay any Incentive Management Fees then due and payable upon Administrative Agent's receipt of satisfactory invoices from the Managers.

(b) Any amounts remaining in the Lockbox Account after the fundings above (and other than the amounts specifically reserved as set forth above) during any Lockbox Period shall accrue and shall be available in the following month to make such payments or fill such reserve deficiencies, and shall be held by the Lockbox Bank on behalf of the Secured Parties, as additional security for the Obligations.

(c) Administrative Agent shall give Borrowers notice of any shortfall of cash available in the Lockbox Account to satisfy the various payments and reserves set forth in Section 3.06(a), and Borrowers shall confirm whether it will deposit additional cash into such Lockbox Account or specific Reserve Accounts through additional investments by FelCor Trust in Borrowers or otherwise by the date that is five (5) Business Days prior to the date on which such payment is due.

3.07. Lockbox Trigger Event Cure.

Upon the occurrence of a Lockbox Trigger Event Cure, and acceptance of such cure by Administrative Agent (such acceptance not to be unreasonably withheld or delayed so long as no Default then exists, the Loan has not been accelerated, and Administrative Agent or Lenders have not moved for a receiver or commenced foreclosure proceedings), the Lockbox Trigger Event shall be deemed cured. In such event, such Lockbox Period shall terminate and any funds contained in the Lockbox Account not otherwise applied or disbursed by Administrative Agent in accordance with this Agreement or the other Loan Documents, shall be distributed to Borrowers, so long as no Event of Default would result therefrom.

3.08. <u>Control</u>.

Borrowers acknowledge and agree that (i) the Accounts (other than the Property Accounts and the Manager Accounts) are subject to the sole dominion, control and discretion of Administrative Agent (for the benefit of the Secured Parties), its authorized agents or designees, including Lockbox Bank, subject to the terms hereof; and Borrowers shall have no right of withdrawal with respect to any Account except as provided herein, as set forth in the Management Agreements as amended by that certain Manager's Consent and Subordination of Management Agreement relating to each Management Agreement or with the prior written consent of Administrative Agent, and (ii) pursuant to the Management Agreements and Lockbox Agreement, once effective, Manager and the Lockbox Bank shall comply with instructions originated by Administrative Agent for the benefit of the Secured Parties directing disposition of the funds in such Account without further consent by Borrowers.

3.09. Security Interest.

Without limitation of the grants set forth in the Mortgages, Borrowers hereby grant to Administrative Agent for the benefit of the Secured Parties a first priority security interest in all

of its right title and interest in and to each of the Accounts and the Account Collateral, whether now existing or hereafter acquired, and any and all monies now or hereafter deposited in each Reserve Fund and related Accounts; in each case as additional security for the Obligations.

3.10. <u>Rights on Default</u>.

Notwithstanding anything to the contrary in this Article III, upon the occurrence of an Event of Default, Administrative Agent shall promptly notify each Manager and Lockbox Bank in writing of such Event of Default and (a) Borrowers shall have no further right in respect of (including, without limitation, the right to instruct any Lockbox Bank or Manager to transfer from) the Accounts, (b) Administrative Agent may direct Lockbox Account to liquidate and transfer any amounts then invested in Permitted Investments to the Accounts or reinvest such amounts in other Permitted Investments as Administrative Agent may determine is necessary to perfect or protect any security interest granted or purported to be granted hereby or pursuant to the other Loan Documents or to enable Lockbox Bank, as agent for Administrative Agent, or Administrative Agent to exercise and enforce Administrative Agent's or Lenders rights and remedies hereunder or under any other Loan Document with respect to any Account or any Account Collateral, (c) Administrative Agent is authorized to endorse, in the name of the applicable Borrower, any item howsoever received by Administrative Agent, representing any payment on or other proceeds of any Account Collateral, and (d) Administrative Agent shall have all rights and remedies with respect to the Accounts and the amounts on deposit therein and the Account Collateral as described in this Agreement and in the Mortgages, in addition to all of the rights and remedies available to a secured party under the UCC, and, notwithstanding anything to the contrary contained in this Agreement or in the Mortgages, Administrative Agent may apply the amounts of such Accounts as Administrative Agent determines in its sole discretion including, but not limited to, payment of the Obligations.

Upon the occurrence of an Event of Default, each Borrower hereby appoints Administrative Agent (and any Person Administrative Agent may designate) as its attorney-in-fact to accomplish the purposes of this <u>Article III</u> hereof, including without limitation the entry into a Lockbox Agreement under <u>Section 3.01(b)</u> hereof, and taking any action and executing or completing any instrument that Administrative Agent may reasonably deem necessary or advisable to accomplish the purposes hereof, which appointment is irrevocable and coupled with an interest.

3.11. Financing Statement; Further Assurances.

Borrowers hereby authorize Administrative Agent to file a financing statement or statements under the UCC in connection with any of the Accounts and the Account Collateral with respect thereto in the form required to properly perfect the security interest of the Secured Parties therein. Borrowers agree that at any time and from time to time, at the expense of Borrowers, Borrowers will promptly execute and deliver all further instruments and documents, and take all further action, that may be necessary or desirable, or that Administrative Agent may request, in order to perfect and protect any security interest granted or purported to be granted hereby (including, without limitation, any security interest in and to any Permitted Investments) or to enable Lockbox Bank or Administrative Agent to exercise and enforce its rights and remedies hereunder with respect to any Account Collateral.

3.12. Borrowers Obligations Not Affected.

The insufficiency of funds on deposit in the Accounts shall not absolve Borrowers of the obligation to make any payments, as and when due pursuant to this Agreement and the other Loan Documents, and such obligations shall be separate and independent, and not conditioned on any event or circumstance whatsoever.

3.13. <u>Deposit Accounts</u>. With respect to each Account, each Borrower represents, covenants and agrees that:

(a) This Agreement, the Mortgages, and the Lockbox Agreement, taken together, shall create valid and continuing security interests (as defined in the UCC) in the Accounts, in favor of the Secured Parties, which security interests shall be prior to all other Liens and are enforceable as such against creditors of and purchasers from Borrowers;

(b) The Accounts, shall each constitute "deposit accounts" within the meaning of the applicable UCC;

(c) The applicable Borrower owns and has good and marketable title to its Accounts, free and clear of any Lien or claim of any Person (other than Administrative Agent) and the applicable Manager (under the applicable Management Agreement);

(d) Other than the security interest granted to Administrative Agent for the benefit of the Secured Parties pursuant to this Agreement, the Mortgages and the agreement with Manager with respect to the Property Account and the Manager Accounts, no Borrower has pledged, assigned, or sold, granted a security interest in, or otherwise conveyed, any of the Accounts;

(e) The Accounts are not in the name of any Person other than a Borrower, a Manager (under the applicable Management Agreement) or Administrative Agent; and

(f) No Borrower has authorized (or will authorize) the banks maintaining the Accounts to comply with instructions of any Person other than the applicable Manager (under its Management Agreement, solely with respect to the Property Account and the Manager Accounts) and Administrative Agent during a Lockbox Period.

ARTICLE IV TAXES, YIELD PROTECTION AND ILLEGALITY

4.01. <u>Taxes</u>.

(a) <u>Payments Free of Taxes</u>. Any and all payments by or on account of any obligation of Borrowers hereunder or under any other Loan Document shall be made free and clear of and without reduction or withholding for any Indemnified Taxes or Other Taxes, provided that if any Borrower shall be required by applicable law to deduct any Indemnified Taxes (including any Other Taxes) from such payments, then (i) the sum payable shall be increased as necessary so that after making all required deductions (including deductions applicable to additional sums payable under this Section) Administrative Agent or any

Lender, as the case may be, receives an amount equal to the sum it would have received had no such deductions been made, (ii) such Borrower shall make such deductions and (iii) such Borrower shall timely pay the full amount deducted to the relevant Governmental Authority in accordance with applicable law.

(b) <u>Payment of Other Taxes by Borrowers</u>. Without limiting the provisions of subsection (a) above, Borrowers shall timely pay any Other Taxes to the relevant Governmental Authority in accordance with applicable law.

(c) <u>Indemnification by Borrowers</u>. Borrowers shall, jointly and severally, indemnify Administrative Agent and each Lender, within ten (10) days after demand therefor, for the full amount of any Indemnified Taxes or Other Taxes (including Indemnified Taxes or Other Taxes imposed or asserted on or attributable to amounts payable under this Section) paid by Administrative Agent or such Lender, as the case may be, and any penalties, interest and reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes or Other Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to Borrowers by a Lender (with a copy to Administrative Agent), or by Administrative Agent on its own behalf or on behalf of a Lender, shall be conclusive absent manifest error.

(d) <u>Evidence of Payments</u>. As soon as practicable after any payment of Indemnified Taxes or Other Taxes by any Borrower to a Governmental Authority, such Borrower shall deliver to Administrative Agent the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to Administrative Agent.

(e) <u>Status of Lenders</u>. Any Foreign Lender that is entitled to an exemption from or reduction of withholding tax under the law of the jurisdiction in which a Borrower is resident for tax purposes, or any treaty to which such jurisdiction is a party, with respect to payments hereunder or under any other Loan Document shall deliver to such Borrower (with a copy to Administrative Agent), at the time or times prescribed by applicable law or reasonably requested by such Borrower or Administrative Agent, such properly completed and executed documentation prescribed by applicable law as will permit such payments to be made without withholding or at a reduced rate of withholding. In addition, any Lender, if requested by applicable law or reasonably requested by applicable law or reasonably requested by applicable law or reasonably requested by account of Administrative Agent, shall deliver such other documentation prescribed by applicable law or reasonably requested by any Borrower or Administrative Agent, shall deliver such other documentation prescribed by applicable law or reasonably requested by such Borrower or Administrative Agent to determine whether or not such Lender is subject to backup withholding or information reporting requirements.

Without limiting the generality of the foregoing, if any Borrower is resident for tax purposes in the United States, any Foreign Lender shall deliver to such Borrower and Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender under this Agreement (and from time to time thereafter upon the request of any Borrower or Administrative Agent, but only if such Foreign Lender is legally entitled to do so), whichever of the following is applicable:

(i) duly completed copies of Internal Revenue Service Form W-8BEN claiming eligibility for benefits of an income tax treaty to which the United States is a party,

(ii) duly completed copies of Internal Revenue Service Form W-8ECI,

(iii) in the case of a Foreign Lender claiming the benefits of the exemption for portfolio interest under section 881(c) of the Code, (A) a certificate to the effect that such Foreign Lender is not (1) a "bank" within the meaning of section 881(c)(3)(A) of the Code, (2) a "10 percent shareholder" of a Borrower within the meaning of section 881(c)(3)(B) of the Code, or (3) a "controlled foreign corporation" described in section 881(c)(3)(C) of the Code and (B) duly completed copies of Internal Revenue Service Form W-8BEN, or

(iv) any other form prescribed by applicable law as a basis for claiming exemption from or a reduction in United States Federal withholding tax duly completed together with such supplementary documentation as may be prescribed by applicable law to permit such Borrower to determine the withholding or deduction required to be made.

Treatment of Certain Refunds. If Administrative Agent or any Lender (f) determines, in its reasonable discretion, that it has received a refund of any Taxes or Other Taxes as to which it has been indemnified by Borrowers or with respect to which a Borrower has paid additional amounts pursuant to this Section, it shall pay to such Borrower an amount equal to such refund (but only to the extent of indemnity payments made, or additional amounts paid, by such Borrower under this Section with respect to the Taxes or Other Taxes giving rise to such refund), net of all out-of-pocket expenses of Administrative Agent or such Lender, as the case may be, and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund), provided that such Borrower, upon the request of Administrative Agent or such Lender agrees to repay the amount paid over to such Borrower (plus any penalties, interest or other charges imposed by the relevant Governmental Authority) to Administrative Agent or such Lender if Administrative Agent or such Lender is required to repay such refund to such Governmental Authority. This subsection shall not be construed to require Administrative Agent or any Lender to make available its tax returns (or any other information relating to its taxes that it deems confidential) to any Borrower or any other Person.

4.02. <u>Illegality</u>. If any Lender determines that any Law has made it unlawful, or that any Governmental Authority has asserted that it is unlawful, for any Lender or its applicable Lending Office to make, maintain or fund Eurodollar Loans, or to determine or charge interest rates based upon the Adjusted LIBO Rate, or any Governmental Authority has imposed material restrictions on the authority of such Lender to purchase or sell, or to take deposits of, Dollars in the London interbank market, then, on notice thereof by such Lender to Borrowers through Administrative Agent, any obligation of such Lender to make or continue Eurodollar Loans or to convert ABR Loans to Eurodollar Loans shall be suspended until such Lender notifies Administrative Agent and Borrowers that the circumstances giving rise to such determination no longer exist. Upon receipt

of such notice, Borrowers shall, upon demand from such Lender (with a copy to Administrative Agent), prepay or, if applicable, convert all Eurodollar Loans of such Lender to ABR Loans, either on the last day of the Interest Period therefor, if such Lender may lawfully continue to maintain such Eurodollar Loans to such day, or immediately, if such Lender may not lawfully continue to maintain such Eurodollar Loans. Upon any such prepayment or conversion, Borrowers shall also pay accrued interest on the amount so prepaid or converted.

4.03. <u>Inability to Determine Rates</u>. If the Required Lenders determine that for any reason in connection with any request for a Eurodollar Loan or a conversion to or continuation thereof that (a) Dollar deposits are not being offered to banks in the London interbank eurodollar market for the applicable amount and Interest Period of such Eurodollar Loan, (b) adequate and reasonable means do not exist for determining the Adjusted LIBO Rate for any requested Interest Period with respect to a proposed Eurodollar Loan, or (c) the Adjusted LIBO Rate for any requested Interest Period with respect to a proposed Eurodollar Loan does not adequately and fairly reflect the cost to such Lenders of funding such Loan, Administrative Agent will promptly so notify Borrowers and each Lender. Thereafter, the obligation of the Lenders to make or maintain Eurodollar Loan shall be suspended until Administrative Agent (upon the instruction of the Required Lenders) revokes such notice. Upon receipt of such notice, Borrowers may revoke any pending request for the Loans or conversion to or continuation of Eurodollar Loans or, failing that, will be deemed to have converted such request into a request for ABR Loans in the amount specified therein.

4.04. <u>Increased Costs; Reserves on Eurodollar Loans</u>. (a) Increased Costs Generally. If any Change in Law shall:

(i) impose, modify or deem applicable any reserve, special deposit, compulsory loan, insurance charge or similar requirement against assets of, deposits with or for the account of, or credit extended or participated in by, any Lender;

(ii) subject any Lender to any tax of any kind whatsoever with respect to this Agreement or any Eurodollar Loan made by it, or change the basis of taxation of payments to such Lender in respect thereof (except for Indemnified Taxes or Other Taxes covered by Section 4.01 and the imposition of, or any change in the rate of, any Excluded Tax payable by such Lender); or

(iii) impose on any Lender or the London interbank market any other condition, cost or expense affecting this Agreement or Eurodollar Loans made by such Lender;

and the result of any of the foregoing shall be to increase the cost to such Lender of making or maintaining any Eurodollar Loan (or of maintaining its obligation to make any such Loan) or to reduce the amount of any sum received or receivable by such Lender hereunder (whether of principal, interest or any other amount) then, upon request of such Lender, Borrowers will pay to such Lender such additional amount or amounts as will compensate such for such additional costs incurred or reduction suffered.

(b) <u>Capital Requirements</u>. If any Lender determines that any Change in Law affecting such Lender or any Lending Office of such Lender or such Lender's holding company, if any, regarding capital requirements has or would have the effect of reducing the rate of return on such Lender's capital or on the capital of such Lender's holding company, if any, as a consequence of this Agreement, the Commitments of such Lender or the Loans made by such Lender to a level below that which such Lender or such Lender's holding company could have achieved but for such Change in Law (taking into consideration such Lender's policies and the policies of such Lender's holding company with respect to capital adequacy), then from time to time Borrowers will pay to such Lender such additional amount or amounts as will compensate such Lender or such Lender's holding company for any such reduction suffered.

(c) <u>Certificates for Reimbursement</u>. A certificate of a Lender setting forth the amount or amounts necessary to compensate such Lender or its holding company, as the case may be, as specified in subsection (a) or (b) of this Section and delivered to Borrowers shall be conclusive absent manifest error. Borrowers shall pay such Lender the amount shown as due on any such certificate within ten (10) days after receipt thereof.

(d) <u>Delay in Requests</u>. Failure or delay on the part of any Lender to demand compensation pursuant to the foregoing provisions of this Section shall not constitute a waiver of such Lender's right to demand such compensation, provided that Borrowers shall not be required to compensate a Lender pursuant to the foregoing provisions of this Section for any increased costs incurred or reductions suffered more than nine (9) months prior to the date that such Lender notifies Borrowers of the Change in Law giving rise to such increased costs or reductions and of such Lender's intention to claim compensation therefor (except that, if the Change in Law giving rise to such increased costs or reductions is retroactive, then the nine (9) month period referred to above shall be extended to include the period of retroactive effect thereof).

4.05. <u>Compensation for Losses</u>. Upon demand of any Lender (with a copy to Administrative Agent) from time to time, Borrowers shall promptly compensate such Lender for and hold such Lender harmless from any loss, cost or expense incurred by it as a result of:

(a) any continuation, conversion, payment or prepayment of any Loan other than a ABR Loan on a day other than the last day of the Interest Period for such Loan (whether voluntary, mandatory, automatic, by reason of acceleration, or otherwise);

(b) any failure by any Borrower (for a reason other than the failure of such Lender to make a Loan) to prepay, borrow, continue or convert any Loan other than a ABR Loan on the date or in the amount notified by such Borrower; or (c) any assignment of a Eurodollar Loan on a day other than the last day of the Interest Period therefor as a result of a request by Borrowers pursuant to <u>Section 12.14;</u>

including any loss of anticipated profits and any loss or expense arising from the liquidation or reemployment of funds obtained by it to maintain such Loan or from fees payable to terminate the deposits from which such funds were obtained. Borrowers shall also pay any customary administrative fees charged by such Lender in connection with the foregoing.

For purposes of calculating amounts payable by Borrowers to the Lenders under this Section 4.05, each Lender shall be deemed to have funded each Eurodollar Loan made by it at the Adjusted LIBO Rate for such Loan by a matching deposit or other borrowing in the London interbank eurodollar market for a comparable amount and for a comparable period, whether or not such Eurodollar Loan was in fact so funded.

4.06. <u>Mitigation Obligations; Replacement of Lenders</u>. xi) <u>Designation of a Different</u> <u>Lending Office</u>. If any Lender requests compensation under <u>Section 4.04</u>, or any Borrower is required to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to <u>Section 4.01</u>, or if any Lender gives a notice pursuant to <u>Section 4.02</u>, then such Lender shall use reasonable efforts to designate a different Lending Office for funding or booking its Loans hereunder or to assign its rights and obligations hereunder to another of its offices, branches or affiliates, if, in the judgment of such Lender, such designation or assignment (i) would eliminate or reduce amounts payable pursuant to <u>Section 4.01</u> or <u>4.04</u>, as the case may be, in the future, or eliminate the need for the notice pursuant to <u>Section 4.02</u>, as applicable, and (ii) in each case, would not subject such Lender to any unreimbursed cost or expense and would not otherwise be disadvantageous to such Lender. Borrowers hereby agree to pay all reasonable costs and expenses incurred by any Lender in connection with any such designation or assignment.

(b) <u>Replacement of Lenders</u>. If any Lender requests compensation under <u>Section</u> <u>4.04</u>, or if any Borrower is required to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to <u>Section 4.01</u>, Borrowers may replace such Lender in accordance with <u>Section 12.14</u>.

4.07. <u>Survival</u>. All of Borrowers' obligations under this Article IV shall survive termination of the Commitments and repayment of all other Obligations hereunder.

4.08. Defaulting Lender.

Notwithstanding any provision of this Agreement to the contrary, if any Lender becomes a Defaulting Lender, then the following provisions shall apply for so long as such Lender is a Defaulting Lender:

(a) unused fees shall cease to accrue on the unfunded portion of the Commitment of such Defaulting Lender;

(b) the Commitment and Revolving Credit Exposure of such Defaulting Lender shall not be included in determining whether the Required Lenders, the Supermajority Lenders or all Lenders have taken or may take any action hereunder (including any consent to any amendment, waiver or other modification pursuant to <u>Section 12.01</u>); except (i) such Defaulting Lender's Commitment may not be increased or extended without its consent and (ii) the principal amount of, or interest or fees payable on, Loans may not be reduced or excused or the scheduled date of payment may not be postponed as to such Defaulting Lender without such Defaulting Lender's consent;

(c) if any Swingline Exposure exists at the time such Lender becomes a Defaulting Lender then:

(i) all or any part of the Swingline Exposure of such Defaulting Lender shall be reallocated among the non-Defaulting Lenders in accordance with their respective Applicable Percentages but only to the extent the sum of all non-Defaulting Lenders' Revolving Credit Exposures plus such Defaulting Lender's Swingline Exposure does not exceed the total of all non-Defaulting Lenders' Commitments;

(ii) if the reallocation described in clause (i) above cannot, or can only partially, be effected, Borrowers shall within one Business Day following notice by Administrative Agent prepay such Swingline Exposure;

(d) so long as such Lender is a Defaulting Lender, Swingline Lender shall not be required to fund any Swingline Loan, unless it is satisfied that the related exposure will be 100% covered by the Commitments of the non-Defaulting Lenders in accordance with <u>Section 4.08(b)</u>, and participating interests in any newly made Swingline Loan shall be allocated among non-Defaulting Lenders in a manner consistent with <u>Section 4.08(b)(i)</u> (and such Defaulting Lender shall not participate therein).

If (i) a Bankruptcy Event with respect to a Parent of any Lender shall occur following the date hereof and for so long as such event shall continue or (ii) Swingline Lender has a good faith belief that any Lender has defaulted in fulfilling its obligations under one or more other agreements in which such Lender commits to extend credit, Swingline Lender shall not be required to fund any Swingline Loan, unless Swingline Lender shall have entered into arrangements with Borrowers or such Lender, satisfactory to Swingline Lender to defease any risk to it in respect of such Lender hereunder.

In the event that Administrative Agent, Borrowers and Swingline Lender agree that a Defaulting Lender has adequately remedied all matters that caused such Lender to be a Defaulting Lender, then the Swingline Exposure of the Lenders shall be readjusted to reflect the inclusion of such Lender's Commitment and on such date such Lender shall purchase at par such of the Loans of the other Lenders (other than Swingline Loans) as Administrative Agent shall determine may be necessary in order for such Lender to hold such Loans in accordance with its Applicable Percentage.

ARTICLE V CONDITIONS PRECEDENT TO LOANS

5.01. <u>Conditions to Closing Date</u>. The obligation of each Lender to make the initial Loans and the occurrence of the Closing Date hereunder is subject to satisfaction of the following conditions precedent:

(a) Administrative Agent's receipt of the following, each of which shall be originals or facsimiles (followed promptly by originals) unless otherwise specified, each properly executed by a Responsible Officer of the signing Loan Party (where applicable), each dated the Closing Date (or, in the case of certificates of governmental officials, a recent date before the Closing Date) and each in form and substance reasonably satisfactory to Administrative Agent and its legal counsel:

(i) counterparts executed by the Loan Parties of each Loan Document, including, without limitation, this Agreement, each Guaranty, Pledge Agreement and with respect to each Collateral Property the Environmental Indemnity and the Collateral Documents, in each case, sufficient in number for distribution to Administrative Agent, each Lender, and Borrowers;

(ii) Notes executed by Borrowers in favor of each Lender requesting a Note;

(iii) fully executed Operating Lease Subordination Agreements;

(iv) such certificates of resolutions or other action, incumbency certificates, and/or other certificates of Responsible Officers of each Loan Party as Administrative Agent may require evidencing the identity, authority, and capacity of each Responsible Officer thereof authorized to act as a Responsible Officer in connection with this Agreement and the other Loan Documents to which such Loan Party is a party;

(v) copies of the Organizational Documents of each Loan Party, together with such other documents and certifications as Administrative Agent may reasonably require to evidence that each Loan Party is duly organized or formed, and that each Loan Party is validly existing, in good standing, and qualified to engage in business in each jurisdiction where its ownership, lease or operation of properties or the conduct of its business requires such qualification, except to the extent that failure to be so qualified could not reasonably be expected to have a Material Adverse Effect;

(vi) a favorable opinion of one or more firms counsel to the Loan Parties, addressed to Administrative Agent and each Lender, including (x) Akin, Gump, Strauss, Hauer & Feld, L.L.P. and (y) such local counsel opinions for each jurisdiction in which the Collateral Properties are located, in each case as to such matters concerning the Loan Parties and the Loan Documents as Administrative Agent may reasonably request, including existence, power and authority; enforceability; non-contravention; no consent; and perfection;

(vii) a certificate of a Responsible Officer of each Loan Party (other than Borrowers) either (A) attaching copies of all consents, licenses, and approvals required in connection with the execution, delivery and performance by such Loan Party and the validity against such Loan Party of the Loan Documents to which it is a party, and such consents, licenses, and approvals shall be in full force and effect, or (B) stating that no such consents, licenses, or approvals are so required;

(viii) completed and executed Officer's Certificate (A) providing calculations of (x) the Loan to Value Ratio, (y) the Debt Service Coverage Ratio and (z) pro forma compliance with the restrictions on indebtedness set forth in Section 4.03 of the Senior Secured Notes Indenture; in each case, after giving effect to the Aggregate Commitments and the Borrowings on the Closing Date; which calculations shall show a Loan to Value Ratio of not more than sixty two and three quarters percent (62.75%) and a pro forma Debt Service Coverage Ratio as of December 31, 2010 of at least 1.30 to 1.00, (B) providing calculations showing that no Collateral Property comprises more than twenty percent (20%) of the aggregate Net Operating Income of all the Collateral Properties as of the Closing Date, except for the Mandalay Beach Property, the Net Operating Income of which does not comprise more than twenty five percent (25%) of the aggregate Net Operating Income of all the Collateral Properties as of the Closing Date, and (C) attaching the Annual Budget for 2011 for each of the Collateral Properties;

(ix) evidence that all insurance required to be maintained pursuant to the Loan Documents has been obtained and is in effect;

(x) unless otherwise agreed or approved by Administrative Agent, (A) two (2) prints of an original ALTA survey (a "<u>Survey</u>") of each Collateral Property and improvements thereon dated not more than sixty (60) days prior to the date of this Agreement (or dated such earlier date, if any, as is satisfactory to Administrative Agent and the Title Company, but in any event not more than one hundred eighty (180) days prior to the date of this Agreement) and otherwise complying with <u>Exhibit H</u> to the extent required by Administrative Agent and the Title Company; and (B) a Flood Insurance Policy for each Collateral Property in an amount required by Administrative Agent, but in no event less than the amount sufficient to meet the requirements of applicable law and the Flood Insurance Acts, or evidence satisfactory to Administrative Agent that none of such Collateral Property is located within a one hundred year flood plain or in a flood hazard area as defined by the Federal Insurance Administration and appropriate flood certificates acceptable to Administrative Agent;

(xi) unless otherwise agreed or approved by Administrative Agent, true and correct copies of all existing plans with respect to the Collateral Properties within the possession or control of any of the Loan Parties (including the site plan) requested by Administrative Agent, together with evidence satisfactory to Administrative Agent that the same comply in all material respects to applicable requirements of Governmental Authorities;

with respect to each Collateral Property: (A) true and correct copies (xii) of each Major Lease, as well as the Operating Leases, any Ground Lease, and (if applicable) Guarantees thereof; (B) estoppel certificates and subordination and attornment agreements (including nondisturbance agreements if and to the extent agreed by Administrative Agent in its discretion), dated within thirty (30) days prior to this Agreement and in form and content satisfactory to Administrative Agent, from the tenants and subtenants as Administrative Agent requires; (C) estoppel and consent agreements from each of the Ground Lessors, dated within thirty (30) days prior to this Agreement and in form and content reasonably satisfactory to Administrative Agent; (D) copies of all personal property leases under which (1) payments by Borrowers exceed \$50,000 in any year, and/or (2) the term exceeds one year; (E) copies of all operating and service agreements under which (1) payments by Borrowers exceed \$50,000 in any year, and/or (2) the term exceeds one year, and if reasonably requested by Administrative Agent estoppel and recognition agreements relating thereto and (F) evidence of the applicable Borrower's or the applicable Loan Party's compliance with each Major Lease delivered pursuant to clause (A) above;

(xiii) evidence satisfactory to Administrative Agent that no portion of any Collateral Property is "wetlands" under any applicable Law and no Collateral Property contains nor is within or near any area designated as a hazardous waste site by any Governmental Authority, that no Collateral Property or any adjoining property contains or has ever contained any Hazardous Material under any Law pertaining to health or the environment, and that no Collateral Property or any use or activity thereon violates or is or could be subject to any response, remediation, clean-up, or other obligation under any Law pertaining to health or the environment including without limitation, a written report of an environmental assessment (including soil assessments) of each Collateral Property, made within thirty (30) days prior to the date of this Agreement (other than for the Boca Raton Property and Dana Point Property, which reports were obtained in connection with the 2010 Term Loan) (in each case, an "Environmental Report"), by an engineering firm, and of a scope and in form and content satisfactory to Administrative Agent, complying with Administrative Agent's established guidelines, showing that there is no evidence of any Hazardous Material which has been generated, treated, stored, released, or disposed of in any Collateral Property, and such additional evidence as may be required by Administrative Agent. All reports, drafts of reports, and recommendations, whether written or oral, from such engineering firm shall be made available and communicated to Administrative Agent;

(xiv) (A) evidence that each Collateral Property abuts and has fully adequate direct and free access to one or more public streets, dedicated to public use, fully installed and accepted by the appropriate Governmental Authority, that all fees, costs and expenses of the installation and acceptance thereof have been paid in full, and that there are no restrictions on the use and enjoyment of such streets which would adversely affect such Collateral Property; (B) evidence that each Collateral Property shall have access to utilities; (C) evidence that all applicable zoning ordinances, restrictive covenants, and Laws affecting each Collateral Property permit the use for which such Collateral Property is intended and have been or will be complied with without the existence of any variance, non-complying use, nonconforming use or other special exception; (D) evidence that each Collateral Property and Improvements comply and will comply with all Laws regarding subdivision and platting and would so comply if such Collateral Property and the Improvements thereon were conveyed as a separate parcel; and (E) evidence of compliance by Borrowers and each Collateral Property, and any proposed construction, use and occupancy of the Improvements, with such other applicable Laws as Administrative Agent may request, including all Laws regarding access and facilities for handicapped or disabled persons including, without limitation and to the extent applicable, The Federal Architectural Barriers Act (42 U.S.C. § 4151 et seq.), The Fair Housing Amendments Act of 1988 (42 U.S.C. § 3601 et seq.), The Americans With Disabilities Act of 1990 (42 U.S.C. § 12101 et seq.), The Rehabilitation Act of 1973 (29 U.S.C. § 794), and any applicable state requirements, with such exceptions therefrom as previously disclosed in writing and accepted by Administrative Agent;

(xv) evidence (A) of the identity of all taxing authorities and utility districts (or similar authorities) currently exercising ad valorem or real property taxing or assessment jurisdiction over any Collateral Property or any portion thereof; (B) that all taxes, standby fees and any other similar charges have been paid, including copies of receipts or statements marked "paid" by the appropriate authority; and (C) that each Collateral Property is a separate tax lot or lots with separate assessment or assessments of the Collateral Property and Improvements, independent of any other Collateral Property or improvements and that each Collateral Property is a separate legally subdivided parcel;

(xvi) (A) executed, acknowledged, and/or sworn to as required counterparts of the Mortgages for each Collateral Property, which shall have been delivered to the Title Company and released for recordation in the official records of the city or county in which each Collateral Property is located, and (B) UCC-1 financing statements which shall have been furnished for filing in all filing offices that Administrative Agent may require;

(xvii) a Title Policy or a Title Policy Commitment (or a Title Policy promulgated by the Laws of the state in which each respective Collateral Property is located if an ALTA insurance policy is not available). No Borrower and none of Borrowers' counsel shall have any interest, direct or indirect, in the Title Company (or its agent) or any portion of the premium paid for the Title Insurance;

(xviii) (A) evidence that immediately prior to the Closing Date and as of the time the Mortgages will be filed for record, except for Permitted Liens: (1) no contract, or memorandum thereof, for construction, design, surveying, or any other service relating to any Collateral Property has been filed for record in the county where such Collateral Property is located; and (2) no mechanic's or materialman's Lien claim or notice, lis pendens, judgment, or other claim or encumbrance against such Collateral Property has been filed for record in the Collateral

Property is located or in any other public record which by Law provides notice of claims or encumbrances regarding such Collateral Property; (B) a certificate or certificates of a reporting service acceptable to Administrative Agent, reflecting the results of searches of each of the Loan Parties made within forty-five (45) days prior to the Closing Date (except in the case of searches of FelCor Op or FelCor Trust, which shall have been made within sixty (60) days prior to the Closing Date), (1) of the central and local Uniform Commercial Code records, showing no filings against any of the collateral for the Obligations or against Borrowers otherwise except as consented to by Administrative Agent; and (2) of the appropriate judgment and tax Lien records, showing no outstanding judgment or tax Lien against Borrowers;

(xix) to the extent reasonably deemed necessary by Administrative Agent, an executed REA estoppel letter from each party to any REA for any applicable Collateral Property;

(xx) a true and correct copy of (A) each Franchise Agreement and fully executed "comfort letter" for each such Franchise Agreement in form and substance satisfactory to Administrative Agent, (B) each Management Agreement for each Collateral Property, together with a fully executed Manager's Consent and Subordination of Management Agreement relating to each such Management Agreement (other than the Myrtle Beach Golf Course Management Agreement); in each case, which includes an agreement to attorn to Administrative Agent in the event Administrative Agent takes possession of any Collateral Property by foreclosure, deed in lieu of foreclosure or otherwise and (C) all Licenses relating to the Collateral Properties, provided that Borrowers obligations to deliver copies of such Licenses prior to the Closing Date shall be limited to an obligation to use commercially reasonable efforts to do so;

(xxi) a Physical Condition Report for each Collateral Property (other than for the Boca Raton Property and Dana Point Property, which reports were obtained in connection with the 2010 Term Loan);

(xxii) an earthquake or seismic condition report for each of the Dana Point Property, the Santa Barbara Property, the Santa Monica Property and the Mandalay Beach Property;

(xxiii) Unaudited Financial Statements of each Borrower;

(xxiv) Acceptable Appraisals of each Collateral Property, which collectively show: (A) a Loan to Value Ratio for all of the Collateral Properties, of no more than sixty two and one half percent (62.5%) after giving effect to the Aggregate Commitments on the Closing Date; (B) no Collateral Property comprises more than twenty percent (20%) of the aggregate Net Operating Income of all the Collateral Properties as of the Closing Date, except for the Mandalay Beach Property, the Net Operating Income of which does not comprise more than twenty five percent (25%) of the aggregate Net Operating Income of all the Collateral Properties as of the Closing Date; (xxv) such payoff letters in respect of the repayment of any Indebtedness outstanding on the Closing Date not permitted pursuant to Section 8.01, including the 2009 Term Loan and the 2010 Term Loan, and evidence of releases of Liens outstanding on the Closing Date and not permitted pursuant to Section 8.03, including those Liens securing the 2009 Term Loan and the 2010 Term Loan;

(xxvi) Administrative Agent's Disbursement and Rate Management Signature Authorization and Instruction Form;

(xxvii) a Certification of Non-Foreign Status for each Borrower; and

(xxviii) such other assurances, certificates, documents, consents, or opinions as Administrative Agent or the Lenders reasonably may require.

(b) Any fees required to be paid on or before the Closing Date shall have been paid.

(c) Borrowers shall have paid all reasonable attorney's costs of Administrative Agent to the extent invoiced prior to or on the Closing Date.

(d) The representations and warranties contained in Article VI or any other Loan Document, or which are contained in any document furnished at any time under or in connection herewith or therewith, shall be true and correct on and as of the Closing Date.

(e) No Default or Event of Default shall exist on or result from the occurrence of the Closing Date.

5.02. <u>Each Credit Event</u>. The obligation of each Lender to make any Loan on the occasion of any Borrowing, is subject to the satisfaction of the following conditions:

(a) Borrowers shall be in compliance with the Debt Service Coverage Ratio set forth in Section 8.07 after giving effect to the Loans to be made;

(b) Borrowers shall be in compliance with the restrictions on indebtedness set forth in Section 4.03 of the Senior Secured Notes Indenture after giving effect to the Loans to be made;

(c) The representations and warranties of Borrowers contained in <u>Article VI</u> or any other Loan Document, or which are contained in any document furnished at any time under or in connection herewith, shall be true and correct in all material respects (unless such representations are qualified as to "materiality", "Material Adverse Effect" or with similar language, in which case such representations shall be true in all respects) on and as of the date of such Borrowing, except to the extent such representations and warranties were made as of a specified date, in which case such representation and warranty shall have been true and correct as of such specified date; (d) No Default or Event of Default shall exist, or would result from the making of Loans on or as of the date of such Borrowing;

(e) Administrative Agent shall have one or more Officer's Certificates, each of which shall be originals or facsimiles or electronic copies unless otherwise specified, each properly executed by a Responsible Officer of the signing Loan Party (where applicable), each dated on and as of the date of such Borrowing and each in form and substance reasonably satisfactory to Administrative Agent, certifying to the matters set forth <u>Sections 5.02(a)</u> through (<u>d</u>) above.

ARTICLE VI REPRESENTATIONS AND WARRANTIES

Borrowers each represent and warrant to Administrative Agent and the Lenders that:

Existence, Qualification and Power. Each Borrower (a) is duly organized or formed, 6.01. validly existing and, as applicable, in good standing under the Laws of the jurisdiction of its incorporation or organization, (b) has all requisite power and authority and all requisite governmental licenses, authorizations, consents and approvals to (i) own or lease its assets and carry on its business and (ii) execute, deliver and perform its obligations under the Loan Documents to which it is a party and consummate the transactions contemplated thereby, and (c) is duly qualified and is licensed and, as applicable, in good standing under the Laws of each jurisdiction where its ownership, lease or operation of properties or the conduct of its business requires such qualification or license; except in each case referred to in clause (b)(i) or (c) with respect to qualification to do business, to the extent that failure to do so would not reasonably be expected to have a Material Adverse Effect or, solely as of the Closing Date and the Extension Effective Date, a Material Property Event. Attached hereto as Schedule 6.01 is an organizational chart of each Borrower. Execution and delivery of each of the Loan Documents by each Borrower party thereto, will not: (i) conflict with or result in any breach or contravention of any provision of law, statute, rule or regulation to any Borrowers is subject or any judgment, order, writ, injunction, license or permit applicable to Borrowers, or (ii) conflict with any provision of the corporate charter or bylaws of, or any agreement or other instrument binding upon, Borrowers.

6.02. <u>Proceedings</u>. Each Borrower has taken all necessary action to authorize the execution, delivery and performance of this Agreement and the other Loan Documents to which it is a party. This Agreement and the other Loan Documents have been duly executed and delivered by or on behalf of each Borrower and constitute legal, valid and binding obligations of such Borrower, enforceable against such Borrower in accordance with their respective terms, subject only to applicable Debtor Relief Laws, and subject, as to enforceability, to general principles of equity (regardless of whether enforcement is sought in a proceeding in equity or at law).

6.03. <u>No Conflicts</u>. The execution, delivery and performance of this Agreement and the other Loan Documents by each Borrower will not conflict with or result in a breach of any of the terms or provisions of, or constitute a default under, or result in the creation or imposition of any Lien, charge or encumbrance (other than pursuant to the Loan Documents) upon any of the property or assets of such Borrower, as applicable pursuant to the terms of its Organizational Documents, any indenture, mortgage, deed of trust, loan agreement, partnership agreement, management agreement, franchise agreement, or other agreement or instrument to which any Borrower is a party or by which any of such Borrower property or assets is subject, nor will such action result in any violation of the provisions of any statute or any order, rule or regulation of any court or Governmental Authority or body having jurisdiction over any Borrower or any of the Collateral Properties or any of such Loan Party's other assets, or any license or other approval required to operate the Collateral Properties, and any consent, approval, authorization, order, registration or qualification of or with any Governmental Authority required for the execution, delivery and performance by any Borrower of this Agreement or by Borrowers of any other Loan Document, have been obtained and is in full force and effect, in each case if such Borrower's noncompliance with this Section 6.03 would reasonably be expected to have a Material Adverse Effect.

6.04. <u>Litigation</u>. There are no actions, suits or proceedings at law or in equity by or before any Governmental Authority or other agency now pending or, to its knowledge, threatened against or affecting any Borrower or any Collateral Property, which actions, suits or proceedings, if determined against such Borrower or any Collateral Property, would reasonably be expected to have a Material Adverse Effect.

6.05. <u>Agreements</u>. No Borrower is party to any agreement or instrument or subject to any restriction which would reasonably be expected to have a Material Adverse Effect or, solely as of the Closing Date and the Extension Effective Date, cause a Material Property Event. No Borrower is in default in any material respect in the performance, observance or fulfillment of any of the obligations, covenants or conditions contained in any Contractual Obligation, Law or Legal Requirement to which it is a party or by which such Borrower or any of the Collateral Properties are bound. No Borrower has any material financial obligation under any indenture, mortgage, deed of trust, loan agreement or other agreement or instrument to which such Borrower is a party or by which such Borrower or any Collateral Property is otherwise bound, other than (a) obligations incurred in the ordinary course of the operation of the Collateral Properties as permitted by this Agreement and (b) obligations under the Loan Documents.

6.06. <u>Solvency</u>. No Borrower has executed the Notes, this Agreement or any other Loan Document with the actual intent to hinder, delay or defraud any creditor. Each Borrower has received reasonably equivalent value in exchange for its obligations under the Loan Documents. Giving effect to the Loans on the Closing Date, the fair saleable value of each Borrower's assets exceeds and will, immediately following the making of the Loans on the Closing Date, exceed such Borrower's total liabilities, including, without limitation, subordinated, unliquidated, disputed and contingent liabilities. Each Borrower's assets do not and, immediately following the making of the Loans on the Closing Date will not, constitute unreasonably small capital to carry out its business as conducted or as proposed to be conducted. No Borrower intends to incur debt and liabilities (including contingent liabilities and other commitments) beyond its ability to pay such debt and liabilities as they mature (taking into account the timing and amounts of cash to be received by such Borrower and the amounts to be payable on or in respect of obligations of such Borrower). No petition under any Debtor Relief Laws has been filed against any Borrower or any constituent Person in the last seven (7) years, and neither such Borrower nor any constituent Person in the last seven (7) years has ever made an assignment for the benefit of creditors or taken advantage of any insolvency act for the benefit of debtors. Neither any Borrower nor any of its constituent Persons are contemplating either the filing of a petition by it under any Debtor Relief Laws or the liquidation of all or a major portion of such Borrower's assets or property, and no Borrower has knowledge of any Person contemplating the filing of any such petition against it or such constituent Persons.

6.07. <u>Full and Accurate Disclosure</u>. No statement of fact made by any Borrower in this Agreement or in any of the other Loan Documents contains any untrue statement of a material fact or omits to state any material fact necessary to make statements contained herein or therein not misleading in any material respect. There is no fact presently known to any Borrower which has not been disclosed to Administrative Agent which would have or could reasonably be expected to cause a Material Adverse Effect or, solely as of the Closing Date and the Extension Effective Date, a Material Property Event.

6.08. <u>No Plan Assets</u>. No Borrower is a Plan and none of the assets of any Borrower constitute or will constitute "Plan Assets" of one or more Plans. In addition, (a) no Borrower is a "governmental plan" within the meaning of Section 3(32) of ERISA and (b) transactions by or with any Borrower are not subject to state statutes regulating investment of, and fiduciary obligations with respect to, governmental plans similar to the provisions of Section 406 of ERISA or Section 4975 of the Code currently in effect, which prohibit or otherwise restrict the transactions contemplated by this Agreement.

6.09. <u>Compliance with Legal Requirements</u>. Except as disclosed in any Environmental Report, Physical Condition Report or Survey, Borrowers and the Collateral Properties and the ownership, operation and use thereof comply in all material respects with all applicable Legal Requirements, including, without limitation, all Environmental Laws, building and zoning ordinances and codes. No Borrower is in default or violation in any material respect of any order, writ, injunction, decree or demand of any Governmental Authority. There has not been committed by any Borrower or any other Person in occupancy of or involved with the operation or use of the Collateral Properties any act or omission affording the Federal government or any other Governmental Authority the right of forfeiture as against any Collateral Property or any part thereof or any monies paid in performance of any Borrower's obligations under any of the Loan Documents.

6.10. <u>Financial Information</u>. All financial data, including, without limitation, the operating statements of revenue and expense, that have been delivered to Administrative Agent or any Lender by or on behalf of Borrowers and the Collateral Properties (a) considered in the aggregate, are true, complete and correct in all material respects, (b) fairly present the financial condition of Borrowers and the Collateral Properties, as of the date of such reports, and (c) have been prepared in accordance with GAAP throughout the periods covered, except as disclosed therein (but subject to normal year-end adjustments). Except for Permitted Liens, no Borrower has any contingent liabilities, liabilities for taxes, unusual forward or long-term commitments or unrealized or anticipated losses from any unfavorable commitments that are known to such Borrower and reasonably likely to have a materially adverse effect on any Collateral Property or the operation

thereof as hotels except as referred to or reflected in said financial statements. Since the date of the most recent such financial statements supplied to Administrative Agent, there has been no materially adverse change in the financial condition, operations or business of Borrowers from that set forth in said financial statements.

6.11. <u>Condemnation</u>. Except as provided on Schedule 6.11, no Condemnation or other similar proceeding has been commenced or, to the best of Borrowers' knowledge, is threatened or contemplated with respect to all or any portion of any Collateral Property or for the relocation of roadways providing access to any Collateral Property.

6.12. <u>Federal Reserve Regulations; Foreign Asset Control Regulations</u>. No part of the proceeds of the Loans will be used for the purpose of purchasing or carry any "margin stock" within the meaning of Regulation U of the Board of Governors of the Federal Reserve System or for any other purpose which would be inconsistent with such Regulation U or any other Regulations of such Board of Governors, or for any purposes prohibited by Legal Requirements or by the terms and conditions of this Agreement or the other Loan Documents. No Borrower owns any "margin stock". Neither making of the Loans to Borrowers, nor Borrowers' use of the proceeds of the Loans, will violate the Trading with the Enemy Act, as amended, or any of the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B, Chapter V, as amended) or any enabling legislation or executive order relating thereto, including without limitation Executive Order 13224 66 Fed. Reg. 49079 (September 25, 2001) (Blocking Property and Prohibiting Transactions with Persons who Commit, Threaten to Commit or Support Terrorism), or the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)).

6.13. <u>Utilities and Public Access</u>. Each Collateral Property has rights of access to public ways and is served by public water, sewer, sanitary sewer and storm drain facilities adequate to service such Collateral Property for its respective intended uses. All public utilities necessary or convenient to the full use and enjoyment of each Collateral Property are located either in the public right-of-way abutting each Collateral Property (which are connected so as to serve each Collateral Property without passing over other property) or in recorded easements serving each Collateral Property and such easements are set forth in and insured by the Title Policy. All roads necessary for the use of each Collateral Property for their current respective purposes have been completed, are physically open and except as disclosed on the Surveys, are dedicated to public use and have been accepted by all Governmental Authorities.

6.14. <u>Foreign Person</u>. No Borrower is a "foreign person" within the meaning of § 1445 (f)(3) of the Code.

6.15. <u>Fee and Leasehold Ownership</u>. Hospitality Owner is the sole legal and equitable fee owner of good and marketable, record title to the Charlotte Property, the Mandalay Beach Property, the Myrtle Beach Property (except for the Myrtle Beach Golf Course), the Nashville Property, the Santa Barbara Property and the Santa Monica Property, in each case subject only to Permitted Liens. Boca Owner is the sole legal and equitable fee owner of good and marketable to title to the Dana Point Property, subject only to Permitted Liens. The Philadelphia Affiliate Ground Lessor is the sole legal and equitable fee owner of the Philadelphia Property, subject only to Permitted

Liens. Boca Raton Affiliate Ground Lessor is the sole legal and equitable fee owner of the Boca Raton Property, subject only to Permitted Lien. Each Owner is the owner of good and marketable, record title to the leasehold estate under its applicable Ground Lease, subject only to Permitted Liens; and each Operating Lessee is the legal and equitable owner of good and marketable, record title to the leasehold estate under its applicable Operating Lease subject only to Permitted Liens.

6.16. <u>Separate Tax Lots; Assessments</u>. Each Collateral Property is comprised of one (1) or more parcels which constitute a separate tax lot or lots and does not constitute a portion of any other tax lot not a part of such Collateral Property. To the best of each Borrower's knowledge, except for Permitted Liens, there are no pending or proposed special or other assessments for public improvements or otherwise affecting any Collateral Property, nor are there any contemplated improvements to any Collateral Property that may result in such special or other assessments, except for Permitted Liens.

6.17. <u>Enforceability</u>. The Loan Documents are not subject to any right of rescission, setoff, counterclaim or defense by any Borrower, including the defense of usury, and no Borrower has asserted any right of rescission, set-off, counterclaim or defense with respect thereto.

6.18. <u>No Prior Assignment</u>. There are no prior assignments of the Leases or any portion of the Rents due and payable or to become due and payable which are presently outstanding.

6.19. <u>Insurance</u>. Borrowers have obtained and have delivered to Administrative Agent certified copies of all insurance policies reflecting the insurance coverages, amounts and other requirements set forth in this Agreement. To the best of each Borrower's knowledge, no Person, including any Borrower, has done, by act or omission, anything which would impair the coverage of any such policy.

6.20. <u>Use of Property</u>. Each Collateral Property is used exclusively for hotel purposes and other appurtenant and related uses including but not limited to restaurants and lounges.

6.21. <u>Certificate of Occupancy; Licenses</u>. All material certifications, permits, licenses and approvals, including without limitation, certificates of completion and occupancy permits required for the legal use, occupancy and operation of each Collateral Property by any Borrower as a hotel for its current uses and amenities, including all liquor, hotel and common victualler's Licenses (collectively, the "Licenses"), have been obtained and are in full force and effect and are not subject to revocation, suspension or forfeiture. Schedule 6.21 lists all Licenses maintained for each Collateral Property and the holder of each such License. Borrowers shall keep and maintain (or shall cause to be kept and maintained) all Licenses necessary for the operation of each Collateral Property is in conformity with the certificate of occupancy issued for such Collateral Property.

6.22. <u>Flood Zone</u>. Except as disclosed on the Surveys, none of the Improvements on any Collateral Property are located in an area as identified by the Federal Emergency Management Agency as an area having special flood hazards and, if so located, the flood insurance required pursuant to Section 9.01(a)(vii) is in full force and effect with respect to each such Collateral Property.

6.23. <u>Physical Condition</u>. Except to the extent that any Borrower's noncompliance with this Section 6.23 would not reasonably be expected to have a Material Adverse Effect or, solely as of the Closing Date and the Extension Effective Date, a Material Property Event, and except as was disclosed in any Environmental Report, Physical Condition Report or Survey: (a) each Collateral Property, including, without limitation, all buildings, improvements, parking facilities, sidewalks, storm drainage systems, roofs, plumbing systems, HVAC systems, fire protection systems, electrical systems, equipment, elevators, exterior sidings and doors, landscaping, irrigation systems and all structural components, are in good condition, order and repair in all material respects; there exists no structural or other material defects or damages in any Collateral Property, whether latent or otherwise, and no Borrower has received notice from any insurance company or bonding company of any defects or inadequacies in any Collateral Property, or any part thereof, which would adversely affect the insurability of the same or cause the imposition of extraordinary premiums or charges thereon or of any termination or threatened termination of any policy of insurance or bond; (b) each Collateral Property is free from damage caused by fire or other casualty; and (c) all liquid and solid waste disposal, septic and sewer systems located on each Collateral Property are, in all material respects, in a good and safe condition and repair and in compliance with all Legal Requirements.

6.24. <u>Boundaries</u>. All of the Improvements which were included in determining the Appraised Value as set forth in the Acceptable Appraisal of each Collateral Property lies wholly within the boundaries and, except as disclosed on the Surveys, building restriction lines of such Collateral Property, and no improvements on adjoining properties encroach upon such Collateral Property, and other than Permitted Liens, no easements or other encumbrances upon the applicable Collateral Property encroach upon any of the Improvements.

6.25. Leases. The Operating Lessees are the lessors under the Leases as indicated in Schedule 6.25. No Person (other than hotel guests) other than with respect to Permitted Liens has any possessory interest in any Collateral Property or right to occupy the same except under and pursuant to the provisions of the Leases and Management Agreements. The current Leases are in full force and effect and, there are no defaults by any Borrower or, to the best of each Borrower's knowledge, any tenant under any Lease, and there are no conditions that, with the passage of time or the giving of notice, or both, would constitute defaults under any Lease. With respect to the Major Leases, no Rent has been paid more than one (1) month in advance of its due date. There are no offsets or defenses to the payment of any portion of the Rents. All work to be performed by any Borrower under each Lease has been performed as required and has been accepted by the applicable tenant, and any payments, free rent, partial rent, rebate of rent or other payments, credits, allowances or abatements required to be given by any Borrower to any tenant has already been received by such tenant. There has been no prior sale, transfer or assignment, hypothecation or pledge of any Lease or of the Rents received therein which is still in effect. No tenant under any Lease has sublet all or any portion of the premises demised thereby, nor does anyone except such tenant and its employees occupy such leased premises other than with respect to Permitted Liens. No tenant under any Lease has a right or option pursuant to such Lease or otherwise to purchase all or any part of Collateral Property. Except as set forth in the Leases, no tenant under any Lease has any right or option for additional space in any Collateral Property.

6.26. <u>Survey</u>. The Survey for each Collateral Property delivered to Administrative Agent in connection with this Agreement does not fail to reflect any material matter affecting such Collateral Property or any Material Title Defect.

6.27. <u>Filing and Recording Taxes</u>. All transfer taxes, deed stamps, intangible taxes or other amounts in the nature of transfer taxes required to be paid by any Person under applicable Legal Requirements currently in effect in connection with the transfer of the Collateral Properties to the respective Borrower have been paid. All mortgage, mortgage recording, stamp, intangible or other similar tax required to be paid by any Person under applicable Legal Requirements currently in effect in connection, delivery, recordation, filing, registration, perfection or enforcement of any of the Loan Documents, including, without limitation, the Mortgages, have been paid.

6.28. Franchise Agreements; Property Improvement Plans.

(a) The Franchise Agreement, for each Franchised Property, is in full force and effect, all franchise fees, reservation fees, royalties and other sums due and payable thereunder have been paid in full to date, and neither any Borrower nor, to the best of any Borrower's knowledge, the respective Franchisor is in default thereunder.

(b) There exists no property improvement plan with respect to any Collateral Property pursuant to which improvements or repairs to such Collateral Property required by the applicable Franchisor remains incomplete or unsatisfied in accordance with such plan or other requirements of such Franchisor.

6.29. <u>Management Agreements</u>. The Management Agreement for each Collateral Property is in full force and effect and there is no default thereunder by any Borrower or, to any Borrower's knowledge, the Manager thereunder and, to the best of each Borrower's knowledge, no event has occurred that, with the passage of time and/or the giving of notice would constitute a default thereunder. No Management Fees under any Management Agreement are accrued and unpaid except as provided or permitted under the express terms of a Management Agreement.

6.30. <u>Illegal Activity</u>. No portion of any Collateral Property has been or will be purchased by Borrowers with proceeds of any illegal activity and to the best of Borrowers' knowledge, there are no illegal activities or activities relating to any controlled substances at any Collateral Property.

6.31. <u>No Change in Facts or Circumstances; Disclosure</u>. All information submitted by Borrowers, or on behalf of Borrowers, to Administrative Agent or any Lender and in all financial statements, rent rolls, reports, certificates and other documents submitted in connection with this Agreement or in satisfaction of the terms thereof and all statements of fact made by any Borrower in this Agreement or in any other Loan Document, considered in the aggregate, are accurate, complete and correct in all material respects. There has been no material adverse change in any condition, fact, circumstance or event that would make any such information inaccurate, incomplete or otherwise misleading in any material respect or that otherwise materially and adversely affects or would have a Material Adverse Effect. 6.32. <u>Investment Company Act</u>. No Borrower is (a) an "investment company" or a company "controlled" by an "investment company," within the meaning of the Investment Company Act of 1940, as amended; (b) a "holding company" or a "subsidiary company" of a "holding company" or an "affiliate" of either a "holding company" or a "subsidiary company" within the meaning of the Public Utility Holding Company Act of 1935, as amended; or (c) subject to any other federal or state law or regulation which purports to restrict or regulate its ability to borrow money.

6.33. <u>Principal Place of Business; State of Organization; Tax Identification Number</u>. (a) each Owner's principal place of business, state of organization, organizational identification number and tax identification number as of the date hereof is set forth in Schedule 6.33 to this Agreement; (b) each Operating Lessee's principal place of business, state of organization, organizational identification number and tax identification number is set forth in Schedule 6.33.

6.34. <u>Single Purpose Entity</u>. Each Borrower and each Principal of each Borrower (other than FelCor Op and FelCor TRS, each of which represents, covenants and agrees solely with respect to the Boca Owner and Boca Operating Lessee, and not with respect to itself or each other) represents, covenants and agrees that it has not, and shall not, and that its Organizational Documents provide that it has not, and shall not:

(a) in the case of each Borrower, engage in any business or activity other than the acquisition, development, ownership, operation, leasing, managing and maintenance of its respective Collateral Properties, and entering into the Loan, and activities incidental thereto or conduct and operate its business in a different manner from how it is presently conducted and operated; and with respect to each Principal of each Borrower (other than FelCor Op and FelCor TRS), engage in any business or activity other than the ownership of its interest in such Borrower, and activities incidental thereto;

(b) with respect to such Borrower, acquire or own any material assets or property other than (i) (A) in the case of each Owner, its respective Collateral Properties and (B) in the case of each Operating Lessee, its interest in its respective Operating Leases, and (ii) such incidental Personal Property as may be necessary for the operation of its Collateral Property or Collateral Properties, as the case may be, and with respect to each Principal of each Borrower (other than FelCor Op and FelCor TRS), acquire or own any material asset other than its interest in such Borrower;

(c) merge into or consolidate with any Person or dissolve, terminate or liquidate in whole or in part, transfer or otherwise dispose of all or substantially all of its assets or change its legal structure;

(d) (i) fail to observe its organizational formalities or preserve its existence as an entity duly formed, validly existing and in good standing (if applicable) under the laws of the jurisdiction of its formation, and qualification to do business in the state where its Collateral Property or Collateral Properties are located, if applicable, or (ii) amend, modify, terminate or fail to comply with the special purpose entity/bankruptcy remoteness provisions of such Borrower's or such Principal's Organizational Documents (other than FelCor Op or FelCor TRS); (e) own any Subsidiary or make any Investment in any Person except (i) in the case of Borrowers, in compliance with the terms of <u>Sections 8.02</u> and <u>8.10</u> of this Agreement, and (ii) in the case of the Principals (other than FelCor Op or FelCor TRS) of Borrowers, such Principals' ownership of Borrowers (and in the case of Hospitality Owner's Principal, the Philadelphia Ground Lessor), in each case, without the prior written consent of Administrative Agent and the Required Lenders;

(f) commingle its funds or its assets with the assets of any of its members, general partners, Affiliates or of any other Person, participate in a cash management system (other than pursuant to a Management Agreement or in accordance with this Agreement) with any other Person or fail to use its own separate stationery, invoices and checks;

(g) incur any Indebtedness, other than (i) Permitted Liens and Permitted Personal Property Liens; (ii) the Obligations, and (iii) except for trade payables in the ordinary course of its business of owning and operating the Collateral Property or Properties as applicable, provided that such trade debt (A) is not evidenced by a note, (B) is paid within sixty (60) days of the date incurred, (C) does not exceed, in the aggregate, four percent (4%) of the outstanding principal balance of the Notes and (D) is payable to trade creditors and in amounts as are normal and reasonable under the circumstances;

(h) (i) fail to maintain its records (including financial statements), books of account and bank accounts (including payroll accounts) separate and apart from those of the members, general partners, Principal or Affiliates of a Borrower, as the case may be, the Affiliates of a member, general partners, Principal or Affiliate of such Borrower, as the case may be, and any other Person, (ii) permit its assets or liabilities to be listed as assets or liabilities on the financial statement of any other Person or (iii) include the assets or liabilities of any other Person on its financial statements; except for consolidated financial statements which contain a note indicating that such Borrower's or Principal's separate assets and liabilities are neither available to pay the debts of the consolidated entity nor constitute obligations of the consolidated entity;

(i) become insolvent and fail to pay its debts and liabilities (including, as applicable, shared personnel and overhead expenses) from its assets as the same shall become due;

(j) seek the dissolution or winding up in whole, or in part, of either such Borrower or Principal;

(k) fail to correct any known misunderstandings regarding the separate identity of any Borrower or any Principal of a Borrower, as the case may be, or any other Borrower, Principal of a Borrower, member, general partner or Affiliate thereof or any other Person; (1) guarantee or become obligated for the debts of any other Person or hold itself out to be responsible for the debts of another Person, other than with respect to the Obligations or any guarantee of any Management Agreement, Replacement Management Agreement, Franchise Agreement or Replacement Franchise Agreement, as set forth therein as the case may be;

(m) make any loans or advances from Collateral to any third party, including any Principal of a Borrower, member, general partner or Affiliate of Borrowers or any member, general partner or Affiliate thereof (other than Restricted Payments permitted by the Loan Documents), and shall not acquire with Collateral any obligations or securities of any Principal, member, general partner or Affiliate of a Borrower, or any member, general partner, or Affiliate thereof; or enter into, or be a party to, any transaction with any Affiliate of Borrowers or its Principal, except for (i) the Operating Leases, the Affiliate Ground Leases, and the Overhead Sharing Agreement, and (ii) transactions which are in the ordinary course of business, with terms no less favorable to any Borrower, its Principal or such Affiliate than would be obtained in a comparable arm's-length transaction with an unrelated third party, in accordance with Section 8.18;

(n) fail to file its own tax returns or be included on the tax returns of any other Person except as required by applicable Law;

(o) fail either to hold itself out to the public as a legal entity separate and distinct from any other Person or to conduct its business solely in its own name or a name franchised or licensed to it by an entity other than an Affiliate of a Borrower, and not as a division or part of any other entity in order not (i) to mislead others as to the identity with which such other party is transacting business, or (ii) to suggest that such Borrower or Principal of a Borrower, is responsible for the debts of any third party (including any Principal, member, general partner, or Affiliate of a Borrower, or any member, general partner or Affiliate thereof);

(p) fail to maintain adequate capital for the normal obligations reasonably foreseeable in a business of its size and character and in light of its contemplated business operations;

(q) hold itself out as or be considered as a department or division of (i) any Principal, general partner, member or Affiliate of a Borrower, (ii) any Affiliate of a Principal, general partner or member of a Borrower, or (iii) any other Person;

(r) fail to allocate fairly and reasonably any overhead expenses that are shared with a Loan Party, including paying for office space and services performed by any employee of an Affiliate of such Loan Party;

(s) pledge its assets that constitute Collateral for the benefit of any other Person other than with respect to the Obligations;

(t) fail to maintain a sufficient number of employees in light of its contemplated business operations;

(u) fail to provide in its Organizational Documents that for so long as the Loans or any other Obligations are outstanding, it shall not file or consent to the filing of any petition, either voluntary or involuntary, to take advantage of any applicable Debtor Relief Law or make an assignment for the benefit of creditors without the affirmative vote of the Independent Director and of all other general partners/managing members/directors;

(v) fail to hold its assets in its own name or maintain its assets in a manner that is costly or difficult to segregate, ascertain, or identify from the assets of any Affiliate, Principal, member, general partner or guarantor of a Borrower, or any Affiliate of such Principal, member, general partner or guarantor, or from any other Person;

(w) have any of its obligations guaranteed by an Affiliate, other than with respect to the Obligations hereunder;

fail at any time (A) in the case of Borrowers, to have at least two (2) (x) independent director/managers (each, an "Independent Director"), or (B) in the case of any Principal of a Borrower (except FelCor Op and FelCor TRS), to have at least one (1) Independent Director, that in either case is not and has not been for at least five (5) years: (i) a direct or indirect legal or beneficial owner of any limited liability company membership interest, stock, partnership interest or other equity interest in a Borrower or Principal or any other Affiliate of either of Borrowers or Principal (other than potential de minimus and indirect interests FelCor Trust incidentally held through mutual fund investments) (an "Equity Owner"), a director, manager or trustee (in each case, other than an independent director, independent manager or independent trustee of FelCor Trust or any other Affiliate of either of Borrowers or Principals), or any officer, employee, partner, member, beneficiary, contractor, accountant, attorney or counsel of a Borrower or Principal or any Affiliate of either of them; (ii) a creditor, customer, supplier or other Person who derives its purchases or revenues (other than any fee paid to such director as compensation for such director to serve as an Independent Director) from its activities with such Borrower, Principal, FelCor Trust or any other Affiliate of either of Borrowers or Principals (a "Business Party"); (iii) a Person or other entity controlling or under common control with any such Equity Owner, partner, member, director, manager, trustee, officer, beneficiary, contractor, employee, accountant, attorney, counsel or Business Party; or (iv) a member of the immediate family of any such Equity Owner, partner, member, director, manager, trustee, officer, beneficiary, contractor, employee, accountant, attorney, counsel or Business Party; provided that if any individual shall cease to be an Independent Director at any time (for any reason), the affected Borrowers and Principals shall immediately, and in any event, within three (3) Business Days after such occurrence, give to Administrative Agent and each Lender notice of such departure and the identity of the replacement Independent Director appointed in accordance with the terms hereof; provided further, that each LLC Agreement shall provide that no removal of any Independent Director shall be effective until such time as a replacement Independent Director meeting all applicable requirements of this Section 6.34 has accepted his or her appointment as an Independent Director of the applicable Borrower or Principal; and provided further that each Borrower's or Principal's (other than FelCor Op's

or FelCor TRS's) LLC Agreement (as defined below) shall provide that each Independent Director and Special Member (as defined below) shall at all times be an employee of National Registered Agents, Inc., Global Securitization Services, LLC, Entity Services Group, LLC, Puglisi & Associates, Stewart Management Co., Lord Securities Corporation, Nevada Holding Services, Inc., CT Corporation, Corporation Service Company, InCorp Services, Inc., or a similar nationally-recognized service company that engages professionally in the business of providing Independent Directors to special purpose entities;

(y) become involved in the day-to-day management of any other Person;

(z) make any investments indirectly or by brokers not engaged and paid by any Borrower or Principal of Borrowers or an agent thereof (provided that if any such agent is an Affiliate of any Borrower or such Principal, it shall be compensated at a fair market rate for its services); or

(aa) permit its board of directors/managers to take any action which, under the terms of any applicable Organizational Documents, requires the unanimous vote of one hundred percent (100%) of the members of the board, without the vote of all of its Independent Directors; including, with respect to itself or to any other Person in which it has a direct or indirect legal or beneficial interest (x) seeking or consenting to the appointment of a receiver, liquidator, assignee, trustee, sequestrator, custodian or other similar official for the benefit of the creditors of such Person or all or any portion of such Person's properties, or (y) taking any action that might cause such Person to become insolvent, petition or otherwise institute insolvency proceedings or otherwise seeking any relief under any laws relating to the relief from debts or the protection of debtors generally, or (z) violating the requirements set forth in this Section 6.34 or in such Person's Organizational Documents, with respect to its operations as a special purpose vehicle.

The limited liability company agreements of each Borrower and each Principal (other than FelCor Op and FelCor TRS) of each Borrower (collectively, the "LLC Agreement") shall provide that (i) to the fullest extent permitted by law, each Independent Director shall consider the interest of the limited liability company, including its respective creditors, in acting or otherwise voting on the single purpose entity matters specified in this Section 6.34 or therein; (ii) that Administrative Agent, together with its successors and assigns as administrative agent hereunder, is an intended beneficiary of the single purpose entity requirements set forth therein and may enforce such provisions, (ii) upon the occurrence of any event that causes the last remaining member of any Borrower or Principal ("Member") to cease to be the member of such Borrower or Principal (other than (1) upon an assignment by Member of all of its limited liability company interest in such Borrower or Principal and the admission of the transferee in accordance with the Loan Documents and the LLC Agreement, or (2) the resignation of Member and the admission of an additional member of such Borrower or Principal in accordance with the terms of the Loan Documents and the LLC Agreement), any Person designated by such Borrower or Principal shall, without any action of any other Person and simultaneously with the Member ceasing to be the member of such Borrower or Principal, automatically be admitted to such Borrower ("Special Member") and shall continue such Borrower or Principal without dissolution and (iv) Special Member may not resign from such Borrower or Principal or transfer its rights as Special Member unless a successor Special Member has been admitted to such Borrower or Principal as Special Member in accordance with requirements

of Delaware law. The LLC Agreement further provides that (v) Special Member shall automatically cease to be a member of such Borrower or Principal upon the admission to such Borrower or Principal of a substitute Member, (w) Special Member shall be a member of such Borrower or Principal that has no interest in the profits, losses and capital of such Borrower or Principal and has no right to receive any distributions of such Borrower's or Principal's assets, (x) pursuant to Section 18-30-1 of the Delaware Limited Liability Company Act (the "Act"), Special Member shall not be required to make any capital contributions to such Borrower or Principal and shall not receive a limited liability company interest in such Borrower, (y) Special Member, in its capacity as Special Member, may not bind such Borrower or Principal and (z) except as required by any mandatory provision of the Act, Special Member, in its capacity as Special Member, shall have no right to vote on, approve or otherwise consent to any action by, or matter relating to, such Borrower or Principal, including, without limitation, the merger, consolidation or conversion of such Borrower or Principal. In order to implement the admission to such Borrower or Principal of Special Member, Special Member shall execute a counterpart to the LLC Agreement. The LLC Agreement provides that prior to its admission to such Borrower or Principal as Special Member, Special Member shall not be a member of such Borrower or Principal. The LLC Agreement further provides the following: (i) upon the occurrence of any event that causes the Member to cease to be a member of such Borrower or Principal, to the fullest extent permitted by law, the personal representative of Member shall, within ninety (90) days after the occurrence of the event that terminated the continued membership of Member in such Borrower or Principal, agree in writing (A) to continue such Borrower or Principal and (B) to the admission of the personal representative or its nominee or designee, as the case may be, as a substitute member of such Borrower or Principal, effective as of the occurrence of the event that terminated the continued membership of Member of such Borrower in such Borrower or Member of such Principal in such Principal, as applicable; (ii) any action initiated by or brought against Member or Special Member under any Debtor Relief Laws shall not cause Member or Special Member to cease to be a member of such Borrower or Principal and upon the occurrence of such an event, the business of such Borrower or Principal shall continue without dissolution; and (iii) that each of Member and Special Member waives any right it might have to agree in writing to dissolve such Borrower or Principal upon the occurrence of any action initiated by or brought against Member or Special Member under any Debtor Relief Laws, or the occurrence of an event that causes Member or Special Member to cease to be a member of such Borrower or Principal.

6.35. <u>Business Purposes</u>. The Loan and the proceeds thereof are to be used solely for the business purposes of Borrowers, and is not for personal, family, household, or agricultural purposes.

6.36. <u>Taxes</u>. Each Borrower has filed all federal, state, county, municipal, and city income and other tax returns required to have been filed by it and has paid, prior to delinquency thereof, all taxes and related liabilities which have become due pursuant to such returns or pursuant to any assessments received by it. No Borrower knows of any basis for any additional assessment in respect of any such taxes and related liabilities for prior years.

Environmental Representations and Warranties. Each Borrower represents and 6.37. warrants, except as specifically disclosed in those certain Environmental Reports identified on Schedule 6.37 with respect to each Collateral Property that: (a) there are no Hazardous Materials or underground storage tanks in, on, or under any of the Collateral Properties, except those that are both (i) in compliance with current Environmental Laws and with permits issued pursuant thereto (if such permits are required), and (ii) either (A) in amounts not in excess of that necessary to operate, clean, repair and maintain the applicable Collateral Property or each tenant's respective business at such Collateral Property as set forth in their respective Leases, or (B) held by a tenant for sale to the public in its ordinary course of business, (b) there are no past, present or threatened Releases of Hazardous Materials in violation of any Environmental Law and which would require remediation by a Governmental Authority in, on, under or from any of the Collateral Properties; (c) there is no threat of any Release of Hazardous Materials migrating to any of the Collateral Properties; (d) there is no present or, to any Borrower's knowledge, prior non-compliance with Environmental Laws, or with permits issued pursuant thereto, in connection with any of the Collateral Properties; (e) each Borrower does not know of, and has not received, any written or oral notice or other communication from any Person (including but not limited to a Governmental Authority) relating to Hazardous Materials located or Released in, on, under or from any of the Collateral Properties or relating to any Environmental Liability; and (f) each Borrower has truthfully and fully provided to Administrative Agent, in writing, any and all information relating to environmental conditions in, on, under or from any of the Collateral Properties known to such Borrower or contained in such Borrower's files and records, including but not limited to any reports relating to Hazardous Materials in, on, under or migrating to or from any of the Collateral Properties and/or to the environmental condition of the Collateral Properties.

6.38. <u>Ground Lease Representations</u>. Except as set forth on <u>Schedule 6.38</u>:

(a) Each Property that is subject to a Ground Lease, and the applicable Ground Lessor thereof and the title and dates with respect to any agreements, amendments or assignments comprising such Ground Lease, are set forth on Schedule 6.15, and (i) Borrower has delivered a true and correct copy of each Ground Lease to Administrative Agent prior to or simultaneously with its execution of the applicable Mortgage, (ii) each Ground Lease is in full force and effect and has not been modified or amended in any manner whatsoever, except as identified on Schedule 6.15, (iii) each Ground Lessee enjoys the quiet and peaceful possession of the property demised thereby, (iv) there are no defaults under any Ground Lease by any party thereunder, and no event has occurred which but for the passage of time, or notice, or both would constitute a default under such Ground Lease, (v) all rents, additional rents and other sums due and payable under each Ground Lease have been paid in full, and (vi) no Ground Lessee or any Affiliate nor the applicable Ground Lessor under each Ground Lease has commenced any action or given or received any notice for the purpose of terminating such Ground Lease, (vii) the consummation of the transactions contemplated hereby will not result in any breach of, or constitute a default under, any Ground Lease, (viii) all actions which must be taken for Administrative Agent to have the rights of a leasehold mortgagee or mortgagee pursuant to each of the Ground Leases have been taken and completed, and (ix) no Ground Lessee or Affiliate has granted any other leasehold mortgage or made any other assignment, pledge or hypothecation of its interest under any of the Ground Leases;

(b) The Mortgage encumbering (i) Hospitality Owner's interest in the Philadelphia Affiliate Ground Lease also encumbers the Philadelphia Affiliate Ground Lessor's fee interest in the Philadelphia Property and (ii) the Boca Owner's interest in the Boca Raton Affiliate Ground Lease also encumbers the Boca Raton Affiliate Ground Lessor's fee interest in the Boca Raton Property, and in each case, the fee interest is subject and subordinate of record to the applicable Mortgage, and such Mortgage does not by its terms provide that it will be subordinated to the Lien of any other mortgage or other Lien upon such fee interest, and upon the occurrence of an Event of Default, Administrative Agent has the right to foreclose or otherwise exercise its rights with respect to the fee interest within a commercially reasonable time;

(c) The Ground Leases or a memorandum thereof have been duly recorded, the Ground Leases permit the interest of the lessee thereunder to be encumbered by the applicable Mortgage. Except as identified on <u>Schedule 6.15</u>, there has not been any change in the terms of the Ground Leases since their recordation. Each of the Ground Leases, either by its express terms or after giving effect to any estoppel and consent agreement described in <u>Section 5.01(a)(xii)(C)</u>, cannot be cancelled, terminated, surrendered or amended without the prior written consent of Administrative Agent;

(d) The Ground Leases are not subject to any Liens (other than Permitted Liens) superior to, or of equal priority with, the applicable Mortgage;

(e) The applicable Ground Lessee's interest in each of the Ground Leases, either by the express terms of the Ground Leases or after giving effect to any estoppel and consent agreement described in Section 5.01(a)(xii)(C), are assignable upon notice to, but without the consent of, the lessor thereunder and, in the event that it is so assigned, it is further assignable upon notice to, but without the need to obtain the consent of, such lessor;

(f) The Ground Leases, either by their express terms or after giving effect to any estoppel and consent agreement described in Section 5.01(a)(xii)(C), require the lessor thereunder to give notice of any default by the applicable Ground Lessee to Administrative Agent; and the Ground Leases, either by their express terms or after giving effect to any estoppel and consent agreement described in Section 5.01(a)(xii)(C), further provide that notice of termination given under the Ground Leases are not effective against Administrative Agent or any Lender unless a copy of the notice has been delivered to Administrative Agent or Lenders in the manner described in the applicable Ground Lease;

(g) Either under the express terms of the Ground Leases or after giving effect to any estoppel and consent agreement described in Section 5.01(a)(xii)(C), Administrative Agent is permitted a reasonable opportunity (including, where necessary, sufficient time to gain possession of the interest of the applicable Ground Leasee under the Ground Leases) to cure any default under the Ground Leases, which is curable after the receipt of notice of any default before the lessor thereunder may terminate such Ground Lease;

(h) The Ground Leases have a term which extends not less than twenty-five (25) years beyond the Maturity Date;

(i) After giving effect to any estoppel and consent agreement described in Section 5.01(a)(xii)(C), the Ground Leases, require the lessor to enter into a new lease on similar terms and conditions upon termination of the applicable Ground Lease for any reason, including rejection of such Ground Lease in a proceeding under any Debtor Relief Law;

(j) Under the terms of each Ground Lease and the applicable Loan Documents, taken together, any Net Proceeds will be applied either to the Restoration of all or part of the Collateral Properties, with Administrative Agent or a trustee appointed by Administrative Agent having the right to hold and disburse such Net Proceeds as the Restoration progresses, or, after giving effect to any estoppel and consent agreement described in Section 5.01(a) (xii)(C), to the payment of the outstanding principal balance of the Loan together with any accrued interest thereon; and

(k) Either under the express terms of the Ground Leases or after giving effect to any estoppel and consent agreement described in $\underline{\text{Section 5.01(a)(xii)(C)}}$, the Ground Leases do not impose restrictions on subletting.

6.39. Operating Lease Representations.

(a) (i) Each Operating Lease is in full force and effect and has not been modified or amended in any manner whatsoever, (ii) except as set forth on <u>Schedule 6.39</u>, there are no defaults under any Operating Lease by any party thereunder, and no event has occurred which but for the passage of time, or notice, or both would constitute a default under such Operating Lease, (iii) except as set forth on <u>Schedule 6.39</u>, all rents, additional rents and other sums due and payable under each Operating Lease have been paid in full, and (iv) no Operating Lessee nor any Owner (Lessor) under any Operating Lease has commenced any action or given or received any notice for the purpose of terminating such Operating Lease;

(b) Each Mortgage which encumbers an Operating Lessee's interest in an Operating Lease also encumbers the respective Owner's interest in the applicable Collateral Property, and the fee interest is subject and subordinate of record to the applicable Mortgage, and such Mortgage does not by its terms provide that it will be subordinated to the Lien of any other mortgage or other Lien upon such fee interest, and upon the occurrence and during the continuance of an Event of Default, Administrative Agent has the right to foreclose or otherwise exercise its rights with respect to the fee interest within a commercially reasonable time;

(c) The Operating Leases cannot be cancelled, terminated, surrendered or amended without the prior written consent of Administrative Agent, except as expressly permitted by $\underline{Section 7.16(d)}$;

(d) The applicable interests of each Operating Lessee and each Owner in the Operating Leases are not subject to any Liens (other than Permitted Liens) superior to, or of equal priority with, the applicable Mortgage;

(e) The Operating Leases require the applicable Owner to give notice of any default by the applicable Operating Lessee to Administrative Agent and the Operating Leases further provide that notice of termination given under the Operating Leases are not effective against Administrative Agent or any Lender unless a copy of the notice has been delivered to Administrative Agent or Lenders in the manner described in the applicable Operating Lease;

(f) Under the terms of each Operating Lease, Administrative Agent is permitted a reasonable opportunity (including, where necessary, sufficient time to gain possession of the interest of the applicable Operating Lessee under the Operating Leases) to cure any default under the Operating Lease, which is curable after the receipt of notice of any default before the lessor thereunder may terminate such Operating Lease;

(g) Under the terms of each Operating Lease and the applicable Loan Documents, taken together, any Net Proceeds will be applied either to the Restoration of all or part of the Collateral Properties, with Administrative Agent or a trustee appointed by Administrative Agent having the right to hold and disburse such Net Proceeds as the Restoration progresses, or to the payment of the outstanding principal balance of the Loan together with any accrued interest thereon; and

(h) The Operating Leases do not impose restrictions on subletting.

6.40. <u>Liens</u>. The Collateral Documents provide first priority Liens in the Collateral in favor of the Secured Parties, subject to no other Liens, other than Permitted Liens.

6.41. <u>Service Contracts</u>. Other than the Leases, the Operating Leases and the Ground Leases, no Borrower has entered into any Contractual Obligation with respect to property or services to be provided by third parties with respect to any Collateral Property except (a) those listed on <u>Schedule 6.41</u>, (b) after the Closing Date, those permitted under <u>Section 8.05</u>, and (c) those that are otherwise not material to Borrowers or any Collateral Property.

6.42. Personal Property Leasing and Financing.

No Personal Property used in the ownership or operation of any hotel, is subject to any financing or leasing arrangements except (a) as disclosed on <u>Schedule 6.42</u>, (b) after the Closing Date, those permitted under <u>Section 8.04</u>, and (c) those that are otherwise not material to Borrowers or any Collateral Property.

6.43. <u>Reciprocal Agreements</u>.

(a) No Loan Party is currently in default in any respect (nor has any notice been given or received with respect to an alleged or current default) under any of the terms and conditions of any REA, and each REA remains unmodified and in full force and effect except where such default, modification or failure to be in full force and effect could not reasonably be expected to cause a Material Property Event.

(b) All sums due and owing by any Loan Party to the other parties to each REA (or by the other parties to each REA to any Loan Party) pursuant to the terms of such REA, have been paid, are current, and no lien has attached on any Collateral Property (or threat thereof been made) for failure to pay any of the foregoing except where such failure could not reasonably be expected to cause a Material Property Event.

6.44. <u>Survival of Representations</u>. Borrowers agree that all of the representations and warranties set forth in this Article VI and elsewhere in this Agreement and in the other Loan Documents (as such representations and warranties are modified or qualified by the Schedules, reports and other certificates and instruments delivered to Administrative Agent pursuant to this Agreement) shall survive for so long as any amount remains owing by any Loan Party under this Agreement or any of the other Loan Documents. All representations, warranties, covenants and agreements made in this Agreement or in the other Loan Documents by any Loan Party shall be deemed to have been relied upon by Administrative Agent and Lenders notwithstanding any investigation heretofore or hereafter made by any of Administrative Agent or Lenders or on its behalf.

ARTICLE VII AFFIRMATIVE COVENANTS

From the date hereof until payment and performance in full of all Obligations under the Loan Documents or, in respect of a specific Collateral Property, until the earlier release of the Liens of all Mortgages encumbering such Collateral Property in accordance with the terms of this Agreement and the other Loan Documents, Borrowers each covenant as follows:

7.01. Existence; Compliance with Legal Requirements.

(a) There shall never be committed by any Borrower or any other Person in occupancy of or involved with the operation or use of the Collateral Properties any act or omission affording the Federal government or any state or local government the right of forfeiture against any Collateral Property or any part thereof or any monies paid in performance of such Borrower's obligations under any of the Loan Documents. Each Borrower hereby covenants and agrees not to commit, permit or suffer to exist any act or omission affording such right of forfeiture. To the extent that any Borrower's noncompliance with this Section 7.01(a) would reasonably be expected to have a Material Adverse Effect: (i) such Borrower shall do or cause to be done all things necessary to preserve, renew and keep in full force and effect its existence, rights, licenses, permits and franchises, and comply, in all material respects, with all Legal Requirements applicable to it and its Collateral Properties; (ii) such Borrower shall at all times maintain, preserve and protect all franchises and trade names and preserve all the remainder of its property used or useful in the conduct of its business and shall keep the Collateral Properties in good working order and repair, and from time to time make, or cause to be made, all reasonably necessary repairs, renewals, replacements, betterments and improvements thereto, all as more fully provided in this Agreement and the Mortgages; (iii) such Borrower shall keep the Collateral Properties insured at all times by financially sound and reputable insurers, to such extent and against such risks, and maintain liability and such other insurance, as is more fully provided in this

Agreement; and (iv) such Borrower shall operate any Collateral Property that is the subject of any O&M Program in accordance with the terms and provisions thereof in all material respects.

(b) Borrowers, at their own expense, may contest by appropriate legal proceeding promptly initiated and conducted in good faith and with due diligence, the validity of any Legal Requirement, the applicability of any Legal Requirement to Borrowers or any Collateral Property or any alleged violation of any Legal Requirement, provided that (i) no Event of Default has occurred and remains uncured; (ii) such proceeding shall be permitted under and be conducted in accordance with the provisions of any Contractual Obligation to which the applicable Borrowers are subject and shall not constitute a default thereunder and such proceeding shall be conducted in accordance with all applicable Laws; (iii) no Collateral Property nor any part thereof or interest therein will be in danger of being sold, forfeited, terminated, cancelled or lost; (iv) Borrowers shall promptly upon final determination thereof comply with any such Legal Requirement determined to be valid or applicable or cure any violation of any Legal Requirement; (v) such proceeding shall suspend the enforcement of the contested Legal Requirement against Borrowers or any Collateral Property; and (vi) Borrowers shall furnish such security as may be required in the proceeding, to insure compliance with such Legal Requirement, together with all interest and penalties payable in connection therewith. Administrative Agent may apply any such security or part thereof, as necessary to cause compliance with such Legal Requirement at any time when, in the judgment of Administrative Agent, the validity, applicability or violation of such Legal Requirement is finally established or any Collateral Property (or any part thereof or interest therein) shall be in danger of being sold, forfeited, terminated, cancelled or lost.

Taxes and Other Charges. Borrowers shall pay all Taxes and Other Charges now or 7.02. hereafter levied or assessed or imposed against the Collateral Properties or any part thereof prior to delinquency thereof. Upon Administrative Agent's request, each Borrower shall furnish to Administrative Agent receipts, or other evidence for the payment of the Taxes and the Other Charges prior to the date the same shall become delinquent. Borrowers shall not suffer and shall promptly cause to be paid and discharged any Lien or charge whatsoever which may be or become a Lien or charge against the Collateral Properties, and shall promptly pay for all utility services provided to the Collateral Properties. Borrowers, at their own expense, may contest by appropriate legal proceeding, promptly initiated and conducted in good faith and with due diligence, the amount or validity or application in whole or in part of any Taxes or Other Charges, provided that (i) no Event of Default has occurred and remains uncured or would result from such contest; (ii) such proceeding shall be permitted under and be conducted in accordance with the provisions of any Contractual Obligation to which a Borrower is subject and shall not constitute a default thereunder and such proceeding shall be conducted in accordance with all applicable Laws; (iii) no Collateral Property nor any part thereof or interest therein will be in danger of being sold, forfeited, terminated, cancelled or lost; (iv) Borrowers shall promptly upon final determination thereof pay the amount of any such Taxes or Other Charges, together with all costs, interest and penalties which may be payable in connection therewith; (v) such proceeding shall suspend the collection of such contested Taxes or Other Charges from the applicable Collateral Properties; and (vi) each Borrower shall furnish to

Administrative Agent for the benefit of the Lenders, such security as may be required in the proceeding or as may be necessary or advisable, to insure the payment of any such Taxes or Other Charges, together with all interest and penalties thereon. Administrative Agent may apply such security or part thereof held by Administrative Agent at any time when, in the judgment of Administrative Agent, the validity or applicability of such Taxes or Other Charges are established or any Collateral Property (or part thereof or interest therein) shall be in danger of being sold, forfeited, terminated, cancelled or lost or there shall be any danger of the Lien of any Mortgage being primed by any related Lien.

7.03. <u>Litigation</u>. Borrowers shall give prompt written notice to Administrative Agent and Lenders of any litigation or governmental proceedings pending or threatened in writing against any Borrower which might have a Material Adverse Effect or could reasonably be expected to result in a judgment in excess of \$10,000,000.

7.04. <u>Access to Collateral Properties</u>. Borrowers shall permit agents, representatives and employees of Administrative Agent and each Lender to inspect the Collateral Properties or any part thereof at reasonable hours upon reasonable advance notice and shall cause each Operating Lessee and each Manager to permit such access.

7.05. <u>Notice of Default</u>. Borrowers shall promptly advise Administrative Agent and the Lenders of the occurrence of any Default, Event of Default or any material adverse change in any Borrower's condition, financial or otherwise reasonably likely to cause a Material Adverse Effect or Material Property Event of which a Borrower has knowledge.

7.06. <u>Cooperation in Legal Proceedings</u>. Except with respect to any claim by Borrowers against Administrative Agent or Lenders, Borrowers shall cooperate fully with Administrative Agent with respect to any proceedings relating to Borrowers, the Collateral Properties or the Loans before any court, board or other Governmental Authority which may in any way adversely affect the rights of Administrative Agent or any Lender hereunder or any rights obtained by Administrative Agent or any Lender under any of the other Loan Documents and, in connection therewith, permit Administrative Agent on behalf of Lenders, at its election, to participate in any such proceeding.

7.07. <u>Award and Insurance Benefits</u>. Borrowers shall cooperate with Administrative Agent in obtaining for Lenders the benefits of any Awards or Insurance Proceeds lawfully or equitably payable in connection with any Collateral Property, and Administrative Agent shall be reimbursed for any expenses incurred in connection therewith (including reasonable attorneys' fees and disbursements, and the payment by Borrowers of the expense of an Acceptable Appraisal on behalf of Lenders in case of Casualty or Condemnation affecting any Collateral Property or any part thereof) out of such Award or Insurance Proceeds.

7.08. <u>Further Assurances</u>. Borrowers shall, at Borrowers' sole cost and expense:

(a) furnish to Administrative Agent all instruments, documents, boundary surveys, footing or foundation surveys, certificates, plans and specifications, appraisals, title and other insurance reports and agreements, and each and every other document, certificate, agreement and instrument required to be furnished by Borrowers pursuant to the terms of the Loan Documents or reasonably requested by Administrative Agent in connection therewith;

(b) execute and deliver to Administrative Agent such documents, instruments, certificates, assignments and other writings, and do such other acts necessary or desirable, to evidence, preserve and/or protect the Collateral at any time securing or intended to secure the Obligations under the Loan Documents and the Liens in favor of the Secured Parties and priority thereof, as Administrative Agent may reasonably require including, without limitation (i) the authorization of Administrative Agent to execute and/or the execution by Borrowers (if applicable) and filing of UCC financing statements, (ii) the execution and delivery of all such writings necessary to transfer any Licenses into the name of Administrative Agent or its designee after the occurrence of any Event of Default, to the extent such Licenses are transferable, (iii) Borrowers shall cause the applicable Manager to cooperate in the transfer of any Licenses held by such Manager to Borrowers upon the termination of the Management Agreement with such Manager, in accordance with the terms of such Management Agreement, and (iv) in respect of Collateral consisting of motor vehicles in excess of \$25,000.00 in value per vehicle only, notification by Borrowers of the existence of such vehicles whether or not requested by Administrative Agent and upon the request of Administrative Agent such information relating to and instruments of title relating thereto as may be necessary to perfect Liens; and

(c) do and execute all and such further lawful and reasonable acts, conveyances and assurances for the better and more effective carrying out of the intents and purposes of this Agreement and the other Loan Documents, as Administrative Agent shall reasonably require from time to time, including cooperating (i) to assist any Lender in obtaining any additional appraisals required under FIRREA, at the sole expense of such Lender, and (ii) to assist Administrative Agent in obtaining updated appraisals at any time and from time to time, provided that unless an Event of Default has occurred and is continuing, Borrowers shall only be responsible for costs of one such updated appraisal for each Collateral Property per calendar year so long as more than six months has elapsed since the then most recent updated appraisal for such Collateral Property was obtained.

7.09. <u>Mortgage and Intangible Taxes</u>. Borrowers shall pay, or reimburse Administrative Agent and each Lender for (in each case, to the extent permitted by applicable Law) all state, county and municipal recording, mortgage, intangible, and all other taxes imposed upon the execution and recordation of the Mortgage and/or upon the execution and delivery of Notes and this Agreement.

7.10. Financial Reporting.

(a) Borrowers will keep and maintain or will cause to be kept and maintained on a Fiscal Year basis, in accordance with GAAP, (or such other accounting basis acceptable to Administrative Agent) proper and accurate books, records and accounts reflecting all of the financial affairs of Borrowers and all items of income and expense in connection with the operation on an individual basis of the Collateral Properties. Administrative Agent, at its own cost and expense (except as provided in the next sentence) shall have the right from time to time at all times during normal business hours upon reasonable notice to examine such books, records and accounts at the office of Borrowers or any other Person maintaining such books, records and accounts and to make such copies or extracts thereof as Administrative Agent shall desire. Upon the occurrence and during the continuance of an Event of Default, Borrowers shall pay any reasonable costs and expenses incurred by Administrative Agent to examine Borrowers' accounting records with respect to the Collateral Properties, as Administrative Agent shall determine to be necessary or appropriate. All operating and profits and loss statements required pursuant to this Section 7.10 shall be prepared for each Collateral Property and for the Collateral Properties taken as a whole. All other statements required pursuant to this Section 7.10 shall be prepared for the Collateral Properties taken as a whole.

(b) Each Borrower will furnish to Administrative Agent and the Lenders annually, within ninety (90) days following the end of each Fiscal Year: a complete copy of (i) Borrower's annual financial statements certified by a Responsible Officer of such Borrower, both in accordance with GAAP (or such other accounting basis acceptable to Administrative Agent) covering the Collateral Properties for such Fiscal Year and containing statements of profit and loss and a balance sheet, (ii) an operating statement certified by a Responsible Officer of Borrower for each Collateral Property and the Collateral Properties taken as a whole which present the operating results of the Collateral Properties in a manner consistent with those operating statements given by Borrower to Administrative Agent prior to the Closing Date, which operating statement shall be in substantially the form of Exhibit I. Such statements referred to in subsection (ii) above shall set forth the financial condition and the results of operations of the Collateral Properties for such Fiscal Year, and shall include, but not be limited to, amounts representing annual Net Operating Income for such period, and (iii) calculations, set forth in reasonable detail of the Debt Service Coverage Ratio, certified by a Responsible Officer of Borrowers. Borrower's annual financial statements shall be accompanied by (1) operating statements for each Collateral Property and the Collateral Properties taken as a whole showing a comparison of the income and expenses contained in the prior Fiscal Year's Approved Annual Budget and the actual income and expenses for the prior Fiscal Year, (2) a certificate executed by a Responsible Officer or other appropriate officer of such Borrower, stating that each such annual financial statement and operating statements present fairly the financial condition and the results of operations of each Borrower and the Collateral Properties being reported upon and has been prepared in accordance with GAAP, and (3) an annual occupancy report for such year, including the average daily room rate for such year.

(c) Each Borrower will furnish, or cause to be furnished, to Administrative Agent and the Lenders on or before forty-five (45) days after the end of each calendar quarter the following items, accompanied by a certificate of a Responsible Officer or other appropriate officer of such Borrower, stating that such items are true, correct, accurate, and complete and fairly present the results of the operations of such Borrower and the Collateral Properties: (i) a report of occupancy for the subject quarter including an average daily rate, and any and all franchise inspection reports received by any Borrower during the subject quarter accompanied by an Officer's Certificate with respect thereto; (ii) quarterly and year-to-date operating statements, and Capital Expenditures presented for each Collateral Property and the Collateral Properties taken as a whole in a form consistent with the operating statements and Capital Expenditures reports delivered by Borrowers to Administrative Agent and Lenders in connection with Lenders' underwriting of the Loan which operating statement shall be in substantially the form of Exhibit I) and prepared for each calendar quarter, noting Net Operating Income for such period, and other information necessary and sufficient to fairly present the results of operation of the Collateral Properties during such calendar quarter, and containing a comparison of budgeted income and expenses and the actual income and expenses, (iii) a detailed explanation of any variances which are both (I) ten percent (10%) or more and (II) in excess of \$50,000 between budgeted and actual amounts for any Collateral Property, all in form satisfactory to Administrative Agent; (iv) calculations, set forth in reasonable detail of the Debt Service Coverage Ratio, certified by a Responsible Officer of Borrower and (v) a Smith Travel Research STAR Report or similar market benchmarking service for each Collateral Property.

(d) Each Borrower will furnish, or cause to be furnished, to Administrative Agent and the Lenders on or before thirty (30) days after the end of each calendar month, monthly profits and loss statements for each Collateral Property, accompanied by a certificate of a Responsible Officer or other appropriate officer of Borrower, stating that such statements are true, correct, accurate, and complete and fairly present the results of the operations of Borrowers and each Collateral Properties.

(e) For Fiscal Year 2012, and for each Fiscal Year thereafter, (I) Borrowers shall submit to Administrative Agent a preliminary Annual Budget for each Collateral Property not later than thirty (30) days prior to the commencement of such Fiscal Year and (II) Borrowers shall submit to Administrative Agent a final proposed Annual Budget for each Collateral Property not later than sixty (60) days after to the commencement of such Fiscal Year. The final budget shall be in form reasonably satisfactory to Administrative Agent, and shall be subject to Administrative Agent's written approval, which approval shall not be unreasonably withheld or delayed (each such Annual Budget after it has been approved in writing by Administrative Agent shall be hereinafter referred to as an "Approved Annual Budget"; provided that each of the Annual Budgets for Fiscal Year 2011 for each Collateral Property, which were delivered on or before the Closing Date, constitutes an Approved Annual Budget for such Collateral Property). In the event that Administrative Agent objects to the preliminary or final proposed Annual Budget submitted by Borrowers, Administrative Agent shall advise Borrowers of such objections within fifteen (15) days after receipt respectively thereof (and deliver to Borrowers a reasonably detailed description of such objections) and Borrowers shall promptly revise such Annual Budget and resubmit the same to Administrative Agent. Administrative Agent shall advise Borrowers of any objections to such revised Annual Budget within ten (10) days after receipt thereof (and deliver to Borrowers a reasonably detailed description of such objections) and Borrowers shall promptly revise the same in accordance with the process described in this subsection until Administrative Agent approves the Annual Budget. Until such time that Administrative Agent approves a proposed Annual Budget, the most recently Approved Annual Budget shall apply; provided that, such Approved Annual Budget shall be adjusted to reflect (A) estimated increases in Taxes, Insurance Premiums and utilities expenses and, (B) any incremental increases provided for under the terms of the Management Agreements. Any such proposed preliminary or final Annual Budget submitted to Administrative Agent for Administrative Agent's approval shall be deemed approved if Administrative Agent shall have failed to notify Borrowers of its approval or disapproval within fifteen (15) Business Days following Administrative Agent's receipt of Borrowers' written

request together with such final proposed Annual Budget, as the case may be, and any and all required information and documentation reasonably required by Administrative Agent to reach a decision, provided, such request to Administrative Agent is marked in bold lettering with the following language: "ADMINISTRATIVE AGENT'S RESPONSE IS REQUIRED WITHIN FIFTEEN (15) BUSINESS DAYS OF RECEIPT OF THIS NOTICE PURSUANT TO THE TERMS OF ATERM LOAN AGREEMENT AMONG THE UNDERSIGNED, AND ADMINISTRATIVE AGENT AND OTHERS" and the envelope containing the request must be marked "PRIORITY". With respect to the Annual Budgets to the extent that the timing and the procedures for approval of such Annual Budgets as set forth herein shall be inconsistent with the timing and the procedures set forth in the Management Agreements, the timing and procedures set forth in the Management Agreement shall control, provided that the foregoing shall not in any way limit Administrative Agent's rights to approve such Annual Budgets as set forth above prior to Borrowers agreement on such Budget with the applicable Manager.

(f) Borrowers shall furnish to Administrative Agent and each Lender, promptly, and in any event, within ten (10) Business Days after written request such further detailed information with respect to the operation of the Collateral Properties and the financial affairs of Borrowers as may be reasonably requested by such Person. Borrowers shall give to Administrative Agent and each Lender notice of the departure of any Independent Director from any Borrower (and the appointment of any replacement therefore) in accordance with Section 6.34(x) above.

(g) Borrowers shall furnish to Administrative Agent and each Lender copies of any periodic reporting on covenant compliance under Section 4.18 of the Senior Secured Notes Indenture, concurrently with their delivery thereunder.

(h) Any reports, statements or other information required to be delivered under this Agreement shall be delivered (i) on a diskette or via email, (ii) if requested by Administrative Agent or a Lender and within the capabilities of Borrowers' data systems without change or modification thereto, in electronic form and prepared using a Microsoft Excel, Microsoft Word for Windows or WordPerfect for Windows files (which files may be prepared using a spreadsheet program and saved as word processing files) and (iii) if requested by Administrative Agent or a Lender, in paper form.

7.11. <u>Business and Operations</u>. Each Borrower will continue to engage in the businesses presently conducted by such Borrower as and to the extent the same are necessary for the ownership, maintenance, management and operation of the Collateral Properties. Each Borrower will exercise commercially reasonable efforts to comply at all times and in all material respects with the applicable brand standards at each Collateral Property. Borrowers will remain in good standing under the laws of each jurisdiction to the extent required for the ownership, maintenance, management and operation of the Collateral Properties.

7.12. <u>Costs of Enforcement</u>. Without limitation of any other provision of this Agreement or any other Loan Document, in the event (a) that any Mortgage encumbering any Collateral Property is foreclosed in whole or in part or that any such Mortgage is put into the hands of an attorney for collection, suit, action or foreclosure, (b) of the foreclosure of any mortgage prior to or subsequent to any Mortgage encumbering any Collateral Property in which proceeding Administrative Agent or any Lender is made a party, or (c) a Bankruptcy Event occurs, in respect of any Borrower, any Affiliate Ground Lessor or any of their constituent Persons or an assignment by any Borrower, any Affiliate Ground Lessors, and their successors or assigns, shall be chargeable with and agree to pay all costs of collection and defense, including attorneys' fees and costs, incurred by Administrative Agent, any Lender, Borrowers or the Affiliate Ground Lessors in connection therewith and in connection with any appellate proceeding or post-judgment action involved therein, together with all required service or use taxes.

7.13. Estoppel Statements.

(a) After written request by Administrative Agent, and no more often than one time during any twelve (12) month period (unless an Event of Default has occurred and is continuing) Borrowers shall within ten (10) Business Days furnish Administrative Agent and the Lenders, with a statement, duly acknowledged and certified, setting forth (i) the amount of the original principal amount of the Obligations, (ii) the unpaid principal amount of the Obligations, (iii) the Applicable Margin, (iv) the date installments of interest and/or principal were last paid, (v) any offsets or defenses to the payment of the Indebtedness hereunder, and (vi) that the Notes, this Agreement, the Mortgages and the other Loan Documents are valid, legal and binding obligations and have not been modified or if modified, giving particulars of such modification.

(b) Borrowers shall use commercially reasonable efforts to deliver to Administrative Agent and the Lenders upon request, and no more often than one time during any twelve (12) month period (unless an Event of Default has occurred and is continuing) tenant estoppel certificates from each commercial tenant under a Major Lease in form and substance reasonably satisfactory to Administrative Agent. As used in this clause (b), "commercially reasonable efforts" shall not, unless an Event of Default has occurred and is continuing or is reasonably likely to occur, require the payment of fees by Borrower to any Ground Lessor, other than reimbursement for nominal expenses associated with its legal review of the estoppel to the extent required under the applicable Ground Lease.

(c) Borrowers shall use commercially reasonable efforts, promptly upon request of Administrative Agent, and no more often than one time during any twelve (12) month period (unless an Event of Default has occurred and is continuing) to deliver to Administrative Agent and the Lenders an estoppel certificate from each Franchisor stating that (i) its Franchise Agreement is in full force and effect and has not been modified, amended or assigned, (ii) neither such Franchisor nor its Operating Lessee is in default under any of the terms, covenants or provisions of the Franchise Agreement and Franchisor knows of no event which, but for the passage of time or the giving of notice or both, would constitute an event of default under such Franchise Agreement, (iii) neither such Franchisor nor such Operating Lessee has commenced any action or given or received any notice for the purpose of terminating such Franchise Agreement and (iv) all sums due and payable to such Franchisor under its Franchise Agreement have been paid in full.

(d) Borrowers shall, promptly upon request of Administrative Agent, and no more often than one time during any twelve (12) month period (unless an Event of Default has occurred and is continuing) deliver to Administrative Agent and the Lenders an estoppel certificate from each Operating Lessee stating that (i) its respective Operating Leases are in full force and effect and have not been modified, amended or assigned, (ii) Borrowers are not in default under any of the terms, covenants or provisions of its Operating Lease(s) and such Operating Lessee does not know of any event which, but for the passage of time or the giving of notice or both, would constitute an event of default under any Operating Lease, (iii) no Borrower has commenced any action or given or received any notice for the purpose of terminating any Operating Lease and (iv) all sums due and payable under any Operating Lease have been paid in full.

(e) Borrowers shall use commercially reasonable efforts, promptly upon request of Administrative Agent, and no more often than one time during any twelve (12) month period (unless an Event of Default has occurred and is continuing) deliver to Administrative Agent and the Lenders an estoppel certificate from each Ground Lessor stating that (i) the applicable Ground Lease is in full force and effect and has not been modified, amended or assigned, (ii) neither Ground Lessor nor the applicable Ground Lessee is in default under any of the terms, covenants or provisions of the Ground Lease and Ground Lessor knows of no event which, but for the passage of time or the giving of notice or both, would constitute an event of default under the Ground Lease, (iii) neither Ground Lessor nor the applicable Ground Lessee has commenced any action or given or received any notice for the purpose of terminating the Ground Lease and (iv) all sums due and payable under the Ground Lease have been paid in full.

(f) Borrowers shall use commercially reasonable efforts, promptly upon request of Administrative Agent, and no more often than one time during any twelve (12) month period (unless an Event of Default has occurred and is continuing) deliver to Administrative Agent an estoppel certificate from each Manager stating that (i) the applicable Management Agreement is in full force and effect and has not been modified, amended or assigned, (ii) neither Manager nor applicable Borrower is in default under any of the terms, covenants or provisions of such Management Agreement and Manager knows of no event which, but for the passage of time or the giving of notice or both, would constitute an event of default under such Management Agreement, (iii) neither Manager nor applicable Borrower has commenced any action or given or received any notice for the purpose of terminating such Management Agreement and (iv) all sums due and payable under such Management Agreement have been paid in full.

7.14. <u>Use of Proceeds</u>. Borrowers shall use the proceeds of the Loans only for (a) Borrowers' working capital, Capital Expenditures and other general corporate purposes not in contravention of any law or any Loan Document and (b) to make certain Restricted Payments

7.15. <u>Performance by Borrowers</u>. Each Borrower shall in a timely manner observe, perform and fulfill each and every covenant, term and provision of each Loan Document executed and delivered by, or applicable to such Borrower.

7.16. Leasing Matters.

(a) With respect to any Collateral Property, each Borrower may enter into a proposed Lease (including the renewal or extension of an existing Lease (a "Renewal Lease")) without the prior written consent of Administrative Agent, provided such proposed Lease or Renewal Lease (i) provides for rental rates and terms comparable to existing local market rates and terms (taking into account the type and quality of the tenant) as of the date such Lease is executed by such Borrower (unless, in the case of a Renewal Lease, the rent payable during such renewal, or a formula or other method to compute such rent, is provided for in the original Lease), (ii) is an arms-length transaction with a bona fide, independent third party tenant, (iii) would not cause a Material Property Event, (iv) is subject and subordinate to the related Mortgage and, upon Administrative Agent's reasonable request, the lessee thereunder agrees to attorn to Administrative Agent and Lenders pursuant to an agreement acceptable to Administrative Agent and (v) is not a Major Lease. All proposed Leases which do not satisfy the requirements set forth in this Section 7.16(a) shall be subject to the prior approval of Administrative Agent, which approval shall not be unreasonably withheld, delayed or At Administrative Agent's request, Borrowers shall promptly deliver to conditioned. Administrative Agent copies of all Leases which are entered into pursuant to this Subsection together with Borrowers' certification that they have satisfied all of the conditions of this Section.

(b) Except as set forth in Schedule 6.39, Borrowers (i) shall observe and perform all the obligations imposed upon the lessor under the Major Leases and shall not do or permit to be done anything to impair the value of any of the Major Leases as security for the Obligations; (ii) shall promptly send copies to Administrative Agent and the Lenders of all notices of default or other material matters which Borrowers shall send or receive with respect to the Major Leases; (iii) shall enforce all of the material terms, covenants and conditions contained in the Major Leases upon the part of the tenant thereunder to be observed or performed (except for termination of a Major Lease which shall require Administrative Agent's prior written approval); (iv) shall not collect any of the Rents more than one (1) month in advance (except Security Deposits shall not be deemed Rents collected in advance); (v) shall not execute any other assignment of the lessor's interest in any of the Leases or the Rents; and (vi) shall not consent (to the extent that lessor's consent is required under a Major Lease) to any assignment of or subletting under any Major Leases not in accordance with their terms, without the prior written consent of Administrative Agent, which consent shall not be unreasonably withheld, delayed or conditioned.

(c) Borrowers may, without the consent of Administrative Agent, amend, modify or waive the provisions of any Lease or terminate, reduce rents under, accept a surrender of space under, or shorten the term of, any Lease (including any guarantee, letter of credit or other credit support with respect thereto) provided that such Lease is not a Major Lease and that such action (taking into account, in the case of a termination, reduction in rent, surrender of space or shortening of term, the planned alternative use of the affected space) would not cause a Material Property Event, and provided that such Lease, as amended, modified or waived, is otherwise in compliance with the requirements of this Agreement and any Lease subordination agreement binding upon Administrative Agent with respect to such Lease. A termination of a Lease (other than a Major Lease) with a tenant who is in default beyond applicable notice and grace periods shall not be considered an action which constitutes a Material Property Event unless it causes a breach, default or failure of performance under a Management Agreement or Franchise Agreement. Any amendment, modification, waiver, termination, rent reduction, space surrender or term shortening which does not satisfy the requirements set forth in this Subsection shall be subject to the prior written approval of Administrative Agent and its counsel, at Borrowers' expense. At Administrative Agent's request, Borrowers shall promptly deliver to Administrative Agent and the Lenders copies of all Leases, amendments, modifications and waivers which are entered into pursuant to this <u>Section 7.16(c)</u> together with each applicable Borrower's certification that it has satisfied all of the conditions of this <u>Section 7.16(c)</u>.

Notwithstanding anything contained herein to the contrary, with respect to (d) any Collateral Property, no Borrower shall, without the prior written consent of Administrative Agent (which consent shall not be unreasonably withheld, delayed or conditioned), enter into, materially amend, materially modify, waive any material provisions of, terminate, reduce rents under, accept a surrender of space under, or shorten the term of, or renew or extend upon terms and conditions less favorable to such Borrower, any Major Lease or any instrument guaranteeing or providing credit support for any Major Lease; provided, however, such Borrower shall not be required to obtain Administrative Agent's written consent to any immaterial, non-economic change or beneficial economic change to a Major Lease (provided that, changes materially altering: the square footage leased, the location or use of the Collateral Property or extending the term including any renewal option for a period in excess of five (5) years shall, in each case, be considered a material change); and provided further, that no consent shall be required for any amendment under any Operating Lease, to the extent such amendment could not reasonably be expected to result in a decrease in the Net Operating Income of any Collateral Property by more than 5% or otherwise result in a Material Adverse Effect. For the avoidance of doubt, to the extent that the lessor's consent is required under the applicable Major Lease, any assignment or sublease of a Major Lease shall require the consent of Administrative Agent.

(e) To the extent actually received by Administrative Agent, Administrative Agent shall hold any and all monies representing security deposits under the Leases (the "<u>Security Deposits</u>") received by Administrative Agent, in accordance with the terms of this Agreement and the respective Lease, and shall only release the Security Deposits in order to return a tenant's Security Deposit to such tenant if such tenant is entitled to the return of the Security Deposit under the terms of the Lease.

(f) To the extent that Administrative Agent's consent or approval is required under this <u>Section 7.16</u>, any such proposed modification, change, supplement, alteration, amendment, assignment or sublease of a Lease or Major Lease submitted to Administrative Agent for approval shall be deemed approved if (i) Borrowers deliver to Administrative Agent a written request for such approval marked in bold lettering with the following language: ADMINISTRATIVE AGENT'S RESPONSE IS REQUIRED WITHIN FIFTEEN (15) BUSINESS DAYS OF RECEIPT OF THIS NOTICE PURSUANT TO THE TERMS OF ATERMLOAN AGREEMENT AMONG THE UNDERSIGNED, AND ADMINISTRATIVE AGENT AND OTHERS" and the envelope containing the request must be marked "PRIORITY"; and (ii) Administrative Agent shall have failed to notify Borrowers of its approval or disapproval within such fifteen (15) Business Days following Administrative Agent's receipt of Borrowers' written request together with such proposed modification, change, supplement, alteration, amendment, assignment or sublease of a Lease or Major Lease, and any and all other information and documentation relating thereto reasonably required by Administrative Agent to reach a decision. In no event shall Administrative Agent be deemed to have approved (1) a surrender, termination or cancellation of a Major Lease or (2) any change having a Material Adverse Effect; provided that Administrative Agent shall not unreasonably delay such decision. Upon Borrowers' request, Administrative Agent shall deliver to Borrower a reasonably detailed description of the reasons for any disapprovals under this <u>Section 7.16</u>.

7.17. Management Agreements.

(a) Each Collateral Property is operated under the terms and conditions of the applicable Management Agreement. Each Operating Lessee shall (i) diligently perform and observe all of the terms, covenants and conditions of the Management Agreements on the part of such Operating Lessee to be performed and observed to the end that all things shall be done which are necessary to keep unimpaired the rights of such Operating Lessee under the Management Agreements and (ii) promptly notify Administrative Agent and the Lenders of the giving of any notice by any Manager to such Operating Lessee of any default by Operating Lessee in the performance or observance of any of the terms, covenants or conditions of any Management Agreement on the part of Operating Lessee to be performed and observed and deliver to Administrative Agent a true copy of each such notice. No Operating Lessee shall surrender any Management Agreement, consent to the assignment by any Manager of its interest under a Management Agreement, or terminate or cancel any Management Agreement, or modify, change, supplement, alter or amend any Management Agreement, in any material respect, either orally or in writing without Required Lenders' prior written consent; provided, however, that Borrowers shall be entitled to terminate or cancel (i) the Myrtle Beach Golf Course Management Agreement or (ii) any other Management Agreement that is replaced with a Replacement Management Agreement with any Qualified Manager, in either case without the prior consent of the Administrative Agent or any Lenders. Borrowers shall deliver notice of such termination or cancellation to the Administrative Agent and the Lenders within five (5) Business Days of the occurrence thereof. Borrowers hereby assign to Administrative Agent for benefit of the Secured Parties as further security for the payment of the Obligations and for the performance and observance of the terms, covenants and conditions of this Agreement and the other Loan Documents, all the rights, privileges and prerogatives of Borrowers to surrender any Management Agreement, or to terminate, cancel, modify, change, supplement, alter or amend such Management Agreements, in any material respect, and any such surrender of such Management Agreements, or termination, cancellation, modification, change, supplement, alteration or amendment of any Management Agreement (other than the Myrtle Beach Golf Course Management Agreement) in any material respect, without the prior consent of Administrative Agent and Required Lenders, shall be void and of no force and effect. Any such proposed modification, change, supplement, alteration or amendment of the Management Agreement submitted to Administrative Agent and the Lenders for approval shall be deemed approved if (i) Borrowers deliver to the Lenders a written request for such approval marked in bold lettering with the following language: "ADMINISTRATIVE AGENT AND **REQUIRED LENDERS' RESPONSE IS REQUIRED WITHIN FIFTEEN (15) BUSINESS** DAYS OF RECEIPT OF THIS NOTICE PURSUANT TO THE TERMS OF A TERM LOAN AGREEMENT AMONG THE UNDERSIGNED, AND ADMINISTRATIVE AGENT AND OTHERS" and the envelope containing the request must be marked "PRIORITY", and (ii) Administrative Agent and Required Lenders shall have failed to notify Borrowers of approval or disapproval within such fifteen (15) Business Days following Administrative Agent's and the Lenders' receipt of Borrowers' written request together with such proposed modification, change, supplement, alteration or amendment of the Management Agreement, and any and all other information and documentation relating thereto reasonably required by Administrative Agent or the Lenders to reach a decision. In no event shall Administrative Agent or the Required Lenders be deemed to have approved (1) a surrender, termination or cancellation of the Management Agreement, (2) any change having a Material Adverse Effect under the Management Agreement, or (3) or a new management agreement with a new property manager. Upon a Borrower's request, Administrative Agent and each of the disapproving Lenders shall deliver to such Borrower a reasonably detailed description of the reasons for any disapprovals under this Section 7.17.

(b) If any Operating Lessee shall default in the performance or observance of any material term, covenant or condition of any Management Agreement on the part of each Operating Lessee to be performed or observed, after expiration of any applicable notice and cure periods provided in such Management Agreement, then, without limiting the generality of the other provisions of this Agreement, and without waiving or releasing such Borrower from any of its obligations hereunder, Administrative Agent shall have the right, but shall be under no obligation, to pay any sums and to perform any act or take any action as may be appropriate to cause all the terms, covenants and conditions of such Management Agreement on the part of such Operating Lessee to be performed or observed to be promptly performed or observed on behalf of such Operating Lessee to the end that the rights of such Operating Lessee in, to and under such Management Agreement shall be kept unimpaired and free from default in all material respects. Administrative Agent and any Person designated by Administrative Agent by written notice to Borrowers shall have, and are hereby granted, the right to enter upon the applicable Collateral Property at any time and from time to time for the purpose of taking any such action. If any Manager shall deliver to Administrative Agent a copy of any notice of default under the Management Agreement, such notice shall constitute full protection to Administrative Agent for any action taken or omitted to be taken by Administrative Agent in good faith, in reliance thereon. No Operating Lessee shall, nor shall it permit any Manager to, sub-contract all or any material portion of its management responsibilities under any Management Agreement to a third-party without the prior written consent of Administrative Agent, which consent shall not be unreasonably withheld, delayed or conditioned. Each Operating Lessee shall request of Managers and deliver to Administrative Agent upon receipt such certificates of estoppel with respect to compliance by such Operating Lessee with the terms of its respective Management Agreement as may be reasonably requested by Administrative Agent. Each Operating Lessee shall exercise each individual option, if any, to extend or renew the term of its respective Management Agreements to the extent required to continue it in full force and effect until after the Maturity Date, and each Operating Lessee hereby authorizes and appoints Administrative Agent its attorney-in-fact to exercise any such option in the name of and upon behalf of such Operating Lessee, which power of attorney shall

be irrevocable and shall be deemed to be coupled with an interest. Any sums expended by Administrative Agent pursuant to this paragraph (i) shall bear interest at the Default Rate from the date such cost is incurred to the date of payment to Administrative Agent, (ii) shall be deemed to constitute a portion of the Obligations, (iii) shall be secured by the lien of the Mortgages and the other Loan Documents and (iv) shall be immediately due and payable upon demand by Administrative Agent therefor.

(c) Without limitation of the foregoing, the applicable Borrower shall, upon request of Administrative Agent or the Required Lenders, and in accordance with the provisions of the applicable assignment of Management Agreement, terminate any Management Agreement and replace the Manager thereunder, without penalty or fee payable by Borrowers, Administrative Agent or Lenders, if at any time during the term of this Agreement: (i) such Manager shall become insolvent or a debtor in bankruptcy or insolvency proceeding or (ii) there exists an event of default by such Manager under the applicable Management Agreement which continues beyond any applicable cure period, except that the requirements of this <u>Section 7.17</u> (c)(ii) shall not apply in the event each and all of the following conditions shall be satisfied (1) the applicable Borrower shall be diligently prosecuting the resolution of such default in a commercially reasonable manner, (2) Administrative Agent and the Lenders shall receive copies of all notices and correspondence sent and/or received by the applicable Borrower with respect to such default, (3) no Event of Default exists and is continuing and (4) no Material Adverse Effect or Material Property Event shall have occurred.

7.18. Environmental Covenants.

Borrowers covenant and agree that so long as any Obligations remain (a) outstanding (i) all uses and operations on or of the Collateral Properties, whether by a Borrower or any other Person, shall be in compliance in all material respects with all Environmental Laws and permits issued pursuant thereto; (ii) there shall be no Releases of Hazardous Materials in, on, under or from any of the Collateral Properties; (iii) there shall be no Hazardous Materials stored or located in, on, or under any of the Collateral Properties, except those that are both (A) in compliance with all Environmental Laws and with permits issued pursuant thereto, if and to the extent required, and (B) (1) in amounts not in excess of that necessary to operate, clean, repair and maintain the applicable Collateral Property as a hotel or (2) fully disclosed to and approved by Administrative Agent in writing; (iv) Borrowers shall keep the Collateral Properties free and clear of all liens and other encumbrances imposed pursuant to any Environmental Law, whether due to any act or omission of Borrowers or any other Person (the "Environmental Liens"); (v) Borrowers shall, at their sole cost and expense, fully and expeditiously cooperate in all activities pursuant to paragraph (b) below, including but not limited to providing all relevant information and making knowledgeable persons available for interviews; (vi) Borrowers shall, at their sole cost and expense, perform any environmental site assessment or other investigation of environmental conditions in connection with any of the Collateral Properties, pursuant to any reasonable written request of Administrative Agent, upon Administrative Agent's reasonable belief that a Collateral Property is not in full compliance with all Environmental Laws or has been the subject of any Release of Hazardous Materials, and deliver to Administrative Agent and the Lenders full and complete copies of the reports and other results thereof, and Administrative Agent, the Lenders and any other Indemnitees shall be entitled to rely on such reports and other results thereof; (vii) Borrowers

shall, at their sole cost and expense (A) promptly and reasonably effectuate remediation of any Hazardous Materials in, on, under or from any Collateral Property in strict compliance with all requirements of Environmental Law; and (B) comply with any Environmental Law, in either case whether or not requested to do so by Administrative Agent; (viii) Borrowers shall not allow any tenant or other user of any of the Collateral Properties to violate any Environmental Law; and (ix) Borrowers shall immediately notify Administrative Agent in writing after any Borrower has become aware of (A) any presence or Release or threatened Releases of Hazardous Materials in, on, under, from or migrating towards any of the Collateral Properties; (B) any non-compliance with any Environmental Laws related in any way to any of the Collateral Properties; (C) any actual or potential Environmental Lien; (D) any required or proposed remediation of environmental conditions relating to any of the Collateral Properties; and (E) any written or oral notice or other communication of which a Borrower becomes aware from any source whatsoever (including but not limited to a Governmental Authority) relating in any way to Hazardous Materials in connection with the Collateral Properties.

(b) Administrative Agent and any other Person designated by Administrative Agent by written notice to Borrowers, including but not limited to any representative of a Governmental Authority, and any environmental consultant, and any receiver appointed by any court of competent jurisdiction, shall have the right, but not the obligation, to enter upon any Collateral Property at all reasonable times to assess any and all aspects of the environmental condition of any Collateral Property and its use. If any Event of Default shall have occurred or if Administrative Agent shall reasonably believe that any Hazardous Materials are located on the Collateral Property in violation of the terms and conditions of this Agreement or that any Release of any Hazardous Material has occurred to, from or onto any Collateral Property, then Administrative Agent and any other Person designated by Administrative Agent shall have the right, at the cost and expense of Borrowers, payable by Borrowers on demand from Administrative Agent, to conduct an environmental assessment or audit (the scope of which shall be determined in Administrative Agent's sole and absolute discretion) of the Collateral Property, including, without limitation, the taking of samples of soil, groundwater or other water, air, or building materials, and conducting other invasive testing. Borrowers shall cooperate with and provide access to Administrative Agent and any such Person or entity designated by Administrative Agent by written notice to Borrowers.

7.19. <u>Alterations</u>.

Other than the purchase, replacement and/or installation of FF&E or Capital Expenditures contemplated by the most recent Approved Annual Budget, Borrowers shall obtain Administrative Agent's prior written consent to (i) any material structural alteration or (ii) with respect to each Collateral Property, any other alteration to any Improvements thereon which is estimated to cost in excess of four (4%) percent of the value of the Collateral Property, which consent shall not be unreasonably withheld, delayed or conditioned, except with respect to alterations that may have a Material Adverse Effect or cause a Material Property Event. To the extent that Administrative Agent's consent or approval is required under this Section 7.19, any such proposed alterations to any Improvements submitted to Administrative Agent for approval shall be deemed approved if (i) Borrowers deliver to Administrative Agent a written request for such approval marked in bold lettering with the following language: "ADMINISTRATIVE AGENT'S RESPONSE IS

REQUIRED WITHIN FIFTEEN (15) BUSINESS DAYS OF RECEIPT OF THIS NOTICE PURSUANT TO THE TERMS OF A TERM LOAN AGREEMENT AMONG THE UNDERSIGNED, AND ADMINISTRATIVE AGENT AND OTHERS" and the envelope containing the request must be marked "PRIORITY"; and (ii) Administrative Agent shall have failed to notify Borrowers of its approval or disapproval within such fifteen (15) Business Days following Administrative Agent's receipt of Borrowers' written request together with a reasonably detailed description of such proposed alteration and any and all other information and documentation relating thereto reasonably required by Administrative Agent to reach a decision. In no event shall Administrative Agent be deemed to have approved alterations that may have a Material Adverse Effect or cause a Material Property Event. Upon a Borrower's request, Administrative Agent shall deliver to such Borrower a reasonably detailed description of the reasons for any disapprovals under this Section 7.19.

7.20. Franchise Agreement.

Each Franchised Property shall be operated under the terms and conditions (a) of the applicable Franchise Agreement. Borrowers shall (i) pay all sums required to be paid by each Operating Lessee under its Franchise Agreement, (ii) diligently perform, observe and enforce all of the terms, covenants and conditions of each Franchise Agreement on the part of each Operating Lessee to be performed, observed and enforced to the end that all things shall be done which are necessary to keep unimpaired the rights of any Borrower and/or each Operating Lessee under any Franchise Agreement, (iii) promptly notify Administrative Agent and the Lenders of the giving of any notice to either Operating Lessee of any default by such Operating Lessee in the performance or observance of any of the terms, covenants or conditions of any Franchise Agreement on the part of each Operating Lessee to be performed and observed and deliver to Administrative Agent a true copy of each such notice, and (iv) promptly deliver to Administrative Agent a copy of each financial statement, business plan, capital expenditure plan, notice, report and estimate received by it under any Franchise Agreement. Borrowers shall not, without the prior consent of Administrative Agent, surrender any Franchise Agreement or terminate or cancel any Franchise Agreement or modify, change, supplement, alter or amend any Franchise Agreement, in any material respect, either orally or in writing, provided, however, that Borrowers shall be entitled to terminate or cancel any Franchise Agreement that is replaced with a Replacement Franchise Agreement. Borrowers hereby assign to Administrative Agent as further security for the payment of the Obligations and for the performance and observance of the terms, covenants and conditions of this Agreement and the other Loan Documents, all the rights, privileges and prerogatives of Borrowers and Operating Lessees to surrender any Franchise Agreement or to terminate, cancel, modify, change, supplement, alter or amend any Franchise Agreement in any material respect, and any such surrender of any Franchise Agreement or termination, cancellation, modification, change, supplement, alteration or amendment of any Franchise Agreement in any material respect without the prior consent of Administrative Agent shall be void and of no force and effect.

(b) If either Operating Lessee shall default in the performance or observance of any material term, covenant or condition of any Franchise Agreement on the part of such Operating Lessee to be performed or observed after expiration of any applicable notice and cure periods provided in the Franchise Agreement, then, without limiting the generality of the other provisions of this Agreement, and without waiving or releasing a Borrower from any of its obligations hereunder, Administrative Agent shall have the right, but shall be under no obligation, to pay any sums and to perform any act or take any action as may be appropriate to cause all the terms, covenants and conditions of any Franchise Agreement on the part of such Operating Lessee to be performed or observed to be promptly performed or observed on behalf of such Operating Lessee, to the end that the rights of such Operating Lessee in, to and under such Franchise Agreement shall be kept unimpaired and free from default in all material respects. Administrative Agent and any Person designated by Administrative Agent by written notice to Borrowers shall have, and are hereby granted, the right to enter upon the Collateral Properties at any time and from time to time for the purpose of taking any such action. If any Franchisor shall deliver to Administrative Agent a copy of any notice of default under any Franchise Agreement, such notice shall constitute full protection to Administrative Agent for any action taken or omitted to be taken by Administrative Agent in good faith, in reliance thereon. Borrowers shall, from time to time, use their best efforts to obtain from each Franchisor such certificates of estoppel with respect to compliance by each Operating Lessee with the terms of its Franchise Agreement as may be requested by Administrative Agent. Each Operating Lessee shall exercise each individual option, if any, to extend or renew the term of its respective Franchise Agreement to the extent required to continue it in full force and effect until after the Maturity Date, and each Operating Lessee hereby expressly authorizes and appoints Administrative Agent as its attorney-in-fact to exercise any such option in the name of and upon behalf of such Operating Lessee, which power of attorney shall be irrevocable and shall be deemed to be coupled with an interest. Any sums expended by Administrative Agent pursuant to this paragraph shall bear interest at the Default Rate from the date such cost is incurred to the date of payment to Administrative Agent, shall be deemed to constitute a portion of the Obligations, shall be secured by the lien of the Mortgages and the other Loan Documents and shall be immediately due and payable upon demand by Administrative Agent therefor.

To the extent that Administrative Agent's consent or approval is required (c) under this Section 7.20, any such proposed modification, change, supplement, alteration or amendment of the Franchise Agreement submitted to Administrative Agent for approval shall be deemed approved if (i) Borrowers deliver to Administrative Agent a written request for such approval marked in bold lettering with the following language: "ADMINISTRATIVE AGENT'S RESPONSE IS REQUIRED WITHIN FIFTEEN (15) BUSINESS DAYS OF RECEIPT OF THIS NOTICE PURSUANT TO THE TERMS OF A TERM LOAN AGREEMENT AMONG THE UNDERSIGNED, AND ADMINISTRATIVE AGENT AND OTHERS" and the envelope containing the request must be marked "PRIORITY"; and (ii) Administrative Agent shall have failed to notify Borrowers of its approval or disapproval within such fifteen (15) Business Days following Administrative Agent's receipt of Borrowers' written request together with such proposed modification, change, supplement, alteration or amendment of the Franchise Agreement and any and all other information and documentation relating thereto reasonably required by Administrative Agent to reach a decision. In no event shall Administrative Agent be deemed to have approved (1) a surrender, termination or cancellation of the Franchise Agreement, (2) any change having a Material Adverse Effect under the Franchise Agreement, or (3) a new franchise agreement with a new franchisor. Upon a Borrower's request, Administrative Agent shall deliver to such Borrower a reasonably detailed description of the reasons for any disapprovals under this <u>Section 7.20</u>.

7.21. Operating Lease.

Other than as set forth in Schedule 6.39, each Borrower shall:

(a) promptly perform and/or observe all of the covenants and agreements required to be performed and observed by it under any Operating Lease and do all things necessary to preserve and to keep unimpaired its rights thereunder;

(b) promptly notify Administrative Agent and the Lenders of any event of default under any Operating Lease;

(c) promptly enforce the performance and observance of all of the covenants and agreements required to be performed and/or observed by either Operating Lessee under each Operating Lease;

(d) maintain each Operating Lease in full force and effect during the term of the Loan.

7.22. <u>OFAC</u>.

At all times throughout the term of the Loan, each Loan Party shall be in compliance in all material respects with all applicable orders, rules, regulations and recommendations of The Office of Foreign Assets Control of the U.S. Department of the Treasury.

7.23. Ground Leases.

With respect to each Ground Lease, each Ground Lessee shall (i) pay all (a) rents, additional rents and other sums required to be paid by such Ground Lessee, as tenant under and pursuant to the provisions of each Ground Lease, (ii) diligently perform and observe all of the terms, covenants and conditions of each Ground Lease on the part of such Ground Lessee, as tenant thereunder, (iii) promptly notify Administrative Agent of the giving of any notice by the landlord under the applicable Ground Lease to any Ground Lessee of any default by any Ground Lessee, as tenant thereunder, and deliver to Administrative Agent a true copy of each such notice within five (5) days of receipt and (iv) promptly notify Administrative Agent of any bankruptcy, reorganization or insolvency of the landlord under the applicable Ground Lease or of any notice thereof, and deliver to Administrative Agent a true copy of such notice within five (5) days of such Ground Lessee's receipt. No Ground Lessee shall, without the prior consent of Administrative Agent and the Required Lenders, surrender the leasehold estate created by the applicable Ground Lease or terminate or cancel any Ground Lease or modify, change, supplement, alter, amend or waive any term of any Ground Lease, either orally or in writing; provided that only the consent of Administrative Agent shall be required for any amendment to a Ground Lease which only extends the term of such Ground Lease and increases the annual rent payable thereunder by 5% or less; and provided further, that Borrowers shall

be entitled to terminate or cancel the Myrtle Beach Golf Course Lease without the prior consent of the Administrative Agent or any Lender. If any Ground Lessee shall default in the performance or observance of any term, covenant or condition of any Ground Lease on the part of such Ground Lessee, as tenant thereunder, Administrative Agent shall have the right, but shall be under no obligation, to pay any sums and to perform any act or take any action as may be appropriate to cause all of the terms, covenants and conditions of such Ground Lease on the part of such Ground Lessee to be performed or observed on behalf of such Ground Lessee, to the end that the rights of such Ground Lessee in, to and under such Ground Lease shall be kept unimpaired and free from default. If the landlord under the applicable Ground Lease shall deliver to Administrative Agent a copy of any notice of default under such Ground Lease, such notice shall constitute full protection to Administrative Agent for any action taken or omitted to be taken by Administrative Agent, in good faith, in reliance thereon. Each Ground Lessee shall exercise its option, if any, to extend or renew the term of each Ground Lease (other than the Myrtle Beach Golf Course Lease) upon demand by Administrative Agent made at any time within one (1) year prior to the last day upon which any such option may be exercised, and each Ground Lessee hereby expressly authorizes and appoints Administrative Agent its attorney-in-fact to exercise any such option in the name of and upon behalf of such Ground Lessee, which power of attorney shall be irrevocable and shall be deemed to be coupled with an interest.

(b) Notwithstanding anything contained in any Ground Lease to the contrary and except for Operating Leases, no Ground Lessee or Affiliate Ground Lessor shall further sublet any portion of the related Collateral Property (other than as permitted pursuant to Section 7.16) without prior written consent of Administrative Agent and the Required Lenders. To the extent that consent or approval of Required Lenders or Administrative Agent is required under this Section 7.23, any such proposed modification, change, supplement, alteration or amendment, or proposed sublease submitted to Administrative Agent and Lenders for approval shall be deemed approved if (i) Borrowers deliver to Administrative Agent and Lenders a written request for such approval marked in bold lettering with the following language: "ADMINISTRATIVE AGENT AND REQUIRED LENDERS' RESPONSE IS REQUIRED WITHIN FIFTEEN (15) BUSINESS DAYS OF RECEIPT OF THIS NOTICE PURSUANT TO THE TERMS OF A TERM LOAN AGREEMENT AMONG THE UNDERSIGNED, AND ADMINISTRATIVE AGENT AND OTHERS" and the envelope containing the request must be marked "PRIORITY", and (ii) Administrative Agent and Required Lenders shall have failed to notify Borrowers of approval or disapproval within such fifteen (15) Business Days following Administrative Agent's and the Lenders' receipt of Borrowers' written request together with such proposed modification, change, supplement, alteration or amendment, or sublease and any and all other information and documentation relating thereto reasonably required by Administrative Agent or the Lenders to reach a decision. In no event shall Administrative Agent or the Required Lenders be deemed to have approved any modification, change, supplement, alteration or amendment or sublease that may have a Material Adverse Effect. Upon a Borrower's request, Administrative Agent and each of the disapproving Lenders shall deliver to such Borrower a reasonably detailed description of the reasons for any disapprovals under this Section 7.23.

(c) In the event of any default by any Ground Lessee in the performance of any of its obligations under any Ground Lease, including, without limitation, any default in the payment of rent and other charges and impositions made payable by the lessee thereunder, then, in the case of a default in the payment of any monetary obligation, Administrative Agent may, at its option and without notice, make payment to cure such default, or, in the case of a default in the performance of any non-monetary obligation, Administrative Agent may, at its option, cause the default or defaults to be remedied and otherwise exercise any and all of the rights of the Ground Lessee thereunder in the name of and on behalf of the Ground Lessee. Each Owner shall, on demand, reimburse Agent for all advances made and expenses incurred by Administrative Agent in curing any such default (including, without limitation, reasonable attorneys' fees), together with interest thereon computed at the Default Rate from the date that an advance is made or expense is incurred, to and including the date the same is paid.

(d) Notwithstanding anything to the contrary contained herein with respect to any Ground Lease:

(i) The Lien of the Mortgages attach to all of the Ground Lessees' rights and remedies at any time arising under or pursuant to subsection 365(h) of the Bankruptcy Code, including, without limitation, all of the Ground Lessees' rights, as debtor, to remain in possession of the related Collateral Property which is subject to a Ground Lease;

(ii) No Ground Lessee shall, without Administrative Agent's written consent, elect to treat a Ground Lease as terminated under subsection 365(h)(1) of the Bankruptcy Code. Any such election made without Administrative Agent's prior written consent shall be void;

(iii) As security for the Obligations, each Ground Lessee unconditionally assigns, transfers and sets over to Administrative Agent for the benefit of the Lenders all of such Ground Lessee's claims and rights to the payment of damages arising from any rejection by any Ground Lessor under the Bankruptcy Code. Administrative Agent and the applicable Owner shall proceed jointly or in the name of such Owner in respect of any claim, suit, action or proceeding relating to the rejection of a Ground Lease, including, without limitation, the right to file and prosecute any proofs of claim, complaints, motions, applications, notices and other documents in any case in respect of a Ground Lessor under the Bankruptcy Code. This assignment constitutes a present, irrevocable and unconditional assignment of the foregoing claims, rights and remedies, and shall continue in effect until all of the Obligations shall have been satisfied and discharged in full. Any amounts received by Administrative Agent or any Owner as damages arising out of the rejection of a Ground Lease as aforesaid shall be applied to all costs and expenses of Administrative Agent (including, without limitation, reasonable attorneys' fees and costs) incurred in connection with the exercise of any of its rights or remedies in accordance with the applicable provisions hereof;

(iv) If pursuant to subsection 365(h) of the Bankruptcy Code, any Ground Lessee seeks to offset, against the rent reserved in a Ground Lease, the amount of any damages caused by the nonperformance by the applicable Ground Lessor of any of its obligations thereunder after the rejection by such Ground Lessor under the Bankruptcy Code, then no Owner shall effect any offset of the amounts so objected to by Administrative Agent. If Administrative Agent has failed to object as aforesaid within ten (10) days after notice from such Ground Lessee in accordance with the first sentence of this subsection, such Ground Lessee may proceed to offset the amounts set forth in such Ground Lessee' notice to Administrative Agent;

(v) In any action, proceeding, motion or notice shall be commenced or filed in respect of any Ground Lessor of all or any part any Collateral Property subject to a Ground Lease in connection with any case under the Bankruptcy Code, Administrative Agent and the applicable Owner shall cooperatively conduct and control any such litigation with counsel agreed upon between such Owner and Administrative Agent in connection with such litigation. Borrowers shall, upon demand, pay to Administrative Agent all costs and expenses (including reasonable attorneys' fees and costs) actually paid or actually incurred by Administrative Agent or any Lender in connection with the cooperative prosecution or conduct of any such proceedings. All such costs and expenses shall be secured by the Lien of the applicable Mortgage; and

(vi) Each Owner shall promptly, after obtaining knowledge of such filing, notify Administrative Agent orally of any filing by or against a Ground Lessor of a petition under the Bankruptcy Code. Each Owner shall thereafter promptly give written notice of such filing to Administrative Agent and the Lenders, setting forth any information available to such Owner as to the date of such filing, the court in which such petition was filed, and the relief sought in such filing. Each Owner shall promptly deliver to Administrative Agent any and all notices, summons, pleadings, applications and other documents received by Administrative Agent in connection with any such petition and any proceedings relating to such petition. (e) Each of Philadelphia Affiliate Ground Lessor and Hospitality Owner acknowledge and confirm that, with respect to the Philadelphia Affiliate Ground Lease and each of the Boca Raton Affiliate Ground Lessor and Boca Owner acknowledge and confirm that with respect to the Boca Raton Affiliate Ground Lease:

(i) the Mortgage covering such Ground Lease Property is and shall be deemed to be a "Leasehold Mortgage" as set forth in the applicable Affiliate Ground Lease, for all purposes thereunder;

(ii) Administrative Agent, as mortgagee, grantee or beneficiary, as applicable, thereunder, together with its successors and assigns in such capacity, shall constitute a "Leasehold Mortgagee" thereunder;

(iii) the Mortgages covering such Ground Lease Property shall be entitled to all the rights, benefits and privileges of a "Leasehold Mortgage" under the applicable Affiliate Ground Lease;

(iv) Administrative Agent shall be entitled to all the rights, benefit and privileges of a "Leasehold Mortgagee" under the applicable Affiliate Ground Lease; and

(v) the notice for address for the "Leasehold Mortgagee" under the applicable Affiliate Ground Lease, shall be the address of Administrative Agent as set forth in this Agreement.

7.24. <u>O&M Program</u>.

With respect to each of the Dana Point Property, the Houston Property, the Myrtle Beach Property and the Mandalay Beach Property, the applicable Borrower shall implement and follow the terms and conditions of its applicable O&M Program during the term of this Agreement, including any extension or renewal thereof. Such requirement that a Borrower develop and comply with its applicable O&M Program shall not be deemed to constitute a waiver or modification of any of Borrowers' covenants and agreements with respect to Hazardous Materials or Environmental Laws.

7.25. Certain Post-Closing Obligations.

As promptly as practicable, and in any event within the time periods after the Closing Date specified in <u>Schedule 7.25</u> or such later date as the Administrative Agent agrees to in writing, Borrowers and each other Loan Party shall deliver the documents or take the actions specified on <u>Schedule 7.25</u>.

ARTICLE VIII NEGATIVE COVENANTS

From the date hereof until payment and performance in full of all Obligations under the Loan Documents or, in respect of a specific Collateral Property, until the earlier release of the Liens of all Mortgages encumbering such Collateral Property in accordance with the terms of this

Agreement and the other Loan Documents, Borrowers each covenant that it will not do, directly or indirectly, any of the following:

8.01. <u>Indebtedness</u>.

No Borrower shall create, incur, assume or suffer to exist any Indebtedness other than the Obligations, except for Indebtedness of the type described pursuant to Section 6.34(g) or permitted by Section 8.04.

8.02. Investments.

No Borrower shall make any Investments from Collateral except:

(a) Investments in the form of Permitted Investments; and

(b) Investments consisting of extensions of credit in the nature of accounts receivable or notes receivable arising from the grant of trade credit in the ordinary course of business, and Investments received in satisfaction or partial satisfaction thereof from financially troubled account debtors to the extent reasonably necessary in order to prevent or limit loss.

8.03. <u>Liens</u>.

No Borrower shall, nor shall any Borrower permit any other Person to, create, incur, assume, or suffer to exist any Lien upon any Collateral, Collateral Property or any Equity Interest in any Restricted Party other than any of the following (each a "<u>Permitted Lien</u>"):

- (a) Liens pursuant to any Loan Document;
- (b) Liens existing on the date hereof and listed on Schedule 8.03;

(c) Liens for taxes not yet due or which are being contested in good faith and by appropriate proceedings diligently conducted, if adequate reserves with respect thereto are maintained on the books of the applicable Person in accordance with GAAP;

(d) carriers', warehousemen's, mechanics', materialmen's, repairmen's, or other like Liens arising in the ordinary course of business which are (i) not in excess of \$2,000,000 in the aggregate for any individual Collateral Property or (ii) remain undischarged of record (by payment, bonding or otherwise) for a period of more than sixty (60) days, provided that in case of (i) and (ii), such Liens are being contested in good faith by appropriate proceedings diligently conducted;

(e) pledges or deposits in the ordinary course of business in connection with workers' compensation, unemployment insurance and other social security legislation, other than any Lien imposed by ERISA; (f) deposits to secure the performance of bids, trade contracts and leases (other than Indebtedness), statutory obligations, surety bonds, performance bonds, and other obligations of a like nature incurred in the ordinary course of business;

(g) Liens set forth in the Title Policies issued with respect to the Mortgages;

(h) other encumbrances on a Collateral Property, which do not constitute a grant by a Loan Party of a mortgage or deed of trust, which in the aggregate, are not substantial in amount, and do not in any case materially detract from the value of any Collateral Property subject thereto or materially interfere with the ordinary conduct of the business of the applicable Mortgagor;

(i) Liens securing judgments for the payment of money not constituting an Event of Default under Section 10.01(m) or securing appeal or other surety bonds related to such judgments;

(j) with respect to Personal Property constituting a part of the Collateral Property, a Permitted Personal Property Lien; and

(k) Liens related to financing or leasing arrangements permitted by <u>Section 8.04</u>; provided that such Liens do not encumber any property other than the property financed or leased under <u>Section 8.04</u>.

Except for such Permitted Liens and Permitted Personal Property Lien provided in <u>Section 8.04</u>, each Borrower will own all parts of the Collateral Properties and will not acquire any fixtures, equipment, or other property (including software embedded therein) forming a part of any Collateral Property pursuant to a Lease, license, security agreement, or similar agreement, whereby any party has or may obtain the right to repossess or remove same, without the prior written consent of Administrative Agent.

8.04. Personal Property Leasing and Financing.

Without the prior consent of Administrative Agent, which consent may be withheld in Administrative Agent's sole discretion, no Borrower shall, nor shall any Borrower permit any other Person to, create, incur, assume, or suffer to exist in connection with the ownership or operation of any hotel, any Personal Property subject to any financing or any leasing arrangements except (each a "<u>Permitted Personal Property Lien</u>"):

(a) those disclosed on <u>Schedule 6.42;</u>

(b) extensions, renewals, replacements and refinancings of any such financing or leasing arrangement referred to in (a) that are on an arm's length basis and that do not materially increase the outstanding principal amount thereof except for amounts attributable to (i) fees and expenses for extensions and renewals and (ii) market increases for refinancings and replacements; or (c) any other financing or leasing arrangements, provided however, that the aggregate amount of debt service or lease payments in respect of all financing and leasing arrangements shall be less than \$400,000 in the aggregate per annum for any individual Collateral Property.

8.05. Operation and Service Agreements.

No Borrower has entered into any material Contractual Obligation with respect to property or services to be provided by third parties with respect to any Collateral Property other than:

(a) those listed on <u>Schedule 6.41;</u>

(b) any renewal, replacement or extension of the foregoing on substantially similar financial terms or reasonable increases thereof made on an arm's length basis; or

(c) any other material Contractual Obligation entered into in the ordinary course of business consistent with past practices.

8.06. <u>Restricted Payments</u>.

No Borrower shall declare or make, directly or indirectly, any Restricted Payment from Collateral, or incur any obligation (contingent or otherwise) to do so, except that:

(a) each Borrower may make Restricted Payments so long as there is no Default or Event of Default has occurred and is continuing or would result therefrom;

(b) each Borrower may declare and make dividend payments or other distributions payable solely in the common stock or other common Equity Interests of such Person;

(c) each Loan Party may purchase, redeem, or otherwise acquire shares of its common stock or other common Equity Interests or warrants or options to acquire any such shares with the proceeds received from the substantially concurrent issue of new shares of its common stock or other common Equity Interests.

provided, however, that notwithstanding the foregoing Borrowers may declare and make any Restricted Payments that are necessary for FelCor Trust to maintain its status as a REIT, but then only in the minimum amount necessary to maintain such status as a REIT.

8.07. Financial Covenants.

(a) Debt Service Coverage Ratio. Borrowers shall not permit the Debt Service Coverage Ratio as of the last day of any fiscal quarter or any other date of testing during any period set forth below, to be less than the amount set forth below:

Period	Debt Service Coverage Ratio
Closing Date until March 4, 2012	1.30 to 1.00
March 5, 2012 until March 4, 2013	1.35 to 1.00
March 5, 2013 until August 1, 2014	1.40 to 1.00
August 2, 2014 and thereafter	1.50 to 1.00

<u>provided that</u> if Borrowers fail to meet such covenant level as of the last day of any fiscal quarter from time to time, Borrowers may make a voluntary prepayment of the Loans pursuant to <u>Section 2.09(a)</u> in an amount sufficient to cure such failure by reducing the amount in the denominator (Debt Service) and bring Borrowers back into pro forma covenant compliance, <u>provided further</u>, that such voluntary prepayment must be made within five (5) Business Days of the date on which financial statements showing such Default are due to be delivered to Administrative Agent pursuant to <u>Section 7.10</u>.

(b) Loan to Value Ratio. As of September 4, 2012, Borrowers shall not permit the Loan to Value Ratio for the Collateral Properties to be more than sixty percent (60%) based on the most recent Acceptable Appraisals for the Collateral Properties (and after giving effect to any permanent reductions of the Aggregate Commitment that may be necessary to cause the Loan to Value Ratio to be equal to or less than sixty percent (60%)).

8.08. Dispositions

No Borrower shall make any Disposition from Collateral or enter into any agreement to make any Disposition from Collateral except:

(a) Dispositions of personal property, whether now owned or hereafter acquired, in the ordinary course of business for fair consideration and on an arms length basis, provided that if such property was required under any Loan Document to be subject to a first priority Lien in favor of Administrative Agent, such personal property shall be replaced by an item of equal or greater value which is subject to a first priority Lien in favor of Administrative Agent;

(b) Dispositions of Collateral Properties, so long as such Collateral Properties are released pursuant to <u>Section 2.17</u> prior to or contemporaneously with such Disposition; and

(c) Dispositions of the Myrtle Beach Golf Course Management Agreement and Myrtle Beach Golf Course Lease as contemplated by Sections 7.17(a) and 7.23(a);

provided, however, that any Disposition pursuant to clauses (a) and (b) shall be for fair market value.

8.09. <u>Dissolution</u>.

No Borrowers shall (a) engage in any dissolution, liquidation or consolidation or merger with or into any other business entity, (b) transfer, lease or sell, in one transaction or any combination of transactions, the assets or all or substantially all of the properties or assets of such Borrower except to the extent expressly permitted by the Loan Documents, (c) except as expressly permitted under the Loan Documents, modify, amend, waive or terminate its organizational documents or its qualification and good standing in any jurisdiction or (d) permit any Borrower's Principal to do any of the following if such action could reasonably be expected to have a Material Adverse Effect: (i) dissolve, wind up or liquidate or take any action, or omit to take any action, as a result of which any Borrower's Principal would be dissolved, wound up or liquidated in whole or in part, or (ii) except as expressly permitted under the Loan Documents, amend, modify, waive or terminate the certificate of incorporation, bylaws or similar organizational documents of any Borrower's Principal (other than FelCor Op and FelCor TRS), in each case, without obtaining the prior written consent of Administrative Agent, which consent (with respect to (c) only) shall not be unreasonably withheld, delayed or conditioned. In connection with clause (c) hereof, in addition to obtaining the consent of Administrative Agent as set forth in the preceding sentence, no Borrower may amend its limited liability company agreement to provide for the membership interests therein to become subject to Article 8 of the UCC, without first giving Administrative Agent at least thirty (30) days prior written notice.

8.10. No Subsidiaries.

Except as may be previously approved by the Required Lenders in writing, no Borrower shall create or suffer to exist any Subsidiary.

8.11. Burdensome Agreements.

No Borrower shall enter into any Contractual Obligation (other than this Agreement or any other Loan Document) that (a) limits the ability (i) of any Guarantor to Guarantee the Indebtedness of Borrowers under the Guaranty or (ii) of any Borrower to create, incur, assume or suffer to exist Liens on property of such Person; or (b) requires the grant of a Lien to secure an obligation of such Person if a Lien is granted to secure another obligation of such Person.

8.12. Change In Business.

No Borrower shall enter into any line of business other than the ownership, acquisition, development, operation, leasing and management of the Collateral Properties (including providing services in connection therewith), or make any material change in the scope or nature of its business objectives, purposes or operations or undertake or participate in activities other than the continuance of its present business. Each Borrower shall not take or fail to take any action relating to its status as a Single Purpose Entity, as further detailed in <u>Section 6.34</u> above.

8.13. <u>Debt Cancellation</u>.

Borrowers shall not cancel or otherwise forgive or release any material claim or debt (other than termination of Leases in accordance herewith) owed to Borrowers by any Person, except for adequate consideration and in the ordinary course of Borrowers' business.

8.14. <u>Zoning</u>.

Borrowers shall not initiate or consent to any zoning reclassification of any portion of any Collateral Property or seek any variance under any existing zoning ordinance or use or permit the use of any portion of any Collateral Property in any manner that could result in such use becoming a non-conforming use under any zoning ordinance or any other applicable Law, without the prior written consent of Administrative Agent.

8.15. No Joint Assessment.

Borrowers shall not suffer, permit or initiate the joint assessment of any Collateral Property with (a) any other real property constituting a tax lot separate from such Collateral Property, or (b) any portion of such Collateral Property which may be deemed to constitute personal property, or any other procedure whereby the Lien of any taxes which may be levied against such personal property shall be assessed or levied or charged to such Collateral Property.

8.16. Name, Identity, Structure, or Principal Place of Business.

No Borrower shall change its name, identity (including its trade name or names), or principal place of business as set forth in Schedule 6.33, without, in each case, first giving Administrative Agent at least thirty (30) days prior written notice. No Borrower shall change its corporate, partnership or other structure, or the place of its organization as set forth in Schedule 6.33, without, in each case, the consent of Administrative Agent. To the extent that Administrative Agent's consent or approval is required under this Section 8.16, any such proposed change submitted to Administrative Agent for approval shall be deemed approved if (i) Borrowers deliver to Administrative Agent a written request for such approval marked in **bold** lettering with the following language: "ADMINISTRATIVE AGENT'S RESPONSE IS REQUIRED WITHIN FIFTEEN (15) BUSINESS DAYS OF RECEIPT OF THIS NOTICE PURSUANT TO THE TERMS OF A TERM LOAN AGREEMENT AMONG THE UNDERSIGNED, ADMINISTRATIVE AGENT AND OTHERS" and the envelope containing the request must be marked "PRIORITY"; and (ii) Administrative Agent shall have failed to notify Borrowers of its approval or disapproval within such fifteen (15) Business Days following Administrative Agent's receipt of Borrowers' written request together with a description of the nature of such proposed change, and any and all other information and documentation relating thereto reasonably required by Administrative Agent to reach a decision. Upon a Borrower's request, Administrative Agent shall deliver to such Borrower a reasonably detailed description of the reasons for any disapprovals under this Section 8.16. Upon Administrative Agent's request, such Borrower shall execute and deliver additional financing statements, security agreements and other instruments which may be necessary to effectively evidence or perfect the Secured Parties' security interest in the Collateral as a result of such change of principal place of business or place of organization.

8.17. <u>ERISA</u>.

(a) During the term of this Agreement or while any Obligations are outstanding, no Borrower shall be a Plan and none of the assets of any Borrower shall constitute Plan Assets.

(b) Borrowers further covenant and agree to deliver to Administrative Agent such certifications or other evidence from time to time throughout the term of the Loan, as requested by Administrative Agent in its sole discretion, and represent and covenant that (A) each Borrower is not and each Borrower and any ERISA Affiliate do not maintain or otherwise have any liability (contingent or otherwise) with respect to an "employee benefit plan" as defined in Section 3(3) of ERISA, which is subject to Title I or Title IV of ERISA, or a "governmental plan" within the meaning of Section 3(3) of ERISA; (B) no Borrower is subject to state statutes regulating investments and fiduciary obligations with respect to governmental plans; and (C) one or more of the following circumstances is true:

(i) Equity interests in such Borrower are publicly offered securities, within the meaning of 29 C.F.R. §2510.3-101(b)(2);

(ii) Less than twenty-five percent (25%) of each outstanding class of Equity Interests in such Borrower are held by "benefit plan investors" within the meaning of 29 C.F.R. 2510.3-101(f)(2), as modified by Section 3(42) of ERISA; or

(iii) Such Borrower qualifies as an "operating company" or a "real estate operating company" within the meaning of 29 C.F.R. §2510.3-101(c) or (e).

8.18. Affiliate Transactions.

Except for the Operating Leases, the Overhead Sharing Agreement and the Affiliate Ground Leases, Borrowers shall not enter into, or be a party to, any transaction with any Affiliate of Borrowers except in the ordinary course of business, with terms no less favorable to Borrowers or such Affiliate than would be obtained in a comparable arm's-length transaction with an unrelated third party.

8.19. Transfers.

(a) Except for (a) Permitted Liens, (b) Permitted Personal Property Liens, (c) sales or dispositions of Personal Property in accordance with the provisions of <u>Section 8.08</u>, (d) in connection with a release of any Collateral Property in accordance with the provisions of Section 2.17 or 2.18, (e) Condemnation of Collateral Property, or (f) as otherwise provided in Sections 8.19(b) and (c) below, no Borrower shall, nor shall any Borrower permit, any Transfer of any Collateral, Collateral Property or any Equity Interest in any Restricted Party or any part thereof or any legal or beneficial interest therein without the prior written consent of the Required Lenders, which may be withheld in their sole discretion. Administrative Agent and Lenders shall not be required to demonstrate any actual impairment of its security or any increased risk of default hereunder in order to declare the Obligations immediately due and payable upon a Transfer in violation of this Section 8.19. This provision shall apply to every Transfer regardless of whether voluntary or not, or whether or not Administrative Agent or Required Lenders have consented to any previous Transfer. Notwithstanding anything to the contrary contained in this Section 8.19, no transfer (whether or not such transfer shall constitute a Transfer) shall be made to any Person which is not permitted pursuant to applicable Law.

(b) The provisions of Section 8.19(a) set forth above shall not apply to Transfers of Equity Interests in any Restricted Party (i) under any will or applicable law of descent, (ii) among any of the holders of direct or indirect Equity Interests in Borrowers that were holders of Equity Interests in Borrowers on the Closing Date, or (iii) to any entity that is wholly owned, directly or indirectly, by FelCor Trust or FelCor Op or any combination thereof, as long as following any Transfer described in (ii) or (iii), of this sentence (1) FelCor Op shall at all times own, directly or indirectly one hundred percent (100%) of the Equity Interests in Borrowers, and FelCor Trust shall at all times be the sole general partner of, and maintain Control over, FelCor Op, (2) Administrative Agent receives at least fifteen (15) days prior written notice of such Transfer, (3) unless the Super-Majority Lenders provide their prior written consent (not to be unreasonably withheld), such Transfer does not result in a Change of Control and (4) within fifteen (15) days following any such Transfer, Borrowers shall deliver to Administrative Agent (a) a statement showing the current ownership of Borrowers, (b) a certification from Borrowers that Borrowers remain in compliance with the ERISA provisions of the Loan Documents, and (c) a certification from Borrowers that Borrowers remain in compliance with the representations, warranties and covenants in the Loan Documents relative to anti-terrorism laws, rules and regulations.

(c) Notwithstanding anything to the contrary in <u>Section 8.19(a)</u> above, (i) a Transfer of Equity Interests in FelCor Op or FelCor Trust shall be permitted without the consent of the Lenders so long as such Transfer does not result in a Change of Control or violate applicable Law and (ii) a Transfer of Equity Interests in FelCor or FelCor Trust that requires the consent of the Lenders pursuant to the preceding subclause (i) shall be permitted so long as the Super-Majority Lenders provide their prior written consent (not to be unreasonably withheld).

8.20. <u>REA</u>. No Borrower shall:

(a) enter into, terminate, or modify any REA in any material respect without Administrative Agent's prior written consent, which consent shall not be unreasonably withheld, conditioned or delayed; or

(b) fail to enforce, comply with, or cause each of the parties to the REA to comply with all of the material economic terms and conditions contained in any REA.

ARTICLE IX INSURANCE; CASUALTY; CONDEMNATION

9.01. <u>Insurance</u>.

(a) Borrowers shall obtain and maintain, or cause to be maintained, Policies for each Borrower and the Collateral Properties providing at least the following coverages:

(i) so called "All Risk" or "Special Form" insurance on the Improvements and the Personal Property, in each case (A) in an amount equal to one hundred percent (100%) of the "Full Replacement Cost," which for purposes of this Agreement shall mean actual replacement value (exclusive of costs of excavations, foundations, underground utilities and footings) with a waiver of depreciation; (B) containing an agreed amount endorsement or its equivalent with respect to the Improvements, business income, rent loss and Personal Property waiving all coinsurance provisions; (C) including windstorm/named storm coverages for the Boca Raton Property, providing for no deductible in excess of \$100,000, other than deductibles for windstorm and earthquake coverage, and in the case of the Myrtle Beach Property only, flood; each of which shall not exceed an amount equal to five percent (5%) of the insured value for each of the Collateral Property, which shall with respect to each Collateral Property, be no more than five percent (5%) of the insurable value of each Collateral Property; and (D) providing coverage for contingent liability from Operation of Building Laws, Demolition Costs and Increased Cost of Construction Endorsements together with customary "Ordinance or Law Coverage" if any of the Improvements or the use of each Collateral Property shall constitute legal non-conforming structures or uses;

(ii) commercial general liability insurance against claims for personal injury, bodily injury, death or property damage occurring upon, in or about each Collateral Property, including "Dram Shop" or other liquor liability coverage if alcoholic beverages are sold from or may be consumed at the Collateral Property such insurance (A) to be on the so-called "occurrence" form with a limit of not less than \$1,000,000 per occurrence and \$2,000,000 in the aggregate; (B) to continue at not less than the aforesaid limit until required to be changed by Administrative Agent in writing by reason of changed economic conditions making such protection inadequate; and (C) to cover at least the following hazards: (1) premises and operations; (2) products and completed operations on an "if any" basis; (3) independent contractors; (4) blanket contractual liability for all written and oral contracts; and (5) contractual liability covering the indemnities contained in Article 10 of the Mortgages to the extent the same is available;

(iii) business interruption/loss of rents insurance (A) with loss payable to Administrative Agent on behalf of Lenders; (B) covering all risks required to be covered by the insurance provided for in Section 9.01(a)(i); (C) in an amount equal to one hundred percent (100%) of the projected gross income from each Collateral Property (on an actual loss sustained basis) for a period continuing until the Restoration of the Collateral Property is completed; the amount of such business interruption/loss of rents insurance shall be determined prior to the Closing Date and at least once each year thereafter based on the greatest of: (x) Borrowers' reasonable estimate of the gross income from each Collateral Property and (y) the highest gross income received during the term of the Notes for any full calendar year prior to the date the amount of such insurance is being determined, in each case for the succeeding twenty-four (24) month period and (D) containing an extended period of indemnity endorsement which provides that after the physical loss to the Improvements and the Personal Property has been repaired, the continued loss of income will be insured until such income either returns to the same level it was at prior to the loss, or the expiration of twelve (12) months from the date that the applicable Collateral Property is repaired and operations are resumed, whichever first occurs, and notwithstanding that the policy may expire prior to the end of such period; all insurance proceeds payable to Administrative Agent pursuant to this Section 9.01(a)(iii) shall be held by Administrative Agent and shall be applied to the obligations secured hereunder from time to time due and payable hereunder and under the Notes and this Agreement; provided, however, that nothing herein contained shall be deemed to relieve any Borrower of its obligations to pay the obligations secured hereunder on the respective dates of payment provided for in the Notes and this Agreement except to the extent such amounts are actually paid out of the proceeds of such business interruption/ loss of rents insurance.

(iv) at all times during which structural construction, repairs or alterations are being made with respect to the Improvements (A) owner's contingent or protective liability insurance covering claims not covered by or under the terms or provisions of the insurance provided for in Section 9.01(c)(ii); and (B) the insurance provided for in Section 9.01(a)(i) shall be written in a so-called builder's risk completed value form (1) on a non-reporting basis, (2) against all risks insured against pursuant to Section 9.01(a)(i), (3) shall include permission to occupy each Collateral Property, and (4) shall contain an agreed amount endorsement waiving co-insurance provisions; provided, however, the insurance required pursuant to this Section 9.01(a)(iv) may be obtained by a Manager for the benefit of the applicable Borrower and the applicable Collateral Property.

(v) workers' compensation, subject to the statutory limits of the state in which each Collateral Property is located, and employer's liability insurance with a limit of at least \$1,000,000 per accident and per disease per employee, and \$1,000,000 for disease aggregate in respect of any work or operations on or about each Collateral Property, or in connection with such Collateral Property or its operation (if applicable);

(vi) comprehensive boiler and machinery insurance covering all mechanical and electrical equipment and boilers and pressure valves, if applicable, in amounts as shall be reasonably required by Administrative Agent on terms consistent with the commercial property insurance policy required under Section 9.01(a)(i);

(vii) if any portion of the Improvements is at any time located in an area identified by the Secretary of Housing and Urban Development or any successor thereto as an area having special flood hazards pursuant to the National Flood Insurance Act of 1968, the Flood Disaster Protection Act of 1973 or the National Flood Insurance Reform Act of 1994, as each may be amended, or any successor law (the "Flood Insurance Acts"), flood hazard insurance of the following types and in the following amounts (A) coverage under Policies issued pursuant to the Flood Insurance Acts (the "Flood Insurance Policies") in an amount equal to the maximum limit of coverage available for the applicable Collateral Property under the Flood Insurance Acts, subject only to customary deductibles under such Policies and (B) coverage under the Flood Act Policies in an amount, which when added to the coverage provided under the Flood Act Policies with respect to an Collateral Property, is not less than the Full Replacement Cost (as defined in <u>Section 9.01(a)</u> (i)) for such Collateral Property;

(viii) if required by Administrative Agent, earthquake, sinkhole and mine subsidence insurance in amounts as determined by Administrative Agent in its sole discretion and in form and substance satisfactory to Administrative Agent, provided that the insurance pursuant to this Section 9.01(a)(viii) shall be on terms consistent with the all risk insurance policy required under Section 9.01(a)(i);

(ix) umbrella liability insurance in an amount not less than 100,000,000 per occurrence on terms consistent with the commercial general liability insurance policy required under <u>Section 9.01(a)(ii)</u>;

insurance against terrorism, terrorist acts or similar acts of sabotage (x) ("Terrorism Insurance") pursuant to a (A) blanket insurance policy with aggregate limits of not less than the lesser of one hundred ten percent (110%) of Obligations, or \$50,000,000 or (B) a stand-alone insurance policy covering only the Collateral Properties with coverage of not less than \$50,000,000, and, in either case with a deductible of not more than \$100,000 (the "Terrorism Insurance Required Amount"). Notwithstanding the foregoing sentence, in the event a Borrower has obtained a stand alone insurance policy pursuant to subsection (B) above, such Borrower shall not be obligated to expend more than \$400,000 on Insurance Premiums for Terrorism Insurance (the "Terrorism Insurance Cap") and if the cost of the Terrorism Insurance Required Amount exceeds the Terrorism Insurance Cap, such Borrower shall purchase the maximum amount of Terrorism Insurance available with funds equal to the Terrorism Insurance Cap; provided, however, in the event it is customary among owners of Class A hotel properties in the United States to have "All Risk" coverage without any exclusion (a "Terrorism Exclusion") from coverage under such Policy for loss or damage incurred as a result of an act of terrorism, terrorist acts or similar acts of sabotage, such Borrower shall (provided the same does not add any material cost to such Borrower's Insurance Premiums) obtain a Policy without any such Terrorism Exclusion. After the occurrence of any event which reduces the amount of insurance available under the Terrorism Insurance required hereunder (whether due to a claim or otherwise), such Borrower shall be obligated to immediately increase the coverage of such Terrorism Insurance so that at least \$50,000,000 of coverage is available thereunder at all times;

(xi) a blanket fidelity bond and errors and omissions insurance coverage insuring against losses resulting from dishonest or fraudulent acts committed by (A) a Borrower's personnel; (B) any employees of outside firms that provide appraisal, legal, data processing or other services for a Borrower or (C) temporary contract employees or student interns; provided, however, the insurance required pursuant to this <u>Section 9.01(a)(xi)</u> may be obtained by a Manager for the benefit of the applicable Borrowers and the applicable Collateral Property; (xii) such other insurance and in such amounts as are required pursuant to any Franchise Agreement or as Administrative Agent from time to time may reasonably request against such other insurable hazards which at the time are commonly insured against for property similar to each Collateral Property located in or around the region in which the each Collateral Property is located; and

(b) All insurance provided for in Section 9.01(a) shall be obtained under valid and enforceable policies (the "Policies" or in the singular, the "Policy"), in such forms and, from time to time after the date hereof, in such amounts as may be satisfactory to Administrative Agent, issued by financially sound and responsible insurance companies authorized to do business in the state in which the property is located, and approved by Administrative Agent. All carriers providing insurance at any level shall have a claims paying ability rating of "A" (or its equivalent) or better by at least two (2) of the Rating Agencies including (i) S&P, (ii) Fitch, if Fitch is rating the Securities and rates the carrier and (iii) Moody's, if Moody's is rating the Securities and rates the carrier or such other Rating Agency approved by Administrative Agent, and a general policy rating of "A-X" or better by A.M. Best Company Inc. (each such insurer shall be referred to as a "Qualified Insurer"). Prior to Closing and not less than thirty (30) days prior to the expiration dates of the Policies theretofore furnished to Administrative Agent pursuant to Section 9.01(a), Borrowers shall deliver, electronically or by hard copy, certificates of insurance in form and substance acceptable to Administrative Agent evidencing the coverages required herein Section 9.01 and certified copies of the Policies marked "premium paid" or accompanied by evidence satisfactory to Administrative Agent of payment of the premiums due thereunder (the "Insurance Premiums"). In the event any Borrower desires to obtain the insurance required hereunder from an insurer not meeting the requirements of this Section 9.01(b), such Borrower may request, in writing, Administrative Agent's approval of such insurer, which approval may not be unreasonably withheld.

(c) Borrowers shall not obtain (i) any umbrella or blanket liability or casualty Policy unless, in each case, such Policy is at least equal in scope of coverage as if a "standalone" Policy meeting all of the requirement noted above is provided as such Policy is approved in advance in writing by Administrative Agent and Administrative Agent's interest is included therein as provided in this Agreement and such Policy is issued by a Qualified Insurer, or (ii) separate insurance concurrent in form or contributing in the event of loss with that required in Section 9.01(a) to be furnished by, or which may be reasonably required to be furnished by, a Borrower. In the event any Borrower obtains separate insurance or an umbrella or a blanket policy, such Borrower shall notify Administrative Agent of the same and shall cause certified copies of each Policy to be delivered as required in Section 9.01(a). Any blanket insurance Policy shall specifically allocate to the Collateral Property the amount of coverage from time to time required hereunder and shall otherwise provide the same protection as would a separate Policy insuring only the Collateral Property in compliance with the provisions of Section 9.01 (a). Notwithstanding Administrative Agent's approval of any umbrella or blanket liability or casualty Policy hereunder, Administrative Agent reserves the right, in its sole discretion, to require any Borrower to obtain a separate Policy in compliance with this Section 9.01.

(d) All liability Policies provided for or contemplated by <u>Section 9.01(a)</u> shall name Borrowers as the named insured and, except for the Policy referenced in Section 9.01 (a)(v), Administrative Agent on behalf of Lenders and Borrowers as the insured or additional insured, as their respective interests may appear, and in the case of Property policies, including but not limited to all-risk property, terrorism, boiler and machinery, earthquake and flood insurance, shall contain a standard non-contributory mortgagee clause in favor of Administrative Agent providing that the loss thereunder shall be payable to Administrative Agent.

(e) All Policies provided for in <u>Section 9.01(a)</u> shall contain clauses or endorsements to the effect that:

(i) no act or negligence of Borrowers, or anyone acting for Borrowers, or failure to comply with the provisions of any Policy which might otherwise result in a forfeiture of the insurance or any part thereof, shall in any way affect the validity or enforceability of the insurance insofar as Administrative Agent is concerned;

(ii) the Policy shall not be materially changed (other than to increase the coverage provided thereby) or cancelled without at least thirty (30) days' written notice to Administrative Agent and any other party named therein as an insured;

(iii) each Policy shall provide that the issuers thereof shall give written notice to Administrative Agent if the Policy has not been renewed thirty (30) days prior to its expiration; and

(iv) Administrative Agent and Lenders shall not be liable for any Insurance Premiums thereon or subject to any assessments thereunder.

(f) Borrowers shall furnish to Administrative Agent, on or before thirty (30) days after the close of Borrowers' Fiscal Year, a statement certified by Borrowers or a duly authorized officer of Borrowers of the amounts of insurance maintained in compliance herewith, of the risks covered by such insurance and of the insurance company or companies which carry such insurance and, if requested by Administrative Agent, verification of the adequacy of such insurance by an independent insurance broker or appraiser acceptable to Administrative Agent.

(g) If at any time Administrative Agent is not in receipt of written evidence that all insurance required hereunder is in full force and effect, Administrative Agent shall have the right, with two (2) Business Days prior notice to Borrowers (unless the giving of such notice would result in there being less than five (5) Business Days remaining until the termination of any insurance coverage, in which case Administrative Agent shall have the immediate right, with notice to Borrowers), to take such action as Administrative Agent deems necessary to protect its interest in the Collateral Properties, including, without limitation, the obtaining of such insurance coverage as Administrative Agent in its sole discretion deems appropriate, and all expenses incurred by Administrative Agent in connection with such action or in obtaining such insurance and keeping it in effect shall be paid by Borrowers to Administrative Agent upon demand and until paid shall be secured by the Mortgages and shall bear interest at the Default Rate. (h) In the event of a foreclosure of any of the Mortgages, or other transfer of title to any Collateral Property in extinguishment in whole or in part of the Obligations all right, title and interest of Borrowers in and to the Policies then in force and all proceeds payable thereunder shall thereupon vest in the purchaser at such foreclosure or Administrative Agent or other transferee in the event of such other transfer of title.

9.02. <u>Casualty</u>.

If a Collateral Property shall be damaged or destroyed, in whole or in part, by fire or other casualty resulting in an amount greater than \$2,000,000 to repair (a "<u>Casualty</u>"), Borrowers shall give prompt notice of such damage to Administrative Agent and the Lenders and shall promptly commence and diligently prosecute the completion of the Restoration of the Collateral Property as nearly as possible to the condition the Collateral Property was in immediately prior to such Casualty, with such alterations as may be reasonably approved by Administrative Agent and otherwise in accordance with <u>Section 9.04</u>. Borrowers shall pay all costs of such Restoration whether or not such costs are covered by insurance. Administrative Agent may, but shall not be obligated to make proof of loss if not made promptly by Borrowers.

9.03. Condemnation.

Borrowers shall promptly give Administrative Agent and the Lenders notice of the actual or threatened commencement of any proceeding for the Condemnation of all or any part of any Collateral Property and shall deliver to Administrative Agent copies of any and all papers served in connection with such proceedings. Administrative Agent may participate in any such proceedings, and Borrowers shall from time to time deliver to Administrative Agent all instruments requested by it to permit such participation. Each Borrower shall, at its expense, diligently prosecute any such proceedings, and shall consult with Administrative Agent, its attorneys and experts, and cooperate with them in the carrying on or defense of any such proceedings. Notwithstanding any taking by any public or quasi-public authority through Condemnation or otherwise (including, but not limited to, any transfer made in lieu of or in anticipation of the exercise of such taking), Borrowers shall continue to pay the Obligations at the time and in the manner provided for its payment in the Notes and in this Agreement and the Obligations shall not be reduced until any Award shall have been actually received and applied by Administrative Agent, after the deduction of expenses of collection, to the reduction or discharge of the Obligations. Administrative Agent shall not be limited to the interest paid on the Award by the condemning authority but shall be entitled to receive out of the Award interest at the rate or rates provided herein or in the Notes. If any Collateral Property or any portion thereof is taken by a condemning authority, Borrowers shall, promptly commence and diligently prosecute the Restoration of the applicable Collateral Property and otherwise comply with the provisions of Section 9.04. If any Collateral Property is sold, through foreclosure or otherwise, prior to the receipt by Administrative Agent of the Award, Administrative Agent shall have the right, whether or not a deficiency judgment on the Obligations shall have been sought, recovered or denied, to receive the Award, or a portion thereof sufficient to pay the Obligations.

9.04. <u>Restoration</u>.

The following provisions shall apply in connection with the Restoration of any Collateral Property:

(a) If the Net Proceeds shall be less than 2,000,000 and the costs of completing the Restoration shall be less than 2,000,000, the Net Proceeds will be disbursed directly to a Borrower, provided that all of the conditions set forth in <u>Section 9.04(c)(i)</u> and (iii) are met and Borrowers deliver to Administrative Agent a written undertaking to expeditiously commence and to satisfactorily complete with due diligence the Restoration in accordance with the terms of this Agreement and a written confirmation of satisfaction of such conditions.

(b) If the Net Proceeds are equal to or greater than \$2,000,000 or the costs of completing the Restoration is equal to or greater than \$2,000,000, Administrative Agent shall make the Net Proceeds available for the Restoration in accordance with the provisions of this Section 9.04. Notwithstanding the immediately preceding sentence, a portion of the Net Proceeds not to exceed \$2,000,000 shall be made available to Borrowers after receipt thereof by Administrative Agent to pay or reimburse a Borrower for any immediate and necessary repair or work required (i) to prevent further damage to the Collateral Property, (ii) to protect life or to provide safety or (iii) to restore hotel operations at the Collateral Property (collectively, the "Emergency Repairs") provided all the conditions set forth in Sections 9.04(c)(i) and (iii) met and Borrowers deliver to Administrative Agent a written undertaking to expeditiously commence and to satisfactorily complete with due diligence the Emergency Repairs in accordance with the terms of this Agreement. The term "Net Proceeds" means: (A) the net amount of all insurance proceeds received by Administrative Agent pursuant to Section 9.01 (a)(i), (iv), (vii) and (viii) as a result of such damage or destruction, after deduction of its reasonable costs and expenses (including, but not limited to, reasonable counsel fees), if any, in collecting same ("Insurance Proceeds"), or (B) the net amount of the Award, after deduction of its reasonable costs and expenses (including, but not limited to, reasonable counsel fees), if any, in collecting same ("Condemnation Proceeds"), whichever the case may be.

(c) The Net Proceeds shall be made available to a Borrower for Restoration provided that each of the following conditions are met:

(i) no Default or Event of Default (unless caused solely by the Condemnation or Casualty) shall have occurred and be continuing;

(ii) (1) in the event the Net Proceeds are Insurance Proceeds, less than thirty percent (30%) of the total floor area of the Improvements on the particular Collateral Property has been damaged, destroyed or rendered unusable as a result of such Casualty or (2) in the event the Net Proceeds are Condemnation Proceeds, less than fifteen percent (15%) of the land constituting the particular Collateral Property is taken, and such land is located along the perimeter or periphery of the Collateral Property, and no portion of the Improvements is located on such land;

(iii) The respective Operating Lease shall remain in full force and effect during and after the completion of the Restoration;

(iv) Such Borrower shall commence the Restoration as soon as reasonably practicable (but in no event later than thirty (30) days after such Casualty or Condemnation, whichever the case may be, occurs) and shall diligently pursue the same to satisfactory completion in compliance with all applicable Laws, including, without limitation, all applicable Environmental Laws and in accordance with the terms and conditions of the applicable Franchise Agreement;

(v) Administrative Agent shall be satisfied that any operating deficits, including all scheduled payments of principal and interest on the Loans under this Agreement, which will be incurred with respect to the Collateral Property as a result of the occurrence of any such Casualty or Condemnation, whichever the case may be, will be covered out of (A) the Net Proceeds, (B) the insurance coverage referred to in Section 9.01(a)(iii) if applicable, or (C) by other funds of Borrowers;

(vi) Administrative Agent shall be satisfied that the Restoration will be completed (with reasonable allowance for time to make such repairs) on or before the earliest to occur of (A) twelve (12) months after the occurrence of such Casualty or Condemnation, or (B) the earliest date required for such completion under the terms of any Leases which are required in accordance with the provisions of this Section 9.04 to remain in effect subsequent to the occurrence of such Casualty or Condemnation and the completion of the Restoration, or (C) the date required for such completion pursuant to the applicable Franchise Agreement, or (D) such time as may be required under applicable Law, in order to repair and restore the applicable Collateral Property to the condition it was in immediately prior to such Casualty or Condemnation or (E) the expiration of the insurance coverage referred to in Section 9.01(a)(ii);

(vii) the Collateral Property and the use thereof after the Restoration will be in compliance with and permitted under all applicable Laws;

(viii) Administrative Agent shall be satisfied that the Debt Service Coverage Ratio for the four (4) fiscal quarters immediately succeeding the completion of the Restoration shall be equal to or greater than 1.25 to 1 on a pro forma basis;

(ix) Such Casualty or Condemnation, as applicable, does not result in the loss of access in any material respect to the Collateral Property or the related Improvements;

(x) Borrowers shall deliver, or cause to be delivered, to Administrative Agent a signed detailed scope of damages and repairs, including cost estimates for each repair item, approved in writing by Borrowers' architect, engineer or project manager stating the entire cost of completing the Restoration, which scope shall be acceptable to Administrative Agent; (xi) the Net Proceeds together with any cash or Permitted Investments deposited by Borrowers with Administrative Agent are sufficient in Administrative Agent's discretion to cover the cost of the Restoration;

(xii) any applicable Management Agreement in effect as of the date of the occurrence of such Casualty or Condemnation, whichever the case may be, shall (A) remain in full force and effect during the Restoration and shall not otherwise terminate as a result of the Casualty or Condemnation or the Restoration or (B) if terminated, shall have been replaced with a Replacement Management Agreement with a Qualified Manager, prior to the opening or reopening of the applicable Collateral Property or any portion thereof for business with the public; and

(xiii) any applicable Franchise Agreement is not terminated as a result of such Casualty or Condemnation.

(d) The Net Proceeds shall be held by Administrative Agent in an interest-bearing account and, until disbursed in accordance with the provisions of this <u>Section 9.04</u>, shall constitute additional security for the Obligations and other obligations under the Loan Documents. The Net Proceeds shall be disbursed by Administrative Agent to, or as directed by, such Borrower from time to time during the course of the Restoration, upon receipt of evidence satisfactory to Administrative Agent that (i) all materials installed and work and labor performed (except to the extent that they are to be paid for out of the requested disbursement) in connection with the Restoration have been paid for in full, and (ii) there exist no notices of pendency, stop orders, mechanic's or materialman's liens or notices of intention to file same, or any other Liens or encumbrances of any nature whatsoever on the Collateral Property which have not either been fully bonded to the satisfaction of Administrative Agent and discharged of record or in the alternative fully insured to the satisfaction of Administrative Agent by the title company issuing the Title Policy.

All plans and specifications required in connection with a Restoration, where (e) the cost of such Restoration is greater than \$2,000,000 shall be subject to prior review and acceptance in all respects by Administrative Agent, and, at the option of Administrative Agent, an independent consulting engineer selected by Administrative Agent (the "Casualty Consultant"), which acceptance shall not be unreasonably withheld, conditioned or delayed in either case. Administrative Agent shall have the use of the plans and specifications and all permits, licenses and approvals required or obtained in connection with the Restoration. The identity of the contractors, subcontractors and materialmen engaged in the Restoration as well as the contracts under which they have been engaged where the cost of such Restoration is greater than \$2,000,000 be subject to prior review and acceptance by Administrative Agent, and, at the option of Administrative Agent, and the Casualty Consultant, which acceptance shall not be unreasonably withheld, conditioned or delayed in either case. Notwithstanding the foregoing, Administrative Agent's prior review and acceptance of plans and specifications, identity of contractors and contracts shall not be required in connection with Restoration which shall be required on an emergency basis (i) to prevent further damage to the Collateral Property, (ii) to protect life or to provide safety or (iii) to restore hotel operations at the Collateral Property; provided that in such cases, such plans, specifications, identity of contractors and contracts shall be made available to Administrative Agent and the Casualty Consultant as soon as practicable. All costs and expenses incurred by Administrative Agent in connection with making the Net Proceeds available for the Restoration including, without limitation, reasonable counsel fees and disbursements and the Casualty Consultant's fees, shall be paid by Borrowers.

(f) In no event shall Administrative Agent be obligated to make disbursements of the Net Proceeds in excess of an amount equal to the costs actually incurred from time to time for work in place as part of the Restoration, as certified by the Casualty Consultant, minus the Casualty Retainage. The term "Casualty Retainage" shall mean an amount equal to ten percent (10%), of the costs actually incurred for work in place as part of the Restoration, as certified by the Casualty Consultant, until the Restoration has been completed. The Casualty Retainage shall in no event, and notwithstanding anything to the contrary set forth above in this Section 9.04(f), be less than the amount actually held back by Borrowers from contractors, subcontractors and materialmen engaged in the Restoration. The Casualty Retainage shall not be released until the Casualty Consultant certifies to Administrative Agent that the Restoration has been completed in accordance with the provisions of this Section 9.04 and that all approvals necessary for the re-occupancy and use of the Collateral Property have been obtained from all appropriate Governmental Authorities, and Administrative Agent receives evidence satisfactory to Administrative Agent that the costs of the Restoration have been paid in full or will be paid in full out of the Casualty Retainage; provided, however, that Administrative Agent will release the portion of the Casualty Retainage being held with respect to any contractor, subcontractor or materialman engaged in the Restoration as of the date upon which the Casualty Consultant certifies to Administrative Agent that the contractor, subcontractor or materialman has satisfactorily completed all work and has supplied all materials in accordance with the provisions of the contractor's, subcontractor's or materialman's contract, the contractor, subcontractor or materialman delivers the lien waivers and evidence of payment in full of all sums due to the contractor, subcontractor or materialman as may be reasonably requested by Administrative Agent or by the title company issuing the Title Policy for the related Collateral Property, and Administrative Agent receives an endorsement to such Title Policy insuring the continued priority of the Lien of the related Mortgage and evidence of payment of any premium payable for such endorsement. If required by Administrative Agent, the release of any such portion of the Casualty Retainage shall be approved by the surety company, if any, which has issued a payment or performance bond with respect to the contractor, subcontractor or materialman.

(g) Administrative Agent shall not be obligated to make disbursements of the Net Proceeds more frequently than once every calendar month.

(h) If at any time the Net Proceeds or the undisbursed balance thereof shall not, in the opinion of Administrative Agent in consultation with the Casualty Consultant, if any, be sufficient to pay in full the balance of the costs which are estimated by the Casualty Consultant to be incurred in connection with the completion of the Restoration, Borrowers shall deposit the deficiency (the "<u>Net Proceeds Deficiency</u>") with Administrative Agent before any further disbursement of the Net Proceeds shall be made. The Net Proceeds Deficiency deposited with Administrative Agent shall be held by Administrative Agent and shall be disbursed for costs actually incurred in connection with the Restoration on the same conditions applicable to the disbursement of the Net Proceeds, and until so disbursed pursuant to <u>Section 9.04(b)</u> shall constitute additional security for the Obligations and other obligations under the Loan Documents.

(i) The excess, if any, of the Net Proceeds and the remaining balance, if any, of the Net Proceeds Deficiency deposited with Administrative Agent after the Casualty Consultant certifies to Administrative Agent that the Restoration has been completed in accordance with the provisions of <u>Section 9.04(b)</u>, and the receipt by Administrative Agent of evidence satisfactory to Administrative Agent that all costs incurred in connection with the Restoration have been paid in full, shall be promptly remitted by Administrative Agent to Borrowers, provided no Default or Event of Default shall have occurred and shall be continuing under the Notes, this Agreement or any of the other Loan Documents.

(j) All Net Proceeds not required (i) to be made available for the Restoration or (ii) to be returned to Borrowers as excess Net Proceeds pursuant to <u>Section 9.04(i)</u>, may be at the discretion of Administrative Agent (x) retained and applied by Administrative Agent toward the prepayment of the Obligations pursuant to Section 2.09(b)(ii), or (y), paid, either in whole or in part, to Borrowers for such purposes as Administrative Agent shall approve, in its discretion. Any Net Proceeds remaining with the Administrative Agent after prepayment of all then outstanding Obligations pursuant to clause (x) above shall be paid to Borrowers. If the subject Casualty or Condemnation shall result in (or could reasonably be expected to result in) a Material Property Event with respect to such Collateral Property, and the Administrative Agent shall have caused the Loans to be repaid pursuant to clause (x) above, the Aggregate Commitment shall be reduced by the amount of such prepayment as set forth in <u>Section 2.07</u> (c). If the Aggregate Commitments are reduced, the Lien of the Mortgages shall be reduced by such amount.

ARTICLE X

EVENTS OF DEFAULT AND REMEDIES

10.01. <u>Events of Default</u>. Any of the following shall constitute an Event of Default ("<u>Event</u> <u>of Default</u>"):

(a) if any portion of the Obligations are not paid on or before the date the same is due and payable; provided, however, with respect to any failure by Borrowers to pay any regularly scheduled installment of interest hereunder when due (other than any installment of interest or any other amount due on the Maturity Date), and up to two (2) times during any twelve month period, Borrowers shall have two (2) Business Days in which to cure such Event of Default, after which time Administrative Agent may, at its option, exercise any of its rights and remedies available to it pursuant to this Agreement and the other Loan Documents or at law or in equity;

(b) if any of the Taxes or Other Charges are not paid prior to delinquency;

(c) if (i) the Policies are not kept in full force and effect or if (ii) certified copies of the Policies are not delivered to Administrative Agent promptly on request;

(d) if any representation or warranty made by any Borrower or any other Loan Party herein or in any other Loan Document, or in any report, certificate, financial statement or other instrument, agreement or document furnished to Administrative Agent or Lenders shall have been false or misleading in any material respect as of the date the representation or warranty was made;

(e) if a Transfer occurs in violation of the provisions of Section 8.19 of this Agreement or <u>Article 5</u> of the Mortgages;

(i) if a receiver, liquidator or trustee shall be appointed for any Borrower or (f) the Philadelphia Affiliate Ground Lessor; or if any Borrower or the Philadelphia Affiliate Ground Lessor shall be adjudicated a bankrupt or insolvent, or if any petition for bankruptcy, reorganization or arrangement pursuant to the Bankruptcy Code, or any similar federal or State law, shall be filed by or against, consented to, or acquiesced in by any Borrower or the Philadelphia Affiliate Ground Lessor, or if any proceeding for the dissolution or liquidation of any Borrower or the Philadelphia Affiliate Ground Lessor shall be instituted, or if any event resulting in the dissolution of any Borrower or the Philadelphia Affiliate Ground Lessor shall occur; provided, however, if such appointment, adjudication, petition or proceeding was involuntary and not consented to by any Borrower or the Philadelphia Affiliate Ground Lessor, upon the same not being discharged, stayed or dismissed within sixty (60) days; or (ii) other than pursuant to a "pre-packaged" proceeding in which (a) the Guarantor's plan or reorganization provides that each Guaranty shall be binding upon, and enforceable against, the reorganized Guarantor so that upon emergence of the Guarantor from the bankruptcy reorganization case, each Guaranty shall be a legal, valid and binding obligation enforceable against the Guarantor in accordance with its terms, and (b) the material terms of the Guarantor's plan of reorganization (1) have been negotiated and agreed by the Guarantor and the necessary percentages of its senior noteholders and other material creditors prior to the filing of such petition or (2) are being diligently negotiated by the Guarantor with the necessary percentages of its senior noteholders and other material creditors and are agreed within 90 days after the filing of such petition, and an order confirming a plan of reorganization which incorporates clauses (a) and (b) is entered within one hundred eighty (180) days of the filing of such petition and within ninety (90) days of the filing of such plan of reorganization (or, in each case, such longer period as shall be permitted by Administrative Agent in its reasonable discretion), the Guarantor shall (A) file a petition under Section 301 of the Bankruptcy Code or otherwise voluntarily commence any proceeding or file any petition seeking liquidation, reorganization or other relief under any similar Federal, state or foreign bankruptcy, insolvency, receivership or similar law now or hereafter in effect, (B) consent to the institution of, or fail to contest successfully within ninety (90) days in an appropriate manner, any involuntary proceeding seeking liquidation, reorganization or other relief in respect the Guarantor or its debts, or of a substantial part of its assets, under any Federal, state or foreign bankruptcy, insolvency, receivership or similar law now or hereafter in effect, (C) apply for or consent to the appointment of a receiver, trustee, custodian, sequestrator, conservator or similar official for the Guarantor or for a substantial part of its assets, (D) file an answer admitting the material allegations of a petition filed against it in any such proceeding, (E) make a general assignment for the benefit of creditors or (F) take any action for the purpose of effecting any of the foregoing;

(g) if any (i) Borrower or Guarantor attempts to assign its rights under this Agreement or any of the other Loan Documents or any interest herein or therein in contravention of the Loan Documents or (ii) Borrower or the Philadelphia Affiliate Ground Lessor shall make an assignment for the benefit of creditors;

(h) other than with respect to a default which is expressly contemplated by another subsection of this Section 10.01, if Borrowers breach their affirmative covenant in the first sentence of Section 7.13(e) or if any Borrower breaches any of its respective negative covenants contained in Sections 8.01 through 8.09, 8.15 through 8.19.

(i) if any Borrower violates or does not comply in any material respect with any of the provisions of <u>Section 7.16</u>, that results in a Material Adverse Effect;

(j) if a (i) default has occurred and continues beyond any applicable cure period under any Management Agreement (or any Replacement Management Agreement) if such default permits the Manager thereunder to terminate or cancel such Management Agreement (or any Replacement Management Agreement) or (ii) any Management Agreement (or Replacement Management Agreement) expires or otherwise terminates and is not replaced with a Replacement Management Agreement or (iii) if any Collateral Property operates for any time without the Management Agreement or a Replacement Management Agreement;

(k) if any Borrower violates or does not comply with the provisions of Section 6.34; provided, however, that such violation or failure to comply shall not constitute an Event of Default if (i) such violation or failure to comply is immaterial, and (ii) to the extent such violation or failure to comply is curable, Borrowers shall have promptly cured such violation or failure to comply within fifteen (15) calendar days of the earlier of (a) notice from Administrative Agent and (b) the date any Borrower or other Loan Party becomes aware of such violation or failure to comply,;

(1) if any federal tax Lien or state or local income tax Lien in excess of \$1,500,000 is filed against any Borrower, any Guarantor or any Collateral Property and same is not discharged of record (by payment or by posting of a bond or cash collateral) within thirty (30) days after same is filed;

(m) There is entered against any Borrower or the Philadelphia Affiliate Ground Lessor (i) a final judgment or order for the payment of money in an aggregate amount exceeding \$1,500,000 (which is not covered by insurance or unless being appealed and such Loan Party has posted a bond or cash collateral), or (ii) any one or more non-monetary final judgments that have, or could reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect and, in either case, (A) enforcement proceedings are commenced by any creditor upon such judgment or order, or (B) there is a period of ten (10)

consecutive days during which a stay of enforcement of such judgment, by reason of a pending appeal or otherwise, is not in effect;

(n) (i) any Borrower or the Philadelphia Affiliate Ground Lessor is a Plan or its assets constitute Plan Asset; or (ii) any Borrower or the Philadelphia Affiliate Ground Lessor consummates a transaction which would cause the Mortgages or Administrative Agent or any Lender's exercise of its rights under the Mortgages, the Notes, this Agreement or the other Loan Documents to constitute a nonexempt prohibited transaction under ERISA or result in a violation of a state statute regulating governmental plans, subjecting Lenders to liability for a violation of ERISA, the Code, a state statute or other similar law;

(o) if any default occurs under any guaranty or indemnity executed in connection herewith (including, without limitation, the Guaranty and the Environmental Indemnity) and such default continues after the expiration of applicable grace periods, if any;

(p) (i) if any Borrower or Affiliate Ground Lessor shall be in default in any material respect beyond applicable notice and grace periods under any other mortgage, deed of trust, deed to secure debt or other security agreement permitted hereunder and covering any part of any Collateral Property whether it be superior or junior in lien to the related Mortgage;

(ii) any event or condition occurs that results in any Material Indebtedness (other than Indebtedness under the Senior Secured Notes Indenture) becoming due prior to its scheduled maturity or that enables or permits (with or without the giving of notice, the lapse of time or both) the holder or holders of such Material Indebtedness or any trustee or agent on its or their behalf to cause such Material Indebtedness to become due, or to require the prepayment, repurchase, redemption or defeasance thereof, prior to its scheduled maturity; <u>provided</u> that this clause (p) (ii) shall not apply to secured Indebtedness that becomes due as a result of the voluntary sale or transfer of the property or assets securing such Indebtedness;

(iii) any event or condition occurs that (x) results in the Senior Secured Notes becoming due prior the scheduled maturity thereof, or (y) that results in the existence of an "Event of Default" under (and as defined in) the Senior Secured Notes Indenture or (z) requires the prepayment, repurchase, redemption or defeasance of the Senior Secured Notes, prior to the scheduled maturity thereof; <u>provided</u> that this clause (p) (iii) shall not apply to Senior Secured Notes that become due as a result of the voluntary sale or transfer of the property or assets securing such Indebtedness, any Offer to Purchase pursuant to and as defined in the Senior Secured Notes Indenture, or any other voluntary prepayment, repurchase, redemption or defeasance of the Senior Secured Notes; or (iv) Guarantor and its Subsidiaries shall fail to comply with the covenants set forth in Sections 4.03 [*Limitations on Indebtedness*], 4.04 [*Limitations on Restricted Payments*], and 4.05 [*Limitation on Dividend and Other Payment Restrictions Affecting Restricted Subsidiaries*] of the Senior Secured Note Indenture (together with all definitions used therein) in each case as in effect as of the Closing Date, without giving effect to any future amendment, modification or termination thereof unless, in the case of an amendment or modification such revision (x) imposes terms or conditions that are more restrictive on FelCor Trust, FelCor Op and their Subsidiaries, or (y) the Required Lenders have given their prior written consent to such revision (an amendment or modification satisfying (x) or (y), a "Permitted Indenture Amendment"); provided that for so long as there have been no amendments or modifications or the only amendments, no Event of Default shall be deemed to occur under this clause (iv) unless and until an Event of Default has occurred under clause (iii) above;

(q) with respect to any term, covenant or provision set forth herein which specifically contains a notice requirement or grace period, if any Borrower shall be in default under such term, covenant or condition after the giving of such notice or the expiration of such grace period;

(r) if any default occurs under an Operating Lease Subordination Agreement, and such default continues after the expiration of applicable grace periods, if any;

(s) if (i) a material default has occurred and continues beyond any applicable cure period under any Franchise Agreement, and such default permits a party to terminate or cancel any Franchise Agreement or (ii) any Franchise Agreement expires or otherwise terminates without Administrative Agent's prior written consent and is not replaced with a Replacement Franchise Agreement;

(t) if any Borrower ceases to operate a hotel on any Collateral Property or terminates such business for any reason whatsoever (other than temporary cessation in connection with any renovations to an Collateral Property or restoration of the Collateral Property after Casualty or Condemnation);

(u) if there shall occur any default by any Ground Lessee, as tenant under any Ground Lease, in the observance or performance of any material term, covenant or condition of such Ground Lease on the part of any Ground Lessee to be observed or performed and said default is not cured following the expiration of any applicable grace and notice periods therein provided, or if the leasehold estate created by any Ground Lease shall be surrendered or if any Ground Lease shall cease to be in full force and effect or any Ground Lease shall be terminated or canceled for any reason or under any circumstances whatsoever, or if any of the material terms, covenants or conditions of any Ground Lease shall in any manner be modified, changed, supplemented, altered, or amended without the consent of Administrative Agent;

(v) if there shall occur any default by either Operating Lessee, under any Operating Lease, in the observance or performance of any material term, covenant or condition of any Operating Lease on the part of either Operating Lessee to be observed or performed and said default is not cured following the expiration of any applicable grace and notice periods therein provided, or if the leasehold estate created by the any Operating Lease shall be surrendered or if any Operating Lease shall cease to be in full force and effect or any Operating Lease shall be terminated or canceled for any reason or under any circumstances whatsoever, or if any of the material terms, covenants or conditions of any Operating Lease shall in any manner be modified, changed, supplemented, altered, or amended without the consent of Administrative Agent;

(w) if any Borrower shall continue to be in Default under any of the other terms, covenants or conditions of this Agreement not specified in (a) - (w) above, for ten (10) Business Days after written notice to Borrowers from Administrative Agent, in the case of any Default which can be cured by the payment of a sum of money, or for thirty (30) days after written notice from Administrative Agent in the case of any other Default; provided, however, that if such non-monetary Default is susceptible of cure but cannot reasonably be cured within such 30-day period and provided further that Borrowers shall have commenced to cure such Default within such thirty (30) day period and thereafter diligently and expeditiously proceeds to cure the same, such thirty (30) day period shall be extended for such time as is reasonably necessary for Borrowers in the exercise of due diligence to cure such Default, such additional period not to exceed sixty (60) days; or

(x) if there shall occur any default by any Borrower or any other Loan Party, or any default in respect of any Collateral Property, under the Mortgages or any of the other Loan Documents beyond any applicable notice and cure periods as contained in such documents.

10.02. <u>Remedies upon Event of Default</u>. If any Event of Default occurs and is continuing, Administrative Agent shall, at the request of, or may, with the consent of, the Required Lenders, take any or all of the following actions:

(a) declare the unpaid principal amount of all outstanding Loans, all interest accrued and unpaid thereon, and all other Obligations to be immediately due and payable, without presentment, demand, protest, notice of protest and non-payment, or other notice of default, notice of acceleration and intention to accelerate or other notice of any kind, all of which are hereby expressly waived by Borrowers; and

(b) exercise on behalf of itself and the Lenders all rights and remedies available to it and the Lenders under the Loan Documents;

provided, however, that upon the occurrence of an actual or deemed entry of an order for relief with respect to any Borrower under any Debtor Relief Laws, the unpaid principal amount of all outstanding Loans and all interest and other Obligations shall automatically become due and payable without further act of Administrative Agent or any Lender.

10.03. <u>Application of Funds</u>. After the exercise of remedies provided for in <u>Section 10.02</u> (or after the Loans have automatically become immediately due and payable), any amounts received on account of the Obligations shall be applied by Administrative Agent in the following order:

<u>First</u>, to payment of that portion of the Obligations constituting fees, indemnities, expenses and other amounts (including fees, charges and disbursements of counsel to Administrative Agent and amounts payable under Article IV) payable to Administrative Agent in its capacity as such;

<u>Second</u>, to payment of that portion of the Obligations constituting fees, indemnities and other amounts (other than principal and interest) payable to the Lenders (including fees, charges and disbursements of counsel to the respective Lenders and including fees and time charges for attorneys who may be employees of any Lender) and amounts payable under <u>Article IV</u>, ratably among them in proportion to the respective amounts described in this clause <u>Second</u> payable to them;

<u>Third</u>, to payment of that portion of the Obligations constituting accrued and unpaid and interest on the Loans and other Obligations, ratably among the Lenders in proportion to the respective amounts described in this clause <u>Third</u> payable to them;

<u>Fourth</u>, to payment of that portion of the Obligations constituting unpaid principal of the Loans ratably among the Lenders in proportion to the respective unpaid principal amounts of the Loans held by them;

Last, the balance, if any, after all of the Obligations have been indefeasibly paid in full, to Borrowers or as otherwise required by Law.

ARTICLE XI ADMINISTRATIVE AGENT

11.01. <u>Appointment and Authority</u>. (a) Each Lender hereby irrevocably appoints JPMC to act on its behalf as Administrative Agent hereunder and under the other Loan Documents and authorizes Administrative Agent to take such actions on its behalf and to exercise such powers as are delegated to Administrative Agent by the terms hereof or thereof, together with such actions and powers as are reasonably incidental thereto. The provisions of this Article are solely for the benefit of Administrative Agent and Lenders and no Borrower shall have rights as a third party beneficiary of any of such provisions.

(b) Administrative Agent shall also act as the "<u>Collateral Agent</u>" under the Loan Documents, and each Lender hereby irrevocably appoints and authorizes Administrative Agent to act as the agent of such Lender for purposes of acquiring, holding and enforcing any and all Liens on Collateral granted by any of the Loan Parties to secure any of the Obligations, together with such powers and discretion as are reasonably incidental thereto. In this connection, Administrative Agent, as "collateral agent" and any co-agents, sub-agents and attorneys-in-fact appointed by Administrative Agent pursuant to Section 11.05 for purposes of holding or enforcing any Lien on the Collateral (or any portion thereof) granted under the Collateral Documents, or for exercising any rights and remedies thereunder at the direction of Administrative Agent, shall be entitled to the benefits of all provisions of this

<u>Article XI and Article XII (including Section 12.04(c)</u>, as though such co-agents, sub-agents and attorneys-in-fact were the "collateral agent" under the Loan Documents) as if set forth in full herein with respect thereto.

11.02. <u>Rights as a Lender</u>. The Person serving as Administrative Agent hereunder shall have the same rights and powers in its capacity as a Lender as any other Lender and may exercise the same as though it were not Administrative Agent and the term "Lender" or "Lenders" shall, unless otherwise expressly indicated or unless the context otherwise requires, include the Person serving as Administrative Agent hereunder in its individual capacity. Such Person and its Affiliates may accept deposits from, lend money to, act as the financial advisor or in any other advisory capacity for and generally engage in any kind of business with any Borrower or Affiliate thereof as if such Person were not Administrative Agent hereunder and without any duty to account therefor to the Lenders.

11.03. <u>Exculpatory Provisions</u>. Administrative Agent shall not have any duties or obligations except those expressly set forth herein and in the other Loan Documents. Without limiting the generality of the foregoing, Administrative Agent:

(a) shall not be subject to any fiduciary or other implied duties, regardless of whether a Default has occurred and is continuing;

(b) shall not have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated hereby or by the other Loan Documents that Administrative Agent is required to exercise as directed in writing by the Required Lenders (or such other number or percentage of Lenders as shall be expressly provided for herein or in the other Loan Documents), provided that Administrative Agent shall not be required to take any action that, in its opinion or the opinion of its counsel, may expose Administrative Agent to liability or that is contrary to any Loan Document or applicable law; and

(c) shall not, except as expressly set forth herein and in the other Loan Documents, have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to Borrowers or any of their Affiliates that is communicated to or obtained by the Person serving as Administrative Agent or any of its Affiliates in any capacity.

Administrative Agent shall not be liable for any action taken or not taken by it (i) with the consent or at the request of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary, or as Administrative Agent shall believe in good faith shall be necessary, under the circumstances as provided in <u>Sections 12.01</u> and <u>10.02</u>) or (ii) in the absence of its own gross negligence or willful misconduct. Administrative Agent shall be deemed not to have knowledge of any Default unless and until notice describing such Default is given to Administrative Agent by Borrowers or a Lender.

Administrative Agent shall not be responsible for or have any duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with this Agreement or any other Loan Document, (ii) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith, (iii) the performance or observance of any of the covenants, agreements or other terms or conditions set forth herein or therein or the occurrence of any Default, (iv) the validity, enforceability, effectiveness or genuineness of this Agreement, any other Loan Document or any other agreement, instrument or document, or the creation, perfection or priority of any Lien purported to be created by the Collateral Documents, (v) the value or the sufficiency of any Collateral, or (vi) the satisfaction of any condition set forth in <u>Article V</u> or elsewhere herein, other than to confirm receipt of items expressly required to be delivered to Administrative Agent.

11.04. <u>Reliance by Administrative Agent</u>. Administrative Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing (including any electronic message, Internet or intranet website posting or other distribution) believed by it to be genuine and to have been signed, sent or otherwise authenticated by the proper Person. Administrative Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. In determining compliance with any condition hereunder to the making of a Loan that by its terms must be fulfilled to the satisfaction of a Lender, Administrative Agent may presume that such condition is satisfactory to such Lender unless Administrative Agent shall have received notice to the contrary from such Lender prior to the making of such Loan. Administrative Agent may consult with legal counsel (who may be counsel for Borrowers), independent accountants and other experts selected by it, and shall not be liable for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts.

11.05. <u>Delegation of Duties</u>. Administrative Agent may perform any and all of its duties and exercise its rights and powers hereunder or under any other Loan Document by or through any one or more sub-agents appointed by Administrative Agent. Administrative Agent and any such sub-agent may perform any and all of its duties and exercise its rights and powers by or through their respective Related Parties. The exculpatory provisions of this Article shall apply to any such sub-agent and to Related Parties of Administrative Agent and any such sub-agent, and shall apply to their respective activities in connection with the syndication of the credit facilities provided for herein as well as activities as Administrative Agent.

11.06. <u>Resignation of Administrative Agent</u>. Administrative Agent may at any time give notice of its resignation to Lenders and Borrowers. Upon receipt of any such notice of resignation, the Required Lenders shall have the right, to appoint a successor, which shall be a bank with an office in the United States, or an Affiliate of any such bank with an office in the United States. <u>Provided</u> there is no Event of Default continuing, such appointment shall be made by the Required Lenders in consultation with Borrowers. If no such successor shall have been so appointed by the Required Lenders and shall have accepted such appointment within thirty (30) days after the retiring Administrative Agent gives notice of its resignation, then the retiring Administrative Agent may on behalf of the Lenders, appoint a successor Administrative Agent meeting the qualifications set forth above; provided that if Administrative Agent shall notify Borrowers and Lenders that no qualifying Person has accepted such appointment, then such resignation shall nonetheless become effective

in accordance with such notice and (a) the retiring Administrative Agent shall be discharged from its duties and obligations hereunder and under the other Loan Documents (except that in the case of any collateral security held by Administrative Agent on behalf of Lenders under any of the Loan Documents, the retiring Administrative Agent shall continue to hold such collateral security until such time as a successor Administrative Agent is appointed) and (b) all payments, communications and determinations provided to be made by, to or through Administrative Agent shall instead be made by or to each Lender directly, until such time as the Required Lenders appoint a successor Administrative Agent as provided for above in this Section. Upon the acceptance of a successor's appointment as Administrative Agent hereunder, such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring (or retired) Administrative Agent, and the retiring Administrative Agent shall be discharged from all of its duties and obligations hereunder or under the other Loan Documents (if not already discharged therefrom as provided above in this Section). The fees payable by Borrowers to a successor Administrative Agent shall be the same as those payable to its predecessor unless otherwise agreed between Borrowers and such successor. After the retiring Administrative Agent's resignation hereunder and under the other Loan Documents, the provisions of this Article XI and Section 12.04 shall continue in effect for the benefit of such retiring Administrative Agent, its sub-agents and their respective Related Parties in respect of any actions taken or omitted to be taken by any of them while the retiring Administrative Agent was acting as Administrative Agent.

11.07. <u>Non-Reliance on Administrative Agent and Other Lenders</u>. Each Lender acknowledges that it has, independently and without reliance upon Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Lender also acknowledges that it will, independently and without reliance upon Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it shall from time to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any other Loan Document or any related agreement or any document furnished hereunder or thereunder.

11.08. <u>No Other Duties, Etc</u>. Anything herein to the contrary notwithstanding, none of the Bookrunners, Arrangers or Syndication Agent listed on the cover page hereof shall have any powers, duties or responsibilities under this Agreement or any of the other Loan Documents, except in its capacity, as applicable, as Administrative Agent or a Lender hereunder.

11.09. <u>Administrative Agent May File Proofs of Claim</u>. In case of the pendency of any proceeding under any Debtor Relief Law or any other judicial proceeding relative to any Loan Party, Administrative Agent (irrespective of whether the principal of any Loan shall then be due and payable as herein expressed or by declaration or otherwise and irrespective of whether Administrative Agent shall have made any demand on Borrowers) shall be entitled and empowered, by intervention in such proceeding or otherwise

(a) to file and prove a claim for the whole amount of the principal and interest owing and unpaid in respect of the Loans and all other Obligations that are owing and unpaid and to file such other documents as may be necessary or advisable in order to have the claims of Lenders and Administrative Agent (including any claim for the reasonable compensation, expenses, disbursements and advances of the Lenders and Administrative Agent and their respective agents and counsel and all other amounts due the Lenders and Administrative Agent under <u>Section 2.11</u> and <u>Section 12.04</u>) allowed in such judicial proceeding; and

(b) to collect and receive any monies or other property payable or deliverable on any such claims and to distribute the same;

and any custodian, receiver, assignee, trustee, liquidator, sequestrator or other similar official in any such judicial proceeding is hereby authorized by each Lender to make such payments to Administrative Agent and, if Administrative Agent shall consent to the making of such payments directly to Lenders, to pay to Administrative Agent any amount due for the reasonable compensation, expenses, disbursements and advances of Administrative Agent and its agents and counsel, and any other amounts due Administrative Agent under <u>Section 2.10</u> and <u>Section 12.04</u>.

Nothing contained herein shall be deemed to authorize Administrative Agent to authorize or consent to or accept or adopt on behalf of any Lender any plan of reorganization, arrangement, adjustment or composition affecting the Obligations or the rights of any Lender to authorize Administrative Agent to vote in respect of the claim of any Lender or in any such proceeding.

11.10. <u>Collateral and Guaranty Matters</u>. Lenders irrevocably authorize Administrative Agent, at its option and in its discretion, to release any Lien on any property granted to or held by Administrative Agent under any Loan Document (i) upon payment in full of all Obligations (other than contingent indemnification obligations), (ii) that is sold or to be sold as part of or in connection with any sale permitted hereunder or under any other Loan Document, or (iii) if approved, authorized or ratified in writing in accordance with <u>Section 12.01</u>.

Upon request by Administrative Agent at any time, the Required Lenders may confirm in writing Administrative Agent's authority to release or subordinate its interest in particular types or items of property, or to release any Guarantor from its obligations under the Guaranty pursuant to this <u>Section 11.10</u>. In each case as specified in this <u>Section 11.10</u>, Administrative Agent will, at Borrowers' expense, execute and deliver to the applicable Loan Party such documents as such Loan Party may reasonably request to evidence the release of such item of Collateral from the assignment and security interest granted under the Collateral Documents or to subordinate its interest in such item, or to release such Guarantor from its obligations under the Guaranty, in each case in accordance with the terms of the Loan Documents and this <u>Section 11.10</u>.

ARTICLE XII MISCELLANEOUS

12.01. <u>Amendments, Etc</u>. No amendment or waiver of any provision of this Agreement or any other Loan Document, and no consent to any departure by any Borrower or any other Loan Party therefrom, shall be effective unless in writing signed by the Required Lenders and Borrowers or the applicable Loan Party, as the case may be, and acknowledged by Administrative Agent, and each such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given; provided, however, that no such amendment, waiver or consent shall:

(a) waive any condition set forth in <u>Section 5.01</u>, without the written consent of each Lender;

(b) extend or increase the Commitment of any Lender without the written consent of each Lender directly affected thereby;

(c) postpone any date fixed by this Agreement or any other Loan Document (other than as expressly permitted in <u>Section 2.16</u> with respect to the one-year extension of the Stated Maturity Date) for (i) any payment (excluding mandatory prepayments) of principal, interest, fees or other amounts due to the Lenders (or any of them) hereunder or under such other Loan Document without the written consent of each Lender entitled to such payment or (ii) any scheduled reduction of the Facility hereunder or under any other Loan Document without the written consent of each Lender;

(d) reduce or forgive the principal of, or the rate of interest specified herein on, any Loan or any fees or other amounts payable hereunder or under any other Loan Document without the written consent of each Lender entitled to such amount; provided, however, that only the consent of the Required Lenders shall be necessary to amend the definition of "Default Rate" or to waive any obligation of Borrowers to pay interest at the Default Rate;

(e) change (i) <u>Section 2.15</u> or <u>Section 10.03</u> in a manner that would alter the pro rata sharing of payments required thereby without the written consent of each Lender or (ii) the order of application of any prepayment of Loans from the application thereof set forth in the applicable provisions of <u>Section 2.12</u>, in any manner that materially and adversely affects the Lenders without the written consent of each Lender so affected thereby;

(f) change (i) any provision of this <u>Section 12.01</u> or the definition of "Required Lenders", "Super-Majority Lenders" or any other provision hereof specifying the number or percentage of Lenders required to amend, waive or otherwise modify any rights hereunder or make any determination or grant any consent hereunder, without the written consent of each Lender;

(g) release all or substantially all of the Collateral in any transaction or series of related transactions (except as provided in <u>Section 11.10</u>) or release any Collateral Property from the Mortgage encumbering such Collateral Property (except as provided in <u>Section 2.17</u>), without the written consent of each Lender except in connection with the repayment in full of the Obligations;

(h) release any Borrower or Guarantor of its obligations hereunder or the Guaranty, or release all or substantially all of the value of a Guaranty, without the written consent of each Lender, except to the extent the release of any Loan Party from such Guaranty is permitted pursuant to <u>Section 11.10</u> (in which case such release may be made by Administrative Agent acting alone); or

(i) impose any greater restriction on the ability of any Lender under the Facility to assign any of its rights or obligations hereunder without the written consent of the Required Lenders, provided that no revisions to Section 12.06(b)(v) or (vi) shall be made without the written consent of each Lender;

and provided, <u>further</u>, that no amendment, waiver or consent shall, unless in writing and signed by Administrative Agent or the Swingline Lender (in addition to such Lenders required above) affect the rights or duties of Administrative Agent or the Swingline Lender, as the case may be, under this Agreement or any other Loan Document; and the Fee Letter may be amended, or rights or privileges thereunder waived, in a writing executed only by the parties thereto.

Notwithstanding anything to the contrary herein, no Defaulting Lender shall have any right to approve or disapprove any amendment, waiver or consent hereunder, provided that any amendment, waiver or consent as set forth in <u>Section 12.01(b)</u>, (c) or (d) above shall require the consent of such Defaulting Lender only to the extent required by Section 4.08(b).

12.02. Notices; Effectiveness; Electronic Communications.

(a) <u>Notices Generally</u>. Except for (i) notices and other communications expressly permitted to be given by telephone hereunder and under each of the other Loan Documents, (ii) any notice given in accordance with the requirements of any applicable statute (including, without limitation, statutes governing foreclosure or notice of foreclosure) which shall be effective when given in accordance with statutory requirements, notwithstanding anything to the contrary contained herein or in any other Loan Document, and as (iii) as provided in subsection (b) below, all notices and other communications, including all reports, statements, certificates and other deliverables provided for herein and in each of the other Loan Documents shall be in writing and shall be delivered by hand or overnight courier service, mailed by certified or registered mail or sent by telecopier as follows, and all notices and other communications expressly permitted hereunder to be given by telephone shall be made to the applicable telephone number, as follows:

(i) if to any Borrower or Administrative Agent, to the address, telecopier number, electronic mail address or telephone number specified for such Person on <u>Schedule 12.02;</u> and

(ii) if to any other Lender, to the address, telecopier number, electronic mail address or telephone number specified in its Administrative Questionnaire; provided, however, that to the extent any notices or other communications are required hereunder to be delivered by Borrowers to Administrative Agent and the Lenders, Borrowers shall satisfy such obligation by delivering a copy of such notice or communication to Administrative Agent with a direction to Administrative Agent to post such notice or communication on an electronic site accessible to all Lenders.

Notices sent by hand or overnight courier service, or mailed by certified or registered mail, shall be deemed to have been given when received or refused; notices sent by telecopier shall be deemed to have been given when sent (except that, if not given during normal business hours for the recipient, shall be deemed to have been given at the opening of business on the next Business Day for the recipient). Notices delivered through electronic communications to the extent provided in subsection (b) below shall be effective as provided in such subsection (b).

(b) <u>Electronic Communications</u>. Notices and other communications to Lenders hereunder may be delivered or furnished by electronic communication (including e-mail and Internet or intranet websites) pursuant to procedures approved by Administrative Agent, provided that the foregoing shall not apply to notices to any Lender pursuant to <u>Article II</u> if such Lender has notified Administrative Agent that it is incapable of receiving notices under such Article by electronic communication. Administrative Agent or Borrowers may, in their discretion, agree to accept notices and other communications to it hereunder by electronic communications pursuant to procedures approved by it, provided that approval of such procedures may be limited to particular notices or communications.

Unless Administrative Agent otherwise prescribes, (i) notices and other communications sent to an e-mail address shall be deemed received upon the sender's receipt of an acknowledgement from the intended recipient (such as by the "return receipt requested" function, as available, return e-mail or other written acknowledgement), provided that if such notice or other communication is not sent during the normal business hours of the recipient, such notice or communication shall be deemed to have been sent at the opening of business on the next Business Day for the recipient, and (ii) notices or communications posted to an Internet or intranet website shall be deemed received upon the deemed receipt by the intended recipient at its e-mail address as described in the foregoing clause (i) of notification that such notice or communication is available and identifying the website address therefor.

(c) The Platform. THE PLATFORM IS PROVIDED "AS IS" AND "AS AVAILABLE." THE AGENT PARTIES (AS DEFINED BELOW) DO NOT WARRANT THE ACCURACY OR COMPLETENESS OF THE BORROWER MATERIALS OR THE ADEQUACY OF THE PLATFORM, AND EXPRESSLY DISCLAIM LIABILITY FOR ERRORS IN OR OMISSIONS FROM THE BORROWER MATERIALS. NO WARRANTY OF ANY KIND, EXPRESS, IMPLIED OR STATUTORY, INCLUDING ANY WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, NON-INFRINGEMENT OF THIRD PARTY RIGHTS OR FREEDOM FROM VIRUSES OR OTHER CODE DEFECTS, IS MADE BY ANY AGENT PARTY IN CONNECTION WITH THE BORROWER MATERIALS OR THE PLATFORM. In no event shall Administrative Agent or any of its Related Parties (collectively, the "Agent Parties") have any liability to any Borrower, any Lender or any other Person for losses, claims, damages, liabilities or expenses of any kind (whether in tort, contract or otherwise) arising out of any Borrower's or Administrative Agent's transmission of Borrower Materials through the Internet, except to the extent that such losses, claims, damages, liabilities or expenses are determined by a court of competent jurisdiction by a final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Agent Party; provided, however, that in no event shall any Agent Party have any liability to any Borrower, any Lender or any other Person for indirect, special, incidental, consequential or punitive damages (as opposed to direct or actual damages).

(d) <u>Change of Address, Etc.</u> Each of Borrowers and Administrative Agent may change its address, electronic mail address, telecopier or telephone number for notices and other communications hereunder and under the other Loan Documents by notice to the other parties hereto. Each other Lender may change its address, electronic mail address, telecopier or telephone number for notices and other communications hereunder by notice to Borrowers and Administrative Agent. In addition, each Lender agrees to notify Administrative Agent from time to time to ensure that Administrative Agent has on record (i) an effective address, contact name, telephone number, telecopier number and electronic mail address to which notices and other communications may be sent and (ii) accurate wire instructions for such Lender.

(e) <u>Reliance by Administrative Agent and Lenders</u>. Administrative Agent and Lenders shall be entitled to rely and act upon any notices (including telephonic Loan Notices) purportedly given by or on behalf of any Borrower or any other party under the other Loan Documents even if (i) such notices were not made in a manner specified herein, were incomplete or were not preceded or followed by any other form of notice specified herein, or (ii) the terms thereof, as understood by the recipient, varied from any confirmation thereof. Borrowers shall indemnify Administrative Agent, each Lender and Related Parties of each of them from all losses, costs, expenses and liabilities resulting from the reliance by such Person on each notice purportedly given by or on behalf of any Borrower or such other Party, **INCLUDING BY SUCH PARTY'S OWN NEGLIGENCE.** All telephonic notices to and other telephonic communications with Administrative Agent may be recorded by Administrative Agent, and each of the parties hereto hereby consents to such recording.

12.03. No Waiver; Cumulative Remedies. No failure by any Lender or Administrative Agent to exercise, and no delay by any such Person in exercising, any right, remedy, power or privilege hereunder or under any other Loan Document shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege, including in each case, the right of Lender to bring any action or exercise any right, remedy, power or privilege in respect of the enforcement of any Loan Document or in respect of Collateral in any State without waiving its rights, remedies, powers or privileges as to any enforcement of any Loan Document or in respect of any Collateral in any other State. In particular, and not by way of limitation, by accepting payment after the due date of any amount payable under this Agreement, the Notes or any other Loan Document, Administrative Agent and any Lender shall not be deemed to have waived any right either to require prompt payment when due of all other amounts due under this Agreement, the Notes, or the other Loan Documents, or to declare a default for failure to effect prompt payment of any such other amount. No waiver of any provision in this Agreement or consent to any departure by Borrower therefrom shall in any event be effective unless the same shall be permitted by Section 12.01 above, and then such waiver or consent shall be effective only in the specific instance

and for the purpose of for which given. The rights, remedies, powers and privileges herein provided, and provided under each other Loan Document, are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law.

12.04. Expenses; Indemnity; Damage Waiver.

(a) <u>Costs</u> and <u>Expenses</u>. Borrowers shall pay (i) all reasonable out-of-pocket expenses incurred by Administrative Agent and its Affiliates (including (X) appraisal fees, title and escrow charges, reasonable fees and charges of other third party consultants performing due diligence on the Collateral Properties and (Y) the reasonable fees, charges and disbursements of counsel for Administrative Agent), in connection with the syndication of the credit facilities provided for herein, the preparation, negotiation, execution, delivery and administration of this Agreement and the other Loan Documents or any amendments, modifications or waivers of the provisions hereof or thereof (whether or not the transactions contemplated hereby or thereby shall be consummated) and (ii) all out-of-pocket expenses incurred by Administrative Agent or any Lender (including the fees, charges and disbursements of any counsel for Administrative Agent or any Lender), in connection with the enforcement or protection of its rights (A) in connection with this Agreement and the other Loan Documents, including its rights under this Section, or (B) in connection with Loans made hereunder, including all such out-of-pocket expenses incurred during any workout, restructuring or negotiations in respect of such Loans.

(b) Indemnification by Borrowers. Each Borrower shall indemnify Administrative Agent (and any sub-agent thereof), Arrangers, each Lender, and each Related Party of any of the foregoing Persons (each such Person being called an "Indemnitee") against, and hold each Indemnitee harmless from, any and all losses, claims, damages, liabilities and related expenses (including the fees, charges and disbursements of any external counsel for any Indemnitee), and shall indemnify and hold harmless each Indemnitee from all reasonable fees and time charges and disbursements for attorneys who may be employees of any Indemnitee, incurred by any Indemnitee or asserted against any Indemnitee by any third party or by any Borrower or any other Loan Party arising out of, in connection with, or as a result of (i) the execution or delivery of this Agreement, any other Loan Document or any agreement or instrument contemplated hereby or thereby, the performance by the parties hereto of their respective obligations hereunder or thereunder or the consummation of the transactions contemplated hereby or thereby, or, in the case of Administrative Agent (and any sub-agent thereof) and its Related Parties only, the administration of this Agreement and the other Loan Documents, (ii) any Loan or the use or proposed use of the proceeds therefrom, (iii) arising out of or relating to any Collateral Property, and any actual or alleged presence or release of Hazardous Materials on or from any Collateral Property or any other property owned or operated by any Borrower or any of its Subsidiaries, or any Environmental Liability related in any way to any Borrower or any of its Subsidiaries, (iv) enforcing any obligations of or collecting any payments due from any Loan Party under this Agreement, the other Loan Documents or with respect to the Collateral Properties or in connection with any refinancing or restructuring of the credit arrangements provided under this Agreement in the nature of Debtor Relief Laws; (v) losses incurred in correcting any prohibited transaction or in the sale of a prohibited loan, and in obtaining any individual prohibited transaction exemption under ERISA, the Code, any state statute or other similar law that

may be necessary in Administrative Agent's sole discretion to correct such prohibited transaction or loan sale and any losses that Administrative Agent or any Lender may incur, directly or indirectly, as a result of a default under Sections 6.08 or 8.17; or (vi) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory, whether brought by a third party or by any Borrower or any other Loan Party or any of Borrowers' or such Loan Party's directors, shareholders or creditors, and regardless of whether any Indemnitee is a party thereto, IN ALL CASES, WHETHER OR NOT CAUSED BY OR ARISING, IN WHOLE OR IN PART, OUT OF THE COMPARATIVE, CONTRIBUTORY OR SOLE NEGLIGENCE OF THE INDEMNITEE; provided that such indemnity shall not, as to any Indemnitee, be available to the extent that such losses, claims, damages, liabilities or related expenses (x) are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Indemnitee or (y) result from a claim brought by any Borrower or any other Loan Party against an Indemnitee for breach in bad faith of such Indemnitee's obligations hereunder or under any other Loan Document, if any Borrower or such Loan Party has obtained a final and nonappealable judgment in its favor on such claim as determined by a court of competent jurisdiction.

(c) <u>Reimbursement by Lenders</u>. To the extent that any Borrower for any reason fails to indefeasibly pay any amount required under subsection (a) or (b) of this Section to be paid by it to Administrative Agent (or any sub-agent thereof) or any Related Party of any of the foregoing, each Lender severally agrees to pay to Administrative Agent (or any such sub-agent) or such Related Party, as the case may be, such Lender's Applicable Percentage (determined as of the time that the applicable unreimbursed expense or indemnity payment is sought) of such unpaid amount, provided that the unreimbursed expense or indemnified loss, claim, damage, liability or related expense, as the case may be, was incurred by or asserted against Administrative Agent (or any such sub-agent) in its capacity as such, or against any Related Party of any of the foregoing acting for Administrative Agent (or any such sub-agent in connection with such capacity).

Waiver of Consequential Damages, Etc. To the fullest extent permitted by (d) applicable law, no Borrower shall assert, and each Borrower hereby waives, any claim against any Indemnitee, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, this Agreement, any other Loan Document or any agreement or instrument contemplated hereby, the transactions contemplated hereby or thereby, any Loan or the use of the proceeds thereof, INCLUDING DUE TO SUCH INDEMNITEE'S OWN NEGLIGENCE. No Indemnitee referred to in subsection (b) above shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed to such unintended recipients by such Indemnitee through telecommunications, electronic or other information transmission systems in connection with this Agreement or the other Loan Documents or the transactions contemplated hereby or thereby, other than for direct or actual damages resulting from the gross negligence or willful misconduct of such Indemnitee as determined by a final and nonappealable judgment of a court of competent jurisdiction.

(e) <u>Payments</u>. All amounts due under this Section shall be payable not later than ten (10) Business Days after demand therefor.

(f) <u>Survival</u>. The agreements in this Section shall survive the resignation of Administrative Agent, the replacement of any Lender, the termination of Commitments and the repayment, satisfaction or discharge of all the other Obligations.

12.05. Payments Set Aside. To the extent that any payment by or on behalf of any Borrower is made to Administrative Agent or any Lender, or Administrative Agent or any Lender exercises its right of setoff, and such payment or the proceeds of such setoff or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by Administrative Agent or such Lender in its discretion) to be repaid to a trustee, receiver or any other party, in connection with any proceeding under any Debtor Relief Law or otherwise, then (a) to the extent of such recovery, the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such setoff had not occurred, and (b) each Lender severally agrees to pay to Administrative Agent upon demand its applicable share (without duplication) of any amount so recovered from or repaid by Administrative Agent, plus interest thereon from the date of such demand to the date such payment is made at a rate per annum equal to the Federal Funds Rate from time to time in effect. The obligations of Lenders under clause (b) of the preceding sentence shall survive the payment in full of the Obligations and the termination of this Agreement.

12.06. Successors and Assigns.

(a) Successors and Assigns Generally. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that Borrowers may not assign or otherwise transfer any of their rights or obligations hereunder without the prior written consent of Administrative Agent and each Lender (other than a Defaulting Lender) and no Lender may assign or otherwise transfer any of its rights or obligations hereunder except (i) to an assignee in accordance with the provisions of Section 12.06(b), (ii) by way of participation in accordance with the provisions of Section 12.06(d), or (iii) by way of pledge or assignment of a security interest subject to the restrictions of Section 12.06(f) (and any other attempted assignment or transfer by any party hereto shall be null and void). Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants to the extent provided in subsection (d) of this Section and, to the extent expressly contemplated hereby, the Related Parties of each of Administrative Agent and Lenders) any legal or equitable right, remedy or claim under or by reason of this Agreement.

(b) <u>Assignments by Lenders</u>. Any Lender may at any time assign to one or more assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Commitment(s) and the Loans at the time owing to it); provided that any such assignment shall be subject to the following conditions:

(i) Minimum Amounts.

(A) in the case of an assignment of the entire remaining amount of the assigning Lender's Commitment under the Facility and the Loans at the time owing to it under the Facility or in the case of an assignment to a Lender, an Affiliate of a Lender or an Approved Fund, no minimum amount need be assigned; and

in any case not described in Section 12.06(b)(i)(A), the **(B)** aggregate amount of the Commitment (which for this purpose includes Loans outstanding thereunder) or, if the Commitment is not then in effect, the principal outstanding balance of the Loans of the assigning Lender subject to each such assignment, determined as of the date the Assignment and Assumption with respect to such assignment is delivered to Administrative Agent or, if "Trade Date" is specified in the Assignment and Assumption, as of the Trade Date, shall not be less than \$5,000,000 unless each of Administrative Agent and, so long as no Event of Default has occurred and is continuing, Borrowers otherwise consent (each such consent not to be unreasonably withheld or delayed); provided, however, that concurrent assignments to members of an Assignee Group and concurrent assignments from members of an Assignee Group to a single Eligible Assignee (or to an Eligible Assignee and members of its Assignee Group) will be treated as a single assignment for purposes of determining whether such minimum amount has been met;

(ii) Proportionate Amounts. Each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Loans or the Commitment assigned;

(iii) Required Consents. No consent shall be required for any assignment except to the extent required by subsection $(\underline{b})(\underline{i})(\underline{B})$ of this Section and, in addition:

the consent of Borrowers (such consent not to be (A) unreasonably withheld or delayed) shall be required, unless (1) an Event of Default has occurred and is continuing at the time of such assignment, or (2) such assignment is to (x) a Lender (other than a Defaulting Lender), or (y) an Affiliate of Lender (other than a Defaulting Lender), or an Approved Fund that, in either case, is regularly "engaged in commercial real estate lending in the normal course of business"; provided, however, that individuals, funds, partnerships, joint ventures and other Persons that are generally engaged in businesses pursuing so-called activist, opportunistic, "vulture" or similar strategic investment objectives, with respect to which making or acquiring commercial real estate loans is ancillary to such other objectives but nonetheless arises in the normal course of business, shall be deemed not engaged in commercial real estate lending in the normal course of business for purposes of this subsection 12.06(iii)(A); provided that Borrowers shall be deemed to have consented to any such assignment unless

Borrowers shall have objected thereto by written notice to Administrative Agent within five (5) Business Days after having received notice thereof; and

(B) the consent of Administrative Agent and the Swingline Lender (in each case, such consent not to be unreasonably withheld or delayed) shall be required for assignments in respect of any Loan to a Person that is not a Lender;

(iv) Assignment and Assumption. The parties to each assignment shall execute and deliver to Administrative Agent an Assignment and Assumption, together with a processing and recordation fee in the amount of \$3,500; provided, however, that Administrative Agent may, in its sole discretion, elect to waive such processing and recordation fee in the case of any assignment. The assignee, if it is not a Lender, shall deliver to Administrative Agent an Administrative Questionnaire;

(v) No Assignment to Borrowers. No such assignment shall be made to any Borrower or any of Borrowers' Affiliates or Subsidiaries; and

(vi) No Assignment to Natural Persons. No such assignment shall be made to a natural Person.

Subject to acceptance and recording thereof by Administrative Agent pursuant to subsection (c) of this Section, from and after the effective date specified in each Assignment and Assumption, the assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Assignment and Assumption, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Assumption, be released from its obligations under this Agreement (and, in the case of an Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto) but shall continue to be entitled to the benefits of Sections 4.01, 4.04, 4.05 and 12.04 with respect to facts and circumstances occurring prior to the effective date of such assignment. Upon request, Borrowers (at their expense) shall execute and deliver one or more Notes to the assignee Lender. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this subsection shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with Section 12.06(d).

(c) <u>Register</u>. Administrative Agent, acting solely for this purpose as an agent of Borrowers, shall maintain at Administrative Agent's Office a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Commitments of, and principal amounts of the Loans owing to, each Lender pursuant to the terms hereof from time to time (the "<u>Register</u>"). The entries in the Register shall be conclusive (absent manifest error) and Borrowers, Administrative Agent and the Lenders may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by Borrowers and any Lender, at any reasonable time and from time to time upon reasonable prior notice.

(d) Participations. Any Lender may at any time, without the consent of, or notice to, Borrowers or Administrative Agent, sell participations to any Person (other than a natural Person or Borrowers or any of Borrowers' Affiliates or Subsidiaries) (each, a "Participant") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of its Commitment and/or the Loans owing to it); provided that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations and (iii) Borrowers, Administrative Agent and Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement. Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; provided that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, waiver or other modification described in the first proviso to Section 12.06 that affects such Participant. Subject to subsection (e) of this Section, Borrowers agree that each Participant shall be entitled to the benefits of Sections 4.01, 4.04 and 4.05 to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to Section 12.06(b). To the extent permitted by law, each Participant also shall be entitled to the benefits of Section 12.08 as though it were a Lender, provided such Participant agrees to be subject to Section 2.15 as though it were a Lender.

(e) <u>Limitations upon Participant Rights</u>. A Participant shall not be entitled to receive any greater payment under <u>Sections 4.01</u> or <u>4.04</u>, than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is made with Borrowers' prior written consent. A Participant that would be a Foreign Lender if it were a Lender shall not be entitled to the benefits of <u>Section 4.01</u> unless Borrowers are notified of the participation sold to such Participant and such Participant agrees, for the benefit of Borrowers, to comply with <u>Section 4.01(e)</u> as though it were a Lender.

(f) <u>Certain Pledges</u>. Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement (including under its Note, if any) to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank; provided that no such pledge or assignment shall release such Lender from any of its obligations hereunder or substitute any such pledge or assignee for such Lender as a party hereto.

(g) <u>Electronic Execution of Assignments</u>. The words "execution," "signed," "signature," and words of like import in any Assignment and Assumption shall be deemed to include electronic signatures or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any applicable law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act.

12.07. Treatment of Certain Information; Confidentiality. Each of Administrative Agent and Lenders agrees to maintain the confidentiality of the Information (as defined below), except that Information may be disclosed (a) to its Affiliates and to its and its Affiliates' respective partners, members, directors, officers, employees, agents, advisors and representatives (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential), (b) to the extent requested by any regulatory authority purporting to have jurisdiction over it (including any self-regulatory authority, such as the National Association of Insurance Commissioners), (c) to the extent required by applicable laws or regulations or by any subpoena or similar legal process, (d) to any other party hereto, (e) in connection with the exercise of any remedies hereunder or under any other Loan Document or any action or proceeding relating to this Agreement or any other Loan Document or the enforcement of rights hereunder or thereunder, (f) subject to an agreement containing provisions substantially the same or at least as restrictive as those of this Section, to (i) any assignee of or Participant in, or any prospective assignee of or Participant in, any of its rights or obligations under this Agreement, or (ii) any actual or prospective counterparty (or its advisors) to any swap or derivative transaction relating to a Borrower and its obligations, (g) with the consent of Borrowers or (h) to the extent such Information (i) becomes publicly available other than as a result of a breach of this Section or (ii) becomes available to Administrative Agent, any Lender or any of their respective Affiliates on a nonconfidential basis from a source other than Borrowers.

For purposes of this Section, "<u>Information</u>" means all information received from any Loan Party or any Subsidiary thereof relating to any Loan Party or any Subsidiary thereof or their respective businesses, other than any such information that is available to Administrative Agent or any Lender on a nonconfidential basis prior to disclosure by any Loan Party or any Subsidiary thereof, provided that, in the case of information received from a Loan Party or any such Subsidiary after the date hereof, such information is clearly identified at the time of delivery as confidential. Any Person required to maintain the confidentiality of Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information. Each of Administrative Agent and Lenders acknowledge that (a) the Information may include material non-public information concerning Borrowers or any Subsidiary, as the case may be, (b) it has developed compliance procedures regarding the use of material non-public information and (c) it will handle such material non-public information in accordance with applicable Law, including United States Federal and state securities Laws.

12.08. Right of Setoff. If an Event of Default shall have occurred and be continuing, each Lender and each of their respective Affiliates is hereby authorized at any time and from time to time, after obtaining the prior written consent of Administrative Agent, to the fullest extent permitted by applicable law, to set off and apply any and all deposits (general or special, time or demand, provisional or final, in whatever currency) at any time held and other obligations (in whatever currency) at any time owing by such Lender or any such Affiliate to or for the credit or the account of any Borrower against any and all of the obligations of Borrowers now or hereafter existing under this Agreement or any other Loan Document to such Lender irrespective of whether or not such Lender shall have made any demand under this Agreement or any other Loan Document and although such obligations of Borrowers or any of them may be contingent or unmatured or are owed to a branch or office of such Lender different from the branch or office holding such deposit or obligated on such indebtedness. The rights of each Lender and their respective Affiliates under this Section are in addition to other rights and remedies (including other rights of setoff) that such Lender or their respective Affiliates may have. Each Lender agrees to notify Borrowers and Administrative Agent promptly after any such setoff and application, provided that the failure to give such notice shall not affect the validity of such setoff and application.

12.09. Exculpation. Notwithstanding anything to the contrary contained in this Agreement or any other Loan Document, no present or future Constituent Member (as hereinafter defined) in any Borrower, nor any present or future shareholder, officer, director, employee, trustee, beneficiary, advisor, member, partner, principal, participant or agent of or in any Borrower or of or in any Person that is or becomes a Constituent Member in any Borrower, shall have any personal or other liability, directly or indirectly, under or in connection with the Loan Documents, except as may occur by virtue of such Person becoming a successor to any Borrower pursuant to Section 12.06(a) or being a Guarantor. Administrative Agent and each Lender each, on behalf of itself and its respective successors and assigns, hereby waives any and all such personal or other liability. The term "Constituent Member," as used herein, shall mean any direct partner or member in any Borrower and any Person that, directly or indirectly through one or more other partnerships, limited liability companies, corporations or other entities, is a partner or member in any Borrower. Notwithstanding anything to the contrary contained in the Loan Documents, neither the negative capital account of any Constituent Member in any Borrower nor any obligation of any Constituent Member in any Borrower to restore a negative capital account or to contribute or loan capital to any Borrower or to any other Constituent Member in any Borrower shall at any time be deemed to be the property or an asset of Borrower (or any such other Constituent Member) and neither any Borrower nor any of their respective successors or assigns shall have any right to collect, enforce or proceed against any Constituent Member with respect to any such negative capital account or obligation to restore, contribute or loan. Nothing contained in this Section 12.09 shall apply to, or be deemed to be a release or exculpation from liability of, any Guarantor or any other Person who executes, or is required by any Loan Document to execute, a Guaranty or the Environmental Indemnity (provided that nothing herein shall limit the exculpation provisions (including Section 16) contained in any Guaranty).

12.10. Interest Rate Limitation. Notwithstanding anything to the contrary contained in any Loan Document, the interest paid or agreed to be paid under the Loan Documents shall not exceed the maximum rate of non-usurious interest permitted by applicable Law (the "Maximum Rate"). If Administrative Agent or any Lender shall receive interest in an amount that exceeds the Maximum Rate, the excess interest shall be applied to the principal of the Loans or, if it exceeds such unpaid principal, refunded to Borrowers. In determining whether the interest contracted for, charged, or received by Administrative Agent or a Lender exceeds the Maximum Rate, such Person may, to the extent permitted by applicable Law, (a) characterize any payment that is not principal as an expense, fee, or premium rather than interest, (b) exclude voluntary prepayments and the effects thereof, and (c) amortize, prorate, allocate, and spread in equal or unequal parts the total amount of interest throughout the contemplated term of the Obligations hereunder.

12.11. <u>Counterparts</u>; <u>Integration</u>; <u>Effectiveness</u>. This Agreement may be executed in counterparts (and by different parties hereto in different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. This Agreement, the Fee Letter, any separate letter agreements with respect to fees payable to Administrative Agent and the other Loan Documents constitute the entire contract among the parties relating to the subject matter hereof and supersede any and all previous agreements and understandings, oral or written, relating to the subject matter hereof. Except as provided in <u>Section 5.01</u>, this Agreement shall become effective when it shall have been executed by Administrative Agent and when Administrative Agent shall have received counterparts hereof that, when taken together, bear the signatures of each of the other parties hereto. Delivery of an executed counterpart of a signature page of this Agreement by telecopy shall be effective as delivery of a manually executed counterpart of this Agreement.

12.12. <u>Survival of Representations and Warranties</u>. All representations and warranties, covenants and agreements made by Borrowers hereunder and in any other Loan Document or other document delivered pursuant hereto or thereto or in connection herewith or therewith shall survive the execution and delivery hereof and thereof. Such representations and warranties have been or will be relied upon by Administrative Agent and each Lender, regardless of any investigation made by Administrative Agent or any Lender or on their behalf and notwithstanding that Administrative Agent or any Lender may have had notice or knowledge of any Default at the time of the Loans, and shall continue in full force and effect as long as any Loan or any other Obligation hereunder shall remain unpaid or unsatisfied. The provisions of Section 4.01, 4.04, 4.05, 12.04 and Article XI shall survive and remain in full force and effect regardless of the consummation of the transactions contemplated hereby, the repayment of the Loans, the expiration or termination of the Commitments or the termination of this Agreement or any provision hereof.

12.13. <u>Severability</u>. If any provision of this Agreement or the other Loan Documents is held to be illegal, invalid or unenforceable, (a) the legality, validity and enforceability of the remaining provisions of this Agreement and the other Loan Documents shall not be affected or impaired thereby and (b) the parties shall endeavor in good faith negotiations to replace the illegal, invalid or unenforceable provisions with valid provisions the economic effect of which comes as close as possible to that of the illegal, invalid or unenforceable provisions. The invalidity of a provision in a particular jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

12.14. <u>Replacement of Lenders</u>. If any Lender requests compensation under Section 4.04, or if Borrowers are required to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to <u>Section 4.01</u>, or if any Lender is a Defaulting Lender, then Borrowers may, at their sole expense and effort, upon notice to such Lender and Administrative Agent, require such Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in, and consents required by, <u>Section 12.06</u>), all of its interests, rights and obligations under this Agreement and the related Loan Documents to an assignee that shall assume such obligations (which assignee may be another Lender, if a Lender accepts such assignment), provided that:

(a) Borrowers shall have paid to Administrative Agent the assignment fee specified in <u>Section 12.06(b);</u>

(b) such Lender shall have received payment of an amount equal to the outstanding principal of its Loans, accrued interest thereon, accrued fees and all other amounts payable to it hereunder and under the other Loan Documents (including any amounts under <u>Section 4.05</u>) from the assignee (to the extent of such outstanding principal and accrued interest and fees) or Borrowers (in the case of all other amounts);

(c) in the case of any such assignment resulting from a claim for compensation under $\underline{Section 4.04}$ or payments required to be made pursuant to $\underline{Section 4.01}$, such assignment will result in a reduction in such compensation or payments thereafter; and

(d) such assignment does not conflict with applicable Laws.

A Lender shall not be required to make any such assignment or delegation if, prior thereto, as a result of a waiver by such Lender or otherwise, the circumstances entitling Borrowers to require such assignment and delegation cease to apply.

12.15. Governing Law; Jurisdiction; Etc.

THIS AGREEMENT SHALL BE DEEMED TO BE A CONTRACT (a) ENTERED INTO PURSUANT TO THE LAWS OF THE STATE OF NEW YORK AND SHALL IN ALL RESPECTS BE GOVERNED, CONSTRUED, APPLIED AND ENFORCED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK (OTHER THAN THOSE CONFLICT OF LAW PROVISIONS THAT WOULD DEFER TO THE SUBSTANTIVE LAWS OF ANOTHER JURISDICTION). WITHOUT IN ANY WAY LIMITING THE PRECEDING CHOICE OF LAW, THE PARTIES ELECT TO BE GOVERNED BY NEW YORK LAW IN ACCORDANCE WITH, AND ARE RELYING (AT LEAST IN PART) ON, SECTION 5-1401 OF THE GENERAL OBLIGATIONS LAW OF THE STATE OF NEW YORK; PROVIDED HOWEVER, THAT WITH RESPECT TO THE CREATION, PERFECTION, PRIORITY AND ENFORCEMENT OF THE LIENS AND SECURITY INTERESTS CREATED BY THIS AGREEMENT, THE SECURITY INSTRUMENTS AND THE OTHER LOAN DOCUMENTS, AND THE DETERMINATION OF DEFICIENCY JUDGMENTS, THE LAWS OF THE STATE WHERE THE PROPERTY IS LOCATED SHALL APPLY.

WITH RESPECT TO ANY CLAIM OR ACTION ARISING HEREUNDER (b) OR UNDER THIS AGREEMENT, THE NOTES, OR THE OTHER LOAN DOCUMENTS, BORROWER (A) IRREVOCABLY SUBMITS TO THE NONEXCLUSIVE JURISDICTION OF THE COURTS OF THE STATE OF NEW YORK AND THE UNITED STATES DISTRICT COURT LOCATED IN THE BOROUGH OF MANHATTAN IN NEW YORK, NEW YORK, AND APPELLATE COURTS FROM ANY THEREOF, AND (B) IRREVOCABLY WAIVES ANY OBJECTION WHICH IT MAY HAVE AT ANY TIME TO THE LAYING ON VENUE OF ANY SUIT, ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE NOTES OR THE OTHER LOAN DOCUMENTS BROUGHT IN ANY SUCH COURT, IRREVOCABLY WAIVES ANY CLAIM THAT ANY SUCH SUIT, ACTION OR PROCEEDING BROUGHT IN ANY SUCH COURT HAS BEEN BROUGHT IN AN INCONVENIENT FORUM. NOTHING IN THIS AGREEMENT, THE NOTES OR THE OTHER LOAN DOCUMENTS WILL BE DEEMED TO PRECLUDE LENDER FROM BRINGING AN ACTION OR PROCEEDING WITH RESPECT HERETO IN ANY OTHER JURISDICTION. WITHOUT IN ANY WAY LIMITING THE PRECEDING CONSENTS TO JURISDICTION AND VENUE, THE PARTIES AGREE TO SUBMIT TO THE JURISDICTION OF SUCH NEW YORK COURTS IN ACCORDANCE WITH SECTION 5-1402 OF THE GENERAL OBLIGATIONS LAW OF THE STATE OF NEW YORK OR ANY CORRESPONDING OR SUCCEEDING PROVISIONS THEREOF.

(c) <u>SERVICE OF PROCESS</u>. EACH PARTY HERETO IRREVOCABLY CONSENTS TO SERVICE OF PROCESS IN THE MANNER PROVIDED FOR NOTICES IN <u>SECTION 12.02</u>. NOTHING IN THIS AGREEMENT WILL AFFECT THE RIGHT OF ANY PARTY HERETO TO SERVE PROCESS IN ANY OTHER MANNER PERMITTED BY APPLICABLE LAW.

12.16. <u>Waiver of Jury Trial</u>. EACH PARTY HERETO HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY HERETO (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PERSON HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PERSON WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION.

12.17. No Advisory or Fiduciary Responsibility. In connection with all aspects of each transaction contemplated hereby (including in connection with any amendment, waiver or other modification hereof or of any other Loan Document), each Borrower acknowledges and agrees, and acknowledges its Affiliates' understanding, that: (i) (A) the arranging and other services regarding this Agreement provided by Administrative Agent and, Arrangers, are arm's-length commercial transactions between Borrowers and their respective Affiliates, on the one hand, and Administrative Agent and Arrangers, on the other hand, (B) each Borrower has consulted its own legal, accounting, regulatory and tax advisors to the extent it has deemed appropriate, and (C) each Borrower is capable of evaluating, and understands and accepts, the terms, risks and conditions of the transactions contemplated hereby and by the other Loan Documents; (ii) (A) each of the Arrangers and Administrative Agent is and has been acting solely as a principal and, except as expressly agreed in writing by the relevant parties, has not been, is not, and will not be acting as an advisor, agent or fiduciary for any Borrower or any of their respective Affiliates, or any other Person and (B) neither Arranger nor Administrative Agent has any obligation to Borrowers or any of their respective Affiliates with respect to the transactions contemplated hereby except those obligations expressly set forth herein and in the other Loan Documents; and (iii) Administrative Agent and Arrangers and their respective Affiliates may be engaged in a broad range of transactions that involve interests that differ from those of Borrowers and their respective Affiliates, and neither Arranger nor Administrative Agent has any obligation to disclose any of such interests to Borrowers or any of their respective Affiliates. To the fullest extent permitted by law, each Borrower hereby waives and releases any claims that it may have against Administrative Agent and Arrangers with respect to any breach or alleged breach of agency or fiduciary duty in connection with any aspect of any transaction contemplated hereby.

12.18. <u>USA PATRIOT Act Notice</u>. Each Lender that is subject to the Patriot Act (as hereinafter defined) and Administrative Agent (for itself and not on behalf of any Lender) hereby notifies Borrowers that pursuant to the requirements of the USA PATRIOT Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)) (the "<u>Patriot Act</u>"), it is required to obtain, verify and record information that identifies each Loan Party, which information includes the name and address of each Loan Party and other information that will allow such Lender or Administrative Agent, as applicable, to identify each Loan Party in accordance with the Patriot Act.

12.19. <u>Discretion</u>. Whenever pursuant to this Agreement, Administrative Agent or a Lender exercises any right given to it to approve or disapprove, or any arrangement or term is to be satisfactory to Administrative Agent or Lenders, respectively, the decision of Administrative Agent or Lenders to approve or disapprove or to decide whether arrangements or terms are satisfactory or not satisfactory shall (except as is otherwise specifically herein provided) be in the sole discretion of Administrative Agent or Lenders, as applicable and shall be final and conclusive.

12.20. <u>Offsets, Counterclaims and Defenses</u>. Any assignee of any Lender's interest in and to this Agreement, the Notes and the other Loan Documents shall take the same free and clear of all offsets, counterclaims or defenses which are unrelated to the Loan Documents which Borrowers may otherwise have against any assignor of the Loan Documents, and no such unrelated counterclaim or defense shall be interposed or asserted by Borrowers in any action or proceeding brought by any such assignee upon such documents and any such right to interpose or assert any such unrelated offset, counterclaim or defense in any such action or proceeding is hereby expressly waived by Borrowers.

12.21. <u>No Joint Venture or Partnership, No Third Party Beneficiaries</u>. xiii) Borrowers, Administrative Agent and Lenders intend that the relationships created hereunder and under the other Loan Documents be solely that of borrower and lender. Nothing herein or therein is intended to create a joint venture, partnership, tenancy-in-common, or joint tenancy relationship between Borrowers and Lenders nor to grant Lenders any interest in any Collateral Property other than that of mortgagee, beneficiary or lender.

(b) This Agreement and the other Loan Documents are solely for the benefit of Lenders and Borrowers and nothing contained in this Agreement or the other Loan Documents shall be deemed to confer upon anyone other than Lenders and Borrowers (or an Affiliate of either of the foregoing acting on behalf of Borrowers or Lenders, as applicable) any right to insist upon or to enforce the performance or observance of any of the obligations contained herein or therein. All conditions to the obligations of Lenders to make the Loan hereunder are imposed solely and exclusively for the benefit of Lenders and no other Person shall have standing to require satisfaction of such conditions in accordance with their terms or be entitled to assume that Lenders will refuse to make the Loan in the absence of strict compliance with any or all thereof and no other Person shall under any circumstances be deemed to be a beneficiary of such conditions, any or all of which may be freely waived in whole or in part by Lenders if, in Lenders' sole discretion, Lenders deem it advisable or desirable to do so.

12.22. <u>Publicity</u>. All news releases, publicity or advertising by Borrowers or their respective Affiliates through any media intended to reach the general public which refers to the Loan Documents or the financing evidenced by the Loan Documents, to Administrative Agent, Lenders or any of their Affiliates shall be subject to the prior written approval of Administrative Agent, which shall not be unreasonably withheld. Notwithstanding the foregoing, disclosure required by any federal or state securities laws, rules or regulations, as determined by Borrowers' counsel, shall not be subject to the prior written approval of Administrative Agent or any Lender.

12.23. <u>Waiver of Marshalling of Assets</u>. To the fullest extent permitted by Applicable Law, each Borrower, for itself and its successors and assigns, waives all rights to a marshalling of the assets of such Borrower and of the Collateral Properties, or to a sale in inverse order of alienation in the event of foreclosure of any of the Mortgages, and agrees not to assert any right under any laws pertaining to the marshalling of assets, the sale in inverse order of alienation, homestead exemption, the administration of estates of decedents, or any other matters whatsoever to defeat, reduce or affect the right of Lenders under the Loan Documents to a sale of the Collateral Properties for the collection of the Obligations without any prior or different resort for collection or of the right of Lenders to the payment of the Obligations out of the net proceeds of the Collateral Properties

in preference to every other claimant whatsoever. In addition, each Borrower, for itself and its successors and assigns, waives in the event of foreclosure of any or all of the Mortgages, any equitable right otherwise available to Borrowers which would require the separate sale of the Collateral Properties or require Lenders to exhaust their remedies against the Collateral Properties or any combination of the Collateral Properties before proceeding against any other Collateral Property or combination of Collateral Properties; and further in the event of such foreclosure Borrowers hereby expressly consent to and authorize, at the option of Lenders, the foreclosure and sale either separately or together of any combination of the Collateral Properties.

12.24. Conflict; Construction of Documents; Reliance. In the event of any conflict between the provisions of this Agreement and any of the other Loan Documents, the provisions of this Agreement shall control. The parties hereto acknowledge that they were represented by competent counsel in connection with the negotiation, drafting and execution of the Loan Documents and that such Loan Documents shall not be subject to the principle of construing their meaning against the party which drafted same. Each Borrower acknowledges that, with respect to the Loan, each Borrower shall rely solely on its own judgment and advisors in entering into, the Loan without relying in any manner on any statements, representations or recommendations of Lenders or any parent, subsidiary or Affiliate of any Lender. Lenders shall not be subject to any limitation whatsoever in the exercise of any rights or remedies available to it under any of the Loan Documents or any other agreements or instruments which govern the Loan by virtue of the ownership by it or any parent, subsidiary or Affiliate of any Lender of any equity interest any of them may acquire in any Borrower, and each Borrower hereby irrevocably waives the right to raise any defense or take any action on the basis of the foregoing with respect to Lenders' exercise of any such rights or remedies. Each Borrower acknowledges that Lenders engage in the business of real estate financings and other real estate transactions and investments which may be viewed as adverse to or competitive with the business of Borrowers or their Affiliates.

12.25. <u>Brokers and Financial Advisors</u>. Borrowers and Lenders hereby represent that they have dealt with no financial advisors, brokers, underwriters, placement agents, agents or finders in connection with the transactions contemplated by this Agreement. Borrowers hereby agree to indemnify, defend and hold Lenders harmless from and against any and all claims, liabilities, costs and expenses of any kind (including Lenders' reasonable attorneys' fees and expenses) in any way relating to or arising from a claim by any Person that such Person acted on behalf of Borrowers in connection with the transactions contemplated herein. The provisions of this <u>Section 12.25</u> shall survive the expiration and termination of this Agreement and the payment of the Obligations.

12.26. <u>Time of the Essence</u>. Time is of the essence in the Loan Documents.

12.27. ENTIRE AGREEMENT. THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT AMONG THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS AMONG THE PARTIES.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first above written.

FELCOR/JPM HOSPITALITY (SPE), L.L.C.,

as a Borrower

By:	/s/Allison S. Navitskas
Name:	Allison S. Navitskas
Title:	Vice President

DJONT/JPM HOSPITALITY LEASING (SPE), L.L.C., as a Borrower

By:	/s/Allison S. Navitskas
Name:	Allison S. Navitskas
Title:	Vice President

FELCOR/JPM BOCA RATON HOTEL, L.L.C.,

as a Borrower

By:	/s/Allison S. Navitskas
Name:	Allison S. Navitskas
Title:	Vice President

DJONT/JPM BOCA RATON LEASING, L.L.C., as a Borrower

By:

/s/Allison S. Navitskas

Name: Title:

Allison S. Navitskas Vice President

Solely with respect to the representations, warranties and covenants set forth in <u>Section 6.34</u> hereof:

FELCOR/JPM HOSPITALITY HOLDCO (SPE), L.L.C.,

as Principal of Hospitality Owner

By:	/s/Allison S. Navitskas
Name:	Allison S. Navitskas
Title:	Vice President

DJONT/JPM HOSPITALITY LEASING HOLDCO (SPE), L.L.C.,

as Principal of Hospitality Operating Lessee

By:	/s/Allison S. Navitskas
Name:	Allison S. Navitskas
Title:	Vice President

FELCOR LODGING LIMITED PARTNERSHIP,

as Principal of Boca Owner

By: FelCor Lodging Trust Incorporated, its General Partner

By:/s/Allison S. NavitskasName:Allison S. NavitskasTitle:Vice President

FELCOR TRS HOLDINGS, LLC,

as Principal of Boca Operating Lessee

By:	/s/Allison S. Navitskas
Name:	Allison S. Navitskas
Title:	Vice President

Solely with respect to the representations, warranties and covenants set forth in <u>Section 7.23(e)</u> hereof:

Philadelphia Affiliate Ground Lessor:

FELCOR PENNSYLVANIA COMPANY, L.L.C.,

By: Name: Title: /s/Allison S. Navitskas Allison S. Navitskas Vice President

Boca Raton Affiliate Ground Lessor:

FELCOR LODGING LIMITED PARTNERSHIP,

By:

FelCor Lodging Trust Incorporated, its General Partner

By:/s/Allison S. NavitskasName:Allison S. NavitskasTitle:Vice President

JPMORGAN CHASE BANK, N.A.,

as Administrative Agent and as a Lender

By: Name: Title: /s/Mohammad S. Hassan

Mohammad S. Hassan Vice President

BANK OF AMERICA, N.A.

By:/sName: \overline{L} Title:S

/s/Lesa J. Butler Lesa J. Butler Senior Vice President

CITICORP NORTH AMERICA, INC.

By: Name: Title: /s/John Rowland John Rowland Director

DEUTSCHE BANK TRUST COMPANY AMERICAS

By: Name: Title: /s/James Rolison James Rolison Managing Director

By: Name: Title: /s/George R. Reynolds George R. Reynolds Director

THE BANK OF NOVA SCOTIA

By: Name: Title: /s/George Sherman George Sherman

Director

GOLDMAN SACHS BANK USA

By: Name: Title: /s/Mark Walton Mark Walton AUthorized Signatory

CREDIT SUISSE AG, CAYMAN ISLANDS BRANCH

By:	
Name:	
Title:	

/s/Mikhail Faybusovich Mikhail Faybusovich Director

By: Name: Title: /s/Vipul Dhadda Vipul Dhadda Associate

EXHIBIT B-1

FORM OF REVOLVING NOTE

\$

March 4, 2011

FOR VALUE RECEIVED, the undersigned (the "<u>Borrowers</u>"), HEREBY JOINTLY AND SEVERALLY PROMISE TO PAY to the order of _______ or its registered assigns (the "<u>Lender</u>"), in accordance with the provisions of the Agreement (as hereinafter defined), the principal amount of the Revolving Loan from time to time made by the Lender to the Borrowers under that certain Revolving Credit Agreement, dated as of March 4, 2011 (as amended, restated, extended, supplemented or otherwise modified in writing from time to time, the "<u>Agreement</u>", the capitalized terms being used herein and not defined, have the meanings set forth in such Agreement as therein defined), among the Borrowers, the Lenders from time to time party thereto, and JPMorgan Chase Bank, N.A., as Administrative Agent.

Each Borrower jointly and severally promises to make payments of principal in the amounts and at the times required by the Agreement to the Administrative Agent for the benefit of Lenders. The Borrowers jointly and severally promise to pay interest on the unpaid principal amount of the Revolving Loan made by the Lender from the date of such Revolving Loan until such principal amount is paid in full, at such interest rates (which shall not exceed the maximum rate permitted by applicable law) and at such times as provided in the Agreement.

All payments of principal and interest shall be made to the Administrative Agent for the account of the Lender in Dollars in immediately available funds at the Administrative Agent's office. If any amount is not paid in full when due hereunder, such unpaid amount shall bear interest, to be paid upon demand, from the due date thereof until the date of actual payment (and before as well as after judgment) computed at the per annum rate set forth in the Agreement. Each Revolving Loan made by the Lender to the Borrowers pursuant to the Agreement, and all payments made on account of principal thereof, shall be recorded by the Lender on its books and records and, if the Lender so elects in connection with any transfer or enforcement hereof, appropriate notations to evidence the foregoing information with respect to each such Revolving Loan then outstanding may be endorsed by the Lender on the <u>Schedule A</u> attached hereto, or on a continuation of such schedule attached to and made a part hereof; <u>provided</u> that the failure of the Lender to make any such recordation or endorsement shall not affect the obligations of the Borrowers hereunder or under the Agreement.

This Note is one of the Notes referred to in the Agreement, is entitled to the benefits thereof and may be prepaid in whole or in part or the Obligations accelerated or extended, subject to the terms and conditions provided therein. This Note is also entitled to the benefits of each Guaranty and is secured by the Collateral. The terms and conditions of the Agreement are hereby incorporated in their entirety herein by reference as though fully set forth herein. Upon the occurrence and continuation of one or more of the Events of Default specified in the Agreement, all amounts then remaining unpaid on this Note shall become, or may be declared to be, immediately due and payable all as provided in the Agreement. The Revolving Loan made by the Lender shall be evidenced by one or more loan accounts or records maintained by the

Lender in the ordinary course of business, including <u>Schedule A</u> hereto. The Lender may also attach additional schedules to this Note and endorse thereon the date, amount and maturity of its Revolving Loans and payments with respect thereto.

The Borrowers, for themselves, and their successors and assigns, hereby waive diligence, presentment, protest and demand and notice of protest, intention to accelerate, acceleration, demand, dishonor and non-payment of this Note.

REMAINDER OF PAGE INTENTIONALLY LEFT BLANK

SIGNATURE PAGE TO FOLLOW

THIS NOTE SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK.

Borrowers:

FELCOR/JPM HOSPITALITY (SPE), L.L.C.,

a Delaware limited liability company

By:		
Name:		
Title:		

DJONT/JPM HOSPITALITY LEASING (SPE), L.L.C.,

a Delaware limited liability company

By:		
Name:		
Title:		

FELCOR/JPM BOCA RATON HOTEL, L.L.C.,

a Delaware limited liability company

By:	
Name:	
Title:	

DJONT/JPM BOCA RATON LEASING, L.L.C.,

a Delaware limited liability company

By:	
Name:	
Title:	

SCHEDULE A

<u>REVOLVING LOANS</u> AND PAYMENTS WITH RESPECT THERETO

	Amour	nt of	Amoun	t of		Aggrega	te Unpaid	
	Revolvin	g Loan	Revolving Lo	an Repaid		Revolving l	Loan Balance	
Date	Mac	le						
	ABR Loan	Eurodollar Loan	ABR Loan	Eurodoll ar Loan	ABR Loan	Eurodollar Loan	Total Aggregate Unpaid Revolving Loan Balance	Notation Made By

Exhibit 10.3

EXECUTION VERSION

New York, New York As of March 4, 2011

GUARANTY Agreement

THIS GUARANTY (this "<u>Guaranty</u>") is entered into as of March 4, 2011, by FELCOR LODGING TRUST INCORPORATED, a Maryland corporation ("<u>FelCor Trust</u>"), and FELCOR LODGING LIMITED PARTNERSHIP, a Delaware limited partnership (jointly and severally referred to herein with FelCor Trust as "<u>Guarantors</u>"), in favor of JPMORGAN CHASE BANK, N.A., as administrative agent (in such capacity, "<u>Administrative Agent</u>") for the banks and other financial institutions ("<u>Lenders</u>") that are parties to the Credit Agreement described below.

1. FelCor/JPM Hospitality (SPE), L.L.C., a Delaware limited liability company, DJONT/ JPM Hospitality Leasing (SPE), L.L.C., a Delaware limited liability company, FelCor/JPM Boca Raton Hotel, L.L.C., a Delaware limited liability company, and DJONT/JPM Boca Raton Leasing, L.L.C., a Delaware limited liability company (each, a "Borrower" and collectively, "Borrowers"), are parties to the Revolving Credit Agreement dated as of March 4, 2011 (as the same may be amended, supplemented or modified from time to time, the "<u>Credit Agreement</u>"), among Borrowers, Administrative Agent, and Lenders pursuant to which Lenders have agreed, among other things, to make a Loan to Borrower.

2. Guarantors have direct and indirect equity ownership interests in Borrowers, and Guarantor will derive substantial direct and indirect benefits from the making of the Loan.

3. It is a condition precedent to the obligations of Lenders to make the Loan to Borrowers that Guarantors shall execute and deliver this Guaranty in favor of Administrative Agent, for the benefit of Lenders.

NOW THEREFORE, FOR VALUE RECEIVED, the sufficiency of which is hereby acknowledged, and in consideration of the Loan, any credit and/or financial accommodation heretofore or hereafter from time to time made or granted to Borrowers under the Loan Documents by Administrative Agent and Lenders, each Guarantor hereby furnishes its guaranty of the Guaranteed Obligations (as hereinafter defined) as follows:

Section 1. <u>Certain Definitions</u>. All capitalized words and phrases not otherwise defined herein shall have the meanings ascribed to them in the Credit Agreement.

Section 2. <u>Nature and Scope of Guaranty</u>. Each Guarantor hereby absolutely and unconditionally guarantees to Administrative Agent the prompt and unconditional payment of the Guaranteed Obligations (as hereinafter defined) when due (after any applicable notice, grace, or cure periods, if any, set forth in the Loan Documents), subject to the terms hereof. The term "<u>Guaranteed</u> <u>Obligations</u>" as used in this Guaranty shall mean all Obligations arising under the Credit Agreement, any Note, and the other Loan Documents.

It is expressly understood and agreed that this is a continuing guaranty and that the obligations of Guarantors hereunder are and shall be absolute under any and all circumstances, without regard to the validity, regularity or enforceability of any Note, the Credit Agreement, or the other Loan

Documents, a true copy of each of said documents Guarantors hereby acknowledge having received and reviewed.

Subrogation. Any indebtedness of a Borrower to either Guarantor now or Section 3. hereafter existing (including, but not limited to, any rights to subrogation Guarantor may have as a result of any payment by such Guarantor under this Guaranty), together with any interest thereon, shall be, and such indebtedness is, hereby deferred, postponed and subordinated to the prior payment in full of the Obligations. Until payment in full of the Obligations (and including interest accruing on any Note after the commencement of a proceeding by or against a Borrower under any Debtor Relief Laws and the regulations adopted and promulgated pursuant thereto, which interest the parties agree shall remain a claim that is prior and superior to any claim of Guarantors notwithstanding any contrary practice, custom or ruling in cases under any Debtor Relief Laws generally), Guarantors agree not to accept any payment or satisfaction of any kind of indebtedness of any Borrower to Guarantors and hereby assign such indebtedness to Administrative Agent, including the right to file proof of claim and to vote thereon in connection with any such proceeding under any Debtor Relief Laws, including the right to vote on any plan of reorganization. Further, each Guarantor agrees that until such payment in full of the Obligations, (a) neither of them shall accept payment from the other by way of contribution on account of any payment made hereunder by such party to Administrative Agent, (b) neither of them will take any action to exercise or enforce any rights to such contribution, and (c) if either Guarantor should receive any payment, satisfaction or security for any indebtedness of a Borrower to either Guarantor or for any contribution by the other to such Guarantor for payment made hereunder by the recipient to Administrative Agent, the same shall be delivered to Administrative Agent in the form received, endorsed or assigned as may be appropriate for application on account of, or as security for, the Obligations, and until so delivered, shall be held in trust for Administrative Agent as security for the Obligations.

Section 4. <u>Expenses of Enforcement</u>. Each Guarantor agrees that such Guarantor will reimburse Administrative Agent, to the extent that such reimbursement is not made by a Borrower or any other Loan Party, for all expenses (including counsel fees and disbursements) incurred by Administrative Agent in connection with the collection of the Guaranteed Obligations or any portion thereof or with the enforcement of this Guaranty.

Section 5. <u>Application of Monies</u>. Subject to the terms of the Credit Agreement, all monies available to Administrative Agent for application in payment or reduction of the Obligations may be applied by Administrative Agent in such manner and in such amounts and at such time or times and in such order and priority as Administrative Agent may see fit to the payment or reduction of such portion of the Obligations as Administrative Agent may elect.

Section 6. <u>Waivers</u>. Each Guarantor hereby waives notice of the acceptance hereof, presentment, demand for payment, protest, notice of protest, or any and all notice of non-payment, non-performance or non-observance, or other proof, notice of intention to accelerate, notice of acceleration, or notice or demand, whereby to charge either Guarantor therefor.

Each Guarantor further agrees that the validity of this Guaranty and the obligations of each Guarantor hereunder shall in no way be terminated, affected or impaired (a) by reason of the assertion by Administrative Agent of any rights or remedies which it may have under or with respect to either

the Credit Agreement, any Note, or the other Loan Documents, against any Person obligated thereunder or the Collateral Properties covered under the Credit Agreement, or (b) by reason of any failure to file or record any of such instruments or to take or perfect any security intended to be provided thereby, or (c) by reason of the release of any of the Collateral Properties covered under the Credit Agreement or other collateral for the Loan) or (d) by reason of Administrative Agent's failure to exercise, or delay in exercising, any such right or remedy or any right or remedy Administrative Agent may have hereunder or in respect to this Guaranty, or (e) by reason of the commencement of a case under any Debtor Relief Law by or against any Person obligated under the Credit Agreement, any Note, or the other Loan Documents, or (f) by reason of any transfer or assignment of the Loan by Administrative Agent or (g) by reason of any payment made on the Obligations or any other indebtedness arising under the Credit Agreement, any Note, or the other Loan Documents, whether made by a Borrower or either Guarantor or any other Person, which is required to be refunded pursuant to any Debtor Relief Law; it being understood that no payment so refunded shall be considered as a payment of any portion of the Obligations, nor shall it have the effect of reducing the liability of either Guarantor hereunder. It is further understood, that if a Borrower shall have taken advantage of, or be subject to the protection of, any provision in any Debtor Relief Law, the effect of which is to prevent or delay Administrative Agent from taking any remedial action against a Borrower, including the exercise of any option Administrative Agent has to declare the Obligations due and payable on the happening of any Event of Default or event by which under the terms of the Credit Agreement, any Note, or the other Loan Documents, the Obligations shall become due and payable, Administrative Agent may, as against either Guarantor, nevertheless, declare the Obligations due and payable and enforce any or all of its rights and remedies against either Guarantor provided for herein.

Section 7. <u>Covenants and Representations</u>. Each Guarantor further covenants that this Guaranty shall remain and continue in full force and effect as to any modification, extension or renewal of the Credit Agreement, any Note, or the other Loan Documents, that Administrative Agent shall not be under a duty to protect, secure or insure any Collateral Property covered under the Credit Agreement, and that other indulgences or forbearance may be granted under any or all of such documents, all of which may be made, done or suffered without notice to, or further consent of, each Guarantor.

Each Guarantor represents and warrants to Administrative Agent and Lenders that:

(a) Each Guarantor (i) is duly organized or formed, validly existing and, as applicable, in good standing under the Laws of the jurisdiction of its incorporation or organization, and (ii) has all requisite power and authority and all requisite governmental licenses, authorizations, consents and approvals to execute, deliver and perform its obligations under this Guaranty and consummate the transactions contemplated hereby;

(b) Each Guarantor has taken all necessary action to authorize the execution, delivery and performance of this Guaranty;

(c) this Guaranty has been duly executed and delivered by or on behalf of each Guarantor and constitutes legal, valid and binding obligations of each Guarantor enforceable against each Guarantor in accordance with its terms, subject only to applicable Debtor Relief Laws, and subject, as to enforceability, to general principles of equity (regardless of whether enforcement is sought in a proceeding in equity or at law); and

(d) the execution, delivery and performance of this Guaranty by each Guarantor will not conflict with or result in a breach of any of the terms or provisions of, or constitute a default under, or result in the creation or imposition of any Lien, charge or encumbrance (other than pursuant to the Loan Documents) upon any of the property or assets of such Guarantor pursuant to the terms of its Organizational Documents, any indenture, mortgage, deed of trust, loan agreement, partnership agreement, management agreement, franchise agreement, or other agreement or instrument to which such Guarantor is a party or by which any of such Guarantor's property or assets is subject, nor will such action result in any violation of the provisions of any applicable statute or any applicable order, rule or regulation of any court or Governmental Authority or body having jurisdiction over such Guarantor or any of such Guarantor's other assets, and any consent, approval, authorization, order, registration or qualification of or with any Governmental Authority required for the execution, delivery and performance by each Guarantor's noncompliance with this <u>clause (d)</u> would reasonably be expected to have a Material Adverse Effect.

As a further inducement to Administrative Agent to make the Loan and in consideration thereof, each Guarantor further covenants and agrees (a) that in any action or proceeding brought by Administrative Agent against such Guarantor on this Guaranty, each Guarantor shall and does hereby waive trial by jury, (b) that a New York State District Court in New York County, New York, or, in a case involving diversity of citizenship, the United States District Court located in the Borough of Manhattan in New York, New York, shall have exclusive jurisdiction of any such action or proceeding, and (c) that service of any summons and complaint or other process in any such action or proceeding may be made by registered or certified mail directed to each Guarantor at such Guarantor's address set forth in <u>Section 9</u> below, each Guarantor waiving personal service thereof. Nothing in this Guaranty will be deemed to preclude Administrative Agent from bringing an action or proceeding with respect hereto in any other jurisdiction.

Section 8. <u>Guaranty of Payment</u>. This is a guaranty of payment, and not of collection, and upon any Event of Default under the Credit Agreement, any Note, or the other Loan Documents, Administrative Agent may, at its option, proceed directly and at once, without notice to any Borrower, against each Guarantor to collect and recover the full amount of the liability hereunder or any portion thereof, without proceeding against any Borrower or any other Person, or foreclosing upon, selling, or otherwise disposing of or collecting or applying against any of the mortgaged property or other collateral for the Loan. Each Guarantor hereby waives the pleading of any statute of limitations as a defense to the obligation hereunder.

Section 9. <u>Notices</u>. All notices required or permitted hereunder shall be given and shall become effective as provided in the Credit Agreement. Notices to each Guarantor shall be addressed as follows:

c/o FelCor Lodging Trust Incorporated 545 E. John Carpenter Freeway, Suite 1300 Irving, Texas 75062 Attention: General Counsel Facsimile No.: (972) 444-4949 With a copy to: Akin Gump Strauss Hauer & Feld LLP 1700 Pacific Avenue, Suite 4100 Dallas, TX 75201-4675 Attention: Robert W. Dockery, Esq. Facsimile No.: (214) 969-4343

Section 10. <u>Successors and Assigns; Change in Organization</u>. Each reference herein to Administrative Agent shall be deemed to include its successors and assigns, to whose favor the provisions of this Guaranty shall also inure. Each reference herein to Lenders shall be deemed to include their successors and assigns, to whose favor the provisions of this Guaranty shall also inure. Each reference herein to Guarantor shall be deemed to include the heirs, executors, administrators, legal representatives, successors and permitted assigns of such Guarantor, all of whom shall be bound by the provisions of this Guaranty.

If Guarantor is a partnership, the agreements herein contained shall remain in force and be applicable, notwithstanding any changes in the individuals or entities comprising the partnership, and the term "Guarantor," as used herein, shall include any alternate or successor partnership, but any predecessor partnership and its partners shall not thereby be released from any liability. If Guarantor is a corporation, the agreements contained herein shall remain in full force and be applicable notwithstanding any changes in the shareholders comprising, or the officers and directors relating to, the corporation, and the term "Guarantor" as used herein, shall include any alternative or successor corporation, but any predecessor corporation shall not be relieved of liability hereunder. Nothing in the foregoing paragraph shall be construed as a consent to, or a waiver of, any prohibition or restriction on transfers of interests in such partnership or corporation, which may be set forth in the Credit Agreement, each Mortgage or any other Loan Document.

Section 11. <u>Authority</u>. Each Guarantor (and its representatives, executing below, if any) has full power, authority and legal right to execute this Guaranty and to perform all its obligations under this Guaranty.

Section 12. <u>Amendments and Waivers</u>. This Guaranty may not be modified, amended, waived, extended, changed, discharged or terminated orally or by any act or failure to act on the part of Administrative Agent or a Borrower, but only by an agreement in writing signed by the party against whom enforcement of any modification, amendment, waiver, extension, change, discharge or termination is sought.

Section 13. <u>GOVERNING LAW</u>. THIS GUARANTY SHALL BE CONSTRUED IN ACCORDANCE WITH, AND THIS GUARANTY AND ALL CLAIMS AND CAUSES OF ACTION ARISING OUT OF THE TRANSACTIONS CONTEMPLATED HEREBY SHALL BE GOVERNED BY, THE LAWS OF THE STATE OF NEW YORK (OTHER THAN THOSE CONFLICT OF LAW PROVISIONS THAT WOULD DEFER TO THE SUBSTANTIVE LAWS OF ANOTHER JURISDICTION). WITHOUT IN ANY WAY LIMITING THE PRECEDING CHOICE OF LAW, THE PARTIES ELECT TO BE GOVERNED BY NEW YORK LAW IN ACCORDANCE WITH, AND ARE RELYING (AT LEAST IN PART) ON, SECTION 5-1401 OF THE GENERAL OBLIGATIONS LAW OF THE STATE OF NEW YORK. Section 14. <u>CONSENT TO JURISDICTION</u>. EACH GUARANTOR HEREBY IRREVOCABLY SUBMITS TO THE NON-EXCLUSIVE GENERAL JURISDICTION OF THE COURTS OF THE STATE OF NEW YORK, THE UNITED STATES DISTRICT COURTS LOCATED IN THE BOROUGH OF MANHATTAN IN NEW YORK, NEW YORK AND APPELLATE COURTS FROM ANY THEREOF. EACH GUARANTOR HEREBY WAIVES ANY RIGHT IT MAY HAVE TO ASSERT THE DOCTRINE OF FORUM NON CONVENIENS, TO ASSERT THAT IT IS NOT SUBJECT TO THE JURISDICTION OF SUCH COURTS OR TO OBJECT TO VENUE TO THE EXTENT ANY PROCEEDING IS BROUGHT IN ACCORDANCE WITH THIS SECTION.

Section 15. <u>Miscellaneous</u>. This Guaranty may be executed in one or more counterparts by some or all of the parties hereto, each of which counterparts shall be an original and all of which together shall constitute a single agreement of Guaranty. The failure of any party hereto to execute this Guaranty, or any counterpart hereof, shall not relieve the other signatories from their obligations hereunder.

This Guaranty constitutes the entire agreement of each Guarantor with respect to the matters set forth herein. The rights and remedies herein provided are cumulative and not exclusive of any remedies provided by law or any other agreement, and this Guaranty shall be in addition to any other guaranty of or collateral security for any of the Guaranteed Obligations. The provisions of this Guaranty are severable, and the invalidity or unenforceability of any one or more sections of this Guaranty shall not affect the validity or enforceability of its remaining provisions. Captions are for the ease of reference only and shall not affect the meaning of the relevant provisions. The meanings of all defined terms used in this Guaranty shall be equally applicable to the singular and plural forms of the terms defined.

Section 16. Exculpation. Notwithstanding anything to the contrary contained in this Guaranty, the only entities or persons with any financial or other obligation under this Guaranty, at law or in equity, are Guarantors and Administrative Agent, and Lenders shall look solely to the assets of Guarantors and its general partner(s) for the satisfaction of any claim arising under, or in connection with, this Guaranty, at law or equity. Except for any general partner of either Guarantor, no present or future Constituent Member (as hereinafter defined) in either Guarantor, nor any present or future shareholder, officer, director, employee, trustee, beneficiary, advisor, member, partner, principal, participant or agent of or in either Guarantor or of or in any person or entity that is or becomes a Constituent Member in either Guarantor, shall have any personal or other liability, directly or indirectly, under or in connection with this Guaranty. Administrative Agent and Lenders, on behalf of themselves and their respective successors and assigns, hereby waive any and all such personal or other liability. The term "Constituent Member", as used herein, shall mean any direct partner or member in the applicable Guarantor and any Person that, directly or indirectly through one or more other partnerships, limited liability companies, corporations or other entities, is a partner or member in such Guarantor except for any general partner of such Guarantor. Notwithstanding anything to the contrary contained in this Guaranty, neither the negative capital account of any Constituent Member in either Guarantor nor any obligation of any Constituent Member in either Guarantor to restore a negative capital account or to contribute or loan capital to either Guarantor or to any other Constituent Member in either Guarantor shall at any time be deemed to be the property or an asset

of either Guarantor (or any such other Constituent Member) and neither Administrative Agent nor Lenders nor any of their respective successors or assigns shall have any right to collect, enforce or proceed against any Constituent Member with respect to any such negative capital account or obligation to restore, contribute or loan.

Section 17. Termination; Reinstatement. This Guaranty is a continuing and irrevocable guaranty of all Guaranteed Obligations now or hereafter existing and shall remain in full force and effect until all Guaranteed Obligations and any other amounts payable under this Guaranty are indefeasibly paid in full in cash and any commitments of the Lenders or facilities provided by the Lenders with respect to the Guaranteed Obligations are terminated. Notwithstanding the foregoing, this Guaranty shall continue in full force and effect or be revived, as the case may be, if any payment by or on behalf of any Borrower or Guarantor is made, or Administrative Agent or any Lender exercises its right of setoff, in respect of the Guaranteed Obligations and such payment or the proceeds of such setoff or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by the Lender in its discretion) to be repaid to a trustee, receiver or any other party, in connection with any proceeding under any Debtor Relief Law, all as if such payment had not been made or such setoff had not occurred and whether or not Administrative Agent or the Lenders are in possession of or have released this Guaranty and regardless of any prior revocation, rescission, termination or reduction. The obligations of each Guarantor under this paragraph shall survive termination of this Guaranty.

Section 18. <u>California State-Specific Waiver</u>. In the event of any inconsistencies between the terms and conditions that follow and the other terms and conditions of this Guaranty, the following terms and conditions shall control and be binding. Each Guarantor hereby waives:

(a) Presentment, demand, protest, notice of protests, notice of dishonor, notice of intention to accelerate, notice of acceleration and notices of non-payment and notice of acceptance of this Guaranty;

(b) The right, if any, to the benefit of or to direct the application of, any security held by Administrative Agent, including any Collateral Properties; and all rights of subrogation, any right to enforce any remedy which either Guarantor now has or hereafter may have against a Borrower and any right to participate in any security now or hereafter held by Administrative Agent;

(c) The right to require Administrative Agent to proceed against any Borrower or to proceed against any security now or hereafter held by Administrative Agent or to pursue any other remedy in Administrative Agent's power;

(d) The benefits, if either Guarantor is entitled to any benefits, of any single-action legislation or of any or all anti-deficiency statutes or regulations or judicial interpretations thereof, including, but not limited to, any protection which may be afforded such Guarantor by California Code of Civil Procedure Sections 580a, 580b, 580d and 726, and any amendments or modifications thereto. Each Guarantor understands and agrees that by waiving the anti-deficiency protections referred to herein, such Guarantor can be held liable for a deficiency judgment following a non-judicial foreclosure sale (including a non-judicial foreclosure sale of a purchase money obligation)

even if the price paid for any Collateral Property at the non-judicial foreclosure sale is less than the fair value of such Collateral Property; and each Guarantor further understands and agrees that such Guarantor is waiving its defense that the price paid for any Collateral Property at a judicial foreclosure sale may not be equal to the fair value of such Collateral Property; and each Guarantor further understands and agrees that by such Guarantor waiving its right to a fair value hearing following the foreclosure sale that Administrative Agent can seek a deficiency against such Guarantor up to the entire amount of the sums guaranteed hereby less the amount paid for any Collateral Property at the non-judicial or judicial foreclosure sale;

(e) Any right of subrogation which either Guarantor may have under California law to seek reimbursement from a Borrower of any sums paid by such Guarantor to Administrative Agent pursuant to this Guaranty until the prior full and indefeasible repayment of the Loan in accordance with the Loan Documents;

(f) Any estoppel defense arising out of Section 580d of the California Code of Civil Procedure;

(g) Any defense arising out of absence, impairment or loss of any right of reimbursement or subrogation or other right or remedy of either Guarantor against a Borrower or against any security resulting from the exercise or election of any remedies by Administrative Agent, including the exercise of the power of sale under any Mortgage, and any defense arising by reason of any disability or other defense of a Borrower or by reason of the cessation, from any cause, of the liability of a Borrower;

(h) The benefit of or right to assert any statute of limitations affecting either Guarantor's liability hereunder or the enforcement thereof, including but not limited to the provisions of California Code of Civil Procedure Sections 580a and 726 that require that any action for a deficiency be brought within three (3) months after a foreclosure under any Mortgage;

(i) Any partial payment by any Borrower or other circumstances which operate to toll any statute of limitations as to any Borrower shall also operate to toll the statute of limitations as to each Guarantor;

(j) Any defense based upon any change in name, location, composition or structure of any Borrower, or any change in the type of business conducted by any Borrower, or any other change in the identity or legal status of any Borrower;

(k) Any defense based upon the failure (if any) of Administrative Agent to (i) obtain a similar guaranty from any other Person, or (ii) file a creditor's claim in the estate (in administration, bankruptcy or any other proceeding) of any Person;

(1) Any rights which either Guarantor may have under California Civil Code Sections 2809, 2810, 2819, 2822(a), 2845, 2849, 2850, 2899 and 3433; and

(m) Without limiting the foregoing, each Guarantor waives all rights and defenses that such Guarantor may have because Borrowers' debt is secured by real property. This means, among other things:

- (i) Administrative Agent may collect from either Guarantor without first foreclosing on any real or personal property collateral pledged by any Borrower.
- (ii) If Administrative Agent forecloses on any real property collateral pledged by any Borrower:
 - (A) The amount of the debt may be reduced only by the price for which that collateral is sold at the foreclosure sale, even if the collateral is worth more than the sale price.
 - (B) Administrative Agent may collect from either Guarantor even if Administrative Agent, by foreclosing on the real property collateral, has destroyed any right such Guarantor may have to collect from any Borrower.

This is an unconditional and irrevocable waiver of any rights and defenses each Guarantor may have because Borrowers' debt is secured by real property. These rights and defenses include, but are not limited to, any rights or defenses based upon Section 580a, 580b, 580d, or 726 of the California Code of Civil Procedure.

Each Guarantor waives all rights and defenses arising out of an election of remedies by Administrative Agent, even though the election of remedies, such as a nonjudicial foreclosure with respect to security for a guaranteed obligation, has destroyed such Guarantor's rights of subrogation and reimbursement against any Borrower by the operation of Section 580d of the California Code of Civil Procedure or otherwise.

[Signatures on Following Page]

IN WITNESS WHEREOF, each Guarantor has duly executed this Guaranty as of the date first above set forth.

FELCOR LODGING TRUST INCORPORATED,

a Maryland corporation

By: /s/ Allison S. Navitskas Name: Allison S. Navitskas Title: Vice President

FELCOR LODGING LIMITED PARTNERSHIP,

a Delaware limited partnership

- By: FelCor Lodging Trust Incorporated, its General Partner
- By: /s/ Allison S. Navitskas Name: Allison S. Navitskas Title: Vice President

Exhibit 10.4

(Space above this line for recording purposes only.) FELCOR LODGING LIMITED PARTNERSHIP,

[_____] and

1,

together, as mortgagor

to

JPMORGAN CHASE BANK, N.A.,

as Administrative Agent, as mortgagee

FEE AND LEASEHOLD MORTGAGE, ASSIGNMENT OF LEASES AND RENTS, SECURITY AGREEMENT AND FIXTURE FILING

Dated:

As of March ____, 2011

Location:

County:

PREPARED BY (WITH ASSISTANCE OF LOCAL COUNSEL)

AND UPON RECORDATION RETURN TO:

Bingham McCutchen LLP One Federal Street Boston, MA 02110-1726 Attention: Maurice H. Sullivan, III, Esq.

FEE AND LEASEHOLD MORTGAGE, ASSIGNMENT OF LEASES AND RENTS, SECURITY AGREEMENT AND FIXTURE FILING

THIS FEE AND LEASEHOLD MORTGAGE, ASSIGNMENT OF LEASES AND RENTS, SECURITY AGREEMENT AND FIXTURE FILING (this "<u>Security Instrument</u>") is made as of March _____, 2011 (the "<u>Effective Date</u>"), by FELCOR LODGING LIMITED PARTNERSHIP, a Delaware limited partnership ("<u>Fee Owner</u>"), [_____], a Delaware limited liability company ("<u>Owner</u>"), and [_____], a Delaware limited liability company ("<u>Owner</u>") (Fee Owner, Owner, and Operator are collectively and together with their permitted successors and assigns,

individually or collectively (as the context requires) referred to herein as "<u>Mortgagor</u>"), whose address is c/o FelCor Lodging Trust Incorporated, 545 E. John Carpenter Freeway, Suite 1300, Irving, Texas 75062, Attention: General Counsel, as mortgagor, for the benefit of JPMORGAN CHASE BANK, N.A. ("<u>Agent</u>"), as Administrative Agent for the benefit of the Lenders (defined below) from time to time parties to the Credit Agreement (defined below) (Agent, in such capacity, together with its successors and assigns, "<u>Mortgagee</u>"), whose address is 270 Park Avenue, New York, New York 10017, Attention: Joseph Geoghan, as mortgagee. All capitalized terms not defined herein shall have the respective meanings set forth in the Credit Agreement (defined below).

RECITALS:

A. Fee Owner is the fee simple owner of the real estate described on **Exhibit A** attached hereto and the lessor under the Ground Lease (as defined below).

B. Owner is the lessee under the Ground Lease and the lessor under the Operating Lease (as defined below), and Operator is the lessee under the Operating Lease.

C. Owner, Operator, [_____], and [____] (collectively, "**Borrower**"), as borrower, and Agent, as administrative agent for the banks and other financial institutions from time to time party thereto as lenders (Agent and/or such banks or such other financial institutions, collectively or individually together with each of their respective successors and assigns, "**Lenders**") have entered into that certain Revolving Credit Agreement dated as of the date hereof (as the same may be amended, restated, replaced, supplemented or otherwise modified and in effect from time to time, the "**Credit Agreement**").

D. Under the terms of the Credit Agreement, the Lenders have agreed to make a loan to and for the account of Borrower (whether one or more, collectively, the "Loan"), which Loan is evidenced by, among other things, certain promissory notes executed in connection with the Credit Agreement (such promissory note or promissory notes, together with all extensions, renewals, replacements, restatements or other modifications thereof, whether one or more being hereinafter collectively referred to as the "Notes"). The Credit Agreement, the Notes and the other documents now or hereafter executed in connection with, or to guaranty, evidence or secure, in whole or in part, the indebtedness and obligations evidenced thereby, as they may from time to time be amended, restated, replaced or otherwise modified and in effect from time to time, to be referred to herein as the "Loan Documents").

E. Mortgagor is required by the Credit Agreement to grant to Mortgagee, as security for the payment and performance of the indebtedness and obligations evidenced by the Loan Documents (collectively, the "**Obligations**") a valid, enforceable, first priority lien and security interest in the Property (as defined below). The Obligations include, without limitation, Borrower's obligations under the Loan Documents and Fee Owner's guarantee of Borrower's obligations under the Loan Documents pursuant to that certain Limited Recourse Guaranty of even date herewith executed by Fee Owner in favor of Agent for the benefit of the Lenders.

F. Lenders' commitments under the Credit Agreement and the Notes to make Revolving Loans shall be in the maximum aggregate amount of up to \$225,000,000, and it is the intention of Mortgagor and Mortgagee that this Security Instrument secure the payment of all such amounts and that all such amounts be included in the Obligations secured hereby.

G. This Security Instrument secures the Obligations.

H. It is in the best interest of Mortgagor to execute this Security Instrument inasmuch as Mortgagor will derive substantial direct and indirect benefits from the Loan.

ARTICLE I. GRANTS OF SECURITY

Section 1.1 <u>Property Mortgaged</u>. Pursuant to the terms of *Section 1.3* and *Section 1.5* below, Mortgagor does hereby irrevocably MORTGAGE, GRANT, BARGAIN, SELL, PLEDGE, ASSIGN, WARRANT, TRANSFER and CONVEY, subject to the Permitted Liens, to Mortgagee, with power of sale, and grant a security interest in and to the following property, rights, interests, and estates now owned or hereafter acquired by Mortgagor (collectively, the "<u>Property</u>"):

(a) <u>Land</u>. The real property described in <u>Exhibit A</u> attached hereto and made a part hereof (collectively, the "<u>Land</u>");

Ground Lease. That certain Ground Lease dated as of October 21, 2003 between (b) Fee Owner and Owner, covering the Land, a memorandum of which is recorded in Official Records Book 16543, Page 1725 of the Public Records of [_____ _], as amended by that certain First Amendment to Ground Lease dated on or about even date herewith between Fee Owner and Owner (as in effect from time to time, the "Ground Lease"), and the leasehold estate created thereby, including all assignments, modifications, extensions and renewals of the Ground Lease and all credits, deposits, options, privileges and rights of Owner as tenant under the Ground Lease, including but not limited to, rights of first refusal, if any, and the right, if any, to renew or extend the Ground Lease for a succeeding term or terms, and also including all the right title, claim or demand whatsoever of Owner either in law or in equity, in possession or expectancy, of, in and to Owner's rights, as tenant under the Ground Lease, to elect under Section 365(h)(1) of the Bankruptcy Code, Title 11 U.S.C.A. §101 et seq. (the "Bankruptcy Code") to terminate or treat the Ground Lease as terminated in the event (i) of the bankruptcy, reorganization or insolvency of the lessor thereunder, and (ii) the rejection of the Ground Lease by the lessor thereunder, as debtor in possession, or by a trustee for the lessor thereunder, pursuant to Section 365 of the Bankruptcy Code;

(c) <u>Additional Land</u>. All additional lands, estates and development rights hereafter acquired by Mortgagor for use in connection with the Land and the development of the Land and all additional lands and estates therein which may, from time to time, by supplemental mortgage or otherwise be expressly made subject to the Lien of this Security Instrument;

(d) <u>Improvements</u>. The buildings, structures, fixtures, additions, enlargements, extensions, modifications, repairs, replacements and improvements now or hereafter erected or located on the Land (collectively, the "<u>Improvements</u>");

(e) <u>Easements</u>. All easements, rights-of-way or use, rights, strips and gores of land, streets, ways, alleys, passages, sewer rights, water, water courses, water rights and powers, air rights and development rights, and all estates, rights, titles, interests, privileges, liberties, servitudes, tenements, hereditaments and appurtenances of any nature whatsoever, in any way now or hereafter belonging, relating or pertaining to the Land and the Improvements including, but not limited to, those arising under and by virtue of the Ground Lease and the reversions and remainders, and all land lying in the bed of any street, road or avenue, opened or proposed, in front of or adjoining the Land, to the center line thereof and all the estates, rights, titles, interests, dower and rights of dower, curtesy and rights of curtesy, property, possession, claim and demand whatsoever, both at law and in equity, of Mortgagor of, in and to the Land and the Improvements, and every part and parcel thereof, with the appurtenances thereto;

(f) Fixtures and Personal Property. All machinery, equipment, fixtures (including, but not limited to, all heating, air-conditioning, plumbing, lighting, communications and elevator fixtures, inventory and goods), inventory and articles of personal property and accessions thereof and renewals, replacements thereof and substitutions therefor (including, but not limited to, beds, bureaus, chiffonniers, chests, chairs, desks, lamps, mirrors, bookcases, tables, rugs, carpeting, drapes, draperies, curtains, shades, venetian blinds, screens, paintings, hangings, pictures, divans, couches, luggage carts, luggage racks, stools, sofas, chinaware, linens, pillows, blankets, glassware, silverware, food carts, cookware, dry cleaning facilities, dining room wagons, keys or other entry systems, bars, bar fixtures, liquor and other drink dispensers, icemakers, radios, television sets, intercom and paging equipment, electric and electronic equipment, dictating equipment, private telephone systems, medical equipment, potted plants, heating, lighting and plumbing fixtures, fire prevention and extinguishing apparatus, cooling and air-conditioning systems, elevators, escalators, fittings, plants, apparatus, stoves, ranges, refrigerators, laundry machines, tools, machinery, engines, dynamos, motors, boilers, incinerators, switchboards, conduits, compressors, vacuum cleaning systems, floor cleaning, waxing and polishing equipment, call systems, brackets, electrical signs, bulbs, bells, ash and fuel, conveyors, cabinets, lockers, shelving, spotlighting equipment, dishwashers, garbage disposals, washers and dryers), other customary hotel equipment and other tangible property of every kind and nature whatsoever owned by Mortgagor, or in which Mortgagor has or shall have an interest, now or hereafter located upon the Land and the Improvements, or appurtenant thereto, or usable in connection with the present or future operation and occupancy of the Land and the Improvements and all building equipment, materials and supplies of any nature whatsoever owned by Mortgagor, or in which Mortgagor has or shall have an interest, now or hereafter located upon the Land or the Improvements, or appurtenant thereto, or usable in connection with the present or future operation and occupancy of the Land and the Improvements (hereinafter collectively called the "Personal Property"), and the right, title and interest of Mortgagor in and to any of the Personal Property which may be subject to any security interests, as defined in the Uniform Commercial Code, as adopted and enacted by the state or states where any of the Property is located (the "Uniform Commercial Code"), or equipment leases superior in priority to the Lien of this Security Instrument and all proceeds and products of all of the above;

Operating Lease. That certain Lease Agreement dated as of February 28, 1996 (g) between FelCor Suites Limited Partnership, as lessor, and DJONT Operations, L.L.C., as lessee, as amended by that certain Amendment to Lease Agreement dated as of September 23, 1996, effective February 28, 2006, as further amended by that certain Omnibus Lease Amendment Agreement dated as of June 30, 1998 among FelCor Lodging Trust Incorporated, Fee Owner, FelCor Suites Limited Partnership, and the other lessors and lessees signatories thereto, as assigned pursuant to that certain Assignment and Assumption of Agreements dated as of October 21, 2003 by and between DJONT Operations, L.L.C., as assignor, and DJONT/JPM Boca Raton Leasing, L.L.C., as assignee, as further assigned pursuant to that certain Assignment and Assumption of Agreements dated as of October 23, 2003 by and between Fee Owner, as assignor, and FelCor/JPM Boca Raton Hotel, L.L.C., as assignee, a memorandum of which is recorded in Official Records Book 16543, Page 1732 of the Public Records of Palm Beach County, Florida (as amended and assigned and in effect from time to time, the "Operating Lease"), and the leasehold estate created thereby, including all assignments, modifications, extensions and renewals of the Operating Lease and all credits, deposits, options, privileges and rights of Operator as tenant under the Operating Lease, including but not limited to, rights of first refusal, if any, and the right, if any, to renew or extend the Operating Lease for a succeeding term or terms, and also including all the right title, claim or demand whatsoever of Operator either in law or in equity, in possession or expectancy, of, in and to Operator's rights, as tenant under the Operating Lease, to elect under Section 365(h)(1) of the Bankruptcy Code to

terminate or treat the Operating Lease as terminated in the event (i) of the bankruptcy, reorganization or insolvency of the lessor thereunder, and (ii) the rejection of the Operating Lease by the lessor thereunder, as debtor in possession, or by a trustee for the lessor thereunder, pursuant to Section 365 of the Bankruptcy Code;

(h) Leases and Rents. All leases, subleases, rental agreements, registration cards and agreements, if any, and other agreements whether or not in writing affecting the use, enjoyment or occupancy of the Land and/or the Improvements heretofore or hereafter entered into (including, without limitation, the Operating Lease) and all extensions, amendments and modifications thereto, whether before or after the filing by or against Mortgagor of any petition for relief under Creditor's Rights Laws (defined below) (collectively, the "Leases"), and all right, title and interest of Mortgagor, its successors and assigns therein and thereunder, including, without limitation, any guaranties of the lessees' obligations thereunder, cash or securities deposited thereunder to secure the performance by the lessees of their obligations thereunder and all rents, additional rents, revenues, issues, registration fees, if any, and profits (including all oil and gas or other mineral royalties and bonuses and all rents, revenues, bonus money, royalties, rights and benefits accruing to Mortgagor under all present and future oil, gas and mineral leases on any parts of the Land and the Improvements) from the Land and the Improvements, all income, rents, room rates, issues, profits, revenues, deposits, accounts and other benefits from the operation of the hotel on the Land and/or the Improvements, including, without limitation, all revenues and credit card receipts collected from guest rooms, restaurants, bars, mini-bars, meeting rooms, banquet rooms and recreational facilities and otherwise, all receivables, customer obligations, installment payment obligations and other obligations now existing or hereafter arising or created out of sale, lease, sublease, license, concession or other grant of the right of the possession, use or occupancy of all or any portion of the Land and/or Improvements, or personalty located thereon, or rendering of services by Mortgagor or any operator or manager of the hotel or the commercial space located in the Improvements or acquired from others including, without limitation, from the rental of any office space, retail space, commercial space, guest room or other space, halls, stores or offices, including any deposits securing reservations of such space, exhibit or sales space of every kind, license, lease, sublease and concession fees and rentals, health club membership fees, food and beverage wholesale and retail sales, service charges, vending machine sales and proceeds, if any, from business interruption or other loss of income insurance relating to the use, enjoyment or occupancy of the Land and/or the Improvements whether paid or accruing before or after the filing by or against Mortgagor of any petition for relief under Creditors Rights Laws (the "Rents"), and all proceeds from the sale or other disposition of the Leases and the right to receive and apply the Rents to the payment of the Obligations. As used herein, the term "Creditors Rights Laws" shall mean any existing or future Laws of any jurisdiction, domestic or foreign, relating to bankruptcy, insolvency, reorganization, conservatorship, arrangement, adjustment, winding-up, liquidation, dissolution, composition or other relief with respect to its debts or debtors;

(i) <u>Insurance Proceeds</u>. All insurance proceeds in respect of the Property under any insurance policies covering the Property whether required by this Security Instrument or the Credit Agreement or not, including, without limitation, the right to receive and apply the proceeds of any insurance, judgments, or settlements made in lieu thereof, for damage to the Property (collectively, the "<u>Insurance Proceeds</u>");

(j) <u>Condemnation Awards</u>. All condemnation awards, including interest thereon, which may heretofore and hereafter be made with respect to the Property by reason of any taking or condemnation, whether from the exercise of the right of eminent domain (including, but not limited

to, any transfer made in lieu of or in anticipation of the exercise of the right), or for a change of grade, or for any other injury to or decrease in the value of the Property (collectively, the "<u>Awards</u>");

(k) <u>Tax Certiorari</u>. All refunds, rebates or credits in connection with reduction in real estate taxes and assessments charged against the Property as a result of tax certiorari or any applications or proceedings for reduction;

(1) <u>Rights</u>. The right, in the name and on behalf of Mortgagor, to appear in and defend any action or proceeding brought with respect to the Property and to commence any action or proceeding to protect the interest of Mortgagee or Lenders in the Property;

(m) <u>Agreements</u>. All other agreements, management agreements, operating agreements, franchise agreements, license agreements, contracts, certificates, chattel paper (whether tangible or electronic), instruments, franchises, permits, licenses, plans, specifications and other documents (including electronic documents), now or hereafter entered into, and all rights therein and thereto, respecting or pertaining to the use, occupation, construction, management or operation of any part of the Land and Improvements or any business or activity conducted on any part of the Land and Improvements including, but not limited to, the Management Agreement and Franchise Agreement and any and all agreements executed in connection therewith, and all right, title and interest of Mortgagor therein and thereunder, including, without limitation, the right, during the occurrence of any Event of Default, to receive and collect any sums payable to Mortgagor thereunder;

(n) <u>Intangibles</u>. All trade names, trademarks, service marks, logos, copyrights, goodwill, books and records, tenant or guest lists, advertising materials, telephone exchange numbers identified in such materials, and all other general intangibles relating to or used in connection with the operation of the Land, the Improvements and the Personal Property;

(0)Mortgagor Accounts. All right, title and interest of Mortgagor, if any, arising from the operation of the Land and the Improvements in and to all payments for goods or property sold, leased or occupied or for services rendered, whether or not yet earned by performance, and not evidenced by an instrument or chattel paper (hereinafter referred to as "Accounts Receivable") including, without limiting the generality of the foregoing, (i) all accounts, contract rights, book debts, and notes arising from the operation of a hotel on the Land and the Improvements or arising from the sale, lease or exchange of goods or other property and/or the performance of services, (ii) Mortgagor's rights to payment from any consumer credit/charge card organization or entities which sponsor and administer such cards as the American Express Card, the Visa Card and the MasterCard or other similar credit cards, (iii) Mortgagor's rights in, to and under all purchase orders for goods, services or other property, (iv) Mortgagor's rights to any goods, services or other property represented by any of the foregoing, (v) monies due to or to become due to Mortgagor under all contracts for the sale, lease or exchange of goods or other property and/or the performance of services including the right to payment of any interest or finance charges in respect thereto (whether or not yet earned by performance on the part of Mortgagor) and (vi) all collateral security and guaranties of any kind given by any Person with respect to any of the foregoing. Accounts Receivable shall include those now existing or hereafter created, substitutions therefor, proceeds (whether cash or non--cash, movable or immovable, tangible or intangible) received upon the sale, exchange, transfer, collection or other disposition or substitution thereof and any and all of the foregoing and proceeds therefrom;

(p) <u>Reserve Accounts</u>. All Accounts, Account Collateral, reserves, working capital, escrows and deposit accounts required under the Credit Agreement, the other Loan Documents, the Ground Lease, the Operating Lease, any management agreement or any other agreement assigned or collaterally assigned to Mortgagee, or otherwise maintained by Mortgagor, and all cash, checks, drafts, certificates, securities, investment property, financial assets, instruments and other property held therein from time to time and all proceeds, products, distributions or dividends or substitutions thereon and thereof;

(q) <u>Causes of Action</u>. All causes of action and claims (including, without limitation, all causes of action or claims arising in tort, by contract, by fraud or by concealment of material fact) against any Person for damages or injury to the Property or in connection with any transactions financed in whole or in part by the proceeds of the Loan ("<u>Cause of Action</u>");

(r) <u>Security Interests</u>. All right, title and interest of the lessor or lessee under the Ground Lease and the lessor or lessee under the Operating Lease as secured party in the personal property and collateral pursuant to any security interest granted by lessees or by operation of Laws thereunder;

(s) <u>Miscellaneous</u>. To the extent not set forth in this *Section 1.1*, the Collateral defined in *Section 1.3* below;

(t) <u>Proceeds</u>. All proceeds of any of the foregoing items set forth in *subsections* (*a*) through (*s*) above including, without limitation, Insurance Proceeds and Awards and Causes of Action which may at any time be converted into cash or liquidation claims; and

(u) <u>Other Rights</u>. Any and all other rights of Mortgagor in and to the items set forth in *subsections (a)* through *(t)* above.

Section 1.2. <u>ASSIGNMENT OF RENTS</u>. Mortgagor hereby absolutely and unconditionally assigns to Mortgagee all of Mortgagor's right, title and interest in and to all current and future Leases and Rents; it being intended by Mortgagor that this assignment constitutes a present, absolute assignment and not an assignment for additional security only. Nevertheless, subject to the terms of the Credit Agreement and *Section 8.1(h)* of this Security Instrument, Mortgagor shall hold the Rents, or a portion thereof sufficient to discharge all current sums due on the Obligations, for use in the payment of such sums, and (ii) enforce the terms of the Leases.

Section 1.3. <u>SECURITY AGREEMENT</u>. This Security Instrument is both a real property mortgage and a "security agreement" within the meaning of the Uniform Commercial Code. The Property includes both real and personal property and all other rights and interests, whether tangible or intangible in nature, of Mortgagor in the Property. By executing and delivering this Security Instrument, Mortgagor hereby grants to Mortgagee, as security for the Obligations, a security interest in the following properties, assets and rights of Mortgagor, wherever located, whether now owned or hereafter acquired or arising, and all proceeds and products thereof (all of the same being hereinafter called the "**Collateral**"): all personal and fixture property of every kind and nature including all goods (including inventory, equipment and any accessions thereto), instruments (including promissory notes), documents (including, if applicable, electronic documents), accounts, chattel paper (whether tangible or electronic), deposit accounts, letter-of-credit rights (whether or not the letter of credit is evidenced by a writing), commercial tort claims, securities and all other investment property, supporting obligations, any other contract rights or rights to the payment of money, insurance claims and proceeds, and all general intangibles (including all payment intangibles) (as each of

the foregoing terms is defined in the Uniform Commercial Code) to the full extent that the Collateral may be subject to the Uniform Commercial Code; <u>provided that</u>, with respect to Fee Owner, "Collateral" shall only include properties, assets and rights of Fee Owner located at or pertaining exclusively to 661 N.W. 53rd Street, Boca Raton, Florida 33487 and used in connection with the present or future operation, ownership, use, occupancy or management of the Property.

FIXTURE FILING. Without in any manner limiting the generality of any of the Section 1.4. other provisions of this Security Instrument: (a) some portions of the goods described or to which reference is made herein are or are to become fixtures on the Land described or to which reference is made herein or on Exhibit A attached to this Security Instrument; (b) this Security Instrument is to be filed of record in the real estate records as a financing statement and shall constitute a "fixture filing" for purposes of the Uniform Commercial Code; and (c) Fee Owner is the record fee owner of the real estate or interests in the real estate constituting the Property hereunder, subject to the Permitted Liens; and (d) Owner is the record owner of the leasehold estate in the real estate or interests in the real estate constituting the Property hereunder, subject to the Permitted Liens. Information concerning the security interest herein granted may be obtained at the addresses set forth on the first page hereof. This Security Instrument shall be effective as a financing statement filed as a fixture filing with respect to all fixtures included within the Property and is to be filed for record in the real property or other applicable records in the office of the Recorder where the Property (including said fixtures) is situated. This Security Instrument shall also be effective as a financing statement covering as-extracted minerals or the like (including oil and gas) and accounts subject to the applicable provisions of the Uniform Commercial Code of the State in which the Property is located. The address of the Debtor (Mortgagor) is set forth on the first page hereof and the address of the Secured Party (Mortgagee) is set forth below. In that regard, the following information is provided:

Name of First Debtor:	FelCor Lodging Limited Partnership
Type of Organization:	Limited partnership
State:	Delaware
Organizational ID Number:	75-2544994
Name of Secured Party:	JPMorgan Chase Bank, N.A., as Administrative Agent
Address of Secured Party:	270 Park Avenue, New York, New York 10017, Attention: Joseph Geoghan
Name of Second Debtor:	FelCor/JPM Boca Raton Hotel, L.L.C.
Type of Organization:	Limited liability company
State:	Delaware
Organizational ID Number:	20-0067256
Name of Secured Party:	JPMorgan Chase Bank, N.A., as Administrative Agent
Address of Secured Party:	270 Park Avenue, New York, New York 10017, Attention: Joseph Geoghan
Name of Third Debtor:	DJONT/JPM Boca Raton Leasing, L.L.C.
Type of Organization:	Limited liability company
State:	Delaware
Organizational ID Number:	20-0067187
Name of Secured Party:	JPMorgan Chase Bank, N.A., as Administrative Agent

Address of Secured Party:	270 Park Avenue, New York, New York 10017, Attention: Joseph Geoghan
Name of Third Debtor:	DJONT/JPM Boca Raton Leasing, L.L.C.
Type of Organization:	Limited liability company
State:	Delaware
Organizational ID Number:	20-0067187
Name of Secured Party:	JPMorgan Chase Bank, N.A., as Administrative Agent
Address of Secured Party:	270 Park Avenue, New York, New York 10017, Attention: Joseph Geoghan

Section 1.5. <u>CONDITIONS TO GRANT</u>. TO HAVE AND TO HOLD the above granted and described Property unto Mortgagee and to the use and benefit of Mortgagee and Lenders and their successors and assigns, forever; PROVIDED, HOWEVER, these presents are upon the express condition that, upon final payment and performance of the Obligations or the full and final release of this Security Instrument, these presents and the estate hereby granted shall cease, terminate and be void.

Section 1.6. <u>GRANTS TOMORTGAGEE</u>. This Security Instrument and the grants, assignments and transfers made to Mortgagee in this *Article 1* shall inure to Mortgagee solely in its capacity as Lenders' administrative agent under the Credit Agreement.

Section 1.7. <u>HOMESTEAD</u>. None of the Property forms any part of any property owned, used or claimed by Mortgagor as a residence or business homestead. None of the Property is exempt from forced sale under the Laws of the State of New York and, to the extent applicable, the state in which Land and Improvements are located. Mortgagor hereby disclaims and renounces each and every claim to the Property as a homestead.

ARTICLE 2. OBLIGATIONS SECURED

Section 2.1. <u>OBLIGATIONS</u>. This Security Instrument and the grants, assignments and transfers made in *Article 1* are given for the purpose of securing the Obligations, including any advances made by Mortgagee or any Lender for the construction, improvement, operation, repair, maintenance, preservation or protection of the Property and the security for payment of the Obligations, whether such future advances are obligatory or are made at Mortgagee's or such Lender's option, for any purpose, any other future advances by Mortgagee and Lenders to Mortgagor described in Section 20.2(d); and any other future advances by Mortgagee and Lenders to Mortgagor that recite that they are secured by this Security Instrument.

Section 2.2 <u>PAYMENT OF OBLIGATIONS</u>. Mortgagor will pay and perform the Obligations at the time and in the manner provided in the Credit Agreement and the other Loan Documents, subject to and as required by the terms and provisions thereof including, without limitation, any non-recourse provisions expressly set forth therein.

Section 2.3 <u>INCORPORATION BY REFERENCE</u>. All the covenants, conditions and agreements contained in (a) the Credit Agreement (including, without limitation, the exculpatory provisions set forth in *Section 12.09* thereof), (b) the Notes and (c) all and any of the other Loan Documents, are hereby made a part of this Security Instrument to the same extent and with the same force as if fully set forth herein.

ARTICLE 3. PROPERTY REPRESENTATIONS AND COVENANTS

Mortgagor represents, warrants, covenants and agrees as follows:

Section 3.1. <u>INSURANCE</u>. Mortgagor shall obtain and maintain, or cause to be obtained and maintained, in full force and effect at all times insurance with respect to Mortgagor and the Property as required pursuant to the Credit Agreement.

Section 3.2. <u>TAXES AND OTHER CHARGES</u>. Mortgagor shall pay all real estate and personal property taxes, assessments, water rates or sewer rents (collectively, "<u>Taxes</u>"), ground rents, maintenance charges, impositions (other than Taxes), and any Other Charges now or hereafter levied or assessed or imposed against the Property or any part thereof in accordance with the Credit Agreement.

Section 3.3. <u>LEASES</u>. Mortgagor shall not (and shall not permit any other applicable Person to) enter into or modify any Leases for all or any portion of the Property except in accordance with the provisions of the Credit Agreement.

Section 3.4.<u>WARRANTY OF TITLE</u>. Mortgagor has good and marketable title to the Property, subject only to the Permitted Liens, and has the right to mortgage, grant, bargain, sell, pledge, assign, warrant, transfer and convey the same. Fee Owner possesses an unencumbered fee simple absolute estate in the Land and the Improvements, subject only to the Permitted Liens. Owner possesses an unencumbered leasehold estate under the Ground Lease, subject only to the Permitted Liens. Operator possesses an unencumbered subleasehold estate under the Operating Lease, subject only to the Permitted Liens. Subject to the Permitted Liens, Mortgagor shall forever warrant, defend and preserve the title and the validity and priority of the Lien of this Security Instrument and shall forever warrant and forever defend the same to Mortgagee and/or Lenders, as applicable, against the claims of all Persons whatsoever. This Security Instrument, when properly recorded in the appropriate records, together with any Uniform Commercial Code financing statements required to be filed in connection therewith, will create (a) a valid, perfected first priority Lien on the Property, subject only to Permitted Liens and the Liens created by the Loan Documents and (b) perfected security interests in and to, and perfected collateral assignments of, all personalty (including the Leases), all in accordance with the terms thereof, in each case subject only to any applicable Permitted Liens.

Section 3.5. <u>PAYMENT FOR LABOR AND MATERIALS</u>.

(a) Subject to *Section 3.5(b)* below, Mortgagor will promptly pay (or cause to be paid) when due all bills and costs for labor, materials, and specifically fabricated materials incurred in connection with the Property (each, a "<u>Work Charge</u>") and never permit to exist beyond the due date thereof in respect of the Property or any part thereof any Lien or security interest other than the Permitted Liens, even though inferior to the Liens and the security interests hereof, and in any event never permit to be created or exist in respect of the Property or any part thereof any other or additional Lien or security interest other than the Liens or security interests hereof except for the Permitted Liens. Mortgagor represents there are no claims for payment for work, labor or materials affecting the Property which are or may become a Lien prior to, or of equal priority with, the Liens created by the Loan Documents.

(b) After prior written notice to Mortgagee, Mortgagor, at its own expense, may contest by appropriate legal proceeding, promptly initiated and conducted in good faith and with due diligence, the validity of any Work Charge, the applicability of any Work Charge to Mortgagor or to the Property or any alleged non-payment of any Work Charge, provided that (i) no Event of Default has occurred and is continuing; (ii) such proceeding shall be permitted under and be conducted in accordance with the provisions of any instrument to which Mortgagor is subject and shall not constitute a default thereunder and such proceeding shall be conducted in accordance with all applicable Legal Requirements; (iii) neither the Property nor any part thereof or interest therein will be in danger of being sold, forfeited, terminated, cancelled or lost during the duration of such legal proceeding; (iv) Mortgagor shall promptly upon final determination thereof pay (or cause to be paid) any such Work Charge determined to be valid, applicable and unpaid; (v) such proceeding shall suspend the collection of such contested Work Charge from the Property or Mortgagor shall have paid the same (or shall have caused the same to be paid) under protest; and (vi) Mortgagor shall furnish (or cause to be furnished) such security as may be required in the proceeding by applicable Laws or Legal Requirements, or as may be reasonably requested by Mortgagee, to insure payment of such Work Charge, together with all interest and penalties payable in connection therewith. Mortgagee may apply any such security or part thereof, as necessary to pay for such Work Charge at any time when, in the sole but reasonable judgment of Mortgagee, the validity, applicability and non-payment of such Work Charge is finally established or the Property (or any part thereof or interest therein) shall be in danger of being sold, forfeited, terminated, cancelled or lost during or as a result of such legal proceeding or Work Charge.

MAINTENANCE AND USE OF PROPERTY, WASTE, USE. Mortgagor shall Section 3.6. cause the Property to be maintained in a good and safe condition and repair in accordance with the terms of the Credit Agreement. Subject to the terms of the Credit Agreement, the Improvements and the Personal Property shall not be removed, demolished or materially altered or expanded (except for normal replacement of the Personal Property) without the consent of Mortgagee. Subject to the terms of the Credit Agreement, Mortgagor shall promptly repair, replace or rebuild any part of the Property which may be destroyed by any casualty, or become damaged, worn or dilapidated or which may be affected by any condemnation and shall complete and pay for any structure at any time in the process of construction or repair on the Land. Subject to the terms of the Credit Agreement, Mortgagor shall not initiate, join in, acquiesce in, or consent to any change in any private restrictive covenant, zoning law or other public or private restriction, limiting or defining the uses which may be made of the Property or any part thereof. Subject to the provisions of the Credit Agreement with respect thereto, if under applicable zoning provisions the use of all or any portion of the Property is or shall become a nonconforming use, Mortgagor will not cause or permit the nonconforming use to be discontinued or the nonconforming Improvement to be abandoned without the express written consent of Mortgagee. Mortgagor shall not commit or suffer any waste of the Property or make any change in the use of the Property which will in any way materially increase the rise of fire or other hazard arising out of the operation of the Property, or take any action that might invalidate or give cause for cancellation of any policy, or do or permit to be done thereon anything that may in any way impair the value of the Property or the security of this Security Instrument. Mortgagor will not, without the prior written consent of Lender, permit any drilling or exploration for or extraction, removal, or production of any minerals from the surface or the subsurface of the Land, regardless of the depth thereof or the method of mining or extraction thereof. The Property shall be used only for a hotel and any ancillary uses relating thereto, and for no other uses without the prior written consent of Mortgagee.

Section 3.7 <u>GROUND LEASE</u>. Mortgagor represents that the description of the Ground Lease in *Section 1.1* above is a true, complete and accurate description of the Ground Lease.

ARTICLE 4. FURTHER ASSURANCES

Section 4.1. <u>COMPLIANCE WITH CREDIT AGREEMENT</u>. Mortgagor shall comply with all covenants set forth in the Credit Agreement relating to acts or other further assurances to be made on the part of Mortgagor in order to protect and perfect the Lien or security interest hereof upon, and in the interest of Mortgagee and Lenders in, the Property.

AUTHORIZATION TO FILE FINANCING STATEMENTS; POWER OF Section 4.2. ATTORNEY. Mortgagor hereby irrevocably authorizes the Mortgagee at any time and from time to time to file in any filing office in any Uniform Commercial Code jurisdiction any initial financing statements and amendments thereto and continuations thereof that (a) indicate the Collateral (i) as all assets of the Mortgagor or words of similar effect, regardless of whether any particular asset comprised in the Collateral falls within the scope of Article 9 of the Uniform Commercial Code of the state of Florida or such jurisdiction, or (ii) as being of an equal or lesser scope or with greater detail, and (b) provide any other information required by part 5 of Article 9 of the Uniform Commercial Code of the state of Florida or such other jurisdiction for the sufficiency or filing office acceptance of any financing statement or amendment, including (i) whether the Mortgagor is an organization, the type of organization and any organizational identification number issued to the Mortgagor and, (ii) in the case of a financing statement filed as a fixture filing or indicating Collateral as as-extracted collateral or timber to be cut, a sufficient description of real property to which the Collateral relates. The Mortgagor agrees to furnish any such information to the Mortgagee promptly upon the Mortgagee's request. Mortgagor also ratifies its authorization for Mortgagee to have filed any like initial financing statements, amendments thereto and continuations thereof, if filed prior to the date of this Security Instrument. Mortgagor hereby irrevocably constitutes and appoints Mortgagee and any officer or agent of Mortgagee, with full power of substitution, as its true and lawful attorneys-in-fact with full irrevocable power and authority in the place and stead of Mortgagor or in Mortgagor's own name to execute in Mortgagor's name any such documents and otherwise to carry out the purposes of this Section 4.2, to the extent that Mortgagor's authorization above is not sufficient and Mortgagor fails or refuses to promptly execute such documents. To the extent permitted by Laws, Mortgagor hereby ratifies all acts said attorneys-in-fact have lawfully done in the past or shall lawfully do or cause to be done in the future by virtue hereof. This power of attorney is a power coupled with an interest and shall be irrevocable until final payment and performance of the Obligations or the full and final release of this Security Instrument.

Section 4.3. <u>RECORDING OF SECURITY INSTRUMENT ETC</u>. Mortgagor, upon execution and delivery of this Security Instrument and thereafter, from time to time, will cause this Security Instrument and any of the other Loan Documents creating a Lien or evidencing the Lien hereof upon the Property and each instrument of further assurance to be filed, registered or recorded in such manner and in such places as may be required by any present or future Laws in order to publish notice of and fully to protect and perfect the Lien hereof upon, and the interest of Mortgagee in, the Property. Mortgagor will pay all taxes, filing, registration or recording fees, and all expenses incident to the preparation, execution, acknowledgment and/ or recording of this Security Instrument and the other Loan Documents, including any instrument of further assurance and any modification or amendment of the foregoing documents, and all federal state, county and municipal taxes, duties, imposts, assessments and charges arising out of or in connection with the execution and delivery of this Security Instrument and the other Loan Documents, including any instrument of further assurance and any modification or amendment of the foregoing documents, except where prohibited by Laws so to do.

ARTICLE 5. DUE ON SALE/ENCUMBRANCE

Section 5.1. <u>NO SALE/ENCUMBRANCE</u>. Except as and to the extent permitted by the Credit Agreement, Mortgagor shall not cause or permit a sale, conveyance, mortgage, deed, grant, bargain, encumbrance, pledge, assignment, or grant of any options with respect to, or any other transfer or disposition (directly or indirectly, voluntarily or involuntarily, by operation of Laws or otherwise, and whether or not for consideration or of record) of a legal or beneficial interest in the Property or any Restricted Party or any part thereof or interest therein.

ARTICLE 6. PREPAYMENT; RELEASE OF PROPERTY

Section 6.1. <u>PREPAYMENT</u>. The Obligations may not be prepaid in whole or in part except in strict accordance with the express terms and conditions of the Notes and the Credit Agreement.

Section 6.2. <u>RELEASE OF PROPERTY</u>. Mortgagor shall not be entitled to a release of any portion of the Property from the Lien of this Security Instrument except in accordance with the express terms and conditions of the Credit Agreement.

ARTICLE 7. DEFAULT

Section 7.1. <u>EVENT OF DEFAULT</u>. The term "<u>Event of Default</u>" as used in this Security Instrument shall have the meaning assigned to such term in the Credit Agreement.

ARTICLE 8. RIGHTS AND REMEDIES UPON DEFAULT

Section 8.1. <u>REMEDIES</u>. Upon the occurrence and during the continuance of any Event of Default, Mortgagor agrees that Mortgagee may take such action, without notice or demand, as it deems advisable to protect and enforce its rights against Mortgagor, and in and to the Property, including, but not limited to, the following actions, each of which may be pursued alternatively, concurrently or otherwise, at such time and in such order as Mortgagee may determine, in its sole discretion, without impairing or otherwise affecting the other rights and remedies of Mortgagee:

a. Mortgagee may declare the Obligations to be immediately due and payable. Notwithstanding the foregoing, if and to the extent the Credit Agreement provides for automatic acceleration of the Loan upon the occurrence of certain Events of Default, such provisions with respect to automatic acceleration shall govern and control, without any further notice, demand or other action by Lender, Mortgagee or any other Person.

b. With respect to foreclosure, judicial or otherwise, with respect to any of the Property:

1. Mortgagee may institute proceedings, judicial or otherwise, for the complete foreclosure of this Security Instrument under any applicable provision of Laws, in which case the Property or any interest therein may be sold for cash or upon credit in one or more parcels or in several interests or portions and in any order or manner.

2. Mortgagee may, with or without entry, to the extent permitted and pursuant to the procedures provided by applicable Laws, institute proceedings for the partial foreclosure of this Security Instrument, conducting the sale as herein provided, and without declaring the whole Obligations due, and provided that if sale is made because of default as hereinabove mentioned, such sale may be made subject to the unmatured part of the Notes and/or the Obligations secured hereby, and it is agreed that such sale, if so made, shall not in any manner affect any other Obligations secured hereby, but as to such other Obligations this Security Instrument and the Liens created hereby shall remain in full force and effect just as though no sale had been made under the provisions of this Section 8.1(b)(ii). It is further agreed that several sales may be made hereunder without exhausting the right of sale for any remaining Obligations secured hereby, it being the purpose to provide for a foreclosure and sale of the Property for any matured portion of any of the Obligations secured hereby or other items provided for herein without exhausting the power to foreclose and to sell the Property for any remaining Obligations secured hereby, whether matured at the time or subsequently maturing.

3. Mortgagee may sell the Property in whole or in part and in such parcels and order as Mortgagee may determine, and the right of sale hereunder shall not be exhausted by one or more sales, but successive sales may be had until all of the Property has been legally sold.

4. Mortgagee and/ or one or more Lenders may become the purchaser at any such sale if it is the highest bidder, and shall have the right, after paying or accounting for all costs of said sale or sales, to credit the amount of the bid upon the amount of the Obligations owing, in lieu of cash payment.

5. It shall not be necessary for Mortgagee to have constructively in its possession any part of the real or personal property covered by this Security Instrument, and the title and right of possession of said property shall pass to the purchaser or purchasers at any sale hereunder as fully as if the same had been actually present and delivered. Likewise, on foreclosure of this Security Instrument whether by power of sale herein contained or otherwise, Mortgagor or any person claiming any part of the Property by, through or under Mortgagor, shall not be entitled to a marshalling of assets or a sale in inverse order of alienation.

6. The recitals and statements of fact contained in any notice or in any conveyance to the purchaser or purchasers at any sale hereunder shall be prima facie evidence of the truth of such facts, and all prerequisites and requirements necessary to the validity of any such sale shall be presumed to have been performed.

7. To the extent permitted by applicable Laws, any sale under the powers granted by this Security Instrument shall be a perpetual bar against Mortgagor, its heirs, successors, assigns and legal representatives.

8. Mortgagee may sell for cash or upon credit the Property or any part thereof and all estate, claim, demand, right, title and interest of Mortgagor therein and rights of redemption thereof, pursuant to power of sale or otherwise, at one or more sales, as an entirety or in parcels, at such time and place, upon such terms and after such notice thereof as may be required or permitted by applicable Laws.

9. In the event of a sale, by foreclosure, power of sale or otherwise, of less than all of Property, this Security Instrument shall continue as a Lien and security interest on the remaining portion of the Property unimpaired and without loss of priority.

(c) In the event any sale hereunder is not completed or is defective in the opinion of Mortgagee or the holder of any part of the Obligations, to the extent permitted by applicable Laws, such sale shall not exhaust the power of sale hereunder, and Mortgagee or such holder shall have the right to cause a subsequent sale or sales to be made by Mortgagee.

(d) In the event of a foreclosure under the powers granted by this Security Instrument, Mortgagor and all other Persons in possession of any part of the Property shall be deemed tenants at will of the purchaser at such foreclosure sale and shall be liable for a reasonable rental for the use of the Property; and if any such tenants refuse to surrender possession of the Property upon demand, the purchaser shall be entitled to institute and maintain the statutory action of forcible entry and detainer and procure a writ of possession thereunder, and Mortgagor expressly waives all damages sustained by reason thereof.

(e) Mortgagee may institute an action, suit or proceeding in equity for the specific performance of any covenant, condition or agreement contained herein or in the Notes, the Credit Agreement or in the other Loan Documents.

(f) Mortgagee may recover judgment on the Notes either before, during or after any proceedings for the enforcement of this Security Instrument or the other Loan Documents.

(g) Mortgagee may apply for and shall be entitled to the appointment of a receiver, trustee, liquidator or conservator of the Property, without notice and without regard for the adequacy of the security for the Obligations and without regard for the solvency of Mortgagor, any Borrower or Guarantor or any other guarantor or indemnitor under the Loan or any other Person liable for the payment of the Obligations.

The license granted to Mortgagor under Section 1.2 hereof shall automatically be (h) revoked and, to the extent permitted by applicable Laws, Mortgagee may enter into or upon the Property, either personally or by its agents, nominees or attorneys and dispossess Mortgagor and its agents and servants therefrom, without liability for trespass, damages or otherwise and exclude Mortgagor and its agents or servants wholly therefrom, and take possession of all books, records and accounts relating thereto and Mortgagor agrees to surrender possession of the Property and of such books, records and accounts to Mortgagee upon demand, and thereupon Mortgagee may (i) use, operate, manage, control, insure, maintain, repair, restore and otherwise deal with all and every part of the Property and conduct the business thereat; (ii) complete any construction on the Property in such manner and form as Mortgagee deems advisable; (iii) make alterations, additions, renewals, replacements and improvements to or on the Property; (iv) exercise all rights and powers of Mortgagor with respect to the Property, whether in the name of Mortgagor or otherwise, including, without limitation, the right to make, cancel, enforce or modify Leases, obtain and evict tenants, and demand, sue for, collect and receive all Rents of the Property and every part thereof; (v) require Mortgagor to pay monthly in advance to Mortgagee, or any receiver appointed to collect the Rents, the fair and reasonable rental value for the use and occupation of such part of the Property as may be occupied by Mortgagor; (vi) require Mortgagor to vacate and surrender possession of the Property to Mortgagee or to such receiver and, in default thereof, Mortgagor may be evicted by summary proceedings or otherwise; and (vii) apply the receipts from the Property to the payment of the Obligations, in such order, priority and proportions as Mortgagee shall deem appropriate in its sole discretion after deducting therefrom all expenses (including reasonable attorneys' fees) incurred in connection with the aforesaid operations and all amounts necessary to pay the Taxes, Other Charges, insurance and other expenses in connection with the Property, as well as just and reasonable

compensation for the services of Mortgagee and Lenders, and their respective counsel, agents and employees. Notwithstanding the provisions of this *Section 8.1(h)* hereof, no credit shall be given by Mortgagee for any sum or sums received from the rents, issues and profits of the Property until the money collected is actually received by Mortgagee at its principal office, or at such other place as Mortgagee shall designate in writing, and no such credit shall be given for any uncollected rents or other uncollected amounts or bills, nor shall such credit be given for any rents, issues and profits derived from the Property after foreclosure or other transfer of the Property (or part thereof from which rents, issues and/or profits are derived pursuant to the Security Instrument or by agreement) to Mortgagee or any other third party. Receipt of rents, issues and/or profits by Mortgagee shall not be deemed to constitute a pro-tanto payment of the indebtedness evidenced by, or arising under, this Security Instrument, the Notes, the Credit Agreement or any of the other Loan Documents, but shall be applied as provided in *Section 8.2*.

(i) To the extent permitted by applicable Laws, Mortgagee may exercise any and all rights and remedies granted to a secured party upon default under the Uniform Commercial Code, including, without limiting the generality of the foregoing: (i) the right to take possession of the Collateral (including, without limitation, the Personal Property) or any part thereof, and to take such other measures as Mortgagee may deem necessary for the care, protection and preservation of the Collateral (including, without limitation, the Personal Property); and (ii) request Mortgagor at its expense to assemble the Collateral (including, without limitation, the Personal Property); and make it available to Mortgagee at a convenient place acceptable to Mortgagee. Any notice of sale, disposition or other intended action by Mortgager in accordance with the provisions hereof at least five (5) days prior to such action shall constitute commercially reasonable notice to Mortgagor.

(j) Mortgagee may apply any sums then deposited or held in escrow or otherwise by or on behalf of Mortgagee in accordance with the terms of the Credit Agreement, this Security Instrument or any other Loan Document to the payment of the following items in any order in its sole discretion: (i) Taxes and Other Charges; (ii) insurance premiums; (iii) interest on the unpaid principal balance of the Notes; (iv) amortization of the unpaid principal balance of the Notes; (v) all other sums payable pursuant to the Notes, the Credit Agreement, this Security Instrument and the other Loan Documents, including without limitation advances made by Mortgagee or any Lender pursuant to the terms of this Security Instrument.

(k) Mortgagee may surrender the insurance policies maintained pursuant to the Credit Agreement, collect the unearned insurance premiums for such insurance policies and apply such sums as a credit on the Obligations in such priority and proportion as Mortgagee in its discretion shall deem proper, and in connection therewith, Mortgagor hereby appoints Mortgagee as agent and attorney-in-fact (which is coupled with an interest and is therefore irrevocable) for Mortgagor to collect such insurance premiums.

(1) Mortgagee may apply the undisbursed balance of any deposit made by Mortgagor with Mortgagee in connection with the restoration of the Property after a casualty thereto or condemnation thereof, together with interest thereon, to the payment of the Obligations in such order, priority and proportions as Mortgagee shall deem to be appropriate in its discretion.

(m) Mortgagee may pursue such other remedies as Mortgagee may have under applicable Laws.

Section 8.2. <u>APPLICATION OF PROCEEDS</u>. The purchase money, proceeds and avails of any disposition of the Property, and or any part thereof, or any other sums collected by Mortgagee on behalf of Lenders pursuant to the Notes, this Security Instrument or the other Loan Documents shall, subject to the terms and conditions of the Credit Agreement, be applied by Mortgagee to the payment of the Obligations in such priority and proportions as Mortgagee in its discretion shall deem proper.

Section 8.3. RIGHT TO CURE DEFAULTS. Upon the occurrence and during the continuance of any Event of Default, Mortgagee may, but without any obligation to do so and without notice to or demand on Mortgagor and without releasing Mortgagor from any obligation hereunder, make any payment or do any act required of Mortgagor hereunder in such manner and to such extent as Lenders may deem necessary to protect the security hereof. Mortgagee is authorized to enter upon the Property for such purposes, or appear in, defend, or bring any action or proceeding to protect its interest in the Property or to foreclose this Security Instrument or collect the Obligations, and the cost and expense thereof (including reasonable attorneys' fees to the extent permitted by Laws), with interest as provided in this Section 8.3, shall constitute a portion of the Obligations and shall be due and payable to Mortgagee on behalf of Lenders upon demand. All such costs and expenses incurred by Mortgagee or any Lender in remedying such Event of Default or such failed payment or act or in appearing in, defending, or bringing any such action or proceeding shall bear interest at any default rate specified in the Credit Agreement, if any (the "Default Rate"), for the period after notice from Mortgagee or any Lender that such cost or expense was incurred to the date of payment to Mortgagee or such Lender. All such costs and expenses incurred by Mortgagee or any Lender together with interest thereon calculated at the Default Rate shall be deemed to constitute a portion of the Obligations and be secured by this Security Instrument and the other Loan Documents and shall be immediately due and payable upon demand by Mortgagee or such Lender therefor.

Section 8.4. <u>ACTIONS AND PROCEEDINGS</u>. Mortgagee has the right to appear in and defend any action or proceeding brought with respect to the Property and to bring any action or proceeding, in the name and on behalf of Mortgagor, which Mortgagee, in its discretion, decides should be brought to protect its interest in the Property.

Section 8.5. <u>RECOVERY OF SUMS REQUIRED TO BE PAID</u>. Mortgagee and Lenders shall have the right from time to time to take action to recover any sum or sums which constitute a part of the Obligations as the same become due, without regard to whether or not the balance of the Obligations shall be due, and without prejudice to the right of Mortgagee and Lenders thereafter to bring an action of foreclosure, or any other action, for a default or defaults by Mortgagor existing at the time such earlier action was commenced.

Section 8.6. <u>ADDITIONAL PROVISIONS</u>. With respect to the Collateral, from the Effective Date until the Obligations are paid and performed in full or this Security Instrument is otherwise released by written instrument executed by Mortgagee and authorized to be recorded in the applicable public records of the jurisdiction in which the Property is located, Mortgagee is hereby irrevocably appointed the true and lawful attorney of the Mortgagor (coupled with an interest), in its name and stead, to make all necessary conveyances, assignments, transfers and deliveries of the Collateral, and for that purpose Mortgagee may execute all necessary instruments of conveyance, assignment, transfer and delivery, and may substitute one or more Persons with such power, Mortgagor hereby ratifying and confirming all that its said attorney or such substitute or substitutes shall lawfully do by virtue hereof. Notwithstanding the foregoing, Mortgagor, if so requested by Mortgagee, shall ratify and confirm any such sale or sales by executing and delivering to Mortgagee, for such purpose, and as may be designated in such request. To the extent permitted by Laws, any such sale or sales made under or by virtue of this *Section 8.6* shall operate to divest all the estate, right,

title, interest, claim and demand whatsoever, whether at law, or in equity, of Mortgagor in and to the properties and rights so sold, and shall be a perpetual bar both at law and in equity against Mortgagor and against any and all Persons claiming or who may claim the same, or any part thereof, from, through or under Mortgagor. Upon any sale made under or by virtue of this *Section 8.6*, Mortgagee may, to the extent permitted by Laws, bid for and acquire the Property or any part thereof and in lieu of paying cash therefor may make settlement for the purchase price by crediting upon the Obligations secured hereby the net sales price after deducting therefrom the expenses of the sale and the cost of the auction and any other sums which Mortgagee is authorized to deduct by Laws or under this Security Instrument. At any sale pursuant to this *Section 8.6*, whether made under power herein granted, under or as otherwise authorized by applicable Laws or pursuant to Legal Requirements, or by virtue of any judicial proceeding or any other legal right, remedy or recourse, it shall not be necessary for Mortgagee to be physically present, or to have constructive possession of, the Property, and the title to and right of possession of any such property shall pass to the purchaser thereof as completely as if the same had been actually presented and delivered to the purchaser at such sale.

Section 8.7. <u>OTHER RIGHTS, ETC</u>.

i. The failure of Lenders or Mortgagee to insist upon strict performance of any term hereof shall not be deemed to be a waiver of any term of this Security Instrument or any other Loan Document. Mortgagor shall not be relieved of Mortgagor's obligations hereunder by reason of (i) the failure of Lenders or Mortgagee to comply with any request of Mortgagor or any guarantor or indemnitor with respect to the Loan to take any action to foreclose this Security Instrument or otherwise enforce any of the provisions hereof or of the Notes or the other Loan Documents, (ii) the release, regardless of consideration, of less than the whole of the Property, or of any Person liable for the Obligations or any portion thereof unless, in connection with such release, Mortgagee releases of record this Security Instrument in its entirety, or (iii) any agreement or stipulation by Mortgagee or Lenders extending the time of payment or otherwise modifying or supplementing the terms of the Notes, this Security Instrument or the other Loan Documents.

ii. It is agreed that the risk of loss or damage to the Property is on Mortgagor, and neither Mortgagee nor Lenders shall have any liability whatsoever for decline in the value of the Property, for failure to maintain the insurance policies required to be maintained pursuant to the Credit Agreement, or for failure to determine whether insurance in force is adequate as to the amount of risks insured. Possession by Mortgagee shall not be deemed an election of judicial relief if any such possession is requested or obtained with respect to any Property or collateral not in Mortgagee's possession.

iii. Mortgagee or Lenders may resort for the payment of the Obligations to any other security held by Mortgagee or Lenders in such order and manner as Mortgagee or Lenders, in their discretion, may elect. Mortgagee or Lenders may take action to recover the Obligations, or any portion thereof, or to enforce any covenant hereof without prejudice to the right of Mortgagee or Lenders thereafter to foreclose this Security Instrument. The rights of Mortgagee and Lenders under this Security Instrument shall be separate, distinct and cumulative and none shall be given effect to the exclusion of the others. No act of Lenders or Mortgagee shall be construed as an election to proceed under any one provision herein to the exclusion of any other provision. None of Lenders or Mortgagee shall be limited exclusively to the rights and remedies herein stated but shall be entitled to every right and remedy now or hereafter afforded at law or in equity.

iv. In the event of a foreclosure sale, whether made under the terms hereof, or under judgment of a court, the Collateral may, at the option of Mortgagee, be sold as a whole with the Land and Improvements.

Section 8.8. <u>RIGHT TO RELEASE ANY PORTION OF THE PROPERTY</u>. Mortgagee may release any portion of the Property for such consideration as Mortgagee may require without, as to the remainder of the Property, in any way impairing or affecting the Lien or priority of this Security Instrument, or improving the position of any subordinate lienholder with respect thereto, except to the extent that the obligations hereunder shall have been reduced by the actual monetary consideration, if any, received by Mortgagee or any Lender for such release, and may accept by assignment, pledge or otherwise any other property in place thereof as Mortgagee may require without being accountable for so doing to any other lienholder. This Security Instrument shall continue as a Lien and security interest in the remaining portion of the Property.</u>

Section 8.9. <u>RIGHT OF ENTRY</u>. Upon reasonable notice to Mortgagor, Mortgagee and its agents shall have the right to enter and inspect the Property at all reasonable times.

Section 8.10 BANKRUPTCY.

(a) Upon the occurrence and during the continuance of an Event of Default, Mortgagee shall have the right to proceed in its own name or in the name of Lenders or in the name of Mortgagor in respect of any claim, suit, action or proceeding relating to the rejection of any Lease, including, without limitation, the right to file and prosecute, to the exclusion of Mortgagor, any proofs of claim, complaints, motions, applications, notices and other documents, in any case in respect of the lessee under such Lease under the Bankruptcy Code.

(b) If there shall be filed by or against Mortgagor a petition under Bankruptcy Code, as the same may be amended from time to time, and Mortgagor, as lessor under any Lease, shall determine to reject such Lease pursuant to Section 365(a) of the Bankruptcy Code, then Mortgagor shall give Mortgagee not less than ten (10) days' prior notice of the date on which Mortgagor shall apply to the bankruptcy court for authority to reject the Lease. Mortgagee shall have the right, but not the obligation, to serve upon Mortgagor within such ten-day period a notice stating that (i) Mortgagee demands that Mortgagor assume and assign the Lease to Mortgagee pursuant to Section 365 of the Bankruptcy Code and (ii) Mortgagee serves upon Mortgagor the notice described in the preceding sentence, Mortgagor shall not seek to reject the Lease and shall comply with the demand provided for in clause (i) of the preceding sentence within thirty (30) days after the notice shall have been given, subject to the performance by Mortgagee of the covenant provided for in clause (ii) of the preceding sentence.

Section 8.11. <u>SUBROGATION</u>. If any or all of the proceeds of the Notes have been used to extinguish, extend or renew any indebtedness heretofore existing against all or any portion of the Property, then, to the extent of the funds so used, Mortgagee and Lenders shall be subrogated to all of the rights, claims, Liens, titles, and interests existing against the Property heretofore held by, or in favor of, the holder of such indebtedness and such former rights, claims, Liens, titles, and interests, if any, are not waived but rather are continued in full force and effect in favor of Mortgagee and Lenders and are merged with the Lien and security interest created herein as cumulative security for the payment and performance of the Obligations.

ARTICLE 9. INDEMNIFICATIONS

Section 9.1 <u>GENERAL INDEMNIFICATION</u>. Mortgagor shall, at its sole cost and expense, protect, defend, indemnify, release and hold harmless the Indemnitees from and against any and all losses, claims, damages, liabilities and related expenses (including the reasonable fees, charges and disbursements

of any counsel for an Indemnitee) (collectively, "Losses") imposed upon or incurred by or asserted against any Indemnitee and directly or indirectly arising out of or in any way relating to any one or more of the following: (a) any accident, injury to or death of persons or loss of or damage to property occurring in, on or about the Property or any part thereof or on the adjoining sidewalks, curbs, adjacent property or adjacent parking areas, streets or ways; (b) any use, nonuse or condition in, on or about the Property or any part thereof or on the adjoining sidewalks, curbs, adjacent property or adjacent parking areas, streets or ways; (c) performance of any labor services or the furnishing of any materials or other property in respect of the Property or any part thereof; (d) any failure of the Property to be in compliance with any Legal Requirements; (e) any and all claims and demands whatsoever which may be asserted against Mortgagee or any Lender by reason of any alleged obligations or undertakings on its part to perform or discharge any of the terms, covenants, or agreements contained in the Ground Lease, the Operating Lease or any Lease; or (f) the payment of any commission charge or brokerage fee to anyone which may be payable in connection with the Loan; provided that such indemnity shall not, as to any Indemnitee, be available to the extent that such losses, claims, damages, liabilities or related expenses (x) are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Indemnitee or (y) result from a claim brought by Mortgagor, any Borrower or any other Loan Party against an Indemnitee for breach in bad faith of such Indemnitee's obligations hereunder or under any other Loan Document, if Mortgagor, any Borrower or such Loan Party has obtained a final and nonappealable judgment in its favor on such claim as determined by a court of competent jurisdiction. Any amounts payable to an Indemnitee by reason of the application of this Section 9.1 shall be immediately due and payable, shall constitute a portion of the Obligations and shall bear interest at the Default Rate from the date the Loss is sustained by an Indemnitee until paid.

Section 9.2. <u>MORTGAGE AND/OR INTANGIBLE TAX</u>. Mortgagor shall, at its sole cost and expense, protect, defend, indemnify, release and hold harmless the Indemnitees from and against any and all Losses imposed upon or incurred by or asserted against any Indemnitee and directly or indirectly arising out of or in any way relating to any tax on the making and/or recording of this Security Instrument or any other Loan Document.

Section 9.3. <u>ENVIRONMENTAL COVENANTS</u>. Certain Loan Parties have provided representations, warranties and covenants regarding environmental matters set forth either in the Credit Agreement or in the Environmental Indemnity defined therein, and Mortgagor shall comply with the aforesaid covenants regarding environmental matters.

Section 9.4. <u>MORTGAGEE'S RIGHTS</u>. Mortgagee and any other Person designated by Mortgagee, including but not limited to any representative of a governmental authority, and any environmental consultant, and any receiver appointed by any court of competent jurisdiction, shall have the right, but not the obligation, to enter upon the Property at all reasonable times to assess any and all aspects of the environmental condition of the Property and its use, including but not limited to conducting any environmental assessment or audit (the scope of which shall be determined in Mortgagee's sole (but reasonable) discretion) and taking samples of soil, groundwater or other water, air, or building materials, and conducting other invasive testing. Mortgagee spreases that it shall not exercise its rights under this *Section 9.4* more frequently than once per calendar year unless Mortgagee reasonably believes that the Property is not in full compliance with all Environmental Laws or if an Event of Default has occurred and is continuing.

ARTICLE 10. WAIVERS AND OTHER MATTERS

Section 10.1. <u>WAIVER OF COUNTERCLAIM</u>. Mortgagor hereby waives the right to assert a counterclaim, other than a mandatory or compulsory counterclaim, in any action or proceeding brought against it by Mortgagee or any Lender arising out of or in any way connected with this Security Instrument, the Property, the Notes, the Credit Agreement, any of the other Loan Documents or the Obligations.

Section 10.2. <u>MARSHALLING AND OTHER MATTERS</u>. Mortgagor hereby waives, to the extent permitted by Laws, the benefit of all Legal Requirements now or hereafter in force regarding appraisement, valuation, stay, extension, reinstatement and redemption and all rights of marshalling in the event of any sale hereunder of the Property or any part thereof or any interest therein. Further, Mortgagor hereby expressly waives any and all rights of redemption from sale under any order or decree of foreclosure of this Security Instrument on behalf of Mortgagor, and on behalf of each and every Person acquiring any interest in or title to the Property subsequent to the date of this Security Instrument and on behalf of all Persons to the extent permitted by Legal Requirements.

Section 10.3. <u>WAIVER OF NOTICE</u>. Mortgagor shall not be entitled to any notices of any nature whatsoever from Lenders, Mortgagee except with respect to matters for which this Security Instrument or the Credit Agreement specifically and expressly provides for the giving of notice by Lenders or Mortgagee to Mortgagor and except with respect to matters for which Mortgagor is not permitted by Legal Requirements to waive its right to receive notice, and Mortgagor hereby expressly waives presentment, demand, protest, notice of protest and non-payment, or other notice of default, notice of acceleration and intention to accelerate or other notice of any kind from Lenders, Mortgagee with respect to any matter for which this Security Instrument or Legal Requirements do not specifically and expressly provide for the giving of notice by Lenders, Mortgagee to Mortgagor.

Section 10.4. <u>WAIVER OF STATUTE OF LIMITATIONS</u>. Mortgagor hereby expressly waives and releases to the fullest extent permitted by applicable Laws, the pleading of any statute of limitations as a defense to payment or performance of the Obligations.

Section 10.5. <u>SOLE DISCRETION OF MORTGAGEE</u>. Whenever pursuant to this Security Instrument, Mortgagee exercises any right given to it to approve or disapprove, or any arrangement or term is to be satisfactory to Mortgagee,-the decision of Mortgagee to approve or disapprove or to decide whether arrangements or terms are satisfactory or not satisfactory shall (except as is otherwise specifically provided herein or in the Credit Agreement) be in the sole discretion of Mortgagee and shall be final and conclusive.

Section 10.6. WAIVER OF TRIAL BY JURY. MORTGAGOR HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAWS, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS SECURITY AGREEMENT OR ANY OTHER LOAN DOCUMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). MORTGAGOR CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PERSON HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PERSON WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER.

Section 10.7. <u>WAIVER OF FORECLOSURE DEFENSE</u>. Mortgagor hereby waives any defense Mortgagor might assert or have by reason of Mortgagee's failure to make any tenant or lessee of the Property a party defendant in any foreclosure proceeding or action instituted by Mortgagee.

Section 10.8. MORTGAGOR'S KNOWLEDGE. MORTGAGOR SPECIFICALLY ACKNOWLEDGES AND AGREES (a) THAT IT HAS A DUTY TO READ THIS SECURITY INSTRUMENT AND THAT IT IS CHARGED WITH NOTICE AND KNOWLEDGE OF THE TERMS HEREOF, (b) THAT IT HAS IN FACT READ THIS SECURITY INSTRUMENT AND IS FULLY INFORMED AND HAS FULL NOTICE AND KNOWLEDGE OF THE TERMS, CONDITIONS AND EFFECTS OF THIS SECURITY INSTRUMENT, (c) THAT IT HAS BEEN REPRESENTED BY LEGAL COUNSEL OF ITS CHOICE THROUGHOUT THE NEGOTIATIONS PRECEDING ITS EXECUTION OF THIS SECURITY INSTRUMENT AND HAS RECEIVED THE ADVICE OF SUCH COUNSEL IN CONNECTION WITH ENTERING INTO THIS SECURITY INSTRUMENT, AND (d) THAT IT RECOGNIZES THAT CERTAIN OF THE TERMS OF THIS SECURITY INSTRUMENT PROVIDE FOR (i) CERTAIN WAIVERS AND FOR (ii) THE ASSUMPTION BY ONE PARTY OF, AND/OR RELEASE OF THE OTHER PARTY FROM, CERTAIN LIABILITIES THAT SUCH PARTY MIGHT OTHERWISE BE RESPONSIBLE FOR UNDER APPLICABLE LAWS. MORTGAGEE FURTHER AGREES AND COVENANTS THAT IT WILL NOT CONTEST THE VALIDITY OR ENFORCE ABILITY OF ANY SUCH PROVISIONS OF THIS SECURITY INSTRUMENT ON THE BASIS THAT THE PARTY HAD NO NOTICE OR KNOWLEDGE OF SUCH PROVISION OR THAT SUCH PROVISIONS ARE NOT "CONSPICUOUS."

Section 10.9. USURY SAVINGS PROVISIONS. It is the intent of Mortgagee, Lenders and Mortgagor in the execution of the Credit Agreement and the other Loan Documents and any other written or oral agreement by Mortgagor in favor of Mortgagee and Lenders to contract in strict compliance with applicable usury Laws. In furtherance thereof, Mortgagee, Lenders and Mortgagor stipulate and agree that none of the terms and provisions contained in the Credit Agreement and the other Loan Documents, or in any other written or oral agreement by Mortgagor or any Loan Party in favor of Mortgagee and Lenders, shall ever be construed to create a contract to pay for the use, forbearance or detention of money, or interest at a rate in excess of the maximum interest rate permitted to be charged by applicable Laws; that neither Mortgagor nor any guarantors, endorsers or other Persons now or hereafter becoming liable for payment of the Obligations are agreeing to pay at a rate in excess of the maximum interest that may be lawfully charged under applicable Laws; and that the provisions of this subsection shall control over all other provisions of the Credit Agreement and the other Loan Documents or any other oral or written agreements which may be in apparent conflict herewith. Mortgagee and Lenders expressly disavow any intention to charge or collect excessive unearned interest or finance charges in the event the maturity of the Loan or the remaining Obligations are accelerated. If the maturity of the Loan or the remaining Obligations shall be accelerated for any reason or if the principal of the Loan or the remaining Obligations are paid prior to the end of the term of the Loan or the Obligations, as applicable, and as a result thereof the interest received for the actual period of existence of the Loan or the Obligations, as applicable, exceeds the applicable maximum lawful rate, Mortgagee and Lenders shall, at Mortgagee's option, either refund to Mortgagor the amount of such excess or credit the amount of such excess against the principal balance of the Obligations then outstanding and thereby shall render inapplicable any and all penalties of any kind provided by applicable Laws as a result of such excess interest. In the event that Mortgagee and Lenders shall contract for, charge or receive any amount or amounts and/or any other thing of value which are determined to constitute interest which would increase the effective interest rate on the Loan or the Obligations to a rate in excess of that permitted to be charged by applicable Laws, an amount equal to interest in excess of the lawful rate shall, upon such determination, at the option of Mortgagee, be either immediately returned to Mortgagor or credited against the Obligations then outstanding, in which event any and all penalties of any kind under applicable Laws as a result of such excess interest shall be inapplicable.

ARTICLE 11. CROSS-COLLATERALIZATION

CROSS-COLLATERALIZATION. Mortgagor acknowledges that the Obligations Section 11.1. are secured by this Security Instrument together with those certain other Mortgages (as defined in the Credit Agreement) now or hereafter given by Borrower (or any one or more of them) to Mortgagee (whether one or more, collectively, the "Other Mortgages") encumbering the real and personal property more particularly described in the Other Mortgages (such real and personal property, collectively, the "Other Properties"), all as set forth in the Credit Agreement. Upon the occurrence of an Event of Default, Mortgagee shall have the right to institute a proceeding or proceedings for the total or partial foreclosure of this Security Instrument and any or all of the Other Mortgages whether by court action, power of sale or otherwise, under any applicable provision of Laws, for all of the Obligations and the Lien and the security interest created by the Other Mortgages shall continue in full force and effect without loss of priority as a Lien and security interest securing the payment of that portion of the Obligations then due and payable but still outstanding. Mortgagor acknowledges and agrees that the Property and the Other Properties are located in one or more states and/ or counties, and therefore Mortgagee shall be permitted to, or as the case may be, to direct certain named trustees to, enforce payment and performance of the Obligations and the performance of any term, covenant or condition of the Notes, this Security Instrument, the Other Mortgages or the other Loan Documents and exercise any and all rights and remedies under the Notes, this Security Instrument, the other Loan Documents or the Other Mortgages or, as provided by law or at equity, by one or more proceedings, whether contemporaneous, consecutive or both, to be determined by Mortgagee, in its sole discretion, in any one or more of the states or counties in which the Property or any of the Other Properties are located. Neither the acceptance of this Security Instrument, the Other Mortgages or the other Loan Documents nor the enforcement thereof in any one state or county, whether by court action, foreclosure, power of sale or otherwise, shall prejudice or in any way limit or preclude enforcement by court action, foreclosure, power of sale or otherwise, of the Notes, this Security Instrument, the Other Mortgages or the other Loan Documents through one or more additional proceedings in that state or county or in any other state or county. Any and all sums received by Mortgagee or any Lender under the Notes, this Security Instrument, and the other Loan Documents shall be applied to the Obligations in such order and priority as Mortgagee shall determine, in its sole discretion, without regard to any portion of the Loan allocated to any Property or any of the Other Properties or the appraised value of the Property or any of the Other Properties.

ARTICLE 12. MORTGAGEE AND NOTICES

Section 12.1. <u>FAILURE TO ACT</u>. Notwithstanding anything to the contrary contained herein or in any other Loan Document, the failure of Mortgagee to take any action hereunder or under any other Loan Document shall not (a) be deemed to be a waiver of any term or condition of this Security Instrument or any of the other Loan Documents, (b) adversely affect any rights of Mortgagee or any Lender hereunder or under any other Loan Document, or (c) relieve Mortgagor of any of Mortgagor's obligations hereunder or under any other Loan Document.

Section 12.2 <u>NOTICES</u>. All notices or other written communications hereunder shall be delivered in accordance with the applicable terms and conditions of the Credit Agreement (including that notices to Mortgagor shall be sent to each of Mortgagor's addresses set forth in the Credit Agreement and this Security Instrument); provided, however, that any notice given in accordance with the requirements of any applicable statute (including, without limitation, statutes governing foreclosure or notices of foreclosure) shall be effective when given in accordance with statutory requirements, notwithstanding anything to the contrary contained herein or in any other Loan Document.

ARTICLE 13. APPLICABLE LAWS

Section 13.1. GOVERNING LAWS; JURISDICTION; ETC.

(a) <u>GOVERNING LAW</u>. THIS SECURITY INSTRUMENT SHALL IN ALL RESPECTS BE GOVERNED, CONSTRUED, APPLIED AND ENFORCED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK (OTHER THAN THOSE CONFLICT OF LAW PROVISIONS THAT WOULD DEFER TO THE SUBSTANTIVE LAWS OF ANOTHER JURISDICTION). WITHOUT IN ANY WAY LIMITING THE PRECEDING CHOICE OF LAW, THE PARTIES ELECT TO BE GOVERNED BY NEW YORK LAW IN ACCORDANCE WITH, AND ARE RELYING (AT LEAST IN PART) ON, SECTION 5-1401 OF THE GENERAL OBLIGATIONS LAW OF THE STATE OF NEW YORK; PROVIDED HOWEVER, THAT WITH RESPECT TO THE CREATION, PERFECTION, PRIORITY AND ENFORCEMENT OF THE LIENS AND SECURITY INTERESTS CREATED BY THIS SECURITY INSTRUMENT AND THE DETERMINATION OF DEFICIENCY JUDGMENTS, THE LAWS OF THE STATE WHERE THE LAND IS LOCATED SHALL APPLY.

(b) SUBMISSION TO JURISDICTION. WITH RESPECT TO ANY CLAIM OR ACTION ARISING UNDER THIS SECURITY INSTRUMENT, MORTGAGOR (A) IRREVOCABLY SUBMITS TO THE NONEXCLUSIVE JURISDICTION OF THE COURTS OF THE STATE OF NEW YORK AND THE UNITED STATES DISTRICT COURT LOCATED IN THE BOROUGH OF MANHATTAN IN NEW YORK, NEW YORK, AND APPELLATE COURTS FROM ANY THEREOF, AND (B) IRREVOCABLY WAIVES ANY OBJECTION WHICH IT MAY HAVE AT ANY TIME TO THE LAYING ON VENUE OF ANY SUIT, ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS SECURITY INSTRUMENT BROUGHT IN ANY SUCH COURT, IRREVOCABLY WAIVES ANY CLAIM THAT ANY SUCH SUIT, ACTION OR PROCEEDING BROUGHT IN ANY SUCH COURT HAS BEEN BROUGHT IN AN INCONVENIENT FORUM. NOTHING IN THIS SECURITY INSTRUMENT WILL BE DEEMED TO PRECLUDE AGENT FROM BRINGING AN ACTION OR PROCEEDING WITH RESPECT HERETO IN ANY OTHER JURISDICTION.

(c) <u>SERVICE OF PROCESS</u>. EACH PARTY HERETO IRREVOCABLY CONSENTS TO SERVICE OF PROCESS IN THE MANNER PROVIDED FOR NOTICES IN *SECTION 12.2*. NOTHING IN THIS AGREEMENT WILL AFFECT THE RIGHT OF ANY PARTY HERETO TO SERVE PROCESS IN ANY OTHER MANNER PERMITTED BY APPLICABLE LAW.

Section 13.2. <u>PROVISIONS SUBJECT TO APPLICABLE LAWS</u>. All rights, powers and remedies provided in this Security Instrument may be exercised only to the extent that the exercise thereof does not violate any applicable provisions of Laws and are intended to be limited to the extent necessary so that they will not render this Security Instrument invalid, unenforceable or not entitled to be recorded, registered or filed under the provisions of any applicable Laws. If any term of this Security Instrument or any application thereof shall be invalid or unenforceable, the remainder of this Security Instrument and any other application of the term shall not be affected thereby.

ARTICLE 14. DEFINITIONS

Section 14.1 <u>GENERAL DEFINITIONS</u>. Unless the context clearly indicates a contrary intent or unless otherwise specifically provided herein, words used in this Security Instrument may be used interchangeably in singular or plural form and:

(a) the word "<u>Mortgagor</u>" shall mean "each Mortgagor and any subsequent owner or owners of the Property or any part thereof or any interest therein";

(b) the word "<u>Mortgagee</u>" shall mean "Mortgagee and any of Mortgagee's successors and assigns";

(c) the word "<u>Lenders</u>" shall mean "all or each of the Lenders and any of their or its respective successors and assigns";

(d) the word "<u>Notes</u>" shall mean "the Notes and any other evidence of indebtedness secured by this Security Instrument";

(e) the word "**<u>Property</u>**" shall include any portion of the Property and any interest therein;

(f) the word "<u>Other Properties</u>" shall include any portion of the Other Properties and any interest therein; and

(g) the phrases "attorneys' fees", "legal fees" and "counsel fees" shall include any and all attorneys', paralegal and law clerk fees and disbursements, including, but not limited to, fees and disbursements at the pre-trial, trial and appellate levels incurred or paid by Lender in protecting its interest in the Property, the Leases and the Rents and enforcing its rights hereunder.

ARTICLE 15. MISCELLANEOUS PROVISIONS

Section 15.1. <u>NO ORAL CHANGE</u>. This Security Instrument, and any provisions hereof, may not be modified, amended, waived, extended, changed, discharged or terminated orally or by any act or failure to act on the part of Mortgagor, Lenders or Mortgagee, but only by an agreement in writing signed by the party against whom enforcement of any modification, amendment, waiver, extension, change, discharge or termination is sought.

Section 15.2. <u>SUCCESSORS AND ASSIGNS</u>. This Security Instrument shall be binding upon and inure to the benefit of Mortgagor, Mortgagee and Lenders and their respective successors and assigns forever.

Section 15.3. <u>INAPPLICABLE PROVISIONS</u>. If any term, covenant or condition of the Credit Agreement, the Notes or this Security Instrument is held to be invalid, illegal or unenforceable in any respect, the Credit Agreement, the Notes and this Security Instrument shall be construed without such provision.

Section 15.4. <u>HEADINGS, ETC</u>. The headings and captions of various Sections of this Security Instrument are for convenience of reference only and are not to be construed as defining or limiting, in any way, the scope or intent of the provisions hereof. **Section 15.5.** <u>NUMBER AND GENDER</u>. Whenever the context may require, any pronouns used herein shall include the corresponding masculine, feminine or neuter forms, and the singular form of nouns and pronouns shall include the plural and vice versa.

Section 15.6. <u>ENTIRE AGREEMENT</u>. This Security Instrument and the other Loan Documents contain the entire agreement of the parties hereto and thereto in respect of the transactions contemplated hereby and thereby, and all prior agreements among or between such parties, whether oral or written, are superseded by the terms of this Security Instrument and the other Loan Documents.

Section 15.7. <u>LIMITATION ON MORTGAGEE'S OR LENDERS' RESPONSIBILITY</u>. No provision of this Security Instrument shall operate to place any obligation or liability for the control, care, management or repair of the Property upon Lenders or Mortgagee, nor shall it operate to make Lenders or Mortgagee responsible or liable for any waste committed on the Property by the tenants or any other Person, or for any dangerous or defective condition of the Property, or for any negligence in the management, upkeep, repair or control of the Property resulting in loss or injury or death to any tenant, licensee, employee or stranger. Nothing herein contained shall be construed as constituting Lenders or Mortgagee a "mortgagee in possession."</u>

ARTICLE 16. STATUS OF MORTGAGOR

Section 16.1 STATUS OF MORTGAGOR. Mortgagor's exact legal name is correctly set forth in the first paragraph of this Security Instrument and the signature block at the end of this Security Instrument. Mortgagor is an organization of the type specified in the first paragraph of this Security Instrument. Mortgagor is incorporated in or organized under the Laws of the state specified in Section 1.4 of this Security Instrument. Mortgagor's principal place of business and chief executive office, and the place where Mortgagor kept its books and records, including recorded data of any kind or nature, regardless of the medium or recording, including software, writings, plans, specifications and schematics, has been for the preceding four months (or, if less, the entire period of the existence of Mortgagor) c/o FelCor Lodging Trust Incorporated, 545 E. John Carpenter Freeway, Suite 1300, Irving, Texas 75062, Attention: General Counsel. Mortgagor's organizational identification number, if any, assigned by the state of incorporation or organization is correctly set forth in Section 1.4 of this Security Instrument. Mortgagor will not change or permit to be changed (a) Mortgagor's name, (b) Mortgagor's identity (including its trade name or names), (c) Mortgagor's principal place of business set forth on the first page of this Security Instrument, (d) the corporate, partnership or other organizational structure of Mortgagor, (e) Mortgagor's state of organization, or (f) Mortgagor's organizational number, without notifying Lender of such change in writing at least thirty (30) days prior to the effective date of such change and, in the case of a change in Mortgagor's structure not permitted by the Credit Agreement, without first obtaining the prior written consent of Lender. If Mortgagor does not now have an organizational identification number and later obtains one, Mortgagor promptly shall notify the Lender of such organizational identification number.

ARTICLE 17. INTENTIONALLY OMITTED

ARTICLE 18. GROUND LEASE AND OPERATING LEASE PROVISIONS.

Section 18.1. <u>NO MERGER OF FEE AND LEASEHOLD ESTATES, RELEASES</u>. So long as any portion of the Obligations remains unpaid or unsatisfied or until this Security Instrument is reconveyed pursuant to the terms hereof or the Credit Agreement, unless Mortgagee shall otherwise consent, the fee title to the Land and the leasehold estates created under the Ground Lease and the Operating Lease shall not merge but shall always be kept separate and distinct, notwithstanding the union of such estates by purchase,

operation of Laws or otherwise. Mortgagee reserves the right, at any time, to release portions of the Property, including, but not limited to, the leasehold estates created by the Ground Lease and the Operating Lease, with or without consideration, at Mortgagee's election, without waiving or affecting any of its rights hereunder or under the Notes or the other Loan Documents and any such release shall not affect Mortgagee's rights in connection with the portion of the Property not so released.

37. <u>MORTGAGOR'S ACQUISITION OF FEE ESTATE</u>. So long as any portion of the Obligations remains unpaid or unsatisfied or until this Security Instrument is reconveyed pursuant to the terms hereof or the Credit Agreement, if any Mortgagor shall be the owner of fee title to and/or a leasehold estate in all or any portion of the Property, the Lien of this Security Instrument shall be spread to cover such additional title and rights and said title and rights shall be deemed to be included in the Property. Mortgagor agrees, at its sole cost and expense, including without limitation, reasonable attorney's fees to (i) execute any and all documents or instruments necessary to confirm that its title and rights to the Property are subject to the Lien of this Security Instrument; and (ii) provide a title insurance policy which shall insure that the Lien of this Security Instrument is a first lien on its title and rights to the Property.

38. <u>Ground Lease Representations</u>. Each Mortgagor represents and warrants to Agent that: (i) the Ground Lease is in full force and effect and has not been modified or amended in any manner whatsoever, except as described above; (ii) the tenant under the Ground Lease enjoys the quiet and peaceful possession of the property demised thereby; (iii) there are no defaults under the Ground Lease by any party thereunder, and no event has occurred which but for the passage of time, or notice, or both would constitute a default under the Ground Lease; (iv) all rents, additional rents and other sums due and payable under the Ground Lease have been paid in full; (v) no party under the Ground Lease; (vi) the consummation of the transactions contemplated hereby will not result in any breach of, or constitute a default under, the Ground Lease; (vii) all actions which must be taken for Agent to have the rights of a leasehold mortgagee or mortgagee pursuant to the Ground Lease have been taken and completed; and (viii) no Mortgagor has granted any other leasehold mortgage or made any other assignment, pledge or hypothecation of its interest under the Ground Lease.

Section 18.4. GROUND LEASE COVENANTS.

Mortgagor shall (i) pay all rents, additional rents and other sums required to be paid i. by it, as tenant under and pursuant to the provisions of the Ground Lease, (ii) diligently perform and observe all of the terms, covenants and conditions of the Ground Lease on its part to be performed thereunder, (iii) promptly notify Agent of the giving of any notice under the Ground Lease of any default by any party thereunder, and deliver to Agent a true copy of each such notice within five (5) days of receipt and (iv) promptly notify Agent of any bankruptcy, reorganization or insolvency of any party to the Ground Lease or of any notice thereof, and deliver to Agent a true copy of such notice within five (5) days of receipt. No Mortgagor shall, without the prior consent of Agent, surrender the leasehold estate created by the Ground Lease or terminate or cancel the Ground Lease or modify, change, supplement, alter, amend or waive any term of the Ground Lease, either orally or in writing, except as expressly set forth in Section 7.23(a) of the Credit Agreement, and if any Mortgagor shall default in the performance or observance of any term, covenant or condition of the Ground Lease, Agent shall have the right, but shall be under no obligation, to pay any sums and to perform any act or take any action as may be appropriate to cause all of the terms, covenants and conditions of the Ground Lease to be performed or observed, to the end that the rights of such Mortgagor in, to and under the Ground Lease shall be kept unimpaired and free from default. If the landlord under the Ground Lease shall deliver to Agent a copy of any notice of default under the Ground Lease, such notice shall constitute full protection to Agent for any action taken or omitted to be taken by Agent, in good faith, in reliance thereon. Each Mortgagor shall exercise its option, if any, to extend or renew the term of the Ground Lease upon demand by Agent made at any time

within one (1) year prior to the last day upon which any such option may be exercised, and each Mortgagor hereby expressly authorizes and appoints Agent its attorney-in-fact to exercise any such option in the name of and upon behalf of such Mortgagor, which power of attorney shall be irrevocable and shall be deemed to be coupled with an interest.

ii. Notwithstanding anything contained in the Ground Lease to the contrary and except for the Operating Lease, no Mortgagor shall further sublet any portion of the Property without prior written consent of Agent, except as expressly permitted by Section 7.16 of the Credit Agreement.

iii. In the event of any default by any Mortgagor in the performance of any of its obligations under the Ground Lease, including, without limitation, any default in the payment of rent and other charges and impositions made payable by the lessee thereunder, then, in the case of a default in the payment of any monetary obligation, Agent may, at its option and without notice, make payment to cure such default, or, in the case of a default in the performance of any non-monetary obligation, Agent may, at its option, cause the default or defaults to be remedied and otherwise exercise any and all of the rights of Mortgagor thereunder in the name of and on behalf of Mortgagor.

iv. If pursuant to subsection 365(h) of the Bankruptcy Code, any Mortgagor seeks to offset, against the rent reserved in the Ground Lease, the amount of any damages caused by the nonperformance by the ground lessor of any of its obligations thereunder after the rejection by such ground lessor under the Bankruptcy Code, then such Mortgagor shall not effect any offset of the amounts so objected to by Agent. If Agent has failed to object as aforesaid within ten (10) days after notice from such Mortgagor in accordance with the first sentence of this subsection, such Mortgagor may proceed to offset the amounts set forth in such Mortgagor's notice to Agent.

v. In any action, proceeding, motion or notice shall be commenced or filed in respect of the ground lessor of all or any part of the Property in connection with any case under the Bankruptcy Code, Agent and Mortgagor shall cooperatively conduct and control any such litigation with counsel agreed upon between Mortgagor and Agent in connection with such litigation. Borrower shall, upon demand, pay to Agent all costs and expenses (including reasonable attorneys' fees and costs) actually paid or actually incurred by Agent or any Lender in connection with the cooperative prosecution or conduct of any such proceedings. All such costs and expenses shall be secured by the Lien of this Security Instrument.

Section 18.5. <u>NO ASSIGNMENT OF GROUND LEASE</u>. Notwithstanding anything to the contrary contained in this Security Instrument, this Security Instrument shall not constitute an assignment of the Ground Lease within the meaning of any provision thereof prohibiting its assignment and Mortgagee shall have no liability or obligation thereunder by reason of its acceptance of this Security Instrument. Mortgagee shall be liable for the obligations of the lessee arising under the Ground Lease for only that period of time which Mortgagee is in possession of the Property or has acquired, by foreclosure or otherwise, and is holding all of Mortgagor's right, title and interest therein.

Section 18.6. <u>GROUND LEASE COVENANTS, REPRESENTATIONS AND WARRANTIES.</u> Supplementing *Sections 18.3* and *18.4* above, the provisions of the Credit Agreement respecting Mortgagor's covenants, representations and warranties with respect to the Ground Lease, including without limitation, Section 6.38 and Section 7.23 thereof, are hereby incorporated herein by reference as if fully set forth herein. Section 18.7. <u>OPERATING LEASE COVENANTS, REPRESENTATIONS AND</u> <u>WARRANTIES.</u> The provisions of the Credit Agreement respecting Mortgagor's covenants, representations and warranties with respect to the Operating Lease, including without limitation, Section 6.39 thereof, are hereby incorporated herein by reference as if fully set forth herein. Without limiting the foregoing, Owner hereby confirms its consent to Operator's grant of the mortgage lien and security interests encumbering the Operating Lease contained herein.

ARTICLE 19. THIRD PARTY MORTGAGOR PROVISIONS

Section 19.1 <u>THIRD PARTY MORTGAGOR</u>. Each Mortgagor agrees to the provisions of this Article of the Security Instrument with respect to its interest in the property encumbered by this Security Instrument and the obligations secured hereby to the extent the proceeds of the secured obligations have been paid to or for the benefit of a person (the "<u>Third Party Borrower</u>") other than such Mortgagor.

Section 19.2 <u>AUTHORIZATIONS TO MORTGAGEE</u>. Mortgagor authorizes Mortgagee, without notice or demand and without affecting their liability hereunder, from time to time to (i) renew, extend, accelerate or otherwise change the time for payment of, or otherwise change the terms of the secured indebtedness or any part thereof, including increase or decrease of the rate of interest thereon; (ii) take and hold other security for the payment of the indebtedness, and exchange, enforce, waive and release any such security; (iii) apply such security and direct the order or manner of sale thereof, including without limitation, a non-judicial sale permitted by the terms of the controlling security agreement or pledge agreement, as Mortgagee in its discretion may determine; and (iv) release or substitute any one or more of the endorsers or guarantors of any indebtedness.

Section 19.3. <u>MORTGAGOR'S WARRANTIES</u>. Mortgagor warrants that: (i) this Security Instrument is executed at the request of Third Party Borrower; and (ii) Mortgagor has adequate means of obtaining from Third Party Borrower on a continuing basis financial and other information pertaining to Third Party Borrower's financial condition without relying on Mortgagee therefor. Mortgagor agrees to keep adequately informed from such means of any facts, events or circumstances which Mortgagor considers material or which might in any way affect Mortgagor's risks hereunder. With respect to information or material acquired in the normal course of Mortgagee's relationship with Third Party Borrower, Mortgagor agrees that Mortgagee shall have no obligation to disclose such information or material to Mortgagor.

Section 19.4 MORTGAGOR'S WAIVERS.

(a) Mortgagor waives any right to require Mortgagee to (A) proceed against any person, including Third Party Borrower; (B) proceed against or exhaust any collateral held from Third Party Borrower, any endorser or guarantor or any other person; (C) give notice of terms, time and place of any public or private sale of personal property or real property security held from Mortgagors; (D) pursue any other remedy in Mortgagee's power; or (E) make any presentments, demands for performance, or give any notices of nonperformance, protests, notices of protests or notices of dishonor in connection with any obligations or evidences of indebtedness held by Mortgagee as security, in connection with any obligations or evidences of indebtedness which constitute in whole or in part the indebtedness secured hereunder, or in connection with the creation of new or additional indebtedness. Mortgagor agrees that, in case of default and foreclosure, Mortgagee may enforce this Security Instrument against any or all interests encumbered by this Security Instrument in a single proceeding.

(b) Mortgagor waives any defense arising by reason of (A) any disability or other defense of Third Party Borrower, any endorser or guarantor or any other person; (B) the cessation from any cause whatsoever, other than payment in full of the indebtedness of Third Party Borrower, of the liability of any endorser or guarantor or any other person; (C) the application by Third Party Borrower of the proceeds of any indebtedness for purposes other than the purpose represented by Third Party Borrower to Mortgagee or intended or understood by Mortgagee or Mortgagor; (D) any act or omission by Mortgagee which directly or indirectly results in or aids the discharge of Third Party Borrower or of any indebtedness by operation of law or otherwise; (E) any modification to any secured indebtedness, and including without limitation the renewal, extension, acceleration or other change in time for payment of the indebtedness, or other change in the terms of the indebtedness or any part thereof, including an increase or decrease of the rate of interest thereon; and (F) all suretyship and guarantor's defenses generally.

(c) Mortgagor shall have no right of subrogation, and Mortgagor further waives any right to enforce any remedy which Mortgagee now has or may hereafter have against Third Party Borrower, any endorser or guarantor or any other person, and waives any benefit of, or any right to participate in any security whatsoever now or hereafter held by Mortgagee from Third Party Borrower or any other person.

Section 19.5. <u>FURTHER EXPLANATION OF WAIVERS</u>. Mortgagor waives all rights and defenses that the Mortgagor may have because the Third Party Borrower's debt is secured by real property. This means, among other things:

(a) The Mortgagee may foreclose against the Mortgagor's interests without first foreclosing on any real or personal property collateral pledged by the Third Party Borrower;

(b) If the Mortgagee forecloses on any real property collateral pledged by the Third Party Borrower: (A) the amount of the debt may reduced only by the price for which that collateral is sold at the foreclosure sale, even if the collateral is worth more than the sale price; and (B) the Mortgagee may foreclose against the Mortgagor's interests even if the Mortgagee, by foreclosing on the Third Party Borrower's interests, has destroyed any right the Mortgagor may have to collect from the Third Party Borrower.

This is an unconditional and irrevocable waiver of any rights and defenses the Mortgagor may have because the Third Party Borrower's debt is secured by the Third Party Borrower's interest in real property.

Section 19.6. MORTGAGOR'S UNDERSTANDINGS WITH RESPECT TO WAIVERS.

(a) Mortgagor warrants and agrees that Mortgagor has had all necessary opportunity to secure any advice which Mortgagor desires with respect to each of the waivers set forth above, that such waivers are made with Mortgagor's full knowledge of their significance and consequences, and that under the circumstances, the waivers are reasonable and not contrary to public policy or law.

(b) Notwithstanding the foregoing, all waivers in this paragraph shall be effective only to the extent permitted by law.

ARTICLE 20. STATE SPECIFIC PROVISIONS

(a) *Intentionally omitted.*

[Remainder of page intentionally blank. Signature page(s) follow.] IN WITNESS WHEREOF, Mortgagor has executed this Security Instrument as of the day and year set forth above.

FELCOR LODGING LIMITED PARTNERSHIP,

a Delaware limited partnership

FelCor Lodging Trust Incorporated, a Maryland By: corporation, its General Partner

By: /s/ Allison S. Navitskas Name: Allison S. Navitskas Title: Vice President

STATE OF) SS: COUNTY OF)

The foregoing instrument was acknowledged before me this _____ day of March, 2011, by ______, as _____ of FelCor Lodging Trust Incorporated, a Maryland corporation, as General Partner of FELCOR LODGING LIMITED PARTNERSHIP, a Delaware limited partnership, who is [] personally known to me or [] has produced ______ as identification.

NOTARY PUBLIC:

(NOTARY SEAL)

Notary Signature

Notary Printed Name My Commission Expires:_____

FELCOR/JPM BOCA RATON HOTEL, L.L.C.

a Delaware limited liability company

By: /s/ Allison S. Navitskas Name: Allison S. Navitskas Title: Vice President

STATE OF) SS: COUNTY OF)

The foregoing instrument was acknowledged before me this _____ day of March, 2011, by Allison Navitskas, as Vice President of FELCOR/JPM BOCA RATON HOTEL, L.L.C., a Delaware limited liability company, who is [] personally known to me or [] has produced ______ as identification.

NOTARY PUBLIC:

Notary Signature

(NOTARY SEAL)

Notary Printed Name My Commission Expires:_____

DJONT/JPM BOCA RATON LEASING, L.L.C.

a Delaware limited liability company

By: /s/ Allison S. Navitskas Name: Allison S. Navitskas Title: Vice President

STATE OF) SS: COUNTY OF)

The foregoing instrument was acknowledged before me this _____ day of March, 2011, by Allison Navitskas, as Vice President of DJONT/JPM BOCA RATON LEASING, L.L.C., a Delaware limited liability company, who is [] personally known to me or [] has produced ______ as identification.

NOTARY PUBLIC:

(NOTARY SEAL)

Notary Signature

Notary Printed Name My Commission Expires:_____

Certification Pursuant to Section 302 of Sarbanes-Oxley Act of 2002

I, Richard A. Smith, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of FelCor Lodging Trust Incorporated;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent function):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 4, 2011

/s/Richard A. Smith Richard A. Smith Chief Executive Officer

Certification Pursuant to Section 302 of Sarbanes-Oxley Act of 2002

I, Andrew J. Welch, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of FelCor Lodging Trust Incorporated;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent function):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 4, 2011

/s/Andrew J. Welch

Section 906 Certification

Certification of Chief Executive Officer and Chief Financial Officer Pursuant to 18 U.S.C. § 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

Pursuant to 18 U.S.C. § 1350, adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, the undersigned officers of FelCor Lodging Trust Incorporated (the "Company") hereby certify, to such officers' knowledge, that:

(i) the accompanying Quarterly Report on Form 10-Q of the Company for the three months ended March 31, 2011 (the "Report") fully complies with the requirements of Section 13(a) or Section 15(d) of the Securities Exchange Act of 1934, as amended;

and

(ii) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: May 4, 2011

/s/Richard A. Smith

Richard A. Smith *Chief Executive Officer*

/s/Andrew J. Welch

Andrew J. Welch Chief Financial Officer